SURROGATE MOTHERHOOD AND ASSISTED REPRODUCTION IN SOUTH AFRICA: AN ANALYSIS OF SELECTED CONSTITUTIONAL AND MEDICO-LEGAL ISSUES

by

ELAINE AURET

submitted in accordance with the requirements for the degree of

DOCTOR OF LAWS

In the College of Law

at the

UNIVERSITY OF SOUTH AFRICA

SUPERVISOR: Prof M. Labuschaigne

CO-SUPERVISOR: Dr N. Mabeka

(31 January 2023 submitted)

DECLARATION

Name: Elaine Auret

Student number: 43062164

Degree: LLD

Surrogate motherhood and assisted reproduction in South Africa: an analysis of selected constitutional and medico-legal issues

I, ELAINE AURET, declare that the above thesis is my own work and that all the sources that I have used or quoted have been indicated and acknowledged by means of complete references.

I further declare that I submitted the thesis to originality checking software and that it falls within the accepted requirements for originality.

I further declare that I have not previously submitted this work, or part of it, for examination at Unisa for another qualification or at any other higher education institution.

<u>31 January 2023</u> DATE

ACKNOWLEDGMENTS

It is with a grateful heart that I can submit this thesis.

I dedicate this thesis to my grandfather, Dr Auret, who has been a great inspiration to me and who always encouraged me to never stop learning. He believed that you are never too old to learn. He also taught me that there will be hard sacrifices that you would have to make to get you where you want to be.

Thank you to my parents who have always encouraged me to follow my dreams no matter what obstacles life may hold. Thank you for believing in me and for all your love, prayers and support over the years with my never-ending studying. I am also thankful to my sisters for their love and support.

I am grateful to my husband for his endless patience with me whilst writing this thesis and through all the years of studying. Thank you for believing in me, always encouraging me and for allowing me to follow my dreams.

Thank you to Professor Labuschaigne for your invaluable assistance and all your patience over the years. The guidance, encouragement and support that I received from both my supervisors, Professor Labuschaigne and Dr Mabeka, kept me firmly on this path.

Without God's love, grace and strength, I would not be at the end of this journey today.

ABSTRACT

The advancement in medical technology over the past few decades has brought hope to infertile people and same-sex couples to have their own families in circumstances where it would not have been possible for them otherwise, except through adoption. These medical technological advances have pushed difficult ethical, social and religious issues to the front as society tries to adjust to new formulations of what constitutes a family. The use of assisted reproductive technologies (ART) and assisted reproductive procedures, especially in the context of surrogate motherhood, raises a range of medico-legal and constitutional issues concerning the parties involved in a surrogate motherhood agreement, namely the commissioning parent(s); the surrogate mother (and her partner, where relevant), and the child to be born following the agreement. Chapter 19 of the Children's Act 38 of 2005 was enacted to, not only protect the best interests of the commissioned child, but also to provide legal clarity regarding the rights and responsibilities of all parties involved in a surrogacy arrangement, including the legal status of the child born as a result of surrogacy and artificial fertilisation. Although different High Court judgments provide some guidance on the scope and type of information that should accompany an application for the confirmation of a surrogacy agreement to enable the court to make a decision on this matter, many gaps and uncertainties remain. Some of the provisions in the Children's Act were recently constitutionally challenged because of critical human rights concerns. Urgent legislative intervention is required to address these lacunae. This thesis proposes that regulations be enacted to chapter 19 of the Children's Act to close the existing regulatory gaps and to assist persons seeking to conceive a child via a surrogacy agreement to comply with the provisions of chapter 19. The law is notorious for often lagging behind the constant advances in medical technology and changes in society, most evident in present times in the context of surrogate motherhood and artificial fertilisation. This thesis also considers what the proposed regulations to the chapter 19 of the Children's Act should consist of with reference to judgments regarding chapter 19 and the Constitution of the Republic of South Africa, 1996. The best interests of children are of paramount importance in all matters concerning children, entrenched in section 28 of the Constitution. This thesis advances that because High Courts have the responsibility to ensure that surrogacy agreements are in the commissioned child's best interests, this obligation will best be dispensed with by the enactment of much needed regulations. The thesis also demonstrates that although South Africa has a comprehensive legal framework regulating surrogate motherhood and artificial fertilisation compared to the United Kingdom, Canada and India, some instructive examples may be taken from these jurisdictions in closing the existing regulatory gaps and ambiguities.

KEY TERMS

Infertility; artificial fertilisation; IVF; surrogate mother; surrogacy; surrogate agreement; best interests of the child; epigenetics; confirmation; reproductive health care

ABREVIATIONS AND ACRONYMS USED

NHA	National Health Act
IVF	In vitro fertilisation
WHO	World Health Organisation
UK	United Kingdom
UNCRC	United Nations Convention on the Rights of a Child
ACWRC	African Charter on the Rights and Welfare of the Child
SALC	South African Law Commission
SALRC	South African Law Reform Commission
AID	Artificial insemination-donor
AIH	Artificial insemination-husband
ART	Assisted reproductive technology
PAIA	Promotion of Access to Information Act

Table of Contents

DECLARATIO	DN	2
ACKNOWLE	DGMENTS	0
ABSTRACT		1
ABREVIATIO	INS AND ACRONYMS USED	3
BACKGROU	ND TO THE STUDY, PROBLEM STATEMENT, OBJECTIVES AND FRAMEWORK	9
1. INTROI	DUCTION	9
2. BACKG	ROUND TO THE STUDY	. 11
2.1	Addressing infertility	
2.2	Fundamental rights considerations	. 13
2.4	Factors leading to the regulation of surrogate motherhood	. 17
3. PROBLI	EM STATEMENT, AIMS AND OBJECTIVES OF THE RESEARCH	. 18
4. RESEAF	RCH APPROACH AND METHODOLOGY	. 20
5. LIMITA	TIONS	. 20
6. FRAME	WORK OF THE THESIS	. 20
7. DEFINI	TIONS	. 21
7.1	Reproductive health care	. 21
7.2	Relevant health service	. 21
7.3	Artificial fertilisation and artificial insemination	. 22
7.4	Assisted reproductive technology (ART)	. 22
7.5	In vitro fertilisation (IVF) and "test-tube babies"	. 23
7.6	Embryo transfer	. 24
7.7	Oocyte	. 24
7.8	Sperm	. 24
7.9	Surrogate mother/surrogate/surrogate motherhood	. 25
7.10	Commissioning parents	. 25
7.11	Children	. 26
7.12	Surrogate motherhood agreement	. 26
7.13	Surrogacy	. 26
7.14	Gestational surrogacy	. 27
7.15	Traditional surrogacy	. 27
7.16	Gamete	. 27
7.17	Gamete donor	. 27
7.18	Artificial insemination-husband (AIH)	. 28
7.19	Artificial insemination-donor (AID)	. 28

7.2	Gamete intra-fallopian transfer (GIFT) and peritoneal oocyte and sperm transfer (POS	-
7.2		
7.2	2 Vaginal intro-peritoneal sperm transfer (VISPER)	29
7.2	3 Parent	29
7.2	4 Commercial surrogacy	29
7.2	5 Altruistic surrogacy	30
8. EXP	ECTED CONTRIBUTIONS AND OUTCOMES OF STUDY	30
CHAPTEI	۲ 2	31
	CAL DEVELOPMENT OF LAWS RELATING TO ASSISTED REPRODUCTION AND SURROGATE	31
1. INT	RODUCTION	31
2. TYP	ES OF SURROGACY	31
2.1	Formal surrogacy	32
2.2	Informal surrogacy	35
2.3	Altruistic surrogacy	38
2.4	Commercial surrogacy	42
2.5	International surrogacy	44
2.6	Concluding observations	46
	COMMON LAW POSITION REGARDING THE LEGAL STATUS OF THE CHILD CONCEIVED AS A ULT OF ARTIFICIAL FERTILISATION WITH DONOR SPERM	
3.1	Introduction	47
3.2	The maxims mater semper certa est and pater is est quem nuptiae demonstrant	47
	AL POSITION OF THE CHILD CONCEIVED BY ARTIFICIAL FERTILISATION PRIOR TO THE ACTMENT OF THE CHILDREN'S ACT	51
4.1	L v J	52
4.2	The Tzaneen triplets (1987)	53
4.3	V v R	54
4.4	J and Another v Director General, Department of Home Affairs and others	55
	TUTES GOVERNING ARTIFICIAL FERTILISATION AND SURROGATE MOTHERHOOD PRIOR TO	
5.1	Introduction	56
5.2	Child Care Act	56
5.3	Children's Status Act	58
5.4	Human Tissue Act	62
6. COI	NCLUSION	64
CHAPTEI	3	65

THE SOUT	TH AFRICAN LEGAL FRAMEWORK GOVERNING SURROGATE MOTHERHOOD	65
1. INTR	ODUCTION	65
2. THE	RIGHTS OF THE CHILD CONCEIVED BY ARTIFICIAL FERTILISATION	66
2.1	Legal status and related matters	66
2.1.1	EJ and others v Haupt NO	68
2.2	Access to information regarding a child's genetic parents	70
2.3	The right to know one's biological or genetic origins	73
3. SALI	ENT ISSUES RELATING TO CHAPTER 19 OF THE CHILDREN'S ACT	76
3.1	Introduction	76
3.2	Nature of surrogate motherhood agreements	77
3.3	Section 292: Requirements for a valid surrogate agreement	79
3.4	Section 293: Consent of the commissioning mother- and surrogate mother's wife or partner	
3.5	Section 294: The genetic link requirement	87
3.6	Section 295: Confirmation of the surrogacy agreement by a High Court	91
3.7	Section 296: Artificial fertilisation of the surrogate mother	103
3.8	Section 297: The consequence of a surrogate motherhood agreement on the the child	
3.9	Section 298: The dissolution of a surrogate agreement	109
3.10	Section 299: The consequence of termination of the surrogate agreement	111
3.11	Section 300: Termination of the pregnancy by the surrogate mother	112
3.12	Section 301: Prohibition of payments in respect of surrogacy	114
3.13	Section 302: The identity of the parties to a surrogate agreement	116
3.14	Section 303: Certain acts are prohibited	117
3.15	Judgments with a focus on chapter 19 of the Children's Act	118
3.16	Criminal offences in terms of chapter 19 of the Children's Act	146
3.17	The medical threshold requirement and permanent illness	
4. CON	CLUSION	152
CHAPTER	4	154
CONSTITU	JTIONAL FRAMEWORK RELEVANT TO SURROGATE MOTHERHOOD	154
1. INTR	ODUCTION	154
2. RIGH	ITS OF PARTIES TO SURROGATE MOTHERHOOD AND ARTIFICIAL FERTILISATION .	155
2.1	The right to bodily and psychological integrity of a person	155
2.2	The right to human dignity	157
2.3	The right to privacy	
2.4	The right to equality	

2.5	The right to reproductive health care	168
2.6	Children's rights and the protection of children	172
2.7	Miscellaneous constitutional arguments relating to artificial fertilisation and su agreements	•
2.8	The constitutionality of selected provisions in chapter 19 of the Children's Act	178
3. THE	INTERNATIONAL AND REGIONAL HUMAN RIGHTS FRAMEWORK RELEVANT TO	
SURRC	OGATE MOTHERHOOD AND ARTIFICIAL FERTILISATION	212
3.1	The United Nations Convention on the Rights of the Child (UNCRC)	213
3.2	The African Charter on the Rights and Welfare of the Child (ACRWC)	215
4. CONCL	USION	217
CHAPTER 5		219
MEDICO-LE	GAL ASPECTS RELATING TO THE REGULATION OF SURROGATE MOTHERHOOD	219
1. INTRO	DUCTION	219
2. INFERT	FILITY AND THE BIOLOGICAL ABILITY TO REPRODUCE	222
2.1	Introduction	222
2.2	Advances in biotechnology	224
2.3	Assisted reproduction procedures	225
3. SURRC	OGACY VERSUS ADOPTION	234
4. ETHICA	AL, SOCIAL AND RELIGIOUS ISSUES	237
5. THE	ROLE OF EPIGENETICS	244
6. CONCL	USION	250
CHAPTER 6		251
	TIVE ANALYSIS OF SURROGACY REGULATION IN THE UNITED KINGDOM, CANADA	
	DUCTION	
	NITED KINGDOM	
2.1	Regulation of surrogate motherhood	
2.2	AB v DE	
2.2	CC v DD	
2.4	JP v LP and others	
2.5	Re A	
	лс А	
3.1	Regulation of surrogate motherhood	
3.1	B.A.N. v. J.H	
3.3	ы.а.п. v. ј.п	
4. INDIA.		

4.	.1 Regulation of surrogate motherhood	280
4.	.2 Baby Manji Yamada v. Union of India	
4.	.3 Jan Balaz vs. Anand Municipality and others	288
5. C	ONCLUSION	289
СНАРТ	ΓER 7	292
CONCL	LUSION AND RECOMMENDATIONS	292
1. IN	NTRODUCTION	292
2. A	NEED FOR LEGAL REFORM	292
3. RI	EGULATORY CLARITY WITH SOME CONSTITUTIONAL CAVEATS	293
4. LE	EGAL COMPARATIVE LESSONS	299
5. RI	ECOMMENDATIONS	300
BIBLIO	OGRAPHY	304
1. B	OOKS AND CONTRIBUTIONS IN BOOKS	304
2. JC	OURNAL ARTICLES	305
3.0	DLD AUTHORITIES	
4. LA	AW REPORTS	
So	outh Africa	
U	JK 313	
Ca	Canada 313	
5. LE	EGISLATION	
5.	.1 REGULATIONS AND ACTS: SOUTH AFRICA	
5.	.2 LEGISLATION: INTERNATIONAL	315
6.	. INTERNATIONAL DOCUMENTS/INSTRUMENTS	
7.	. REGIONAL DOCUMENTS/INSTRUMENTS	
8.	. REFERENCE WORKS	
9.	. MISCELLANEOUS AND OTHER DOCUMENTS	
10	0. WEBSITES AND INTERNET SOURCES	
ATT	ACHMENT 1:	
ETH	IICAL CLEARANCE	321
ATT	ACHMENT 2:	323
TUR	NITIN REPORT	323

CHAPTER 1

BACKGROUND TO THE STUDY, PROBLEM STATEMENT, OBJECTIVES AND FRAMEWORK

1. INTRODUCTION

Infertility is a very sensitive subject and it is also a reality which is part of so many people's lives, both women and men. Technological advances in the field of assisted reproduction have led to many possibilities for infertile persons to achieve parenthood. One such example is that of surrogate motherhood or surrogacy, which raises unique issues and rights, for example, those of society on the one hand, and those of persons on the other hand, for example the right to procreate, the right of women to make decisions regarding their bodies, and the right of children to establish their biological or genetic origins.¹ The difficulty lies in finding a balance that does justice to the expectations of society, the interests and desires of the infertile and same-sex couples and the best interests of the child.²

This thesis will consider, among others, the constitutionality of assisted reproduction and surrogate motherhood. The pertinent medico-legal issues regarding assisted reproduction and surrogacy will also be canvassed, especially with regard to the legal consequences following the artificial fertilisation of the surrogate mother. In order to adequately contextualise the topic of this thesis, selected aspects concerning infertility and the different forms of infertility treatment, including relevant human rights considerations regarding human reproduction, will be explored. Finally, the legality of surrogate motherhood agreements, and legal consequences of these, will also be considered.

The impact of infertility and the promises of surrogacy in resolving the impact of infertility, is aptly summarised by Khampepe J as follows:

¹ South African Law Commission Report, Project 65, Surrogate Motherhood (11 November 1992) at par 7.6.3.

² SALC Project 65 at par 7.6.3.

Infertility therefore has the capacity to detrimentally affect the psychological integrity and wellbeing of a person at least partly because it restricts their ability to make reproductive decisions, within the context of strong social expectations. In my view, much of the harm infertility brings has to do with the forced deprivation of choice in an area of life that humans consider particularly significant. Infertility cruelly dispossesses a person of the capacity to decide whether or not to have a child; where making this decision has extensive social implications. Surrogacy meaningfully contributes to ameliorating the harms of infertility because it provides a pathway through which infertile people can exercise their right to make reproductive decisions. Those who use surrogacy are able to elect to join the ranks of parenthood. The great benefit of surrogacy is that it opens reproductive avenues for those who would otherwise be unable to have children of their own. Surrogacy *itself* allows the infertile to ameliorate the psychological harms of infertility.³ (own emphasis)

Infertility can further be seen as an unenviable and psychologically harmful condition. The harm lies in the fact that it prevents a person from having his or her own children, which may furthermore result in serious social exclusion and stigma:⁴

The psychological trauma experienced by all infertile people is further heightened by an abiding sense of social shame, leading them to conceal their infertility. The stigma that attaches to infertility is damaging and pervasive, especially in developing countries like our own.⁵

Respect for another person's individual autonomy, if applied to the context of surrogacy, means that despite one's own preferences, the choice of another person to use the advances in medical technology to address infertility, including surrogate motherhood, should be respected as much as the choice of a fertile person to have children or not. These choices are closely related to the rights to dignity, equality, and the right to reproductive freedom, as this thesis will discuss. With the development of new technologies over the past few decades, it has become possible to reproduce by noncoital means and the right to engage in these new technologies is becoming more prevalent.⁶ Both artificial fertilisation and surrogacy are forms of noncoital reproduction.⁷ Modern surrogacy arrangements are possible today as a result of modern medical technology.⁸ Artificial fertilisation of the surrogate would thus not have been possible

³ AB and another v Minister of Social Development 2017 (3) BCLR 267 (CC); 2017 (3) SA 570 (CC) at par 89 and 90.

⁴ AB and another v Minister of Social Development par 86.

⁵ AB and another v Minister of Social Development par 84.

⁶ Van Niekerk, C "Assisted reproductive technologies and the right to reproduce under South African law" 2017 *Potchefstroom Electronic Law Journal* 20 1 and 2.

⁷ Van Niekerk 2017 *PELJ* 19.

AB and another v Minister of Social Development at par 37. Tager, L "Surrogate Motherhood, Legal Dilemma" 1986 South African Law Journal 103(3) 381. The writer states that we have been carried forward, willingly or otherwise, into a new era of genetics as a result of advances in biotechnology. The writer refers to the first baby which was conceived outside of the mother's womb and which was born in England in 1978. This was said to have been a revolution in the

without the said technology and it is important to regulate the procedure to protect the different parties and the child born as a result thereof.⁹ It is trite that with everchanging technology, the law is limping along and lagging behind technology.¹⁰ This is also the case with surrogate motherhood, as this thesis will attempt to illustrate.

2. BACKGROUND TO THE STUDY

Over the past decades, the idea of what 'a family' entails have changed dramatically with the moral changes brought about in society. Single parents, same-sex couples and commissioning parents can now all have a family through the services of a surrogate. By adopting a child or having a child through *in vitro* fertilisation using donor gametes, a family may be established, although there is no genetic link between the parents and the child. There is clearly no right or wrong form of a family. It is each person's choice of how he or she chooses to have a family. Meyerson makes the following important observation regarding parenthood:

Many reasonable people subscribe to more pluralistic ideas about good families. These opposing reasonable conceptions of family do not see biological parenthood as inherently superior to social parenthood or blood relationships as the foundation of our humanity. They see value in the social aspects of kinship and the many family forms, both biological and non-biological, that can provide children with love and care and meet their emotional needs.¹¹

2.1 Addressing infertility

Assisted reproduction, through artificial fertilisation for example, is a form of medical technology that enables the medical profession, more specifically fertility specialists and embryologists, to assist an infertile or fertile person in conceiving a child. Examples

human reproduction in that it was feasible to separate conception and gestation. Biko J and Nene Z "Ethics aspects of third-party reproduction" 2017 *Obstetrics & Gynaecology Forum* 3 12.

⁹ It is important to understand that it is possible for a child conceived by *in vitro* fertilisation to have no less than five (5) parents. Tager 1986 *SALJ* 382. The different parents are the genetic parents (the donors of the sperm and the egg), the woman in whose womb the fertilised embryo was implanted and the commissioning parents (the childless couple who commission the surrogate to bear the child which they intend to rear as their own). It is pointed out by the writer that procreation has been divorced from sexual intercourse and from child-bearing.

¹⁰ Denton, JA and Reynolds CS. "Limping along and lagging behind: The law and emerging gene technologies" (2018) 24 *James Cook University Law Review* 61.

¹¹ Meyerson, D^{*} "Surrogacy, Geneticism and Equality: The case of AB Minister of Social Development" 2019 *Constitutional Court Review* 9 340.

would be where the husband or partner of the women is infertile, or in the case of lesbian couples, where a sperm donor is necessary for the process of conception, or where a woman needs to undergo insemination or *in vitro* fertilisation as she is unable to conceive naturally. Infertility may be viewed from two different perspectives, namely, where a person is what is referred to as "conception infertile" or "pregnancy infertile". "Conception infertile" refers to the situation where a person is unable to contribute a gamete for the purposes of conception through artificial fertilisation, whereas "pregnancy infertile" applies to a person who is permanently and irreversibly unable to carry a pregnancy to term.¹² No matter the type of infertility that the person may suffer from, the longing to have a child is the same.

"Assisted reproductive technologies" (ART) are defined as all treatments or procedures that include the *in vitro* handling of both human oocytes and sperm, or embryos, for the purpose of establishing a pregnancy.¹³ The Centres for Disease Control and Prevention (CDC) describe assisted reproductive technologies (ART) broadly as all fertility treatments in which either eggs or embryos are handled.¹⁴ It is further explained that, in general, the assisted reproductive procedures involve the surgical removal of eggs from a woman's ovaries, combining them with sperm in the laboratory, whereafter they are returned to the woman's body or donated to another woman.¹⁵

ART's, however, do not include treatments in which only sperm are handled, for example artificial insemination, or procedures in which a woman only takes medicine to stimulate her egg production without the intention of having her eggs retrieved.¹⁶ "Artificial fertilisation" in South African law is defined as the introduction by other than natural means of a male gamete or gametes into the internal reproductive organs of a

¹² AB and another v Minister of Social Development at par 8, fn 9 and fn 205. Florescu, S and Sloth-Nielsen, J "Visions on surrogacy – From North to South: The approach of the Netherlands and South Africa to the issue of surrogacy and the child's right to know his origin" 2017 International Survey of Family Law 2017 247. Meyerson, D "Surrogacy, Geneticism and Equality: The case of AB Minister of Social Development" 2019 Constitutional Court Review 9 318.

¹³ Van Niekerk 2017 *PELJ* 4.

¹⁴ What is assisted reproductive technology? accessed from <u>https://www.cdc.gov/art/whatis.html</u> (accessed 1 July 2022).

¹⁵ What is assisted reproductive technology? accessed from <u>https://www.cdc.gov/art/whatis.html</u> (accessed 1 July 2022).

¹⁶ What is assisted reproductive technology? accessed from <u>https://www.cdc.gov/art/whatis.html</u> (accessed 1 July 2022).

female person for the purpose of human reproduction.¹⁷ Artificial fertilisation includes artificial insemination (with the use of donor sperm or sperm of the husband or male partner) and *in vitro* fertilisation.¹⁸ Legislative change in South Africa in 1997 made it possible for both married and unmarried women to have legal access to donor sperm and to subject themselves to artificial insemination procedures.¹⁹

2.2 Fundamental rights considerations

The Bill of Rights, set out in chapter 2 of the Constitution of South Africa, 1996,²⁰ is the cornerstone of democracy in South Africa. In terms of section 7(1), it enshrines the rights of all people and affirms the democratic values of human dignity, equality and freedom. Section 7(2) provides that the state must respect, protect, promote and fulfil the rights in the Bill of Rights. Section 7(3) provides that the rights in the Bill of Rights are subject to the limitations contained or referred to in section 36, or elsewhere in the Bill. Every person has the right to make decisions concerning reproduction in terms of section 27(1)(b) of the Constitution. Section 12(2)(a) gives every person the right to bodily and psychological integrity, which includes the right to make decisions concerning reproduction (section 12(2)(b)).

The advances in medical technology can assist the infertile or fertile person in more than one way and selecting the type of assistance is a choice that a person is entitled to make regardless of his or her sexual orientation, sex or gender. The recent Constitutional Court judgment of *AB and another v Minister of Social Development*,²¹ discussed in detail in this thesis, concerned the constitutionality of section 294 of the Children's Act, and more specifically, the requirement of a genetic link between the commissioning parent or parents and the commissioned child. In this case, the applicants challenged section 294 of the Children's Act which requires the commissioning parents to use their own gametes for the conception of a child, instead

¹⁷ Reg 1 of Regulations relating to artificial fertilisation of persons published under GN R175 in *GG* 35099 of 2 March 2012. Sec 1 of the Children's Act 38 of 2005.

¹⁸ Reg 1 of Regulations relating to artificial fertilisation of persons 2012.

¹⁹ GN R1354 GG 18362 of 17 October 1997. Reg 8(1) of the Human Tissue Act 1983 was deleted and regs 5(d) and 9(e) was amended. Strauss and Maré *Mediese Reg* 164 and 166.

²⁰ Constitution of South Africa ch 2, sec 7.

²¹ 2017 (3) SA 570 CC; (2017 (3) BCLR 267; [2016] ZACC 43).

of donor gametes. The majority of the Constitutional Court rejected the challenge on the ground that a child has the right to know his or her genetic origins, as this is important to a child's development of a positive self-identity. This judgment is also controversial for other reasons. For example, with reference to the different categories of infertility, section 294 prevents "conception infertile" persons from concluding a surrogacy agreement.²² Meyerson refers to this differentiation as the "no-double-donor requirement"²³ and furthermore argues that, because infertility is generally regarded as a form of disability and whilst disability is a specified ground listed in section 9(3) of the Constitution,²⁴ section 294 of the Children's Act may be said to unfairly differentiate on the basis of disability.²⁵ She states that the court should have found that section 294 is discriminatory and presumptively unfair in terms of section 9(5) of the Constitution, thereby infringing section 9(3).²⁶

It is clearly the law and not the person's medical condition that prevents the category of conception infertile person(s) from having a child through surrogacy.²⁷ Moreover, to exclude conception infertile persons from using surrogacy as a form of reproduction has a highly detrimental impact on them and section 294 may be said to impair the dignity of conception infertile people, not to mention that it is unjustifiably unfair to them.²⁸ Meyerson contends that although section 294 does not impose a burden on a person, it withholds an opportunity, and by withholding an opportunity, it is also indirectly discriminatory.²⁹

Discussing section 294 of the Children's Act, Van Niekerk argues with reference to section 12(2)(a) of the Constitution, that whilst the said section protects the right to

²² Meyerson, D "Surrogacy, Geneticism and Equality: The case of AB Minister of Social Development" 2019 Constitutional Court Review 9 318. 23 Meyerson, D "Surrogacy, Geneticism and Equality: The case of AB Minister of Social Development" 2019 Constitutional Court Review 9 318. 24 Meyerson, D "Surrogacy, Geneticism and Equality: The case of AB Minister of Social Development" 2019 Constitutional Court Review 9 321. 25 Meyerson, D "Surrogacy, Geneticism and Equality: The case of AB Minister of Social Development" 2019 Constitutional Court Review 9 321. 26 Meyerson, D "Surrogacy, Geneticism and Equality: The case of AB Minister of Social Development" 2019 Constitutional Court Review 9 321 and 341. 27 Meyerson, D "Surrogacy, Geneticism and Equality: The case of AB Minister of Social Development" 2019 Constitutional Court Review 9 322. 28 Meyerson, D "Surrogacy, Geneticism and Equality: The case of AB Minister of Social Development" 2019 Constitutional Court Review 9 325. 29 Meyerson, D "Surrogacy, Geneticism and Equality: The case of AB Minister of Social Development" 2019 Constitutional Court Review 9 325.

noncoital reproduction, it does so only if the parties themselves are physically involved in the reproductive process (presumably by providing gametes) or by carrying the child.³⁰ Thus, this effectively excludes the persons who cannot meet the criteria in terms of section 294 of the Children's Act.³¹ It can then be said to suggest that section 12(2)(a) is not flexible enough to accommodate medical technological advancements and it may thus need to be amended or interpreted more broadly in future.³² As section 12(2) of the Constitution refers to both bodily and psychological integrity, she regards section 12(2) as protecting both the body and the psyche.³³ Hence, it is possible to inflict harm in a way that infringes both a person's bodily integrity and psychological integrity, however, it is also possible to compromise a person's psychological integrity without an accompanying infringement of the person's bodily integrity.³⁴

2.3 Children born as result of assisted reproduction

Assisted reproduction has far-reaching consequences for children born as a result of assisted reproduction, for a number of reasons. One such consequence relates to the situation where legal ties will exist between children born as a result of the assisted procedure and the persons with whom these children will have a legal connection and why. This thesis will explore the different factors that need to be considered to determine the legal consequences of and the legal ties arising between the different parties and the child or children born as a result of artificial fertilisation.

Less considered, however, is the fact that children born as a result of surrogacy arrangements are also influenced by the science of epigenetics, which refers to an understanding of "the way the genome integrates environmental signals and alters the expression of genes as a result". ³⁵ Epigenetic research shows that the genetic makeup of a surrogate has a substantial impact on the child that she carries, irrespective of whether she is an ovum donor or not. Whilst the two biological parents of the child contribute to their child's genetic make-up, the manner in which their child's genes are

³⁰ Van Niekerk 2017 *PELJ* 15.

³¹ Van Niekerk 2017 *PELJ* 15.

³² Van Niekerk 2017 *PELJ* 16.

³³ Van Niekerk 2017 *PELJ* 15.

³⁴ Van Niekerk 2017 *PELJ* 16.

³⁵ Nicholson, S and Nicholson, C "I used to have two parents and now I have three? When science (fiction) and the law meet: Unexpected complications" 2016 *Medicine and Law* 35 426.

"switched on" in the womb, is substantially influenced by the surrogate mother's transcription factors. This third-party impact has led Nicholson and Nicholson to argue that where it was previously believed that the sperm and ovum donors were the two biological parents of the child, in the case of surrogacy, the womb of the surrogate mother itself should be considered to make her a "third biological parent" whose input is experienced both in utero and after birth.³⁶ Thus, even where the surrogate mother does not provide the egg for the commissioning parent's child that she carries, Nicholson and Nicholson believe that the surrogate mother's physical contribution obscures to some extent questions regarding in whom parental rights and responsibilities vis-à-vis the child that is born should vest. They recommend that if the prenatal womb environment has this significant impact on the health and welfare of the child after birth, a full medical and environmental history of the surrogate should be required.³⁷ This new biological discovery adds a further layer of complexity to the legal consequences of surrogacy that justifies consideration and possible reference in regulations to be drafted regarding surrogate motherhood in terms of the Children's Act.

Section 28 of the Constitution of South Africa captures the best interests of the child and provides that the child's best interest is of paramount importance in all matters concerning the care, protection and well-being of the child.³⁸ It is, however, not easy to identify the best interests of a commissioned child in surrogacy agreements. Such an exercise will require a thorough consideration of several social, ethical, financial and medical issues, including significant choices regarding the future of an unborn child.³⁹

³⁶ Nicholson, S and Nicholson, C "I used to have two parents and now I have three? When science (fiction) and the law meet: Unexpected complications" 2016 *Medicine and Law* 35 426.

³⁷ Nicholson, S and Nicholson, C "I used to have two parents and now I have three? When science (fiction) and the law meet: Unexpected complications" 2016 *Medicine and Law* 35 434.

³⁸ Constitution of South Africa. *AB and another v Minister of Social Development* 2017 (3) SA 570 (CC) par 294, 253 and 281 for acknowledging the paramountcy of the best interests of the commissioned child in respect of the genetic link-requirement in surrogacy agreements. Florescu, S and Sloth-Nielsen, J "Visions on surrogacy – From North to South: The approach of the Netherlands and South Africa to the issue of surrogacy and the child's right to know his origin" 2017 *International Survey of Family Law* 2017 251.

³⁹ Oluwaseyi, O.O. and Oladimeji, O "Surrogacy agreements and the rights of children in Nigeria and South Africa" 2021 *Obiter* 42(1) 21.

The Children's Act provides in section 6(1)(a) guidance on the implementation of legislation relevant to children,⁴⁰ complemented by section 6(1)(b) which requires that certain principles guide all proceedings, actions and decisions by any organ of state in any matter concerning a child or children in general.⁴¹ Section 7 lists some of the factors which must be taken into account when determining the best interests of the child.⁴² Moreover, section 9 determines that the best-interests standard concerning a child must be applied *in all matters* concerning the care, protection and well-being of a child.⁴³ These provisions should be considered in the context of section 28 of the Constitution when a decision has to be made in respect of legal consequences of children born as a result of artificial fertilisation and surrogacy. The best interests of the child principle should thus guide courts not only in confirming surrogacy agreements, but also in respect of disputes arising from surrogacy.

2.4 Factors leading to the regulation of surrogate motherhood

The most common reason for enlisting the services of a surrogate is that the couple want to have children but the woman is either unwilling or unable to become pregnant or carry a pregnancy to term.⁴⁴ Further, an increasing number of young women are infertile today and the supply does not meet the demand for children that are available for adoption.⁴⁵ More unmarried mothers appear to be raising their own children and thus not giving them up for adoption as used to be the position in the past.⁴⁶ It can be said that these changes, together with the greater demands by society, gave rise to the remarkable progress that has been made in the medical field to overcome infertility.⁴⁷ Surrogacy can be described as another technique to provide a childless

⁴⁰ Sec 6(1)(a) of the Act. Florescu, S and Sloth-Nielsen, J "Visions on surrogacy – From North to South: The approach of the Netherlands and South Africa to the issue of surrogacy and the child's right to know his origin" 2017 *International Survey of Family Law* 2017 251.

⁴¹ Sec 6(1)(b) of the Act. Florescu, S and Sloth-Nielsen, J "Visions on surrogacy – From North to South: The approach of the Netherlands and South Africa to the issue of surrogacy and the child's right to know his origin" 2017 International Survey of Family Law 2017 251.

⁴² Sec 7 of the Act. Florescu, S and Sloth-Nielsen, J "Visions on surrogacy – From North to South: The approach of the Netherlands and South Africa to the issue of surrogacy and the child's right to know his origin" 2017 International Survey of Family Law 2017 251.

⁴³ Sec 9 of the Act.

⁴⁴ SALC Project 65 at par 2.1.1.

⁴⁵ SALC Project 65 at par 2.1.1.

⁴⁶ SALC Project 65 at par 2.1.1.

⁴⁷ SALC Project 65 at par 2.1.1.

couple with a child (who is in most instances biologically related to at least one of the persons) through artificial fertilisation.⁴⁸

As this thesis will discuss in detail, that despite the advances in medical technology facilitating the birth of a child through assisted reproduction procedures, numerous legal consequences associated with assisted reproduction arise, and even more so in the case of surrogacy. It is thus extremely important for each party to the surrogate motherhood agreement to fully understand the attendant legal consequences, including the relevant rights and duties of each of the parties, and finally, the legal status of the commissioned child born as a result of the said agreement.

Because of the complexity of surrogacy, difficult ethical, philosophical and social issues arise.⁴⁹ Already in 1993, the SALC pointed out that in a country like South Africa with different population and religious groups, ethical values of one group in the community may not necessarily be acceptable to other groups.⁵⁰ This observation foregrounds the importance of legislation and regulations to guide parties to a surrogacy agreement in light of the fact that not everyone will agree with a person's choice to use surrogacy as an arrangement to have his or her or their own child.

3. PROBLEM STATEMENT, AIMS AND OBJECTIVES OF THE RESEARCH

It follows from the discussion above that contextualised the legal problem regarding the regulation of surrogate motherhood and artificial fertilisation, that several gaps and pertinent questions arise in the legal framework governing surrogacy and related consequences in South Africa. These issues not only include medico-legal and legal status concerns, but also complex fundamental rights issues affecting the rights and obligations of each of the parties to surrogacy arrangements, not to mention those of the commissioned children born as a result of these arrangements.

An interrogation of the human rights aspects relating to surrogate motherhood requires a critical review of the constitutionality of chapter 19 of the Children's Act, which

⁴⁸ SALC Project 65 at par 2.1.1.

⁴⁹ SALC Project 65 at par 2.1.3.

⁵⁰ SALC Project 65 at par 2.2.2.

regulates surrogate motherhood in South Africa. This will include an examination of the rights and duties of parties to the surrogate agreement, including the legal consequences following legal and illegal surrogate motherhood agreements. Relevant case law concerning surrogacy will also be considered in the attempt to identify legal lacunae and constitutional inconsistencies within the existing legal framework. Where relevant, recommendations will be suggested that may hopefully assist in closing some of the identified gaps.

The reasons why this study is important, are manifold: firstly, because despite the ostensible clear statutory framework governing surrogate motherhood (which includes artificial fertilisation), the brief contextual background above has alluded to a few areas of concern, which the thesis will explore in detail. Secondly, the legal consequences of the identified legal ambiguities or lacunae are significant, in that they either relate to important human rights considerations, or significantly affect the legal position and status of parties to the surrogacy arrangements. Thirdly, recommendations should be proposed to address these identified gaps and inconsistencies, which this thesis will attempt to do.

How will the thesis approach the research question in this thesis? This study will first canvass the legal historical path leading to the promulgation of chapter 19 of the Children's Act in order to pinpoint the legal considerations prompting this development. This overview will be followed by a close analysis of the existing legal framework regulating surrogacy and artificial fertilisation in order to identify the gaps and inconsistencies in the framework. A consideration of the constitutional framework and relevant rights pertaining to parties in surrogacy arrangements follows next, including an analysis of the constitutionality of controversial provisions in the Children's Act. Where relevant, comparable provisions in the United Kingdom, Canada and India will also be assessed, with the view that these may aid in providing guidance on how some of the regulatory gaps in the South African legal framework may be addressed or closed.

4. RESEARCH APPROACH AND METHODOLOGY

The methodology which I will follow will consist of a desk-top literature study, which includes a legal comparative component which will draw on legal comparisons with India, the United Kingdom (UK) and Canada with the hope that these may clarify or address some of the unresolved or ambiguous issues relating to assisted reproduction and surrogate motherhood in South Africa. The rationale for including the UK, Canada and India is based on the fact that the UK, Canada and India, similar to South Africa, ban any form of commercial surrogacy but allow altruistic surrogacy. The UK, Canada and India observe a human rights agenda, similar to the position in South Africa. Legal scholarly articles, relevant textbooks, case law and internet sources relevant to the topic will be considered. This will consist of primary sources of law (such as statutes, cases, regulations) and secondary authorities (for example, law reviews, legal

dictionaries, legal treaties, scholarly articles and textbooks).

5. LIMITATIONS

This thesis will not include a discussion on commercial surrogacy or the possibility thereof for South Africa in the future. The regulation of surrogacy in international law will only be briefly alluded to. The thesis will also not delve into a discussion on the various options of assisted reproduction but will focus on artificial fertilisation only. The discussion on pertinent human rights will be limited to those most relevant to the regulation of surrogacy and artificial fertilisation only.

6. FRAMEWORK OF THE THESIS

The thesis is structured in a manner that best supports a logical arrangement of the contents to address the research question. Following this chapter, chapter 2 discusses the historical legal position with regards to assisted reproduction, especially surrogacy, as well as the legislative changes that shaped the current regulatory system governing surrogate motherhood. Chapter 3 focuses on the current legal framework in South Africa with specific reference to chapter 19 of the Children's Act and the changes that this Act brought to the regulation of surrogacy.

status of the commissioned child born of surrogacy or artificial fertilisation, including the issue of access to information about a child's genetic origin and the right to know one's biological origins. Relevant case law pertaining to the confirmation of surrogacy agreements will also be considered, especially those in which the courts provided guidance on the implementation of chapter 19 of the Children's Act. This chapter will also turn to the rights of the child born of assisted reproductive technology. Chapter 4 details the constitutional framework relevant to surrogacy, with reference to case law, and will focus on those rights of parties involved in surrogacy arrangements. Chapter 5 consists of an examination of the medico-legal aspects regarding surrogacy and artificial fertilisation. The ability to reproduce, advances in biotechnology and aspects surrounding assisted reproduction will be discussed with reference to the National Health Act⁵¹ and the Regulations relating to the artificial fertilisation of persons.⁵² Chapter 6 consists of a comparative study of relevant provisions of the United Kingdom, Canada and India, with the aim of identifying solutions for gaps identified in the South African legal framework. Chapter 7 concludes with a brief summary of the key issues originating from the thesis, as well as the recommendations made with regard to the identified gaps.

7. DEFINITIONS

In order to promote a clear understanding of terminology employed in this thesis, the key terms relating to surrogate motherhood and assisted reproduction need to be defined at the outset.

7.1 Reproductive health care:

Reproductive health care refers to those health services connected with reproduction and the right to make decisions concerning reproduction.⁵³

7.2 Relevant health service:

⁵¹ 61 of 2003 (Hereinafter referred to as NHA).

⁵² Government Notice R175 in Government Gazette 35099 of 2 March 2012.

⁵³ Carstens P and Pearmain D *Foundational Principles of South African Medical Law* (LexisNexis 2007) 178.

The description by Carstens and Pearmain below captures the meaning of relevant

health service well:

Relevant health services not only mean medical treatment but also relates to health education and information. Since the decision to reproduce implies the right to decide not to reproduce, reproductive health care must include services concerning contraception and termination of pregnancy as much as it includes services relating to fertility, conception and giving birth.⁵⁴

7.3 Artificial fertilisation and artificial insemination:

Section 1 of the Children's Act defines artificial fertilisation as:

The introduction, by means other than natural means, of a male gamete into the internal reproductive organs of a female person for the purpose of human reproduction, including-

(a) the bringing together of a male and female gamete outside the human body with a view to placing the product of a union of such gametes in the womb of a female person; or
(b) the placing of the product of a union of male and female gametes which have been brought together outside the human body, in the womb of a female person.⁵⁵

Artificial fertilisation includes artificial insemination, *in vitro* fertilisation⁵⁶, gamete intrafallopian tube transfer, embryo intrafallopian transfer or intracytoplasmic sperm injection.⁵⁷ Artificial insemination is more narrowly defined as the placing of male gametes (sperm) into the female reproductive tract by means other than copulation.⁵⁸

7.4 Assisted reproductive technology (ART):

The following two definitions define ART narrowly and broadly:

[A]RT includes all fertility treatments in which either eggs or embryos are handled. In general, ART procedures involve surgically removing eggs from a woman's ovaries,

⁵⁴ Carstens and Pearmain *Foundational Principles* 178.

⁵⁵ Sec 1 of Children's Act 38 of 2005. Report of the Ad Hoc Committee on Report of South African Law Commission on Surrogate Motherhood (11 February 1999) 6. Dhai, A and McQuoid-Mason D Bioethics, Human Rights and Health Law, Principles and Practice (2 ed Juta & Co 2020) 142.

⁵⁶ Strauss SA *Doctor, Patient and the Law* (3 ed JL Van Schaik 1991) 188 and 189. Strauss, SA "Triplets to a Surrogate Grandmother in South Africa: Legal Issues" 1989 *International Legal Practitioner* 14(3) 70. Heaton, J "The Pitfalls of international surrogacy: A South African family law perspective" 2015 *Journal for Contemporary Roman-Dutch Law* 78(1). Heaton states that artificial fertilisation is defined so broadly that it includes all forms of surrogacy.

⁵⁷ Reg 1 of Regulations relating to artificial fertilisation of persons 2012. Intracytoplasmic sperm injection is defined as the process of microscopic technology to bring about fertilisation of an ovum with a male sperm outside the body in an authorised institution. Jordaan, DW "A Constitutional critique on the regulations relating to artificial fertilisation of persons" 2017 *South African Journal of Bioethics and Law* 10(1) 29.

⁵⁸ Reg 1 of Regulations relating to artificial fertilisation of persons 2012. Strauss and Maré *Mediese Reg* 169.

combining them with sperm in the laboratory, and returning them to the woman's body or donating them to another woman. 59

All treatments or procedures that included the in vitro handling of human oocytes and sperm or embryos for the purpose of establishing a pregnancy.⁶⁰

ART does not include intrauterine - or artificial insemination.⁶¹

7.5 In vitro fertilisation (IVF) and "test-tube babies":

Simply put, IVF refers to fertilisation in a glass.⁶² The Regulations relating to artificial fertilisation of persons define IVF as "[t]he process of spontaneous fertilisation of an ovum with a male sperm outside the body in an authorised institution."⁶³ The skill lies in conceiving a child *in vitro* and then successfully implanting the resulting blastocyst into the uterus of the mother/surrogate mother.⁶⁴ IVF thus is the uniting of the male and female gametes outside the woman's body in a test-tube and then transplanting the resulting the resulting the resulting the is loosely described by lay persons as a "test-tube baby".

The Ad Hoc Committee on Surrogate Motherhood defined IVF is as follows:

The placing of the product of a union of a male and female gamete or gametes, which have been brought together outside the human body, in the womb of a female person for the purpose of human reproduction.⁶⁶

⁵⁹ What is assisted reproductive technology? accessed from <u>https://www.cdc.gov/art/whatis.html</u> (accessed 4 January 2023). Biko J and Nene Z "Ethics aspects of third-party reproduction" 2017 *Obstetrics & Gynaecology Forum* 3 12.

⁶⁰ Van Niekerk 2017 PELJ 4.

⁶¹ What is assisted reproductive technology? accessed from <u>https://www.cdc.gov/art/whatis.html</u> (accessed 4 January 2023). Van Niekerk 2017 *PELJ* 5.

⁶² Tager 1986 SALJ 381. Tager describes the glass as a petri-disch, which is a shallow, saucershaped dish. Strauss and Maré *Mediese Reg* 169.

⁶³ Reg 1 of Regulations relating to artificial fertilisation of persons 2012. Reg 1, of the proposed Regulations relating to assisted conception of persons published under Government Gazette No 44321 on 25 March 2021, defines in vitro fertilisation as "the process of fertilisation of an egg with a sperm outside the body in an authorised institution by a competent person."

⁶⁴ LexisNexis *Family Law service* at par J115.

⁶⁵ Strauss *Doctor, Patient and the Law* 188. Strauss further explains that a zygote is defined as the product of the union of a male and female gamete outside the human body. Mason JK, McCall Smith RA and Laurie GT *Law and Medical Ethics* 6th ed (Butterworths 2002) 84. Palm MT and Hirsch HL "Infertility and Sterility: Legal Implications of artificial conception" 1982 *Medicine and Law* 1 48. Tager 1986 *SALJ* 382. South African Research Council: *Guidelines on Ethics for Medical Research* 2nd Book (Reproductive Biology and Genetic Research) 2002 2. There is consensus that there is no moral problem inherent in using this technique in circumstances where gametes from the husband and the wife are used. Strauss and Maré *Mediese Reg* 168. "Oorplasing van 'n bevrugte eier (ovum) vanaf "moeder" A na "gasvrou" B".

⁶⁶ Report of the Ad Hoc Committee 6.

7.6 Embryo transfer:

Embryo transfer refers to the process where an embryo (human offspring in the first eight weeks from conception)⁶⁷ is placed into the uterus or fallopian tube of the recipient.⁶⁸

7.7 Oocyte:

An oocyte refers to the female gamete;⁶⁹ a developing human egg cell,⁷⁰ or more specifically, it is described as a female germ cell involved in reproduction. It is also known as "an immature ovum, or egg cell."⁷¹

7.8 Sperm:

Sperm refers to the human male gamete.⁷² It is however important to distinguish between the terms sperm and semen as they differ although they are used interchangeably.⁷³

Sperm is a motile reproductive cell, not visible with the naked eyes, which is transferred to the female reproductive system during sexual intercourse. During the fertilization process, the sperm's nucleus fuses with the egg's nucleus to form an embryo.⁷⁴

⁶⁷ Sec 1 of the National Health Act 61 of 2003.

⁶⁸ Reg 1 of Regulations relating to artificial fertilisation of persons 2012. Mason, McCall Smith and Laurie *Law and Medical Ethics* 84. The authors draw a distinction between IVF and embryo transfer in that he states popular usage equites IVF with standard treatment for childlessness due to blockage of the fallopian tubes and embryo transfer with the implantation of an embryo which has no genetic relationship either to the recipient or to her husband.

⁶⁹ Reg 1 of Regulations relating to artificial fertilisation of persons 2012. Biko J and Nene Z "Ethics aspects of third-party reproduction" 2017 *Obstetrics & Gynaecology Forum* 3 12.

⁷⁰ Sec 1 of the National Health Act 61 of 2003.

⁷¹ The definition can be accessed from <u>https://fertilitypedia.org/edu/reproductive-cells/oocyte#/Fertilitypedia-description</u> (accessed 4 January 2023). Section 1 of the Human Tissue Act 65 of 1983 defines a gamete as either of the two generative cells essential for human reproduction, which is the male sperm and the female ovum.

⁷² Reg 1 of Regulations relating to artificial fertilisation of persons 2012. Reg 1, of the proposed Regulations relating to assisted conception of persons published under Government Gazette No 44321 on 25 March 2021, defines sperm as "the male gamete procured for the purpose of assisted conception."

⁷³ Semen vs Sperm <u>http://www.andrologycenter.in/2018/10/24/semen-vs-sperm/</u> (accessed on 4 January 2023).

⁷⁴ The definition can be accessed from <u>http://www.andrologycenter.in/2018/10/24/semen-vs-sperm/</u> (accessed on 4 January 2023).

Semen is the seminal fluid which contains the sperm cells along with other plasma liquid. Semen is responsible for transmission of the motile sperm cells into the reproductive organ of the female.⁷⁵

7.9 Surrogate mother/surrogate/surrogate motherhood:⁷⁶

Multiple definitions exist for the above terms. A surrogate mother is described as "[a]n adult woman who enters into a surrogate motherhood agreement with the commissioning parent."⁷⁷ A surrogate is a voluntary recipient of an embryo who will carry such embryo to birth for contractual parents;⁷⁸ one woman who bears a child for another;⁷⁹ or the woman delivering the commissioned child.⁸⁰ "Surrogate mother" refers to the woman who bears the child rather than to the woman who rears the child."⁸¹ The Ad Hoc Committee on Surrogate Motherhood describes the surrogate mother as:

[a]ny competent woman who bears a child or children for another person or persons (the commissioning parent or parents), as the result of an agreement to this effect entered into prior to the conception of the child or children.⁸²

7.10 Commissioning parents:

- ⁸¹ SALC Project 65 at par 2.1.1.
- ⁸² Report of the Ad Hoc Committee 6.

⁷⁵ The definition can be accessed from <u>http://www.andrologycenter.in/2018/10/24/semen-vs-sperm/</u> (accessed on 4 January 2023).

⁷⁶ Throughout the thesis reference will be made to surrogacy although both surrogate motherhood and surrogacy may be used interchangeably.

⁷⁷ Sec 1 of the Children's Act 38 of 2005.

⁷⁸ Reg 1 of Regulations 2012. Mason, McCall Smith and Laurie *Law and Medical Ethics* 96. The authors describe surrogate motherhood as "requires the active co-operation of an otherwise uninvolved woman in the process of pregnancy and birth. It thus introduces a third party into marriage." Tager 1986 *SALJ* 383. Canner, S "Navigating surrogacy law in the non-united states: Why all states should adopt a uniform surrogacy statute" 2019 *Journal of Civil Rights and Economic Development* 33(2) 117. SALC Project 65 at par 2.1.1.

⁷⁹ Nicholson, C "When moral outrage determines a legal response: Surrogacy as labour" 2013 *South African Journal on Human Rights* 29(3) 497. At 499 the author explains that other terms are also being used to describe a surrogate which are 'hostess mother', 'host mother', 'renting a womb' or 'plumbing'. Nene Z "Ethics aspects of third-party reproduction" 2017 Obstetrics & *Gynaecology Forum* 3 13. The authors explain that "third party reproduction or donor assisted reproduction is a form of reproduction in which DNA or gestation is provided by a third party or donor other than the two intended parents who will raise the resulting child." The gestational carrier will be known to the commissioning parents but it is possible that the sperm, egg or embryo donor may be anonymous. Nicholson, C and Bauling, A "Surrogate motherhood agreements and their confirmation: A new challenge for practitioners?" 2013 *De Jure* 46(2) 510. The word surrogate means substitute.

⁸⁰ Oluwaseyi and Oladimeji 2021 *Obiter* 21.

The commissioning parents are the persons who enter into a surrogate motherhood agreement with a surrogate mother.⁸³ They are thus the couple to whom the surrogate will hand over the commissioned child after she has given birth.⁸⁴ Spouses, civil union partners, life or permanent partners of the same or the opposite sex and a single person are all eligible to be commissioning parents.⁸⁵

7.11 Children:

A child is a person under the age of 18 years.⁸⁶

7.12 Surrogate motherhood agreement:

The Children's Act defines this as follows:

An agreement between a surrogate mother and a commissioning parent in which it is agreed that the surrogate mother will be artificially fertilised for the purpose of bearing a child for the commissioning parent and in which the surrogate mother undertakes to hand over such a child to the commissioning parent upon its birth, or within a reasonable time thereafter, with the intention that the child concerned becomes the legitimate child of the commissioning parent.⁸⁷

7.13 Surrogacy:

Broadly, surrogacy refers to an arrangement whereby a woman agrees to become pregnant and deliver a child for a contracted party.⁸⁸ This may be done to overcome infertility because it gives couples and single parents a method of conceiving a child with genetic connections to their family.⁸⁹ A narrower definition describes it as follows:

⁸³ Sec 1 of the Children's Act 38 of 2005. Report of the Ad Hoc Committee 6.

⁸⁴ Oluwaseyi and Oladimeji 2021 *Obiter* 21.

⁸⁵ Heaton 2015 *THRHR* 29.

⁸⁶ Bosman-Sadie, H and Corrie, L *Practical approach to the new Children's Act* (Van Schaik 2012) 5.

⁸⁷ Sec 1 of Children's Act 38 of 2005. Report of the Ad Hoc Committee 6.

⁸⁸ Oluwaseyi and Oladimeji 2021 *Obiter* 21. Nicholson and Bauling 2013 *De Jure* 510. Dhai and McQuoid-Mason *Bioethics, Human Rights and Health Law* 144.

⁸⁹ Iannacci,B.R "Why New York should legalize surrogacy: A comparison of surrogacy legislation in other states with current proposed surrogacy legislation in New York" 2018 *Touro Law Review* 34(4) 1239. The author explains that surrogates are commonly used by couples and single women who are unable to conceive their own genetic children (women who have had their uterus removed or have a uterus that cannot bear a child) and male couples who wish to have a child which will have a genetic connection to one or both partners.

Where a woman other than the husband's wife is impregnated with the husband's semen by way of artificial insemination and the child when born is handed over to the commissioning parents. 90

7.14 Gestational surrogacy:

Gestational surrogacy is an arrangement that utilises *in vitro* fertilisation to fertilise an egg; which usually belongs to the commissioning woman; outside of the surrogate mother's body whereafter the fertilised egg is then implanted in the surrogate mother's uterus.⁹¹ This type of surrogacy is also known as full surrogacy.⁹²

7.15 Traditional surrogacy:

In the case of traditional surrogacy, the surrogate mother provides the egg and carries the baby.⁹³ This is also known as partial surrogacy.⁹⁴ The surrogate is artificially inseminated with the sperm of the commissioning husband or that of a donor.⁹⁵

7.16 Gamete:

Gametes refer to either of two generative cells (male or female) essential for human reproduction.⁹⁶

7.17 Gamete donor:

⁹⁰ LexisNexis Law of South Africa 30(2) at par 63. LexisNexis Family Law service at par J107.

⁹¹ Hisano, EY "Gestational surrogacy maternity disputes: Refocusing on the child" 2011 *Lewis & Clark Law Review* 15(2) 520; Strauss *Doctor, Patient and the Law* 188. Strauss explains it as the situation where the genetic material (sperm and ova) is provided by the commissioning couple and the surrogate merely acts as a hostess ("gestational" or "birth mother"). Iannacci,B.R "Why New York should legalize surrogacy: A comparison of surrogacy legislation in other states with current proposed surrogacy legislation in New York" 2018 *Touro Law Review* 34(4) 1247. The gestational mother is not the biological mother of the commissioned child. SALC Project 65 at par 2.1.2. Report of the Ad Hoc Committee 7. The surrogate is only the gestational mother.

⁹² Tager 1986 SALJ 383. Full surrogacy is discussed in detail in par 2.1.2, ch 2 below.

⁹³ Hisano, EY "Gestational surrogacy maternity disputes: Refocusing on the child" 2011 Lewis & Clark Law Review 15(2) 520.

⁹⁴ Strauss *Doctor, Patient and the Law* 188. Partial surrogacy is when the surrogate mother is artificially inseminated with semen from the commissioning husband. Tager 1986 *SALJ* 383. Partial surrogacy is discussed in detail in par 2.9.1 below.

⁹⁵ SALC Project 65 at par 2.1.2. Report of the Ad Hoc Committee 7. The surrogate is both the genetic and gestational mother.

⁹⁶ Sec 1 of the Children's Act 38 of 2005; Sec 1 of the National Health Act 61 of 2003; Strauss *Doctor, Patient and the Law* 181.

A gamete donor is the living person from whose body a gamete or gametes are removed or withdrawn, for the purpose of artificial fertilisation.⁹⁷

7.18 Artificial insemination-husband (AIH):

AIH refers to artificial insemination homologous (AIH),⁹⁸ which is the artificial insemination of the wife in using the sperm of her husband.⁹⁹

7.19 Artificial insemination-donor (AID):

AID refers to artificial insemination heterologous (AID),¹⁰⁰ which is the artificial fertilisation of a woman in using the sperm donated by a sperm donor.¹⁰¹

7.20 Gamete intra-fallopian transfer (GIFT) and peritoneal oocyte and sperm transfer (POST):

The GIFT procedure refers to the process where ova from a woman are collected and, together with sperm from the husband or a donor, are then put into the woman's fallopian tubes (GIFT) or the peritoneam (POST).¹⁰²

⁹⁷ Reg 1 of Regulations 2012. Reg 1 of the proposed Regulations relating to assisted conception of persons published under Government Gazette No 44321 on 25 March 2021, defines gamete donor as: "A living person from whose body a gamete or gametes are withdrawn or procured after stimulation, for the purpose of donation for assisted conception."

⁹⁸ Palm MT and Hirsch HL "Infertility and sterility: Legal implications of artificial conception" 1982 *Medicine and Law* 1 44.

⁹⁹ Strauss Doctor, Patient and the Law 181; Mason, McCall Smith and Laurie Law and Medical Ethics 70. South African Research Council: Guidelines on Ethics for Medical Research 2nd Book (Reproductive Biology and Genetic Research) 2002 3. Mason, McCall Smith and Laurie Law and Medical Ethics 71. Micro-manipulative techniques are used and in particular those of intracytoplasmic sperm injection (ICSI) or sub-zonal insemination (SUZI). A single spermatozoon is then injected into an egg.

Palm MT and Hirsch HL "Infertility and sterility: Legal implications of artificial conception" 1982 Medicine and Law 1 44.

Strauss SA Doctor, Patient and the Law 181. South African Research Council: Guidelines on Ethics for Medical Research 2nd Book (Reproductive Biology and Genetic Research) 2002 2. Mason, McCall Smith and Laurie Law and Medical Ethics 70. The authors point out that many couples who have recourse to AID are loath to abandon all hope of their own children. In such circumstances, the semen of the apparently infertile male partner is mixed with that of the donor so that the couple can still believe that the ovum may have been fertilised by the male partner's sperm (this form of artificial fertilisation is also known as AIHD).

¹⁰² LexisNexis *Law of South Africa* 30(2) par 63. Reg 1 of the proposed Regulations relating to assisted conception of persons published under Government Gazette No 44321 on 25 March 2021, defines intra fallopian tube embryo transfer as "the transfer of zygotes or embryos in the fallopian tube."

7.21 Infertility:

Infertility is a disease of the male or female reproductive system defined by the failure to achieve a pregnancy after 12 months or more of regular unprotected sexual intercourse.¹⁰³ Primary infertility is a problem of the production of gametes or of implantation of the embryo.¹⁰⁴

Secondary infertility is when at least one prior pregnancy has been achieved.¹⁰⁵

7.22 Vaginal intro-peritoneal sperm transfer (VISPER):

The VISPER procedure entails the transfer of sperm into the peritoneal cavity of the woman.¹⁰⁶

7.23 Parent:

A parent generally refers to a father or mother of a person.¹⁰⁷ The legal definition of a parent in the Children's Act includes the adoptive parent of a child, but excludes the biological father of a child where the child is conceived through the rape of or incest with the child's mother, and further excludes any person who is only biologically related to a child as a result of being a gamete donor for purposes of artificial fertilisation. A person's whose parental responsibilities and rights in respect of a child have been terminated is no longer considered as the parent of said child.¹⁰⁸

7.24 Commercial surrogacy:

Commercial surrogacy refers to the arrangement where a woman is compensated for giving birth to a child whom she hands over to the commissioning parents in return for payment.¹⁰⁹

¹⁰³ Infertility can be accessed from <u>https://www.who.int/news-room/fact-sheets/detail/infertility</u> (accessed on 3 January 2023). Kennedy, I and Grubb, A *Medical Law* (2 ed Butterworths 1994) 758.

¹⁰⁴ Mason, McCall Smith and Laurie *Law and Medical Ethics* 69.

¹⁰⁵ Infertility can be accessed from <u>https://www.who.int/news-room/fact-sheets/detail/infertility</u> (accessed on 3 January 2023).

¹⁰⁶ LexisNexis *Law of South Africa* 30(2) at par 63.

¹⁰⁷ The definition can be accessed from <u>https://www.oxfordlearnersdictionaries.com/definition/english/parent_1?q=parent</u> (accessed on 4 January 2023).

¹⁰⁸ Sec 1 of Children's Act 38 of 2005.

¹⁰⁹ Oluwaseyi and Oladimeji 2021 *Obiter* 29.

7.25 Altruistic surrogacy:

In altruistic surrogacy, the surrogate mother receives no payment for carrying the commissioned child for the commissioning parents and does this as a selfless act.¹¹⁰

8. EXPECTED CONTRIBUTIONS AND OUTCOMES OF STUDY

We live in an era where medical technology is constantly changing, providing new options and alternatives for those who were eager to conceive children, but unable to do so a few years ago. Surrogate motherhood now offers such a chance for commissioning mothers who are pregnancy infertile or where a same-sex couple wants to have a child. The commissioning parents, in both examples, will then be able to have their own genetically related child. Law notoriously limps behind technology,¹¹¹ and as technological changes keep happening rapidly, the law regulating these new options does not always exist or needs to be amended or put in force to provide regulatory clarity and consistency. Ethical questions also arise through these new developments, as the discussion above has alluded to. There are also further constant changes in moral standards across time in societies.¹¹² All these have a direct effect on the choices we as individuals make in our lives.

The contribution that this thesis is expected to make, is to provide legal certainty and guidance in areas where the legal framework governing surrogate motherhood and artificial fertilisation is still lacking or ambiguous in South Africa. Surrogacy arrangements are on the increase, as may be gleaned from the increasing number of applications that courts are receiving for the confirmation of surrogacy agreements. Discussed next, is the historical development of laws, including the context, that shaped the development of the legal regulation of surrogate motherhood in South Africa.

¹¹⁰ Oluwaseyi and Oladimeji 2021 *Obiter* 29.

¹¹¹ World Economic Forum: The law can't keep up with new tech. Here's how to close the gap (21 June 2018). <u>https://www.weforum.org/agenda/2018/06/law-too-slow-for-new-tech-how-keep-up/</u> (accessed 4 January 2023).

¹¹² Wheeler MA, McGrath MJ, Haslam N (2019) Twentieth century morality: The rise and fall of moral concepts from 1900 to 2007. PLOS ONE 14(2): e0212267. https://doi.org/10.1371/journal.pone.0212267 (accessed 4 January 2023).

CHAPTER 2

HISTORICAL DEVELOPMENT OF LAWS RELATING TO ASSISTED REPRODUCTION AND SURROGATE MOTHERHOOD

1. INTRODUCTION

We are fortunate, however, to live in an era where the effects of infertility can be ameliorated to a large extent through assistive reproductive technologies. The technological advances seen over the last half century have greatly expanded the reproductive avenues available to the infertile. These reproductive avenues should be celebrated as they allow our society to flourish in ways previously impossible.¹¹³

This chapter has many objectives. Firstly, it will explain the different types of surrogacy arrangements that are available, including the important difference between altruistic and commercial surrogacy. The common law position regarding the legal status of the child conceived through donor sperm will also be discussed, as well as the legal position of the child conceived through artificial fertilisation prior to the enactment of the Children's Act (with reference to the statutes that governed artificial fertilisation and surrogate motherhood before the enactment of the Children's Act). It is important to understand the changes that were brought by the enactment of the Children's Act and especially the legal consequences in respect of the child born following artificial fertilisation and surrogacy. In order to contextualise the legal historical development of laws relating to surrogacy, it is necessary to briefly consider the types of surrogacy and how these have developed in the last few decades.

2. TYPES OF SURROGACY

Surrogacy may be described by reference to different types of surrogacy, such as informal surrogacy and formal surrogacy.¹¹⁴ Formal surrogacy consists of full and partial surrogacy.

AB and another v Minister of Social Development 2017 (3) BCLR 267 (CC); 2017 (3) SA 570 (CC) at par 3.
 CC) at par 3.

¹¹⁴ Report of the Ad Hoc Committee 13.

2.1 Formal surrogacy

2.1.1 Partial surrogacy

The process of partial surrogacy refers to the case where the surrogate mother's own ovum is fertilised (naturally or through artificial fertilisation) through using the sperm of the commissioning father or that of a donor.¹¹⁵ It is also known as genetic, traditional or straight surrogacy.¹¹⁶ Another example is where the surrogate's ovum is extracted, fertilised *in vitro* by using the sperm of the commissioning father or that of a donor and the resultant embryo placed in her womb.¹¹⁷ In this scenario, the surrogate is both the gestational mother and the genetic mother of the child that she bears.¹¹⁸ Thus, it is important to note that depending on the technique applied, a child born as a result of a surrogate agreement could have as many as six potential parents, namely the genetic parents (the donors of the sperm and ovum), the commissioning parents, the surrogate who carries the baby to term, and the surrogate's husband.¹¹⁹ I will now briefly turn to the situation involving a married and unmarried surrogate.

SALC Project 65 at par 2.1.2. Report of the Ad Hoc Committee 8. At 56 the Committee submitted that partial surrogacy should only be available where it is impossible for biological or medical reasons, to use the female gamete of the commissioning parent for the purpose of artificial fertilisation. Chapter 19 Surrogate motherhood (ss 292-303), Terminology. Pretorius, D *Surrogate motherhood: a worldwide view of the issues* (Charles C Thomas, 1994) 15. It is stated that partial surrogacy may be accomplished without medical expertise or specialised hospital facilities. Clarke, B "A "golden thread"? Some aspects of the application of the standard of the best interest of the child in South African family law" 2000 *Stellenbosch Law Review* 2000(11) 12. Strauss *Doctor, Patient and the Law* 188 Dada and McQuoid-Mason *Introduction to Medico-Legal Practice* 66. The surrogate provides genetic and gestational components. *Ex parte KAF and others* 2019 (2) SA 510 (GJ) at par 14. Skosana, T "A donor-conceived child's right to know its genetic origin: A South African perspective" 2017 *Obiter* 38(2) 263. Oluwaseyi and Oladimeji 2021 *Obiter* 22. Van Niekerk, C "Section 294 of the Children's Act: Do roots really matter?" 2015 *Potchefstroom Electronic Law Journal* 18(2) 400.

¹¹⁶ Nosarka S, Kruger TF "Surrogate Motherhood" 2005 South African Medical Journal 95(12) 942. South African Law Reform Commission Issue paper 32, Project 140, The Right to know one's own biological origins (20 May 2017) 81. Pyrce, C "Surrogacy and Citizenship: A Conjunctive solution to a global problem" 2016 Indiana Journal of Global Legal Studies 23(2) 929.

¹¹⁷ LexisNexis *Family Law service* at par J134. Clarke 2000 *Stell LR* 12.

¹¹⁸ SALC Project 65 at par 2.1.2. Tager 1986 SALJ 383. Pretorius Surrogate motherhood 16. The genetic or biological parents are therefore from two different families. Nosarka and Kruger 2005 SAMJ 942. Three women will be involved when an independent egg donor is used, namely the genetic mother, the surrogate mother and the commissioning mother. SALRC Project 140 1. Hisano, EY "Gestational surrogacy maternity disputes: Refocusing on the child" 2011 Lewis & Clark Law Review 15(2) 520. Louw, A "Surrogacy in South Africa: Should we reconsider the current approach?" 2013 Journal for Contemporary Roman-Dutch Law 76 575. Biko J and Nene Z "Ethics aspects of third-party reproduction" 2017 Obstetrics & Gynaecology Forum 3 14. Skosana 2017 Obiter 263. Nicholson and Bauling 2013 De Jure 512.

¹¹⁹ Clarke 2000 *Stell LR* 12.

2.1.1.1 The unmarried surrogate

In this instance the unmarried surrogate will be both the genetic and the biological mother of the child born as a result of partial surrogacy.¹²⁰ The child will be her child that is born out of wedlock (or extra-marital) in the case where the surrogate is unmarried.¹²¹ The surrogate will thus be the child's legal guardian.¹²²

2.1.1.2 The married surrogate

The legal maxim, *Pater est quem nuptiae demonstrant* refers to a legal presumption that the father is he to whom the marriage points and it is rebuttable.¹²³ Until the presumption has been rebutted by the husband of the surrogate, both the surrogate and her husband would have to consent to the adoption of the child as a child conceived by a married woman is presumed to be legitimate.¹²⁴

2.1.2 Full surrogacy

In contrast to partial surrogacy where the ovum of the surrogate is used, the surrogate will not be genetically linked to the child to be born in full surrogacy.¹²⁵ The surrogate mother in full surrogacy is only the carrier of the child to be born. This type of surrogacy, however, is fraught with potential conflict relating to maternal rights, as Hisano explains below:

¹²⁰ Tager 1986 *SALJ* 384.

¹²¹ Tager 1986 *SALJ* 384.

¹²² Tager 1986 *SALJ* 384 - 385. The surrogate would be entitled to consent to the child's adoption and not the natural genetic farther of the illegitimate child. Tager pointed out that the genetic farther would have little, if any, claim to the child although he would have to support it. An adoption was possible under the Children's Act of 1960. At 386 it is stated that the natural father's interest in respect of the child is only protected if the child is legitimate.

¹²³ Dig. 2.4.5; *Surmon v Surmon* 1926 AD 47 at p.53 Kotzé J.A pointed out that "While the Roman-Dutch writers held the presumption *pater est quem nuptiae demonstrant* to be rebuttable, they on the other hand adhered to the principle that neither the father nor the mother could bastardise their issue. But this latter principle appears to have been confined to cases where the child would be directly prejudiced by the declaration or evidence of the parent. It did not extend to a case of adultery, where such a declaration or evidence merely affects, for instance, the mother, whose confession of adultery would not directly prejudice her child, the latter not being a party to the suit". Tager 1986 *SALJ* 388.

¹²⁴ Tager 1986 *SALJ* 388.

¹²⁵ Van Niekerk 2015 *PELJ* 400. The vessel that carries the foetus from conception to birth. Nicholson and Bauling 2013 *De Jure* 512.

The growing problem of infertility coupled with increasingly sophisticated reproductive technology has produced an unfamiliar problem: the identification of a child's legal mother. This issue of legal motherhood is exacerbated in the situation where an infertile couple uses a gestational surrogate as a means of having a child. Many times, a gestational surrogacy agreement goes smoothly. However, in some cases, the arrangement results in a maternal rights dispute. In a gestational surrogacy arrangement, there are three potential women with maternal rights claims: the gestational surrogate, the genetic donor, and the woman for whom the baby is intended.¹²⁶

This arrangement is also known as gestational or host surrogacy. Here the surrogate merely acts as the gestational mother without being genetically related to the child that she gives birth to.¹²⁷ In this instance, the pregnancy of the surrogate is achieved through the implantation of an embryo into her uterus, which has been 'created' through the use of the gametes of the commissioning person(s) or of donors or of a combination of these persons.¹²⁸ The surrogate will thus give birth to a child to whom she is not genetically related.¹²⁹

¹²⁶ Hisano, EY "Gestational surrogacy maternity disputes: Refocusing on the child" 2011 *Lewis* & *Clark Law Review* 15(2) 517.

¹²⁷ Clarke 2000 Stell LR 12. SALC Project 65 at par 2.1.2. Report of the Ad Hoc Committee 7. At 56 the Committee submitted that full surrogacy is the preferred option. Tager 1986 SALJ 383. Nosarka and Kruger 2005 SAMJ 942. SALRC Project 140 16, 81 and 102. Examples of gestational surrogacy are: (a) the intended father's sperm and the intended mother's egg are used to create the embryo. The child will be genetically linked to both parents; (b) the intended father's sperm and a donor egg are used (where the surrogate is not the donor) to create the embryo. The child is genetically linked to the intended father and the biological mother is the donor; (c) the intended mother's eqg and donor sperm is used to create the embryo. The child is genetically linked to the mother and the biological father is the donor: (d) a donor embryo where both gametes are donated. The child will not be genetically linked to the intended parents and the biological mother and father are the donors. At 103. Sec 294 make it impossible to use donor embryos as the child must be genetically related to one of the intended parents. Hisano, EY "Gestational surrogacy maternity disputes: Refocusing on the child" 2011 Lewis & Clark Law Review 15(2) 520. Dada and McQuoid-Mason Introduction to Medico-Legal Practice 66. The surrogate gestates an embryo without providing the oocyte. Ex parte KAF and others 2019 (2) SA 510 (GJ) at par 14. Skosana 2017 Obiter 263. Oluwaseyi and Oladimeji 2021 Obiter 22. Forman, D.L. "Abortion clauses in surrogacy contracts: Insights from a case study" 2015 Family Law Quarterly 49(1) 32. Canner, S "Navigating surrogacy law in the non-united states: Why all states should adopt a uniform surrogacy statute" 2019 Journal of Civil Rights and Economic Development 33(2) 117. Pyrce, C "Surrogacy and Citizenship: A Conjunctive solution to a global problem" 2016 Indiana Journal of Global Legal Studies 23(2) 929. lannacci, B.R "Why New York should legalize surrogacy: A comparison of surrogacy legislation in other states with current proposed surrogacy legislation in New York" 2018 Touro Law Review 34(4) 1247. Van Niekerk 2015 PELJ 400.

¹²⁸ LexisNexis Family Law service at par J134. Pretorius Surrogate motherhood 15. Nosarka and Kruger 2005 SAMJ 942. Clarke 2000 Stell LR 12. Van Niekerk 2015 PELJ 401. The author gives the following ways of how the gametes are provided in full surrogacy: the permitted ways are (a) husband's sperm and the wife's egg(s); (b) donor sperm and wife's egg(s) and the non-permitted ways in terms of sec 294 of the Children's Act: donor sperm and donor egg(s).

¹²⁹ Clarke 2000 *Stell LR* 12. Louw 2013 *THRHR* 575. Strauss *Doctor, Patient and the Law* 188. The genetic material is provided by the commissioning couple and the surrogate merely acts as a hostess.

lannacci argues that the lack of biological connection between the surrogate and the child allows for a less complicated legal procedure because the commissioning parent(s)' biological connection to the child is not questioned.¹³⁰ It is difficult to state who the real mother of the child is as it can be the genetic mother (who supplied the egg) or the biological mother (who carried the child and gave birth to it).¹³¹ It may be argued that full surrogacy is more beneficial to the child from a psychological point of view in that the child is genetically linked to both parents.¹³² Nicholson explains that some view full surrogacy as potentially more exploitative of the surrogate, compared to partial surrogacy, as it is potentially more attractive to wealthy couples who want a genetically related child and who have no real interest in the socio-economic or cultural background of the surrogate.¹³³ Van Niekerk points out that full surrogacy is more often preferred while partial surrogacy is more commonly used.¹³⁴

2.2 Informal surrogacy

Informal surrogacy (more often practised by private agreement between family members or people known to each other) was and continues to be practised in many societies.¹³⁵ Informal surrogacy can be explained as surrogacy regulated by cultural norms and practices within communities.¹³⁶ It is often performed privately by the parties without the involvement of a medical doctor or clinic.¹³⁷ In this type of surrogacy, the

¹³⁰ Iannacci,B.R "Why New York should legalize surrogacy: A comparison of surrogacy legislation in other states with current proposed surrogacy legislation in New York" 2018 *Touro Law Review* 34(4) 1247. The author points out that this procedure makes it possible for single parents, infertile couples and members of the LGBT community to expand their families.

¹³¹ Tager 1986 *SALJ* 388 - 389. The writer explains that the genetic mother would rely on her genetic link with the child, which affects its physical and other attributes, and also on the fact that the foetus has a separate blood system from that of the surrogate. The surrogate, on the other hand, could claim that her contribution to the child is vital as the growing foetus is dependent on her not only for its nourishment and development but also for the removal of its waste products. It is stated that it seems that neither mother would succeed in proving that the other mother is not the real mother. At 390 "The donation of gametes, that is sperm or ova, cannot be said to be equivalent to the donation of a child, yet life begins as soon as the sperm penetrates the ovum and the resulting zygote or embryo is implanted in a womb."

¹³² Pretorius Surrogate motherhood 15. Iannacci, B.R "Why New York should legalize surrogacy: A comparison of surrogacy legislation in other states with current proposed surrogacy legislation in New York" 2018 Touro Law Review 34(4) 1248.

¹³³ Nicholson 2013 *SAJHR* 499.

¹³⁴ Van Niekerk 2015 *PELJ* 400.

¹³⁵ Ex parte WH and others at par 2. Nicholson 2013 SAJHR 497-498.

¹³⁶ Report of the Ad Hoc Committee 13.

¹³⁷ Report of the Ad Hoc Committee 8 and 12.

surrogate is both the genetic and the gestational mother.¹³⁸ An example of this type of surrogacy will be where a family member (the sister of an infertile woman) agrees to act as surrogate for the infertile woman and her husband and a home insemination kit is used to artificially fertilise the surrogate with the sperm of the husband. A further example will be where a female friend of a same-sex male couple agrees to act as surrogate for them and a home insemination kit is used to artificially fertilise the surrogate of a same-sex male couple agrees to act as surrogate for them and a home insemination kit is used to artificially fertilise the surrogate of a same-sex male couple agrees to act as surrogate with sperm of one of the couple or both.

The South African Law Commission correctly describes surrogacy not as a new technology, but as a phenomenon that was made possible by new medical technology.¹³⁹ It is interesting to note that surrogacy was actually not recognised in South African law before the enactment of the Children's Act,¹⁴⁰ although many reported instances of informal surrogacy were reported, sometimes accompanied by surrogate motherhood agreements drafted by legal professionals.¹⁴¹ Although the former Human Tissue Act 65 of 1983 and the Children's Status Act 82 of 1987 dealt with various aspects contingent upon artificial fertilisation, as will be discussed elsewhere in this thesis in detail, neither of the Acts however dealt specifically with the issue of children born following the conclusion of a surrogacy arrangement.

In 1987, the South African Law Commission initiated an investigation into the scope of surrogate motherhood practices in South Africa, with the objective of legislating the practice. The SALC's Report and the draft legislation on Surrogate Motherhood were tabled in 1993, followed by the establishment of an Ad Hoc Select Parliamentary Committee with the task of further investigating surrogate motherhood and related issues. The report of the Ad Hoc Committee was completed in 1999 and amended draft

¹³⁸ Report of the Ad Hoc Committee 8 and 12. The Ad Hoc Committee however recommended that surrogacy should be practiced within families only.

¹³⁹ SALC Project 65 at par 8.1.2.

¹⁴⁰ 38 of 2005. Heaton 2015 *THRHR* 29.

Ex parte WH and others at par.3. AB and another v Minister of Social Development at par 246. Informal surrogacy agreements were entered into seemingly because of the perceived advantage over the legally prescribed adoption procedure that was available. Pretorius Surrogate motherhood 15. The reasons for the use of surrogate motherhood were explained as being when the woman is unable to produce oocytes or to carry a baby full term. Further, in situations where pregnancy carries an abnormally high risk in the case of severe high blood pressure or diabetes. Report of the Ad Hoc Committee 8 and 54. Informal surrogacy is defined as the insemination of the surrogate with the gametes of the commissioning parent. This is performed privately by the parties and according to accepted customary practices without the intervention of medical doctors or clinics. Thus, the surrogate is both the genetic and the gestational mother.

legislation on surrogacy subsequently referred to the Minister of Justice for final review and drafting. The Ad Hoc Committee proposed that surrogacy could provide a legitimate alternative for permanently infertile people who wish to have children.¹⁴² The full culmination of the Committee's report is best observed in the Children's Act 38 of 2005, which provides a mechanism for many people who desire a child to start a family.¹⁴³

The practice of informal surrogacy is impacted by the regulations to the NHA¹⁴⁴ through the requirement of a competent person as the only person authorised to perform artificial fertilisation.¹⁴⁵ Attempts by a single woman to inseminate herself with sperm donated by a homosexual male couple wishing to have a child, despite the requirements relating to artificial fertilisation, are not criminally punishable. It is clear, though, that failure to comply with chapter 19 of the Children's Act when such woman wishes to act as a surrogate for her male friend, will have serious legal consequences that will also affect parenthood and the legal status of the child that is born, as will be discussed in more detail in chapter 5.

Surrogacy allows a woman to exercise her reproductive rights within prescribed limits as she sees fit; it enables infertile people to have a child who is genetically related to them and it gives childless people the joy of raising a child.¹⁴⁶ The parties to a surrogate agreement are the commissioning parents (the couple who are unable to conceive), the surrogate,¹⁴⁷ and if she is married, her husband.¹⁴⁸ Unfortunately, infertility is real

¹⁴² Report of the Ad Hoc Committee 54.

¹⁴³ Ex parte WH and others at par.3 and par 43. The court stated at par.6: "In very much the same way as society's architecture is structured to advance the best interests of the child, so too does it reflect and give response to the desire of many to have children of their own. For some it represents the fulfilment of the agency of their own lives and existence as they seek to continue their lineage and their legacy, while for others the vision of a family living and loving together is rendered complete with the arrival of a child." Report of the Ad Hoc Committee 55. Surrogacy should be regulated in such a way that it minimises the risks inherent to surrogacy and it ensures the best interests of all the parties to the agreement. The interests of the commissioned child is the overriding factor.

¹⁴⁴ 61 of 2003.

¹⁴⁵ This requirement will be discussed in ch 5.

¹⁴⁶ Heaton 2015 *THRHR* 26.

¹⁴⁷ The word "surrogate" derives from the Latin word *surrogatus* (substituted) and meaning "appointed to act in the place of". SALRC Project 140 80.

¹⁴⁸ Pretorius *Surrogate motherhood* 17. It is important to note that the husband of the surrogate was also a party to the agreement as he had to consent to the medical procedures and the termination of the parental power.

and with many suffering from infertility, desperation may set in, prompting them to enter into an informal surrogacy agreement without understanding the consequences of failing to observe the legal requirements set out in chapter 19 of the Children's Act.

2.3 Altruistic surrogacy

The practice of altruistic surrogacy has been around for some time and was not uncommon or explicitly prohibited before the enactment of chapter 19 of the Children's Act.¹⁴⁹ With altruistic surrogacy, the surrogate acts purely with an altruistic motive to be of assistance, without any prospect of making a profit from her decision.¹⁵⁰ Often the surrogate's main objective is to do good for others and to assist without financial benefit to realise an infertile couple's dream to become parents¹⁵¹ The surrogate therefore does not profit from her gestational services in any manner.¹⁵² Altruistic surrogacy is more socially acceptable as it is seen to display socially accepted virtues such as generosity, selflessness, concern and sacrifice.¹⁵³ The only legally recognised surrogacy arrangement in South Africa is altruistic surrogacy, regulated by chapter 19 of the Children's Act.¹⁵⁴

Before the introduction of chapter 19 of the Children's Act, altruistic surrogacy arrangements in South Africa were subject to the law of contract and legislation and regulations which pertained to artificial fertilisation.¹⁵⁵ In 1985, the South African Law Commission (SALC) commenced with investigating the uncertainties regarding the legal position of illegitimate children (as they were then known) and their parents, considering, among others, the extent to which an illegitimate child's legal position was

¹⁴⁹ Slabbert M and Roodt C 'South Africa' in Trimmings K and Beaumont P *International Surrogacy Arrangements Legal Regulation at the International Level* (Hart Publishing 2013) 325. *AB and another v Minister of Social Development* at par 246.

¹⁵⁰ Pretorius *Surrogate motherhood* 86.

¹⁵¹ Report of the Ad Hoc Committee 9.

¹⁵² Pyrce, C "Surrogacy and citizenship: A conjunctive solution to a global problem" 2016 *Indiana Journal of Global Legal Studies* 23(2) 929.

¹⁵³ SALC Project 65 at par 2.1.6.

¹⁵⁴ Sec 295(c)(iv) and (v). Nicholson, S and Nicholson, C "I used to have two parents and now I have three? When science (fiction) and the law meet: Unexpected complications" 2016 *Medicine and Law* 35 432. Sloth-Nielsen, J "Surrogacy in South Africa" in Scherpe J, Fenton-Glynn C and Kaan T *Eastern and Western perspectives on surrogacy* (*Intersentia* 2019) 186. Sloth-Nielson, J and van Heerden, B "The constitutional family developments in South African child and family law 2003-2013" 2014 *International Journal of Law*, *Policy and the Family* 28(1) 114.

¹⁵⁵ Slabbert and Roodt "South Africa" 325. Mills, L "Certainty about surrogacy" 2010 *Stellenbosch Law Review* 21(3) 429. Oluwaseyi and Oladimeji 2021 *Obiter* 31.

compromised under the law, as well as how this may be eased in the light of norms and prevailing ideas of the time.¹⁵⁶ The main purpose of the investigation was to make recommendations on how to improve the legal status of illegitimate children, including children born as a result of artificial fertilisation.¹⁵⁷ It is significant that the SALC set out to make recommendations that would not only provide the illegitimate child with better legal protection, but also to provide legal certainty for this cohort of children in South African law.¹⁵⁸

Prior to the promulgation of chapter 19 of the Children's Act, laws governing surrogacy consisted of: (a) the Human Tissue Act¹⁵⁹, (b) the Child Care Act¹⁶⁰, (c) the Children's Status Act¹⁶¹, and (d) the Births, Marriages and Deaths Registration Act.¹⁶² These acts are discussed in more detail later in paragraph 5 of this chapter.

Briefly, the Human Tissue Act and Regulations regulated artificial fertilisation of persons; the Child Care Act's aim was the protection and welfare of certain children and the adoption of children; the Children's Status Act dealt with children born as a result of artificial fertilisation; and the Births, Marriages and Deaths Registration Act provided for the registration of births, including the registration of the birth of an illegitimate child. Section 10 of this Act provided that no person was required to provide information as to the father of the child and only upon the joint request of the mother and the person who acknowledged himself in writing to be the father of the child, would the registrar enter the name of the father in the birth register of the child. The Children's Status Act determined the status of the child born of such a surrogacy arrangement by providing that the child born as a result of artificial fertilisation will be legitimate in circumstances where donor gametes were utilised and the woman's husband consented to the procedure.¹⁶³

¹⁵⁶ SALC Project 65 at par 1.1.1 and par 1.2.1.

¹⁵⁷ SALC Project 65 at par 1.1.2.

¹⁵⁸ SALC Project 65 at par 1.1.2.

¹⁵⁹ 65 of 1983. SALC Project 65 at par 4.2.

¹⁶⁰ 74 of 1983. SALC Project 65 at par 4.3.

¹⁶¹ 82 of 1987. SALC Project 65 at par 4.4.

¹⁶² 81 of 1963. SALC Project 65 at par 4.5.

¹⁶³ Sec 5 of the Children's Status Act.

The SALC argued that the Children's Status Act reflected the maxims *mater semper certa est* (the mother is always certain) and *pater est quem nuptiae demonstrant* (the father is he whom the marriage points out).¹⁶⁴ Section 17(a) of the Child Care Act stipulated that the commissioning parents could become the legal parents of a child in a surrogacy arrangement but only if they followed the route of legal adoption.¹⁶⁵

The regulations to the Human Tissue Act, namely the Regulations regarding the artificial insemination of persons, and related matters¹⁶⁶ governed the artificial fertilisation and *in vitro* insemination in South Africa. The Regulations provided that the procedures of removal or withdrawal of a gamete and the artificial insemination had to be carried out by a medical practitioner or persons under his or her supervision (a competent person);¹⁶⁷ the competent person was only allowed to effect artificial insemination of a married woman where her husband had provided written consent for the procedure.¹⁶⁸ Under these Regulations, requirements were set regarding: detailed records of donors¹⁶⁹ and recipients;¹⁷⁰ the personal details of the donor, including the donor's family history, had to be placed at the disposal of the medical practitioner intending to effect the artificial insemination;¹⁷¹ the medical practitioner, intending to remove or withdraw gametes, had to ascertain that the donor had been tested for sexually transmitted diseases and that his sperm had been subjected to analysis.¹⁷² Furthermore, if a medical practitioner knew or suspected that two or more artificially conceived pregnancies existed or that the donations of one donor had led to at least five pregnancies, no further donations by that donor were allowed;¹⁷³ recipients had to be screened to ensure their medical, social, and mental fitness for artificial insemination;¹⁷⁴ the medical practitioner had to ensure that patients received advice

¹⁷¹ Reg 4(d)(iv) of the Regulations regarding artificial insemination of persons 1986.

¹⁶⁴ SALC Project 65 at par 4.4.2.

¹⁶⁵ Slabbert and Roodt "South Africa" 326. *AB and another v Minister of Social Development* at par 247. Clarke 2000 *Stell LR* 13. Nicholson 2013 *SAJHR* 498.

¹⁶⁶ Regulations regarding the artificial insemination of persons and related matters, published in GG No R 1182 of 20 June 1986 (GG 10283).

¹⁶⁷ Reg 3 of Regulations regarding the artificial insemination of persons 1986.

¹⁶⁸ Reg 8(1) of the Regulations regarding artificial insemination of persons 1986.

¹⁶⁹ Reg 4 and 5 of the Regulations regarding artificial insemination of persons 1986.

¹⁷⁰ Reg 9 and 10 of the Regulations regarding artificial insemination of persons 1986.

¹⁷² Reg 5(a) of the Regulations regarding artificial insemination of persons 1986.

¹⁷³ Reg 8(a) and Reg 4(e).

¹⁷⁴ Reg 9(e)(ii).

and information from experts on the implications of artificial insemination;¹⁷⁵ and finally, that confidentiality had to be maintained.¹⁷⁶

Thus, as can be seen from the above, until the promulgation of the Children's Act, surrogacy was not explicitly and comprehensively regulated by legislation per se.¹⁷⁷ The Human Tissue Act and the Children's Status Act, however, governed a range of legal issues relating to artificial fertilisation and the status of children conceived as a result of artificial fertilisation.¹⁷⁸ Chapter 8 of the NHA replaced the Human Tissue Act and the Children's Status Act, as well as the Child Care Act. The Human Tissue Act and regulations were the first attempt in South Africa's legal history to regulate human artificial fertilisation.¹⁷⁹

For those who wished to pursue surrogacy at the time, these statutes referred to regulated surrogacy indirectly and inadequately, in a very artificial and piecemeal manner. For example, the child was considered illegitimate if the woman's husband did not consent to her being artificially fertilised with donor sperm or illegitimate in the event that his consent to the procedure had been obtained. In circumstances where a surrogate gave birth to a commissioned child, the commissioning parents had to adopt the child as the child was seen as the child of the surrogate. All the different provisions caused anxiety and confusion regarding the status of the commissioned child and made the regulation of surrogacy difficult, if not almost impossible, as there was no "one set" of provisions in one place that could guide parties who desired to enter into a surrogacy arrangement.

¹⁷⁵ Reg 9(e)(bb).

¹⁷⁶ Reg 6(e) and Reg 10(2)(b).

¹⁷⁷ SALC Project 65 at par 7.1. The SALC pointed out that there were strong arguments that the effect of surrogacy on the child, the parties and society is covered in uncertainty. It was therefore risky to recognise surrogacy locally at that stage.

¹⁷⁸ There was no reference to artificial insemination in South African statutes prior to The Human Tissue Act. The only other possible reference could have been sec 7 of the Anatomical and Donations and Post Mortem Examinations Act 24 of 1970 which provision prohibited artificial fertilisation by a donor. *V v R* 1979(3) SA 1006 (T) at p1009. *AB and another v Minister of Social Development* at par 246. Pretorius *Surrogate motherhood* 88. Skosana 2017 *Obiter* 262. Oluwaseyi and Oladimeji 2021 *Obiter* 31. SALC Project 65 at par 4.2.1. The SALC submitted that the definition of artificial fertilisation did not stipulate in whose womb the embryo (*in vitro* or *in vivo*) must be placed and therefore, it can be lawfully transferred to the womb of the surrogate. Further, the regulations neither excluded surrogacy as the definition of *in vitro* insemination did not preclude the placing of the embryo in a surrogate's womb.

¹⁷⁹ LexisNexis Law of South Africa 30(2) at par 63. Strauss Doctor, Patient and the Law 184-185. Slabbert, M.N "Are the human embryo and the foetus extra uterum sufficiently protected in terms of South African law?" 2001 Journal of South African Law 2001(3) 496.

2.4 Commercial surrogacy

The practice of commercial surrogacy, in contrast to altruistic surrogacy, was (and still is) seen as *contra bones mores* and is explicitly prohibited in South Africa.¹⁸⁰

Commercial surrogacy refers to a situation where a woman is compensated for giving birth to a child whom she hands over to the commissioning parents in return for payment.¹⁸¹

Commercial surrogacy, as defined in chapter 1, refers to surrogacy where the surrogate's motivation is financial. The SALC argues that payment of the surrogate amounts to the buying of the commissioned baby.¹⁸² Commercial surrogacy is undertaken in exchange for payment in that the commissioning person or couple undertake to pay the surrogate a fee for conceiving and bearing the child (which is greater than the actual costs incurred).¹⁸³ In terms of the commercial agreement, the surrogate enters into the agreement with the prospect of compensation or a source of income.¹⁸⁴ The commissioning parent(s) undertake(s) to pay the surrogate any compensation *of any nature other* than those specifically provided for in section 301 of the Children's Act, such as the medical expenses relating to the birth.¹⁸⁵ Commercial surrogacy points to a rent-a-womb scenario.¹⁸⁶ The risk for exploitation in this kind of arrangement is greater and that is also why it is regarded as unacceptable in many jurisdictions.¹⁸⁷ Over the years, ethical, legal and policy issues have been considered to determine if commercial surrogacy in effect treats children as commodities and

AB and another v Minister of Social Development at par 246. Ex parte HPP and others; Ex parte DME and others and others 2017 (4) SA 528 (GP) at par 24. Pretorius Surrogate motherhood 17. At 86 where large amounts of money are involved. SALRC Project 140 8. Gamete donation and surrogate motherhood should be altruistic in the South African legal scheme and not for commercial purposes. Bonthuys, E and Broeders, N "Guidelines for the approval of surrogate motherhood agreements: Ex parte WH" 2013 South African Law Journal 130(3) 490. Mills 2010 Stell LR 429. Florescu, S and Sloth-Nielsen, J "Visions on surrogacy – From North to South: The approach of the Netherlands and South Africa to the issue of surrogacy and the child's right to know his origin" 2017 International Survey of Family Law 2017 247. Heaton 2015 THRHR 29 and 34. The consequences of commercial surrogacy are not recognised. Commercial surrogacy is illegal, against public policy and in some respects a crime.

¹⁸¹ The term is defined in par 7.24, ch 1 above Oluwaseyi and Oladimeji 2021 *Obiter* 29.

¹⁸² SALC Project 65 at par 2.5.

¹⁸³ Report of the Ad Hoc Committee 9. Cascão, R "The challenges of international commercial surrogacy: from paternalism towards realism?" 2016 *Medicine and Law* 35(2) 152.

¹⁸⁴ Sec 295(iv) specifically makes reference to the words 'source of income'.

¹⁸⁵ *Ex parte WH and others* at par 65. Sec 301 will be discussed in detail below.

¹⁸⁶ Lupton, M.L "Artificial wombs: medical miracle, legal nightmare" 1997 *Medicine and Law* 16 624.

¹⁸⁷ Pyrce, C "Surrogacy and citizenship: A conjunctive solution to a global problem" 2016 *Indiana Journal of Global Legal Studies* 23(2) 930.

violates their rights.¹⁸⁸ Commercial surrogacy has also been linked to human trafficking in circumstances where the parties live in different countries and the transfer of the commissioned child is from one country to another.¹⁸⁹

With commercial surrogacy regarded as illegal in South Africa, surrogacy agreements that contain financial incentives beyond the allowed pregnancy -and birth related expenses are hence unenforceable.¹⁹⁰ Cascão rightly states that it is a deep-rooted principle in bioethics that "the human body and its parts shall not give rise to financial gain".¹⁹¹ Commercial surrogacy has the potential to lead to a serious infringement of the human rights and dignity of surrogate mothers, and ultimately to the rise of an exploited "breeding caste" of underprivileged women and large scale transnational "baby selling" to wealthier families from developed countries.¹⁹²

The United Nations Convention on the Rights of a Child (UNCRC) determines in section 3 that all actions concerning a child shall take full account of the child's best interests. Section 35 of the UNCRC determines that the state has an obligation to make every effort to prevent the sale, trafficking and abduction of children. Commercial surrogacy is fraught with many risks, for example, the surrogate can continuously extract more money from the commissioning parents by manipulating them and using the commissioned child as the reason. The commercialisation of surrogacy makes the practice a business, which may negate the best interests of the commissioned child, as profit will become the main driving factor. For lower- and middle income countries such as South Africa, a further risk is that with high unemployment surrogacy may become a viable option for desperate women to generate an income. The danger exists that some women introduced into prostitution and slavery through promises of a better life, may be convinced to enter into a commercial surrogacy arrangement and to sell

¹⁸⁸ Oluwaseyi and Oladimeji 2021 *Obiter* 29.

¹⁸⁹ Oluwaseyi and Oladimeji 2021 *Obiter* 30.

Ex parte HPP and others; Ex parte DME and others at par 24, 27 and 67. AB and another v Minister of Social Development at par 134. Nicholson 2013 SAJHR 496. Sloth-Nielsen "Surrogacy in South Africa" 187. Sloth-Nielson, J and van Heerden, B "The constitutional family developments in South African child and family law 2003-2013" 2014 International Journal of Law, Policy and the Family 28(1) 114.

¹⁹¹ Cascão, R "The challenges of international commercial surrogacy: from paternalism towards realism?" 2016 *Medicine and Law* 35(2) 152.

¹⁹² Cascão, R "The challenges of international commercial surrogacy: from paternalism towards realism?" 2016 *Medicine and Law* 35(2) 154.

their babies.¹⁹³ It is thus in the best interests of the commissioned children that commercial surrogacy never forms part of South African law and that the state observes the obligations under the UNCRC and section 28 of the Constitution.

2.5 International surrogacy

Two possible scenarios of international surrogacy are where the commissioning parent(s) are not domiciled in South Africa and they obtain a child through surrogacy and a surrogate mother in South Africa, or, where South African commissioning parent(s) obtain a commissioned child through surrogacy in a foreign country.¹⁹⁴

International surrogacy causes a lot of confusion and various complexities regarding the legal position of the child born as a result of surrogacy.¹⁹⁵ Heaton refers to various cultural, moral, ethical and religious differences in perspectives on surrogacy and, importantly, the parent-child relationship.¹⁹⁶ For example, where the rules of the commissioning parents' country of origin are different from the foreign country where the commissioned child is born, conflicting legal outcomes may follow.¹⁹⁷ The possibility exists that either the commissioning couple's country of origin does not recognise surrogacy, or the country does not recognise the surrogacy in the circumstances in which it occurred in the surrogate's country.¹⁹⁸ The country where the commissioned child is born may perhaps not recognise the commissioning couple as the parents of the commissioned child, which means that the commissioned child will not be able to obtain travel documents to travel with the commissioning parents to their country of origin.¹⁹⁹

Currently, due to jurisdictional limitations, chapter 19 of the Children's Act does not contain any provisions relating to international surrogacy and its consequences.²⁰⁰

¹⁹³ Oluwaseyi and Oladimeji 2021 *Obiter* 30.

¹⁹⁴ Heaton 2015 *THRHR* 34.

¹⁹⁵ Heaton 2015 *THRHR* 28.

¹⁹⁶ Heaton 2015 *THRHR* 27.

¹⁹⁷ Heaton 2015 *THRHR* 27.

¹⁹⁸ Heaton 2015 *THRHR* 27. Cascão, R "The challenges of international commercial surrogacy: from paternalism towards realism?" 2016 *Medicine and Law* 35(2) 157.

¹⁹⁹ Heaton 2015 *THRHR* 27-28.

²⁰⁰ Heaton 2015 *THRHR* 28. At 46 the author states that greater predictability and legal certainty regarding legal parentage of children in international surrogacy situations must be ensured and

Heaton argues that the Children's Act seeks to limit international surrogacy by the requirement that a surrogacy agreement has to be entered into in South Africa and the parties to the agreement must be domiciled in the country.²⁰¹

There is, however, no public or private international law instrument on international surrogacy that may direct parties on the legal consequences of the various scenarios that may arise.²⁰² Conflicting legal rules in the various jurisdictions may (looking at the worst-case scenario) result in the commissioned child being parentless or stateless and the commissioning parents being unable to take the child to their country of origin.²⁰³ Although the commissioning parents may succeed in taking the child to their country, they may not be recognised as their child's legal parents in their jurisdiction.²⁰⁴ This result creates an untenable situation for commissioning parents who already have to contend with a range of uncertainties and legal risks, not to mention additional financial and emotional issues. Regarding the emotional issues-in the 2017 Constitutional Court judgment of AB and another v Minister of Social Development,²⁰⁵ which dealt with the constitutionality of section 294 of the Children's Act, Khampepe J observes that the psychological trauma that all infertile people experience may further be heightened by an abiding sense of social shame, which may prompt them to conceal their infertility.²⁰⁶ She further observes that the stigma that attaches to infertility is damaging and pervasive, especially in developing countries like South Africa.²⁰⁷ Infertility is an unenviable and psychologically harmful condition because of its potential of preventing people from having children of their own, and because it may result in serious social exclusion and stigma:²⁰⁸

the legislature should expressly regulate the consequences of the identified instances of international surrogacy. The best interests of the children born as a result of surrogacy are not served by shying away from the task.

²⁰¹ Heaton 2015 *THRHR* 28. Sec 292(1)(b), sec 292(1)(c) and sec 292(1)(d) of the Children's Act.

²⁰² Heaton 2015 *THRHR* 28.

²⁰³ Heaton 2015 *THRHR* 28. At 41 Heaton explains that there are three criteria which is commonly used to assign legal parentage to a person: (a) a genetic link between the person and the child, (b) the person's intention to become a parent (commissioning parent commissioning a child through a surrogate) and (c) giving birth to the child. Importantly, different jurisdictions apply different criteria which has the result that a person may qualify as the child's legal parent in one country but not in another. Cascão, R "The challenges of international commercial surrogacy: from paternalism towards realism?" 2016 *Medicine and Law* 35(2) 157. The writer referred to international commercial surrogacy.

²⁰⁴ Heaton 2015 *THRHR* 28 and 41.

²⁰⁵ 2017 (3) SA 570 (CC).

AB and another v Minister of Social Development at par 84.

²⁰⁷ AB and another v Minister of Social Development at par 84.

AB and another v Minister of Social Development at par 86.

Infertility cruelly dispossesses a person of the capacity to decide whether or not to have a child; where making this decision has extensive social implications.²⁰⁹

2.6 Concluding observations

It is apparent from the preceding discussion that prior to the enactment of chapter 19 of the Children's Act, South Africa needed proper legislation in respect of surrogacy especially taking into account the advancements in medical technology. The uncertainty created by the combined effect of the Children's Status Act and the Child Care Act further demanded the urgent attention of the legislature of South Africa. Before the enactment of the Children's Act, some uncertainties remained regarding the status of the child conceived by artificial fertilisation or in vitro fertilisation of an unmarried woman. Although the Human Tissue Act limited artificial fertilisation and in vitro fertilisation to a married woman with the consent of her husband,²¹⁰ the limitations did not provide certainty regarding the relationship between the child and its genetic father in the event of an unmarried woman being artificially fertilised. The law, shaped by social policy, needs to define the relationship between the parties involved in medically assisted reproduction and their children and what their respective rights and duties are.²¹¹ For Dada and McQuoid-Mason, the right to procreate, which extends to coital and noncoital choices, should be characterised as the right to parent, rather than the right to achieve and maintain a biological tie with a child.²¹²

The lacunae in South African law relating to surrogacy and artificial fertilisation prior to the enactment of chapter 19 of the Children's Act led to a legal vacuum requiring urgent legal intervention. The promulgation of chapter 19 of the Children's Act which regulates surrogacy specifically, addressed some of the legal gaps regarding surrogate motherhood. Artificial fertilisation remained regulated in terms of the Human Tissue Act prior to the enactment of the Children's Act and the National Health Act 61 of 2003.

AB and another v Minister of Social Development at par 89.

²¹⁰ Reg 8(1) of the Regulations regarding the artificial insemination of persons 1986.

²¹¹ Dada, MA and McQuoid-Mason, DJ *Introduction to Medico-Legal Practice* (Butterworths 2001) 66.

²¹² Dada and McQuoid-Mason *Introduction to Medico-Legal Practice* 66.

3. THE COMMON LAW POSITION REGARDING THE LEGAL STATUS OF THE CHILD CONCEIVED AS A RESULT OF ARTIFICIAL FERTILISATION WITH DONOR SPERM

3.1 Introduction

The relevant laws and regulations governing surrogacy prior the Children's Act, discussed above, did not specifically prohibit surrogacy agreements.²¹³ The general opinion was that a surrogacy agreement was likely to be unenforceable as it was seen to be *contra bones mores*.²¹⁴ The SALC admitted that public policy naturally differs from one community to another without remaining static within a community.²¹⁵ Public policy grows and changes as the community changes and technology develops.²¹⁶

3.2 The maxims mater semper certa est and pater is est quem nuptiae demonstrant

The common law saying (maxim) of *mater semper certa est* is based on the assertion that the mother of a child is always certain.²¹⁷ This maxim is informed by the fact that historically, the possibility did not exist that anyone other than the gestational (birth) mother, who is also the genetic mother, could be the legal mother of the child.²¹⁸ The maxim regarding the assumption of paternity, e.g *pater is est quem nuptiae demonstrant*, is different. This maxim means that "the father is he to whom the marriage point".²¹⁹ This meant that a child born to, or conceived by a married woman, was considered legitimate and there was a presumption that the woman's husband was the father of the child.²²⁰

SALC Project 65 at par 4.7.2.

²¹⁴ SALC Project 65 at par 4.7.3.

²¹⁵ SALC Project 65 at par 4.7.3.

SALC Project 65 at par 4.7.3.

²¹⁷ Pretorius *Surrogate motherhood* 17.

²¹⁸ Pretorius *Surrogate motherhood* 134. This maxim has always been irrebuttable.

²¹⁹ Pretorius *Surrogate motherhood* 139. The presumption was rebuttable at any stage on a balance of probabilities. SALC Project 65 at par 4.8.3.

Pretorius Surrogate motherhood 139-140. It was not necessary for conception to have taken place during the subsistence of the marriage and neither was it necessary that the children were conceived from the husband of the mother. The prerequisite was the subsistence of a marriage. SALC Project 65 at par 4.8.3.

The criterion for determining legal parenthood has always been consanguinity.²²¹ The statutory provisions did allow for exceptions in certain circumstances, for instance in the case of adoption, as the courts were empowered to terminate the parental rights of the genetic parents. Consanguinity, however, does not always provide satisfactory results.²²² The advances in modern reproductive technology made it possible that as many as three women could claim some form of maternal rights to a child, causing the basic beliefs of family law to become uncertain.²²³ The possibility of numerous persons claiming parental rights to a child has questioned conventional notions of what a mother and a father are or should be.²²⁴ Modern technology has the potential to wreak havoc with the common law *maxim mater semper certa* est.²²⁵

The Children's Act of 1960,²²⁶ which preceded the Child Care Act, provided in terms of section 71(2) that a child born as a result of artificial fertilisation using the sperm of an unknown donor was regarded as illegitimate even though both the husband and wife had provided their consent to the procedure.²²⁷ This seems to have endorsed and confirmed the common law position.²²⁸ Thus, in terms of the common law, a child conceived by means of artificial fertilisation with the use of donor sperm (AID) would be illegitimate.²²⁹ This is also because of the weight attached to kinship in the late Middle Ages, the 17th and 18th centuries.²³⁰ This is attached to the privileges of kinship, nobility and inheritance privileges, which exclusively depended on the specific blood bond between father and child.²³¹

²²¹ Pretorius *Surrogate motherhood* 138.

²²² Pretorius *Surrogate motherhood* 138.

²²³ Pretorius *Surrogate motherhood* 134. There was an uncertainty of legal motherhood. Who should be regarded legally as the mother of the child? Should it be the ova donor who is genetically linked to the child, or the commissioning mother, who is socially linked to the child or is it the surrogate mother who has a gestational link to the child? Thus, the *mater semper certa est* could no longer provide certainty in this regard. SALC Project 65 at par 4.8.2.

Pretorius Surrogate motherhood 17.

SALC Project 65 at par 4.8.2.

²²⁶ 33 of 1960.

²²⁷ *V v R* 1979(3) SA 1006 (T) at p 1015. Pretorius *Surrogate motherhood* 91.

²²⁸ *V v R* at p 1015.

²²⁹ V v R at p 1014 Lupton, M.L "Artificial wombs: medical miracle, legal nightmare" 1997 *Medicine and Law* 16 628. This was the position prior to 14 October 1987 when the Children's Status Act 82 of 1987 was enacted.

²³⁰ *V v R* at p 1014.

²³¹ *V v R* at p 1014.

It is clear that the legal opinion, as it was in South Africa in 1979, followed the basic principles laid down in our common law by the old Roman-Dutch law writers.²³² At the time, artificial fertilisation did not exist, but has become a reality in modern times as a result of the development of medical technology.²³³ Public policy in South Africa at the time also undeniably favoured the restriction of surrogate motherhood to married couples who were in a stable relationship.²³⁴ Public opinion regarding surrogacy has shown to shift gradually. After the birth of the Tzaneen triplets in South Africa in 1987,²³⁵ the pre-dominant view was that full surrogacy in a family arrangement without any commercial gain for any of the parties was generally acceptable.²³⁶

In the judgment of $V v R^{237}$, the court refers to three legal scholars who confirmed the common law position regarding artificial fertilisation, ranging from the 1950's to the 1970s:

(a) Mentioned first is Professor Hahlo who observed that:²³⁸

The second major issue in connection with heterologous artificial insemination is the legal status of the children. There can be little doubt that in our law, irrespective of whether or not AID is adultery and irrespective of whether or not the husband has consented to it, the children born of this procedure would be illegitimate.

- (b) A similar sentiment was expressed by N.C Masters who asserted that:²³⁹ Where the child is born as a result of AID then it seems equally clear that the child is illegitimate: the fact that the legal father had consented to the operation would not in any way affect its status.
- (c) Professor Spiro came to the same conclusion:²⁴⁰

If a child is born as the result of its mother having been artificially inseminated with the semen of a person other than her husband, the child must be ordinarily illegitimate, and this is so even if the husband consented.

It is evident from the above references, which affirmed the common law position, that a child conceived by means of artificial fertilisation from the sperm of a donor (AID)

²³² *V v R* at p 1015.

²³³ *V v R* at p 1014 – 1015.

²³⁴ Pretorius *Surrogate motherhood* 137.

²³⁵ This matter will be discussed in par 4.2 below.

²³⁶ Pretorius *Surrogate motherhood* 137.

²³⁷ 1979 (3) SA 1006 (T). This judgment will be discussed in detail in par 4.3 below.

²³⁸ Hahlo, H.R "Some Legal Aspects of Human Artificial Insemination" 1957 *SALJ* 167 at 174; *V v R* at p 1015.

²³⁹ Masters, N.C "Artificial insemination" 1953 *SALJ* 375 at 379; *V v R* at p 1016

²⁴⁰ Spiro, E "Artificial insemination and the law" 1972 *Acta Juridica* 213 at 214; *V v R* at p 1016.

was, for quite some time, considered to be illegitimate (Afrikaans: "buite-egtelik").²⁴¹ The effect of this, for example, in the context of intestate succession in South African law was that illegitimate children who had not been legitimated or adopted were deemed to be related to their mother but not to their father.²⁴² This is in accordance with the principle "*een wijf maakt geen bastaard*."²⁴³ Thus, illegitimate children did not succeed upon intestacy to their father and the father's relations but did succeed to their mother and the mother's relations.²⁴⁴ Conversely, the mother and her relations inherited *ab intestato* from the mother's illegitimate child, but not the father and the father's relations.²⁴⁵

The Intestate Succession Act²⁴⁶ commenced on 18 March 1988 and was later amended by the Law of Succession Amendment Act.²⁴⁷ The purpose of the Act was to regulate intestate succession and to provide for matters connected therewith. Section 1(2) of the Intestate Succession Act provided that notwithstanding the provisions of any law or the common law, but subject to section 5(2) of the Children's Status Act of 1987, illegitimacy will not affect the capacity of one blood relation to inherit the intestate estate of another blood relation. Section 8 of the Reform of Customary Law of Succession and Regulation of Related Matters Act²⁴⁸, which commenced on 20 September 2010, substituted section 1(2) of the Intestate Succession Act by providing that:

(2) Notwithstanding the provisions of any law or the common or customary law, but subject to the provisions of this Act and [section 5(2) of the Children's Status Act, 1987, illegitimacy] sections 40(3) and 297(1)(f) of the Children's Act, 2005 (Act No. 38 of 2005), having been born out of wedlock shall not affect the capacity of one blood relation to inherit the intestate estate of another blood relation.

²⁴¹ Judge Steyn also referred to the writings of Hugo de Groot and Van Leeuwen at 1012 - 1013. Van Leeuwen wrote that "these persons are legitimate who are procreated by their parents in lawful wedlock. Steyn stated: "In hierdie vertaalde vorm is die toetssteen van die kind se wettigheid duidelik die moment en omstandighede van die verwekking van die kind. Waar die kind verwek word ingevolge die instemming van die gewaande vader, is beide die wil en oogmerk van die wettig getroude man en vrou gerig op die verwekking van 'n kind uit hulle huwelik. Die tegniek van bevrugting is nie natuurlik nie, maar is dieselfde as by die nie-natuurlike bevrugting met die eggenoot se saad. Die enigste grondslag waarop so 'n kind as buite-egtelik beskou kan word is op grond daarvan dat die moeder owerspel sou gepleeg het deur die ontvangs van 'n onbekende skenker se saad in 'n nie-natuurlike proses van bevrugting."

²⁴² Grotius Inleidinge 2 27 28, Green v Fitzgerald, Fitzgerald v Green 1914 AD 88.

²⁴³ Grotius Inleidinge 2 27 28, Green v Fitzgerald, Fitzgerald v Green 1914 AD 88.

Grotius Inleidinge 2 27 28, Green v Fitzgerald, Fitzgerald v Green 1914 AD 88.

²⁴⁵ Grotius Inleidinge 2 27 28, Green v Fitzgerald, Fitzgerald v Green 1914 AD 88.

²⁴⁶ 81 of 1987.

²⁴⁷ 43 of 1992.

²⁴⁸ 11 of 2009.

The situation is different where a child is conceived by the use of donor gametes. Section 5(2) of the Children Status Act²⁴⁹ provided that no right, duty or obligation shall arise between the child and the donor of the gametes. Section 40(3) of the current Children's Act which determines the rights of a child conceived through artificial fertilisation, provides similarly that (subject to section 296 of the Act) no right, responsibility, duty or obligation arises between a child born of a woman as a result of artificial fertilisation and any person whose gamete(s) were used for the artificial fertilisation (gamete donor), and neither the blood relations of the gamete donor. Two exceptions to this legal position exist, namely: (a) when the gamete donor is the woman who gave birth to the child; or (b) when the gamete donor was the husband of such woman at the time of her artificial fertilisation.²⁵⁰

Section 297(1)(f) of the Children's Act (which determines the effect of a surrogacy agreement on the status of the child) provides that the child born as a result of a surrogacy arrangement will not have a claim for maintenance or of succession against the surrogate, her husband or partner or any of their relatives.

The legal-historical position of a child conceived through artificial fertilisation will be discussed in the next part with reference to legislation and case law.

4. LEGAL POSITION OF THE CHILD CONCEIVED BY ARTIFICIAL FERTILISATION PRIOR TO THE ENACTMENT OF THE CHILDREN'S ACT

Parenthood has traditionally been determined on the basis of a biological connection.²⁵¹ Already in 1991, Pretorius, who wrote the first legal thesis on surrogate motherhood in South Africa, argued that in the absence of legislation, the psychological intent of the parents in a surrogacy agreement should transcend the notion that biological ties should be the only criterion for parenthood.²⁵² The psychological and legal intent of the parties as the commissioning couple entering into a surrogacy

²⁴⁹ 82 of 1987.

²⁵⁰ Sec 40(3)(a) and (b) of the Act.

²⁵¹ Pretorius *Surrogate motherhood* 17.

²⁵² Pretorius *Surrogate motherhood* 17.

arrangement should be the primary consideration in determining their legal status (as parents) to the commissioned child.²⁵³

I will next discuss a few key judgments relating to the legal position of the child conceived by artificial fertilisation preceding the implementation of the Children's Act. Where relevant, the legal position regarding the parents of such child will also be highlighted.

4.1 $L v J^{254}$

In this 1985 case, the defendant, unable to father a child, was left with three options: to remain childless; to adopt a child or to have the plaintiff artificially fertilised with donor sperm (AID).²⁵⁵ The plaintiff underwent artificial fertilisation and she fell pregnant as a result thereof.²⁵⁶ The plaintiff instituted divorce proceedings in 1983 after a son was born.²⁵⁷ It emerged that the defendant never gave his consent to the plaintiff to undergo the AID procedure at the time it was performed.²⁵⁸ The plaintiff claimed sole guardianship and sole custody of the child, as well as maintenance for the child.²⁵⁹ The court concluded that the child is illegitimate, since the husband did not provide his consent for the wife to conceive a child via artificial fertilisation with donor sperm.²⁶⁰ The court made it clear that there is no duty on the husband to provide maintenance for the child of his wife if he is not that child's natural father.²⁶¹ For the court, the sole difference between the case where a child is born by AID or alternatively, out of an adulterous relationship, is that in the first-mentioned instance no accusation of adultery can be levelled against the wife.²⁶² The child's illegitimacy in L v J remains unaffected whether or not the child's mother underwent an artificial fertilisation procedure or enjoyed the thrills of illicit passion.²⁶³ The court explained that the obligation to provide

- ²⁵⁸ *L v J* at p 25.
- ²⁵⁹ *L v J* at p 21.
- 260 *L v J* at p 26.
- 261 *L v J* at p 26.
- ²⁶² *L v J* at p 26.
- ²⁶³ *L v J* at p 26.

²⁵³ Pretorius *Surrogate motherhood* 17.

²⁵⁴ 1985 (4) All SA 20 (C).

 $^{^{255}}$ L v J at p 21. Definition of AID in par 8.19 of ch 1.

²⁵⁶ *L v J* at p 21.

²⁵⁷ *L v J* at p 21.

maintenance is based on consanguinity and inasmuch as the husband is not obliged in law to support a step-child, he can certainly not be obliged to support a child born to his wife by AID without his consent.²⁶⁴ Berman J concluded that considerations of public policy do not dictate that a man in the defendant's position bears any obligation to maintain, upon divorce, the child of his wife, born *stante matrimonio* to her by AID undertaken without his consent.²⁶⁵ No legal obligation hence rests on the defendant to support the child.²⁶⁶

4.2 The Tzaneen triplets (1987)²⁶⁷

The first case of its kind in the world took place in 1987 in South Africa when a married grandmother gave birth to triplets as their surrogate mother.²⁶⁸ The genetic parents were her own daughter and the daughter's husband (son-in-law).²⁶⁹ As the children were born thirteen days before the Children's Status Act²⁷⁰ was promulgated, the children had to be registered as the children of their genetic parents and not that of the grandmother (who gave birth) and her husband.²⁷¹ It was stated that it would appear that the decision was in line with the traditional view that consanguinity determines legal parenthood and the genetic mother was therefore considered the legal mother of the children.²⁷² The Children's Status Act,²⁷³ enacted shortly after the birth of the triplets, aimed to provide legitimacy to children who had previously been considered

²⁶⁴ *L v J* at p 26.

²⁶⁵ *L v J* at p 28.

²⁶⁶ *L v J* at p 28.

²⁶⁷ Strauss, SA "Triplets to a Surrogate Grandmother in South Africa: Legal Issues" 1989 International Legal Practitioner 14(3) 71. Pretorius Surrogate motherhood 85. This case sparked a controversy amongst church leaders, medical practitioners and other interested parties. It is mentioned that the Federal Council of the Medical Association of South Africa concluded in 1986 that surrogate motherhood was undesirable.

²⁶⁸ Slabbert and Roodt "South Africa" 326. Strauss *Doctor, Patient and the Law* 190. Oluwaseyi and Oladimeji 2021 *Obiter* 31.

²⁶⁹ Pretorius *Surrogate motherhood* 135.

²⁷⁰ 82 of 1987.

Pretorius Surrogate motherhood 135. The legal questions were: (a) who the legal parents of the triplets are, and (b) in whose name should the triplets be registered. Strauss Doctor, Patient and the Law 190.

²⁷² Pretorius *Surrogate motherhood* 135.

²⁷³ 82 of 1987.

illegitimate.²⁷⁴ Interestingly, had the triplets been born two weeks later, they would have been considered the legal children of their grandmother.²⁷⁵

4.3 $V v R^{276}$

This was the first known decision in South Africa regarding the legal status of a child conceived by means of artificial fertilisation with sperm obtained from an unknown donor (AID).²⁷⁷ The wife (applicant) and the respondent were married on 31 December 1965.²⁷⁸ The parties divorced on 12 November 1974 after a child (son) was born on 16 September 1971.²⁷⁹ The applicant conceived by means of artificial fertilisation with sperm obtained from an unknown donor.²⁸⁰ Both the husband and wife agreed to the procedure.²⁸¹ After the divorce, the applicant brought an application wherein she asked the court to declare that the respondent is not the natural and/or legal father of the minor child and that only her (as applicant's) consent will be necessary to enable her new husband to adopt the child.²⁸²

The court stated that a child conceived by means of artificial fertilisation with the semen of an unknown donor (AID) that took place on request of the wife, with the consent of the husband, is regarded as an illegitimate child for the purposes of section 71 of the Children's Act 33 of 1960.²⁸³ The court further observed that a child born through artificial fertilisation with consent of the putative father would be considered illegitimate in terms of the old writers' literal judgments, since:²⁸⁴

(1) the child is born to a putative father during a lawful marriage;

²⁷⁴ Pretorius *Surrogate motherhood* 135.

²⁷⁵ Pretorius *Surrogate motherhood* 136. Strauss *Doctor, Patient and the Law* 190. SALC Project 65 at par 4.6.2.

²⁷⁶ 1979 (3) SA 1006 (T).

Strauss, SA "Triplets to a Surrogate Grandmother in South Africa: Legal Issues" 1989 International Legal Practitioner 14(3) 70. Strauss Doctor, Patient and the Law 183.
 V v P at p 1008

 ²⁷⁸ V v R at p 1008.
 ²⁷⁹ V v R at p 1008.

V V R at p 1008.V V R at p 1008.

V V R at p 1008.V V R at p 1008.

V V R at p 1008.V V R at p 1008. Strauss Doctor, Patient and the Law 183.

V v R at p 1000. Strauss Doctor, Patient and the Law 103.
 V v R at p 1015. Strauss Doctor, Patient and the Law 183.

²⁸⁴ *V v R* at p 1015.

- (2) the mother of the child at its conception had committed no act contrary to the marriage contract but an act in execution of the marriage contract where she had acted in consultation with and with the consent of the putative father;
- no man other than the putative father has maintenance obligations in respect of the child;
- (4) the married mother had not committed adultery in the conception of the child by means of artificial fertilisation as the result of the absence of "all wrongful sensuality at its conception" and as the result of the consent obtained from her husband to be regarded as the father of her child.

Moreover, the court held that there is no authority in our law which prevents the consenting husband of the wife who brings a child into the world by means of such artificial fertilisation being placed in a special relationship to the child born as a result thereof.²⁸⁵ The term "putative father" is an appropriate description of such a spouse, who, as a putative father, has obligations in respect of the maintenance, including rights of guardianship in respect of such a child.²⁸⁶ The mother of such a child has a right, which the court cannot limit, to consent to the adoption of her child.²⁸⁷ The consent of the respondent was thus not required for the adoption of the parties' child.²⁸⁸

4.4 J and Another v Director General, Department of Home Affairs and others²⁸⁹

This matter, heard by the Constitutional Court, deals with same-sex partners who had twins that were conceived through artificial fertilisation.²⁹⁰ Both applicants wanted to be registered and recognised as the parents of the twins.²⁹¹ Unfortunately, section 5 of the Children's Status Act did not permit the first applicant to become a legitimate parent

²⁸⁵ *V v R* at p 1016.

²⁸⁶ *V v R* at p 1016. Strauss *Doctor, Patient and the Law* 183.

²⁸⁷ *V v R* at p 1017.

²⁸⁸ *V v R* at p 1017.

²⁸⁹ 2003 (5) BCLR 463; 2003 (5) SA 621 (CC).

²⁹⁰ *J and Another v Director General, Department of Home Affairs and others* at par 2. The ova of the first applicant were used together with donor sperm.

²⁹¹ J and Another v Director General, Department of Home Affairs and others at par 3.

of the twins.²⁹² The court found the provisions of section 5 to be in conflict with the provisions of section 9(3) of the Constitution.²⁹³ Section 5 referred to "married" and "husband", which excluded some permanent same-sex life partners.²⁹⁴ The Constitutional Court declared section 5 of the Children's Status Act to be inconsistent with the Constitution to the extent that the word "married", in conjunction with "husband" did not make provision for permanent same-sex life partners.²⁹⁵ The court further struck the word "married" from the section and inserted the words "or permanent same-sex life partner" after the word "husband".²⁹⁶ The court thus found that the twins born to the applicants, through artificial fertilisation of the second applicant, is deemed for all purposes to be the legitimate children of the applicants.

5. STATUTES GOVERNING ARTIFICIAL FERTILISATION AND SURROGATE MOTHERHOOD PRIOR TO THE ENACTMENT OF THE CHILDREN'S ACT

5.1 Introduction

The preceding discussion described the legal historical development that influenced the way artificial fertilisation and surrogate motherhood were regulated prior to the implementation of the current Children's Act. The following section seeks to contextualise these developments, which need to be read in conjunction with certain provisions from other statutes to get a clear grasp of the exact legal position at the time. The regulation of artificial fertilisation and surrogate motherhood at the time may rightly be characterised as ambiguous and unsatisfactory. Those who had wanted to conceive a child via artificial fertilisation or surrogate motherhood had to patch the legal position together having regard to a patchwork framework characterised by piecemeal and haphazard amendments and revisions.

5.2 Child Care Act²⁹⁷

²⁹² J and Another v Director General, Department of Home Affairs and others at par 13. The section unfairly discriminated between married persons and persons in a permanent same-sex life partnership.

²⁹³ J and Another v Director General, Department of Home Affairs and others at par 14.

²⁹⁴ J and Another v Director General, Department of Home Affairs and others at par 7.

²⁹⁵ J and Another v Director General, Department of Home Affairs and others at par 28.

²⁹⁶ J and Another v Director General, Department of Home Affairs and others at par 28.

²⁹⁷ 74 of 1983.

The aim of the Child Care Act was primarily the protection and welfare of certain children and the regulation of adoption procedures.²⁹⁸ The Child Care Act was repealed by section 313 of the Children's Act 38 of 2005, with effect from 1 April 2010. Under the Child Care Act's dispensation, the commissioning parents in a surrogate relationship could only become the legal parents of the child if they followed the adoption procedure in terms of chapter 4 of the Child Care Act.²⁹⁹ Thus, the adoption procedure as set out in section 18 of the said Act would be relevant if the surrogate mother consented to the adoption of the child.³⁰⁰

Section 17 of the Child Care Act, however, prevented a biological parent from adopting his or her own child, as section 17(a) of the Act prevented the adoption of a child if the child was born of one of the parents.³⁰¹ Those eligible to adopt were: a husband and his wife jointly; a widower or widow or unmarried or divorced person; a married person whose spouse was the parent of the child; or the natural father of a child born out of wedlock.³⁰² Section 18(4) provided for the requirements that had to be met in an application for an order of adoption (in terms of subsection 18(2)) to the Children's Court before such an order may be granted. For instance, the applicant(s) had to be qualified to adopt the child in terms of section 17 and he/she/they had to have adequate means to maintain and educate the child;³⁰³ the applicant(s) had to be or must have been of good repute and person(s) fit and proper to be assigned with the custody of the child;³⁰⁴ and, importantly, that the planned adoption would serve the interests and conduce to the welfare of the child.³⁰⁵ Sections 24(1) and (2) prohibited any compensation for adoption.³⁰⁶ The ostensible object of this provision was to protect young and inexperienced women from financial coercion (by third parties) to give up their children for adoption. This section was never intended to regulate surrogacy.³⁰⁷

³⁰⁵ Sec 18(4)(c).

²⁹⁸ SALC Project 65 at par 4.3.1.

AB and another v Minister of Social Development at par 247. Ex parte WH and others at par 35. Slabbert and Roodt "South Africa" 326.

³⁰⁰ AB and another v Minister of Social Development at par 247. Ex parte WH and others at par 57.

 ³⁰¹ Pretorius Surrogate motherhood 141. It was however possible for a married person to adopt a child that was born from the spouse (stepchild scenario).
 ³⁰² See 17 of Child Care Act

³⁰² Sec 17 of Child Care Act.

³⁰³ Sec 18(4)(a).

³⁰⁴ Sec 18(4)(b).

³⁰⁶ Sec 24. Pretorius *Surrogate motherhood* 90.

³⁰⁷ Pretorius *Surrogate motherhood* 93.

Although surrogacy was not expressly regulated by the Child Care Act *per se*, the Act did provide guidance as to the legal situation should a child have been born following a surrogacy agreement. Such a child could only be considered the child of the commissioning parents after the legal adoption of the child. The possibility existed that the surrogate mother could refuse to consent to the adoption of the child by the commission parents, which would have left the latter without a legal remedy or recourse.

5.3 Children's Status Act³⁰⁸

The *mater semper certa est* maxim was codified in this Act in respect of children conceived by means of artificial fertilisation or *in vitro* fertilisation.³⁰⁹ Section 5 of the Children's Status Act determined the legal consequences of artificial fertilisation regarding the status of children born as a result of artificial fertilisation.³¹⁰ Section 5(1)(a) provided that children born as a result of artificial fertilisation with donor sperm and ova are considered the legitimate children of the women giving birth to the child and her husband, provided that the husband gave his consent to the artificial fertilisation.³¹¹ Section 5(2) determined that no right, duty or obligation would arise between the child and the donor of the gametes.³¹² Such right, duty or obligation would only arise where (a) a person is the woman who gave birth to that child; or (b) a person is the husband of such a woman at the time of the said artificial fertilisation.³¹³ Thus,

³⁰⁸ 82 of 1987. (Die Wet op Status van Kinders) The act was repealed by Act 38 of 2005.

³⁰⁹ Lupton, M.L "Artificial wombs: medical miracle, legal nightmare" 1997 *Medicine and Law* 16 628. Slabbert and Roodt "South Africa" 326.

³¹⁰ Carstens and Pearmain *Foundational Principles* 181. SALC Project 65 at par 4.4.1.

³¹¹ Sec 5(1) of the Children's Status Act 82 of 1987. Lupton, M.L "Artificial wombs: medical miracle, legal nightmare" 1997 *Medicine and Law* 16 628. Pretorius *Surrogate motherhood* 135. It is stated that surrogate motherhood falls within the ambit of this section. Slabbert and Roodt "South Africa" 326. South African Law Commission, project 110, Review of the Child Care Act (December 2002), to be accessed from https://www.justice.gov.za/salrc/reports/r.pr110_01_2002dec.pdf_Mills_2010_*Stell_LR_430*

https://www.justice.gov.za/salrc/reports/r pr110 01 2002dec.pdf. Mills 2010 Stell LR 430. Strauss Doctor, Patient and the Law 184 and 189.

³¹² Sec 5(2) of the Children's Status Act 82 of 1987. Lupton, M.L "Artificial wombs: medical miracle, legal nightmare" 1997 *Medicine and Law* 16 628. Pretorius *Surrogate motherhood* 91. It will thus not be possible for a donor to claim any parental rights to the child and the child will not be able to claim maintenance from the donor Mills 2010 *Stell LR* 430. The writer explained that this meant that in circumstances where a surrogate had changed her mind in not wishing to give her consent to the adoption of the child by the commissioning parent(s), that she was entitled to do so in light of the fact that the surrogate agreement would in all probability have been considered as *contra bones mores*.

³¹³ Sec 5(2) of the Children's Status Act 82 of 1987. Sec 3 determined that artificial insemination of a woman meant introduction by other than natural means of a male gamete(s) into the internal

section 5(2) provided no guidance as to the implementation of a surrogacy agreement.³¹⁴

The Children's Status Act aimed to provide legitimacy to children who were previously considered illegitimate in South African law³¹⁵ by statutorily codifying both the common law maxims of *mater sempter certa est* and *pater is est quem numptiae demonstrant* (presumption of paternity) regarding artificial fertilisation with donor sperm and/or embryo transfer.³¹⁶ A child born as a result of artificial fertilisation with donor sperm had to be registered as the child of the woman giving birth and her husband, and both were considered to be the legal parents of the child.³¹⁷

The codification of the presumption of the paternity maxim (*pater is est quem numptiae demonstrant*) in the Children's Status Act, however, provided an additional obstacle to a surrogacy agreement in the sense that the surrogate and her husband were considered the legal parents of the child, which is contrary to the expectations and the intention of all the parties to the surrogate motherhood agreement.³¹⁸ Motherhood was hence attributed to a mother who never had the intention to keep the child, and fatherhood to a father whose involvement was minimal at best.³¹⁹

reproductive organs of the woman or the placing of the product of a union of a male and a female gamete(s) which have been brought together outside the human body in the womb of that woman for the purpose of human reproduction. Pretorius *Surrogate motherhood* 141. It was possible for a commissioning father to rebut the presumption of paternity before the enactment of the Act. After the enactment of the Act, his position was equated with that of a sperm donor and his rights towards the child (his natural offspring) were terminated by sec 5(2)(a). Although it was seen to be in the child's best interest that the rights of a sperm donor be terminated, the effect was unreasonable for a commissioning father in a surrogate agreement. He was in danger of losing his child as he was no longer in a position to rebut the presumption of paternity. The only available option for the commissioning parents was to adopt the child, provided that the surrogate and her husband gave the requisite consent. Strauss *Doctor, Patient and the Law* 184.

³¹⁴ SALC Project 65 at par 4.4.2.

³¹⁵ Pretorius *Surrogate motherhood* 93 and 152. To afford the children protection against the detrimental effects of illegitimacy in general.

³¹⁶ Pretorius *Surrogate motherhood* 91. SALC Project 65 at par 4.4.2.

³¹⁷ Pretorius *Surrogate motherhood* 91.

³¹⁸ Pretorius *Surrogate motherhood* 140-141. The child's best interests was clearly not served in the circumstances where parental rights were awarded to a man who does not want a child and who played a minor role in the agreement. Should the surrogate mother have handed the child over to the commissioning parents after the birth, her husband's consent was a prerequisite for adoption by the commissioning parents. Slabbert and Roodt "South Africa" 326. Mills 2010 *Stell LR* 430. Strauss *Doctor, Patient and the Law* 189.

³¹⁹ Nicholson and Bauling 2013 *De Jure* 513.

Both married and unmarried women had legal access to donor sperm since 1997 and would thus have been able to opt for and participate in artificial fertilisation procedures.³²⁰ Clark argues that although the issue of surrogacy was not addressed directly by the Children's Status Act or its Regulations, the Act had far-reaching implications for many surrogacy cases, as the definition of artificial fertilisation in the Children's Status Act was broad enough to cover many of the procedures used to implement surrogacy agreements.³²¹ Although Clark does not clarify this conclusion, one would assume that the reference to "a woman" in the definition of artificial fertilisation was wide enough to include the artificial fertilisation of the surrogate mother. Section 5(3) defined artificial fertilisation in respect of a woman as the introduction by other than natural means of a male gamete(s) into the internal reproductive organs of the woman;³²² or placing of the product of a union of a male and female gamete(s) which have been brought together outside the human body in the womb of that woman for the purpose of human reproduction.³²³ Thus, this definition included both artificial insemination and *in vitro* fertilisation.³²⁴

Thus, in terms of the Children's Status Act, a woman, if she was married and gave birth to a child, and her husband (if he consented to her artificial fertilisation), were regarded as the legal parents of the child that was born.³²⁵ On the other hand, in circumstances where the husband did not give his consent to the artificial fertilisation and a child was born, the child would have been deemed born out of wedlock and only the woman would acquire automatic parental rights and responsibilities.³²⁶ It followed that the commissioning parents in a surrogate relationship could only become the legal parents of such a child by following the legal route of adoption in terms of the Child Care Act or to approach the High Court for guardianship or custody of or access to the child.³²⁷

³²⁰ GN R1354 GG 18362 of 17 October 1997 deleting regulation 8(1) of the Human Tissue Act 1983 and amending reg 5(d) and reg 9(e). LexisNexis *Family Law service* at par J103.

³²¹ Clarke 2000 *Stell LR* 13.

³²² Sec 5(3)(a) of the Children's Status Act.

³²³ Sec 5(3)(b) of the Children's Status Act.

³²⁴ The definitions of artificial insemination and *in vitro* fertilisation is discussed in par 8.3 and 8.5 in ch 1.

³²⁵ Slabbert and Roodt "South Africa" 326. *AB and another v Minister of Social Development* at par 247.

³²⁶ Strauss, SA and Maré, MC *Mediese Reg* (Study Guide, LCR404-U, University of South Africa) 166.

³²⁷ Ch 4, sec 20 of the Child Care Act. Clarke 2000 *Stell LR* 13. Mills 2010 *Stell LR* 431. Nicholson 2013 *SAJHR* 498.

Section 5(1)(b) of the Children's Status Act created the presumption that if a surrogate mother was married, both she and her husband consented to the artificial fertilisation and thus, the child born of such fertilisation was deemed to be their legitimate child.³²⁸ Thus, a right, duty or obligation did not arise between a child born as a result of the artificial fertilisation of a woman and a person whose gamete or gametes has/have been used for the artificial fertilisation and the blood relations of that person, unless that person is the woman who gave birth to the child or is the husband of the woman at the time of the artificial fertilisation.³²⁹

It was therefore possible, however, that the commissioning parents would have faced legal uncertainty in the circumstances where the surrogate refused to hand over the child to them for adoption.³³⁰ Clarke argues that where the surrogate backed out of a surrogate agreement, the commissioning couple would probably not have been able to enforce the said agreement, since the agreement would in all likelihood have been considered *contra bones mores* on the grounds that it constituted a possible devaluation or alteration of the concept of the family and the marriage relationship.³³¹ The SALC pointed out that section 5 of the Children's Status Act did not make provision for the implementation of the intention of the parties to a surrogacy agreement.³³² It would appear that the legislature had recognised that creating a process that allowed for the separation of so-called 'genetic' parentage from 'social' parentage, is a process best dealt with via legal recognition of the social parentage in the context of pre-existing legal obligations to children.³³³

A consideration of the provisions considered above, as well as a reading together of the Children's Status Act and the Child Care Act, points to a regime that obstructed and complicated contractual surrogacy agreements in South Africa. Chapter 19 of the Children's Act, discussed in detail elsewhere in this thesis, was based on some of the recommendations contained in the report of the South African Law Commission (as it

³²⁸ Slabbert and Roodt "South Africa" 326 Strauss, SA "Triplets to a Surrogate Grandmother in South Africa: Legal Issues" 1989 *International Legal Practitioner* 14(3) 70. Lupton, M.L "Artificial wombs: medical miracle, legal nightmare" 1997 *Medicine and Law* 16 628.

³²⁹ Sec 5(2).

³³⁰ Slabbert and Roodt "South Africa" 326.

³³¹ Clarke 2000 *Stell LR* 13.

³³² SALC Project 65 at par 1.1.3.

Lupton, M.L "Artificial wombs: medical miracle, legal nightmare" 1997 *Medicine and Law* 16 628.

was known at the time). Many of these recommendations were the direct result of the challenges observed and outlined in the discussion above.

5.4 Human Tissue Act³³⁴

The period between 12 July 1985³³⁵ and 2 May 2005³³⁶ marked the time during which the Human Tissue Act and Regulations³³⁷ governed artificial fertilisation procedure in South Africa. The Human Tissue Act was repealed and replaced by the NHA and the Regulations, discussed elsewhere in this thesis. It is important to note that the Human Tissue Act required that consent (by the husband) to artificial fertilisation had to be in writing.³³⁸ If the consent requirement was disregarded, the child was deemed illegitimate.³³⁹ Only married women could be artificially fertilised in terms of regulation 8(1) of the Regulations to the Human Tissue Act.³⁴⁰ These Regulations (regarding the artificial insemination of persons, and related matters promulgated in terms of the Human Tissue Act), made no reference to surrogate motherhood within their ambit and they only referred to artificial insemination by a donor (AID).³⁴¹

Section 22 of the Human Tissue Act dealt with artificial fertilisation and determined that in the event that a gamete is removed from the body of a living person, that gamete shall not be used for the artificial fertilisation of another person unless the person affecting the artificial fertilisation acts in accordance with a code of practice for artificial fertilisation, published by the then Department of Health and Welfare, specified in the regulations.³⁴² Section 23 regulated the control of the removal and use of tissue and blood, and of artificial fertilisation. The removal of tissue from the body of a living person or the use thereof or to transplant tissue so removed in the body of another living

³³⁴ 65 of 1983. This act was repealed by Act 61 of 2003.

³³⁵ Human Tissue Act. The commencement date of the Human Tissue Act 65 of 1983.

³³⁶ National Health Act. The commencement date of the National Health Act 61 of 2003.

³³⁷ Regulations regarding artificial insemination and related matters, published in GG R 1182 of 20 June 1986.

³³⁸ Reg 8(1) of Regulations regarding artificial insemination, 1986.

³³⁹ Strauss, SA "Triplets to a Surrogate Grandmother in South Africa: Legal Issues" 1989 International Legal Practitioner 14(3) 70.

Reg 8(1) of Regulations regarding artificial insemination, 1986. SALC Project 65 at par 4.2.6.

³⁴¹ Strauss *Doctor, Patient and the Law* 181 and 188. The provisions of the Human Tissue Act and the Regulations affected certain important aspects of surrogacy.

³⁴² Sec 22. This section was repealed by section 15 of Act 51 of 1989. Strauss *Doctor, Patient and the Law* 181.

person was only allowed by a medical practitioner or dentist or a person acting under a medical practitioner's or dentist's supervision.³⁴³ The withdrawal of any blood from the body of a living person or the administration of blood or a blood product to a living person was also only allowed by a medical practitioner or dentist or a person acting under their supervision.³⁴⁴ Importantly, no person was allowed to affect the artificial fertilisation of a person except a medical practitioner or a person acting under his supervision.³⁴⁵

Regulations 6 and 10 contained specific provisions concerning the donor and recipient files respectively, and provided that strict confidentiality applied relating to the content of the files whose content may not have been made available to any other person for inspection.³⁴⁶ It is important to note that the only information that the medical practitioner was allowed to disclose or make available to the recipient and her husband, was set out in regulation 6(1)(a)(ii), and these were the prospective donor's age, height, mass, eye colour, hair colour, complexion, population group, nationality, sex, religion, occupation, highest educational qualification, and fields of interest.

The above reference to the disclosure of information relating to the gamete donor raises the important issue regarding the rights of children to know their genetic heritage, considering, for example, that article 7 of the United Nations Convention on the Rights of the Child (1989) provides that children have a fundamental right to know the truth about their genetic origin. One may argue that denying a child born as a result of artificial fertilisation access to information about his or her genetic parent may possibly be contrary to the child's rights in this regard. This issue will be discussed in more detail elsewhere in the thesis.

³⁴³ Sec 23(1)(a).

³⁴⁴ Sec 23(1)(b).

³⁴⁵ Sec 23(2). This section was substituted by section 16 of Act 51 of 1989. Strauss *Doctor, Patient and the Law* 181-182.

³⁴⁶ Reg 6(2)(e) and reg 10(2)(b) of Regulations regarding the artificial insemination of persons, 1986. Except where any other law otherwise provides or any court so orders.

6. CONCLUSION

This chapter has shown that the legal historical development regarding the legal status of children born as a result of artificial fertilisation was in a regulatory flux, compounded by the confusion regarding the legal consequences for the parties to a surrogacy agreement, including the surrogate mother who was artificially fertilised. As surrogate motherhood was not directly regulated, some of the legal consequences following the implementation of certain statutory legal amendments were outright unjust or bizarre.

The current legal position in South Africa will next be discussed, with specific reference to chapter 19 of the Children's Act³⁴⁷ and how this Act improved legal certainty to those who wish to conceive a child via artificial fertilisation and surrogate motherhood.

³⁴⁷ 38 of 2005.

CHAPTER 3

THE SOUTH AFRICAN LEGAL FRAMEWORK GOVERNING SURROGATE MOTHERHOOD

1. INTRODUCTION

The common law position regarding artificial fertilisation in South Africa prior to the promulgation of the Children's Act,³⁴⁸ discussed in chapter two, provided that a child born to a married couple where the child conceived by means of artificial fertilisation from the sperm of a donor (AID) was considered illegitimate. The need for legal reform to address the regulation of surrogacy and artificial fertilisation to protect the interests and rights of the different parties, and especially the child born as a result thereof, was self-evident. The SALC rightly pointed out that the statutory position and the common law raised more questions than it could provide answers in the implementation of surrogacy agreements³⁴⁹ and that the legal position at the time was clearly not intended to regulate surrogate motherhood.³⁵⁰ The introduction of the Children's Act consisting of a separate section on surrogate motherhood was indeed welcomed by legal practitioners and those considering conceiving a child via a surrogacy arrangement.³⁵¹ This chapter will explore chapter 19 of the Children's Act in detail, in addition to recent case law regarding surrogacy agreements that served before the courts in recent times.³⁵² The NHA and some of the Regulations promulgated in terms of chapter 8 of the NHA that are indirectly relevant to issues regarding artificial fertilisation will also briefly be referred to, where relevant. Chapter 5 of this thesis will canvass the role and impact of the NHA and regulations in more detail.

³⁴⁸ 38 of 2005.

³⁴⁹ SALC Project 65 at par 4.10.

³⁵⁰ SALC Project 65 at par 7.1.

³⁵¹ Nicholson and Bauling 2013 *De Jure* 516. Nöthling Slabbert, M "Legal issues relating to the use of surrogate mothers in the practice of assisted conception" 2012 *South African Journal of Bioethics and Law* 5(1) 34. There are however still some practical legal and ethical issues that remain although there has been some guidance in recent judgments. Tager 1986 *SALJ* 386. There was no expressive reference to surrogate mother in both the Children's Act of 1960 and the Child Care Act of 1983. Baase M "The ratification of inadequate surrogate motherhood agreements and the best interest of the child" 2019 *Potchefstroom Electronic Law Journal* 22 1. South Africa has developed domestic legislation that governs surrogacy matters within the country. SALRC Project 140 82. Florescu, S and Sloth-Nielsen, J "Visions on surrogacy – From North to South: The approach of the Netherlands and South Africa to the issue of surrogacy and the child's right to know his origin" 2017 *International Survey of Family Law* 2017 247.

³⁵² Jordaan DW "Surrogate Motherhood in illness that does not cause infertility" 2016 SAMJ 684.

This chapter will first turn to the rights of the child conceived by artificial fertilisation with reference to the Children's Act.

2. THE RIGHTS OF THE CHILD CONCEIVED BY ARTIFICIAL FERTILISATION

The Children's Act expressly regulates the rights of children born from a surrogacy agreement. The Children's Act was passed to give effect to the provisions of section 28 of the Constitution of South Africa.³⁵³ Further, this Act enforces South Africa's obligations regarding the well-being of children in terms of the international standards and Conventions, which promotes children's rights.³⁵⁴

The pertinent sections of the Children's Act relating to artificial fertilisation are sections 40 and 296 of the Act. In analysing these sections, reference is made to some of the recommendations of the SALC and the Ad Hoc Committee to compare to what extent these suggestions for reform have been incorporated into the Children's Act.³⁵⁵

2.1 Legal status and related matters

Section 40(1)(a) of the Children's Act determines the rights of the child conceived by artificial fertilisation and states the following:

Whenever the gamete or gametes of any person other than a married person or his or her spouse have been used with the consent of both such spouses for the artificial fertilisation of one spouse, any child born of that spouse as a result of such artificial fertilisation must for all purposes be regarded to be the child of those spouses as if the gamete or gametes of those spouses had been used for such artificial fertilisation.³⁵⁶

A child born from artificial fertilisation to a woman will be seen as legitimate, provided that both spouses gave their consent to the artificial fertilisation process.

The surrogacy process legally starts when the surrogate agreement is concluded, followed by the fertilisation of the surrogate mother. Not only must the surrogate mother consent to the procedure, but it is also presumed that both spouses have consented

³⁵³ Sec 2(a) and (b) of Children's Act.

³⁵⁴ Sec 2(d) of the Children's Act.

³⁵⁵ The SALC report on surrogate motherhood in terms of project 65 and the report of the Ad Hoc Committee on the report of the SA Law Commission on surrogate motherhood, dated 11 February 1999.

³⁵⁶ Sec 40 (1)(a) of the Act. Dhai and McQuoid-Mason *Bioethics, Human Rights and Health Law* 142.

to the procedure until the contrary is proved.³⁵⁷ The same applies to partners in a samesex life partnership.³⁵⁸

Section 40(2) furthermore provides, subject to section 296 of the Act, that a child born of a woman who has been artificially fertilised, by the gamete(s) of any person, must be regarded to be the child of that woman for all purposes.³⁵⁹ Moreover, section 40(3) determines (also subject to section 296 of the Act), that no right, responsibility, duty or obligation in law arises between a child born of a woman as a result of artificial fertilisation and any person whose gametes were used for the fertilisation, and neither his or her blood relations.³⁶⁰ Two exceptions to this rule is: (a) when that person is the woman who gave birth to the child; or (b) when that person was the husband of such woman at the time of her artificial fertilisation.³⁶¹ Importantly, the result of an invalid surrogate agreement is that the commissioning parents are not considered the legal parents of the child regardless of whether either or both of them have donated gametes for the artificial fertilisation of the surrogate.³⁶² Section 26(2)(b) of the Children's Act furthermore prevents a person who is biologically related to a child as a result of being a gamete donor for purposes of artificial fertilisation, to make application as a biological father claiming paternity of the child.³⁶³

³⁵⁷ Sec 1(b) of the Act. Bosman-Sadie and Corrie *Practical approach to the new Children's Act* 60. It is stated that it is accepted until disputed that both spouses gave permission for the artificial fertilisation and that both parties will, after the birth of the child, obtain full parental rights and responsibilities. Dhai and McQuoid-Mason *Bioethics, Human Rights and Health Law* 142. The authors recommend that the practitioner must ensure that there is proper informed consent from both the parties if the husband is going to be held responsible for the expenses of the procedure and the maintenance of the child before proceeding with the artificial fertilisation procedure.

³⁵⁸ Sec 40(1)(a). Both the partners to a same-sex life partnership are deemed to be the parents of a child born to either one of the partners as a result of artificial fertilisation where the gamete or gametes of any person other than that woman was used. Thus, the child is deemed to be their legitimate child providing both partners consented to the use of artificial fertilisation. LexisNexis *Family Law service* at par R18.

³⁵⁹ Sec 40(2) of the Act. Dhai and McQuoid-Mason *Bioethics, Human Rights and Health Law* 143.

³⁶⁰ Sec 40(2) of the Act. Heaton 2015 *THRHR* 35. Thus, if sec 296 does not apply, the woman who was artificially fertilised and her husband or partner (if any) acquire parental responsibilities and rights to the exclusion of the gamete donors. An invalid surrogate agreement falls outside the ambit of sec 296 and thus, the provision in sec 40(3) does not apply to it.

³⁶¹ Sec 40(3) of the Act. Bosman-Sadie and Corrie *Practical approach to the new Children's Act* 60. It would appear that where the husband has failed to consent to the artificial fertilisation of his wife with gametes of a donor, the child born will be regarded as extra-marital. Sec 40(3)(a) and (b). The current rule is defensible on the basis that the burdens of fatherhood should not be forced upon a person who is not the child's biological father in the absence of some act indicating a voluntary assumption of those obligations. Dhai and McQuoid-Mason *Bioethics, Human Rights and Health Law* 143.

³⁶² Heaton 2015 *THRHR* 35.

³⁶³ Sec 26(2)(b) of the Act. SALRC Project 140 147.

Palm and Hirsh rightly observe that the most thorny issues arising from the use of artificial fertilisation are those in respect of the legal status of the child and the parental rights and obligations at divorce.³⁶⁴ Depending on the type of artificial insemination method that is employed, it is possible for a commissioned child to have up to six parents.³⁶⁵ This explains why it is important that the rights and obligations of each party to the surrogacy agreement be well defined, as it is possible that parentage, custody and guardianship of the child may become contested issues in the event of a disagreement between the parties.³⁶⁶ The evolving world of multi-parenting is increasingly challenging the traditional notions of a parent's rights and responsibilities with regard to a child.³⁶⁷ Regardless of the consequences following surrogacy agreements, children born as a result of a surrogate agreement should enjoy the same rights as children born through natural means.³⁶⁸

A recent High Court case, discussed next, provides guidance on parental rights and responsibilities of same-sex couples in respect of a child born as a result of assisted reproduction to them.

2.1.1 EJ and others v Haupt NO³⁶⁹

The applicants brought an application for the acquisition of responsibilities and rights regarding their child, born on 10 March 2021 and conceived through artificial fertilisation using a home insemination kit.³⁷⁰ The first and second applicant ("the applicants") were married on 16 November 2019 in terms of the Civil Union Act.³⁷¹

³⁶⁴ Palm MT and Hirsch HL "Infertility and sterility: Legal implications of artificial conception" 1982 *Medicine and Law* 1 47. It is stated that courts have alluded to the notion that AIH (artificial insemination husband) does not create legal problems as the child is the natural offspring of both the husband and wife.

³⁶⁵ Report of the Ad Hoc Committee 29. The genetic parents, the surrogate and her husband or partner and the commissioning parents.

³⁶⁶ Report of the Ad Hoc Committee 29.

³⁶⁷ Quinn, CM "Mom, Mommy & Daddy and Daddy, Dad & Mommy: Assisted Reproductive Technologies & the Evolving Legal Recognition of Tri-Parenting" 2018 *Journal of the American Academy of Matrimonial Lawyers* 31(1) 177.

³⁶⁸ Oluwaseyi and Oladimeji 2021 *Obiter* 23.

³⁶⁹ 2022 (1) SA 514 (GP).

³⁷⁰ EJ and others v Haupt NO at par 1.

³⁷¹ EJ and others v Haupt NO at par 15.

The applicants requested a good friend (DHR) to become a gamete donor for them.³⁷² The first applicant became pregnant through the use of a home insemination kit.³⁷³ The applicants entered into a donor agreement with DHR.374 The question before the court was whether section 40 of the Children's Act makes provision for a spouse in a civil union marriage to automatically obtain parental rights and responsibilities of a child born to the other spouse during the course of their marriage where there is no genetic link between the said spouse and the child.³⁷⁵ The applicants argued that section 40 does not properly deal with the issue of same-sex female couples obtaining parental rights and responsibilities by operation of law.³⁷⁶ They further argued that the wording of sections 40(1) and 40(3) relate to a married heterosexual couple and not to a samesex female couple.³⁷⁷ Section 292 to section 303 make provision for the acquisition of parental rights and responsibilities in respect of surrogacy, and although these refer to a single person, husband, wife or partner, they do not apply to the partner or spouse of a child that is born as a result of artificial fertilisation.³⁷⁸ Another issue for the court to consider was whether the interests of the parties and the child are adequately protected and provided for by the current prevailing legislation and specifically section 40.379

Section 40(1) clearly relates to spouses only.³⁸⁰ Section 40(3)(b) makes reference to the word "husband", which means that only a man will be able to provide the second gamete for fertilisation of the ovum.³⁸¹ This is aligned with the provision that no right, responsibility, duty or obligation arise between the sperm donor (unless he is the husband of the woman who is artificially fertilised) and the child (born from artificial fertilisation) to prevent any common law obligation that arise between the sperm donor and the child.³⁸² *In casu*, Neukircher J rightly observes that because the word "marriage" includes a Civil Union, the word "spouse" as referred to in section 40(1),

³⁷² *EJ and others v Haupt NO* at par 18.

³⁷³ *EJ and others v Haupt NO* at par 20.

³⁷⁴ *EJ and others v Haupt NO* at par 21.

³⁷⁵ *EJ and others v Haupt NO* at par 4. The court also had to answer the question of whether sec 40 is unconstitutional. This aspect will be dealt with in ch 4.

³⁷⁶ EJ and others v Haupt NO at par 28.2.

³⁷⁷ EJ and others v Haupt NO at par 28.3.

³⁷⁸ *EJ and others v Haupt NO* at par 28.4.

³⁷⁹ *EJ and others v Haupt NO* at par 40.

³⁸⁰ EJ and others v Haupt NO at par 59.

³⁸¹ EJ and others v Haupt NO at par 60.

³⁸² EJ and others v Haupt NO at par 60.

should include a person married in terms of the Civil Union Act.³⁸³ Section 40(1) should thus apply to a same-sex couple too.³⁸⁴

The court explains that the purpose of section 40 is to create legal certainty in respect of acquiring parental rights and responsibilities where the child is born as a result of artificial fertilisation.³⁸⁵ The use of the word "spouse" in section 40(1) furthermore does not contradict the use of the word "husband" in section 40(3)(b), as each provision clearly has its own separate legal focus.³⁸⁶ In view of the conclusion that section 40(1) includes same-sex female couples, it automatically confers rights and responsibilities on the spouses of a child that is born as a result of artificial fertilisation.³⁸⁷ Both the applicants thus have the same rights and responsibilities in respect of the child from the moment of the child's birth and no adoption application has to be brought by the second applicant in respect of the said child.³⁸⁸

2.2 Access to information regarding a child's genetic parents

Where a child is born because of a surrogacy agreement, it is possible that the child may be genetically related to only one of the commissioning parents. There is a chance that the child may want to know more about his or her genetic parents, especially the genetic "donor" parent, at some stage.

In South African law, different laws apply to a child's right to access information relating to his or her gamete donor "parent", which include the Constitution (chapter 2); the Promotion of Access to Information Act,³⁸⁹ (hereafter PAIA), as well as the Children's Act. Section 32 of the Constitution provides that every person has the right of access to any information held by the state and any information that is held by another person, which is essential for the exercise or protection of any rights of such person.³⁹⁰ Section

³⁸³ *EJ and others v Haupt NO* at par 63.

³⁸⁴ EJ and others v Haupt NO at par 64 and 73.

³⁸⁵ *EJ and others v Haupt NO* at par 65.

³⁸⁶ EJ and others v Haupt NO at par 71.

³⁸⁷ *EJ and others v Haupt NO* at par 74. The limited application of sec 40(3)(b) does not have a bearing on the present factual matrix.

³⁸⁸ EJ and others v Haupt NO at par 74.

³⁸⁹ 2 of 2002.

³⁹⁰ Sec 32(1) of the Constitution of South Africa.

3 of the PAIA³⁹¹ applies to a record of a private body and a public body irrespective of when it came into existence.³⁹² Section 9 sets out the objects of the Act³⁹³ and provides in section 9(a) that the Act's purpose, among others, is to give effect to the constitutional right of access in terms of section 32 of the Constitution,³⁹⁴ subject to justifiable limitations,³⁹⁵ aimed at the reasonable protection of privacy.³⁹⁶ The manner in which the Act should give effect to the constitutional right should balance this right with any other rights, including those rights as set out in the Bill of Rights in chapter 2 of the Constitution.³⁹⁷

Section 30 of PAIA provides for the regulation of access to health and other records. In the event that the disclosure of the record (held by a medical practitioner), requested by the relevant person may cause serious harm to the person's mental- or physical health or well-being in the opinion of the information officer, he or she may consult with a medical practitioner before providing the record or information so requested.³⁹⁸ A person with parental responsibilities may make a request for a record on behalf of a child younger than sixteen years old.³⁹⁹ In the event that the medical practitioner—with who was consulted—is of the opinion that the disclosure of the record would be likely to cause serious harm to the relevant person's physical or mental health or well-being, the information officer may only provide access to the record if the relevant person proves that adequate provision is made for such counselling or other arrangements as are reasonably practicable before, during or after the disclosure of the record.⁴⁰⁰ The person responsible for such counselling or arrangements must be given access to the record before access is provided to the relevant person requesting the access.⁴⁰¹ Section 50 determines the right to access any record held by a private body⁴⁰² by providing that access to any record of a private body must be given to the requester if

³⁹⁴ Sec 9(a) of PAIA.
 ³⁹⁵ Sec 9(b) of PAIA.

³⁹⁷ Sec 9(b)(ii) of PAIA.

- ³⁹⁹ Sec 30(2)(a) of PAIA.
- ⁴⁰⁰ Sec 30(3)(a) of PAIA.
- ⁴⁰¹ Sec 30(3)(b) of PAIA.
- ⁴⁰² Sec 50 of PAIA.

³⁹¹ 2 of 2002.

³⁹² Sec 3 of PAIA.

³⁹³ Sec 9 of PAIA. ³⁹⁴ Sec 9(2) of PAIA

 $^{^{396}}$ Sec 9(b) (i) of PAIA.

³⁹⁸ Sec 30(1) of PAIA. Sec 30(1)(a) and 30(1)(b). The relevant person is the requester or a person making a request on behalf of a person to whom the information relates to. The medical practitioner is nominated by the relevant person.

the record requested is required for the exercise or protection of any rights (section 50(1)(a)), subject to the proviso that the relevant person has complied with the relevant procedural requirements set out in the Act.⁴⁰³ Section 50(3) determines that a request for a record from a private body includes a request for access to a record containing personal information about the relevant person.⁴⁰⁴

The reason why these sections are relevant for this chapter is that they may apply to records kept in hospitals where children are born because of a surrogacy agreement or artificial fertilisation. These children may justifiably want to have access to information necessary for the protection of their rights. They may request access to hospital files themselves (if older than sixteen years old) or other persons with parental rights, may request the information on their behalf.

Against the background of PAIA, it is also necessary to consider section 41(1) of the Children's Act, titled "Access to biographical and medical information concerning genetic parents". This provision directs that a child born because of artificial fertilisation in general and surrogacy, including the guardian of such child, is entitled to have access to any medical information regarding such child's genetic parents (section 41(1)(a)). Section 41(1)(b) furthermore provides that such child (or his or her guardian) is entitled to access *any other information* concerning the child's genetic parents, provided that the child is eighteen years old or older (section 41(1)(b)). Section 41 must be read together with sections 292 to 302 of the Children's Act.⁴⁰⁵ Section 41 makes it clear that the information that may be provided to the child is limited to medical information relating to his or her genetic parents.⁴⁰⁶ Only after the child turns eighteen years old is he or she allowed access to any other information relating to his or her genetic parents.⁴⁰⁷ The child is not allowed to have access to information that will reveal the identity of the gamete donors or that of the surrogate mother.⁴⁰⁸

⁴⁰³ Sec 50(1)(b) of PAIA.

⁴⁰⁴ Sec 50(3) of PAIA.

⁴⁰⁵ Children's Act 38 of 2005. Davel, CJ and Skelton, AM Commentary on the Children's Act (Jutastat e-publications, revision service 9, 2018 chapter 19) 36. Sloth-Nielsen "Surrogacy in South Africa" 196.

⁴⁰⁶ Sec 41(1) of the Act. Dhai and McQuoid-Mason *Bioethics, Human Rights and Health Law* 143.

 $^{^{407}}$ Sec 41(1) of the Act.

⁴⁰⁸ Sec 41(2) of the Act. SALRC Project 140 103 – 104 and 147. It is unclear whether the child will be able to access medical information only regarding the surrogate mother and if it is any other non-identifying information. Davel and Skelton *Commentary* 36. It is important to note that where

It is commendable that section 41(3) determines that the relevant authorities providing the information may require a person to undergo counselling prior to the disclosure as a result of the potential sensitive nature of the information being revealed.⁴⁰⁹ Contrary to section 41, the SALC was of the view that because the commissioned child is from the moment of birth regarded as the commissioning parents' own legitimate child, the question of the child's origins should be left in the hands of the commissioning parents.⁴¹⁰ The right to know one's biological or genetic origins will be discussed next, so as to better understand the challenges and limitations imposed on this right.

2.3 The right to know one's biological or genetic origins

The Constitution does not expressly provide for the right to know one's origins,⁴¹¹ despite the observation by the SALRC in the Issue Paper on the right to know one's

the child is genetically linked to only one of the commissioning parents, the child will not be allowed to know the identity of his or her other genetic parent. Further, the child will not be allowed to find out who gave birth to him or her. Thaldar D "Surrogate Motherhood and scaremongering: Is South Africa entering an age of post-truth jurisprudence?" 2018 De Rebus 29. The issue of knowing the identity of one's egg or sperm donor is ethically controversial and since the AB judgment, the South African Law Reform Commission has commenced with an investigation into whether a donor-conceived child should be given a right to find out the identity of his or her donor. Sloth-Nielsen "Surrogacy in South Africa" 196. The writer pointed out that sec 41 is consistent with the object of sec 294 to ensure that the child becomes aware of its genetic origin and this is so even if the provision does not allow access to information regarding the identity of the surrogate mother in terms of sec 41(2). Skosana 2017 Obiter 265 and 266. The author states that the parties can voluntarily provide for such disclosure. He explains that it is however obvious that a child who is not aware of the fact that his social parent(s) is or are not his or her genetic parents cannot ask for information about his or her genetic parent. An obvious situation where a child may have questions about his or her origin is where his or her parents are same-sex partners. At p.267. Thus, the decision to inform the child of his or her genetic origin lies with the persons who has parental rights and responsibilities of the child Oluwasevi and Oladimeji 2021 Obiter 34. The authors stated that this is however contrary to section 7 of the UNCRC which provides that a child has the right to know his or her origins. AB and another v Minister of Social Development at par 30. At par 155 the court stated that the effects of sec 41 is that a child born as a result of surrogacy is, as of birth, barred by law from finding out the identity of a gamete donor who contributed to his or her conception Dhai and McQuoid-Mason Bioethics, Human Rights and Health Law 143.

⁴⁰⁹ Sec 41(3) of the Children's Act. Davel and Skelton *Commentary* 37. *AB and another v Minister of Social Development* at par 254.

⁴¹⁰ SALC Project 65 at par 8.4.3. Report of the Ad Hoc Committee 75. The Committee recommended that if the child is younger than ten years, the parents should have the discretion as to whether or not to inform him or her about his or her birth as a result of a surrogacy agreement. If the child is older than ten years, he or she should be informed about his or her genetic background. In all instances emphasis should be placed on the best interests of the child.

⁴¹¹ Rosenberg, W "Does the right to know one's origins exist and can it be limited?" 2020 *Journal of South African Law* 2020(4) 724 and 726.

own biological origins that the well-being of a child is clearly linked to his or her right to identity.⁴¹²

Disclosure of the identities of tissue and gamete donors is prohibited in terms of the NHA and the Regulations in respect of artificial fertilisation.⁴¹³ This means that the commissioning parent(s) has(have) no right to learn the identity of the donor or solicit donor-identifying information from any other source.⁴¹⁴ Biko and Nene state that a child born as a result of artificial fertilisation has a right to non-identifying medical and genetic information about his or her biological parents that is relevant to their own health status and risks.⁴¹⁵ The authors point out that a failure to protect the anonymity of gamete donors may have the result of a collapsing third-party reproductive program.⁴¹⁶

The right to know one's biological or genetic origins should also be considered with reference to Article 7 of the United Nations Convention on the Rights of the Child,⁴¹⁷

⁴¹² SALRC Project 140 12 and 104. At 31 it is stated "It is precisely because genetic information is thought to be of particular relevance that one might believe that access to this information is of fundamental value. A defence of a right to know one's genetic parentage thus presupposes the particular significance of genetic information for people's lives. Given the increasing emphasis on the idea that genetics can explain all kinds of things about human beings, stressing the importance of genetic information might well promote problematic beliefs about genetic essentialism." Importantly, at 33 the SALRC stated: "Anonymous donations prevent donorconceived individuals from gaining access to identifying information about their donors. This constraint does not thwart a vital interest in forming a healthy identity; it might threaten an interest in developing a particular identity, but the moral weightiness of an interest in forging a particular identity is questionable. Because of the privileging of genetic relationships presupposed by a defence of a right to know one's genetic origins, the defence could have negative effects on the well-being of donor-conceived people and their families." It was also pointed out that there is a lack of robust empirical evidence that donor-conceived people suffer certain harms and that even if such harms are present, they do not provide strong enough justification to ground the right to know one's own biological origins. It is further explained that some donor-conceived individuals who are unable to know their genetic origins may suffer great harms where others may suffer no harm at all. At 45 it is pointed out that once it is shown that no harm, no foul does not apply in a child's right to disclosure, the child's right to know their genetic parents is comprised of two distinct claims, namely: (a) the right to access identifying information regarding one's donor based on one's claim to be free from psychological harm arising from lack of access to identifying information; and (b) the right to be told about the nature of one's conception based on one's claim to respect as an identity-holding individual. Van Niekerk 2015 PELJ 421. A right to know one's genetic origins plays an important role in forming one's identity.

⁴¹³ SALRC Project 140 16. The relevant regulations will be discussed in ch 5. Skosana 2017 *Obiter* 268. The regulations will be discussed in detail in ch 5.

⁴¹⁴ SALRC Project 140 12.

⁴¹⁵ Biko J and Nene Z "Ethics aspects of third-party reproduction" 2017 *Obstetrics & Gynaecology Forum* 3 15.

⁴¹⁶ Biko J and Nene Z "Ethics aspects of third-party reproduction" 2017 *Obstetrics & Gynaecology Forum* 3 15.

⁴¹⁷ Reference to the CRC is important as it became the first legally binding international convention to confirm human rights for all children. The Convention was signed by South Africa in 1993 and ratified it on 16 June 1995. The CRC was the first international treaty that the incoming South

which provides not only that a child should be registered immediately after birth, but also that he or she shall have the right to a name (from birth), the right to acquire a nationality and, importantly, the right to know and be cared for by his or her parents as far as possible.⁴¹⁸ The UNCRC's recognition of the right to know one's origins⁴¹⁹ is justified by psychological studies that emphasise that a child who is denied information regarding his or her genetic or biological origins may experience feelings of sorrow, emptiness or isolation, including an incomplete self-image.⁴²⁰ Although the psychological impact of not knowing one's origins is a subjective question,⁴²¹ it is generally agreed that the right to knowledge of one's origins may assist in fostering a child's sense of emotional security and to develop a healthy identity and self-image.⁴²² Further, giving effect to this right may strengthen the child's sense of completeness and belonging; self-respect; a sense of security and the ability to cope with life situations.⁴²³ The right to knowledge of one's origins also protects a person's right to know.⁴²⁴

Since the psychological impact of not knowing your origins is ultimately a subjective question, each child's individual circumstances will determine the effect of these on the child's or person's sense of emotional security or development of a healthy identity. Not every child or person born from artificial fertilisation or surrogacy may necessarily have the need or the longing to find out who his or her biological parent is.

African democratic government ratified. "The United Nations Convention on the Rights of the Child (commonly abbreviated as the CRC or UNCRC) is a human rights treaty which sets out the civil, political, economic, social, health and cultural rights of children. The CRC consists of 54 articles that set out children's rights and how governments should work together to make them children." Fast available to all facts UNCRC to be accessed from https://www.parliament.gov.za/storage/app/media/Pages/2019/november/19-11-2019 30 Year Commemoration of the United Nations Convention on the Rights of the Child/docs/FAST FACTS UNCRC draft 2 19 November 2019final.pdf (accessed on 10

August 2021).
 ⁴¹⁸ A copy of the United Nations Convention on the Rights of a child is to be accessed from https://www.ohchr.org/en/professionalinterest/pages/crc.aspx (accessed on 10 August 2021).

⁴¹⁹ Rosenberg 2020 *TSAR* 734. Oluwaseyi and Oladimeji 2021 *Obiter* 24. The authors stated that the right to know one's parents in article 7 has been interpreted to mean providing a child with information concerning his or her biological origins and the circumstances surrounding his or her bioth. Further, failure to avail children of this information affects their ability to develop a sense of identity and identity is a person's unique profile of which genetic origin is a key feature. Thus, a child's right to identity cannot be achieved if the child is not aware of his or her biological origins as this is one of the determining factors that make a child understand who they are.

⁴²⁰ Rosenberg 2020 *TSAR* 724.

⁴²¹ Rosenberg 2020 *TSAR* 728.

⁴²² Rosenberg 2020 *TSAR* 740.

⁴²³ Rosenberg 2020 *TSAR* 740.

⁴²⁴ Rosenberg 2020 *TSAR* 745.

The next section will comprehensively discuss the salient legal issues in chapter 19 of the Children's Act.

3. SALIENT ISSUES RELATING TO CHAPTER 19 OF THE CHILDREN'S ACT

3.1 Introduction

The rationale behind the promulgation of chapter 19 of the Children's Act has been set out in chapter one of this thesis. It should be borne in mind that the inclusion of chapter 19 into the Children's Act pertains to surrogacy specifically, which, as this chapter will illustrate, applies to couples who are infertile only and not for those who are fertile and able to procreate without assistance.⁴²⁵

Chapter 19 of the Children's Act, and more specifically sections 293 to 303, were enacted to regulate the practice of surrogacy in South Africa and to give legal certainty regarding the rights of the children born as a result of surrogacy and that of the different parties involved.⁴²⁶ Chapter 19 thus regulates the legal consequences of human reproduction by artificial fertilisation of women acting as surrogate mothers.⁴²⁷ In an attempt to regulate surrogate motherhood instead of to outlaw it, the Act contains a number of restrictions on surrogacy agreements.⁴²⁸

Since artificial fertilisation plays an important role in surrogate motherhood, two aspects relevant to artificial fertilisation are worth mentioning.⁴²⁹ Previously, in terms of the Human Tissue Act and attendant Regulations published in 1986, artificial

⁴²⁵ Van Niekerk 2015 *PELJ* 416. The author states that the only slight on ch 19 is the existence of sec 294. This section effectively distinguishes between persons suffering from infertility which prevents them from providing genetic material to produce a child and those who are infertile but able to provide genetic material. In both these cases the parties are infertile and it is only the origin of their infertility which distinguishes them.

⁴²⁶ Ch 19 came into operation on 1 April 2010. Nöthling Slabbert 2012 SAJBL 27. Slabbert and Roodt "South Africa" 325. Louw 2013 THRHR 564. Florescu, S and Sloth-Nielsen, J "Visions on surrogacy – From North to South: The approach of the Netherlands and South Africa to the issue of surrogacy and the child's right to know his origin" 2017 International Survey of Family Law 2017 246.

⁴²⁷ Nöthling Slabbert 2012 SAJBL 27.

⁴²⁸ Bonthuys and Broeders 2013 SALJ 490.

LexisNexis Bill of Rights Compendium at par 3E27.

fertilisation could only be performed by a competent person upon a married woman.⁴³⁰ The exclusion of unmarried women from receiving artificial fertilisation would probably have been held to constitute a violation of the constitutional right to equality in terms of section 9(3) of the Constitution of South Africa, had this still been the case today.⁴³¹ Fortunately and rightly so, the Children's Act has since removed the restriction that artificial fertilisation may only be performed upon married women.⁴³²

3.2 Nature of surrogate motherhood agreements

In the judgment of *Ex Parte MS and Others; In Re: Confirmation of surrogate motherhood agreement*,⁴³³ Keightley AJ observes that surrogate motherhood agreements, or surrogacy arrangements as they are known in short, are multifaceted and that they require a consideration of many interconnected concerns, ranging from those that are deeply personal to the parties' involvement, to those that deal with the necessary practical aspects of the arrangement.⁴³⁴ The judgment emphasises that the legal complexities of surrogacy arrangements are of overriding concern, as these arrangements involve an intricate relationship of interests, rights and obligations on the part of all parties concerned.⁴³⁵

The first and second applicants in this case, and the third applicant, the surrogate mother, sought confirmation of their surrogate motherhood agreement in terms of sections 292 and 295 of the Children's Act. An issue raised before the court was whether it is competent for the high court to confirm a surrogate motherhood agreement where the written agreement between the parties was only entered into (and confirmation sought) *after* the artificial fertilisation and pregnancy of the surrogate mother. The Children's Act requires that surrogate motherhood agreements must be

⁴³⁰ Reg 8(1) of the regulations issued in terms of sec 37 of the Human Tissue Act 65 of 1983 and which was published in the Government Gazette 10238 GN R1182 20 June 1986. Nicholson and Bauling 2013 *De Jure* 513.

⁴³¹ 108 of 1996. The state may not discriminate unfairly against anyone on the grounds of his or her marital status. LexisNexis *Bill of Rights Compendium* at par 3E27.

⁴³² Sec 292 and 293 of the Children's Act only makes reference to the surrogate mother's husband and partner should there be one. Sec 40(2) makes reference to a 'woman'. There is no provision that states that the surrogate mother (sec 292 and 293) or the woman (sec 40) must be married. LexisNexis *Bill of Rights Compendium* at par 3E27.

⁴³³ This case will be discussed in par 3.7.8.

⁴³⁴ 2014 (2) All SA 312 (GNP) at par 7.

⁴³⁵ Ex parte MS and others; In re: Confirmation of surrogate mother agreement at par 7.

formally constructed, vetted and confirmed by the High Court before any steps are taken that might lead to the conception of a child. Since the artificial fertilisation of a surrogate is expressly prohibited until a surrogate motherhood agreement has been confirmed by the court, the parties in this case were therefore in breach of the Act. So, the question whether, despite the breach, the court could confirm the surrogate motherhood agreement, had to be answered. The granting of such relief would depend on the best interests of the child. The court confirmed the surrogacy agreement after finding that the best interests of the child would be best served by confirming the agreement.

The judgment echoes the legal position that artificial fertilisation of a surrogate mother is specifically prohibited unless and until a surrogate agreement has been confirmed by the court.⁴³⁶ In the above case, the reason for the confirmation of the surrogacy agreement is obvious: the best interests of a child are of paramount importance in every matter regarding the child as in terms of our Constitution.⁴³⁷ It can be said that, in essence, surrogacy agreements are primarily concerned with the child to be born, explained by Keightly AJ as follows:

[...] although the hoped-for-child is not a party to the surrogate motherhood agreement, his or her future rights and interests are the most important of all the rights and interests involved. To ensure that they are adequately protected, the law requires certainty and judicial scrutiny of the proposed surrogacy arrangements before there is even any prospect of a child coming into being.⁴³⁸

The two important questions which arose in the matter of MS, in which the court was asked to confirm a surrogacy agreement, were the following:

First, is it competent under the Act for a court to confirm the surrogacy agreement notwithstanding this breach? The Act does not deal with this question in any express terms, and I must derive the answer from interpretation of all the relevant provisions. Second, if, on a proper interpretation of these provisions I find that a court is so competent, what is the correct approach to be adopted in cases like the present? In other words, what is required of the applicants, and on what basis should courts exercise their discretion regarding confirmation of surrogacy agreements in such cases?⁴³⁹

The discussion that follows next will explore some pertinent provisions in chapter 19 of the Children's Act, notably the requirements for a valid surrogacy agreement (sections 292 - 296), including the relevant consent requirements; the controversial and

⁴³⁶ Ex parte MS and others; In re: Confirmation of surrogate mother agreement at par 8.

⁴³⁷ Ex parte MS and others; In re: Confirmation of surrogate mother agreement at par 9.

⁴³⁸ Ex parte MS and others; In re: Confirmation of surrogate mother agreement at par 9.

⁴³⁹ Ex parte MS and others; In re: Confirmation of surrogate mother agreement at par 10.

contested genetic link requirement; the High Court's confirmation of the surrogacy agreement and the artificial fertilisation of the surrogate.

The consequences of a valid surrogate agreement on the status of the child born because of the agreement will be discussed, as well as the termination of the surrogate agreement (section 298) and the consequences thereof (section 299). A discussion on the termination of the pregnancy by the surrogate mother is canvassed next (section 300), as well as the prohibitions relating to surrogacy (sections 301 - 302). Relevant judgments on these components will also be discussed.

3.3 Section 292: Requirements for a valid surrogate agreement⁴⁴⁰

Parties who intend to exercise their reproductive rights by using surrogacy are required to conclude a written agreement that results in the complete transfer of parental rights and responsibilities from the surrogate to the commissioning parents when the commissioned child is born.⁴⁴¹ Section 292 of the Children's Act sets out the requirements for a validly concluded and confirmed surrogate agreement.⁴⁴² It provides that the agreement between the parties must be in writing and signed by all the parties

⁴⁴⁰ The Act. See further *In re confirmation of three surrogate agreements* 2011 (6) SA 22 at par 15. *Ex parte WH and others* at par 38. At par 71 the court stated that: "While a surrogacy agreement is a contract whose validity is dependent upon the confirmation of the High Court, it is a contract of a special kind, unique if regard is being had to its subject-matter. The arrangement that comes into place when a surrogacy agreement is arrived at and the consequences that may follow have far-reaching and sometimes unintended consequences." *Ex parte MS and others; In re: Confirmation of surrogate mother agreement* at par 18. Bosman-Sadie and Corrie *Practical approach to the new Children's Act* 293. Nöthling Slabbert 2012 *SAJBL* 28. Nicholson and Bauling 2013 *De Jure* 516. *AB and another v Minister of Social Development* at par 39.

⁴⁴¹ Baase 2019 *PELJ* 2. Nicholson, S and Nicholson, C "I used to have two parents and now I have three? When science (fiction) and the law meet: Unexpected complications" 2016 *Medicine and Law* 35 433.

⁴⁴² Oluwaseyi and Oladimeji 2021 *Obiter* 31.

involved;⁴⁴³ the agreement must be concluded in the Republic of South Africa;⁴⁴⁴ at least one of the commissioning parents (or if it is a single commissioning parent, that person) must be domiciled in South Africa at the time of conclusion of the agreement;⁴⁴⁵ the surrogate mother (and her husband or partner if any) must be domiciled in South Africa at the time of concluding the agreement;⁴⁴⁶ and the High Court within whose area of jurisdiction the commissioning parent(s) are domiciled or habitually residing, must confirm the agreement.⁴⁴⁷

⁴⁴³ Sec 292(1)(a). SALC Project 65 at par 8.3.1. It is compulsory for the parties that they enter into a written agreement Report of the Ad Hoc Committee 65. It is important to establish the intention of the parties as to parental rights and their willingness to proceed with this intention. The Committee recommended that the following provisions be contained in the agreement: (a) The nature of the surrogacy (full/partial) should be identified and the parties should be provided with the current status of the law in each instance; (b) The surrogate and commissioning parent(s) should understand and agree to the terms of the agreement and the legal position relevant to their particular surrogacy agreement; (c) Specific responsibilities for each party should be established; (d) The financial responsibilities of the parties should be established regarding the expenses related to pregnancy and requiring funds to be placed in a trust account to cover the anticipated expenses; (e) The commissioning parent(s) should state that they will accept parental responsibility for the commissioned child, despite any medical or physical handicaps that the child might suffer; (f) Health insurance policies to be maintained by all the parties throughout the agreement must be stated; (g) The parties must deal with all the eventualities which could lead to either the surrogate or the commissioning parent(s) requesting an abortion and in the even that the surrogate terminate the pregnancy for non-therapeutic reasons that she is responsible for reimbursing and repaying the commissioning parent(s) for the necessary expenses incurred; (h) Any arrangements regarding the child should be specified, for example, visitation rights, access to the child etcetera. Provision should be made for custody of the child in the event of divorce or death of the commissioning parent(s); (i) It can be jointly decided between the parties if the surrogacy agreement is to be publicised. It is however important that the identity of the child is not revealed; and (j) Social disease testing (including HIV) must be performed on all the parties. The surrogate must be medically examined and declared suitable. The agreement may include optional clauses but it may not be contra bones mores. Slabbert and Roodt "South Africa" 331 SALRC Project 140 84. Mills 2010 Stell LR 435. Sloth-Nielsen "Surrogacy in South Africa" 187.

⁴⁴⁴ Sec 292(1)(b). SALC Project 65 at par 8.3.7. The agreement should be concluded in the Republic of South Africa and all the parties concerned should be domiciled in the country at the time of the conclusion of the agreement. LexisNexis *Family Law service* at par W3.

Sec 292(1)(c). LexisNexis Family Law service at par W3. It is argued that by opening South African surrogacy practices to foreign nationals may lead to complex questions being raised on the nationality of the child in instances where the commissioning parent(s) are foreign nationals. Sloth-Nielsen "Surrogacy in South Africa" 187. At 199 the writer states that domicile is a legal concept and it is not equitable to habitual residence. It can be established quite quickly provided that the necessary intention to establish domicile is in place. Florescu, S and Sloth-Nielsen, J "Visions on surrogacy – From North to South: The approach of the Netherlands and South Africa to the issue of surrogacy and the child's right to know his origin" 2017 International Survey of Family Law 2017 247. The writers explain that the requirement of domicile is found where a person is legally deemed to be constantly present even if factually absent. It is important to note that South African citizenship or permanent residence in South Africa does not, on its own, enable a person to become a commissioning parent or a surrogate mother in South Africa.

⁴⁴⁶ Sec 292(1)(d). Report of the Ad Hoc Committee 58. Nöthling Slabbert 2012 *SAJBL* 27. Through this limitation, South Africa is a less attractive reproductive tourism destination.

⁴⁴⁷ Sec 292(1)(e). Report of the Ad Hoc Committee 65. At p.70 the Committee submitted that once the court has confirmed the agreement, it should be a valid and enforceable document. Thus, in this way, the parties are legally bound by the agreement and the interests of the child is protected. SALRC Project 140 84. LexisNexis *Family Law service* at par W3. It is important that sufficient

Non-compliance with the requirements will cause the agreement to be invalid and thus unenforceable between the parties.⁴⁴⁸ This is aligned with the Ad Hoc committee's response to the South African Law Commission's (SALC) Project 65, that a pre-approved agreement would assist to ensure that the interests of the parties to the surrogate agreement could be considered in advance before any problems would arise.⁴⁴⁹ The pre-approved agreement can further be seen to protect the best interests of the commissioned child, as the rights and obligations in respect of the child will also be considered in such agreement and therefore further endorsed through the confirmation by the court of the agreement. Possible future issues may be anticipated in the agreement before the agreement is confirmed by the court.

The rights of children and their interests must always be placed first, even in circumstances where the rights of the prospective parents may be compromised.⁴⁵⁰ The best interests of the children is the overriding factor.⁴⁵¹ It is for this reason that the requirement of a *prior, formally approved surrogate agreement* has always been considered vital in order to minimise the risks intrinsic in surrogacy.⁴⁵² The conclusion to be drawn from this (i.e the requirement of confirmation of the surrogacy agreement by the court) is that the general principles of the law of contract were deemed inadequate by the legislator to exclusively regulate the rights and the duties of the parties concerned in a surrogate agreement.⁴⁵³ Despite the importance of an

facts be set out in the founding affidavit in support of the commissioning parent(s) application to support their domicile or the place of their habitual residence that will give the court jurisdiction.

⁴⁴⁸ Sec 292. See also Nicholson and Bauling 2013 *De Jure* 518. Sloth-Nielsen "Surrogacy in South Africa" 199. It is important to note that a surrogacy agreement concluded between foreign nationals who are not domiciled in South Africa would be invalid and unenforceable and this is regardless of whether it is an altruistic or commercial surrogacy agreement. Thus, the commissioning parents would acquire no rights in respect of the child born and would also likely be ineligible for adoption with reference to the requirements of sec 25 of the Children's Act which foresees such applications as being treated as inter-country adoptions, which must be dealt with in accordance with the Hague Convention on Intercountry Adoption (1993) and which South Africa has ratified.

⁴⁴⁹ The South African Law Commission's *Report on Surrogate Motherhood* (1999) was referred to a parliamentary *Ad Hoc* Select Committee for further investigation and a report. Davel and Skelton *Commentary* 10.

⁴⁵⁰ *Ex parte CJD and others* 2018 (3) SA 197 (GP) at par 10.

⁴⁵¹ Sec 28 of the Constitution of the Republic of South Africa 1996. *Ex parte CJD* at par 10. C and Others v Department of Health and Social Development, Gauteng and Others (CCT 55/11) [2012] ZACC 1; 2012 (2) SA 208 (CC); 2012 (4) BCLR 329 (CC). Article 3(1) of United Nations Convention on the rights of the child (CRC). Article 4 of African Charter on the rights and welfare of the child (ACRWC).

⁴⁵² Davel and Skelton *Commentary* 10.

⁴⁵³ Nicholson and Bauling 2013 *De Jure* 517.

agreement between the parties, the SALC held the view that an agreement should merely be a starting point for the parties, as legislation is better suited to determine the surrogacy procedure, including the rights and duties of the parties and the legal consequences.⁴⁵⁴ The SALC never insisted on a formal confirmation of the agreement. The underlying aim of the confirmation by the court-requirement in section 292(1)(b) was to prevent couples from concluding surrogacy agreements in other jurisdictions where the procedure is less cumbersome.⁴⁵⁵ The domicile requirement pertaining to the commissioning couple and the surrogate mother applies at the time of the conclusion of the agreement and thus also excludes the possibility of foreigners abusing legalised surrogacy in South Africa.⁴⁵⁶ Clarke argues that persons who are temporarily living abroad and who are still domiciled in the Republic will have to ensure that they are physically present in the Republic for the purposes of signing of the surrogate agreement.⁴⁵⁷ Heaton rightly points out that South African citizenship or permanent residence in South Africa does not, on its own, permit a person to become a commissioning parent or a surrogate in South Africa.⁴⁵⁸ The provision in section 292(2), however, allows the court to dispose of the domicile requirement for good reason in the case of the surrogate mother or her partner.⁴⁵⁹ For example, where the commissioning parent or couple can only find a relative not domiciled in South Africa to act as the surrogate, it may arguably be justifiable to dispense with the domicile requirement.⁴⁶⁰ The domicile requirement does not prevent the pregnant surrogate from leaving South Africa to evade the legal consequences of the valid surrogate

⁴⁵⁷ LexisNexis *Family Law service* at par W3. Sec 292(1)(b).

⁴⁵⁴ SALC Project 65 at par 8.3.

⁴⁵⁵ Heaton 2015 *THRHR* 30. The author stated that the domicile requirement is clearly an effort to discourage international surrogacy and to avoid that South Africa become a destination for reproductive tourism.

⁴⁵⁶ Sec 292(1)(c) and (d). SALC Project 65 at par 8.3.7. Davel and Skelton *Commentary* 10. The *lexi loci contractus* will thus be South African law. Heaton 2015 *THRHR* 30. It is thus important to note that the surrogate agreement may only be concluded after the commissioning parent has established a domicile in South Africa. Domicile of choice is at issue where a foreigner comes to South Africa for the purposes of entering into a surrogacy agreement to obtain a child.

⁴⁵⁸ Heaton 2015 *THRHR* 30. At 46 the author pointed out that a provision should be inserted in the Children's Act 38 of 2005 which expressly requires a thorough investigation into whether foreign commissioning parents really do meet the domicile requirement. Detailed information is thus necessary.

⁴⁵⁹ Sec 292(2).

⁴⁶⁰ Davel and Skelton Commentary 9. Florescu, S and Sloth-Nielsen, J "Visions on surrogacy – From North to South: The approach of the Netherlands and South Africa to the issue of surrogacy and the child's right to know his origin" 2017 International Survey of Family Law 2017 249. The writers point out that the High Court can dispense with the domicile requirement in relation to the surrogate mother but not the commissioning parents.

agreement.⁴⁶¹ There is unfortunately no provision in the relevant sections prohibiting the surrogate from leaving the country after the surrogate agreement is confirmed and after she is artificially fertilised. The commissioning parent(s) must thus accept this risk, although they may be unaware of this possibility. Prospective commissioning parents should be alerted to this possibility and that there is basically no protection for them in this regard.

Florescu and Sloth-Nielsen argue that the domicile requirement seems to imply that surrogacy agreements concluded abroad will not be recognised in South Africa.⁴⁶² Further, South Africans are not prevented from obtaining a child through surrogacy abroad. However, despite this possibility, they may encounter considerable difficulties in having their parentage recognised domestically.⁴⁶³

Section 292(1)(d) requires the High Court (and no other court) to confirm the surrogacy agreement. Section 292 does not provide any guidance to the parties as to what information is needed before the court can confirm such an agreement. This section only refers to the conclusion of the agreement in the Republic and the domicile requirement. In the *matter* of *Ex parte CJD and others*,⁴⁶⁴ Tolmay J emphasises that it is important that all aspects of relevance to the issue of whether a surrogacy agreement should be confirmed by the court should be set out in the affidavit.⁴⁶⁵ The best interests of the commissioned children are at stake and the applicants must play open cards

⁴⁶¹ Davel and Skelton *Commentary* 9. Heaton 2015 *THRHR* 33. Heaton points out that the surrogate can escape the enforcement of a valid surrogate agreement by leaving South Africa for a country where surrogacy is not recognised and where the rules of private international law will not permit the recognition and the enforcement of the South African confirmation court order.

⁴⁶² Florescu, S and Sloth-Nielsen, J "Visions on surrogacy – From North to South: The approach of the Netherlands and South Africa to the issue of surrogacy and the child's right to know his origin" 2017 International Survey of Family Law 2017 249.

 ⁴⁶³ Florescu, S and Sloth-Nielsen, J "Visions on surrogacy – From North to South: The approach of the Netherlands and South Africa to the issue of surrogacy and the child's right to know his origin" 2017 International Survey of Family Law 2017 249.

⁴⁶⁴ 2018 (3) SA 197 (GP).

⁴⁶⁵ *Ex parté CJD and others* at par 11. The court stated that the applicants must act with the utmost good faith in an application to confirm a surrogacy agreement. SALC Project 65 at par 8.3.2. The commission submitted that the parties should provide the court with all the evidence necessary to prove that all legislation has been or will be complied with. This includes medical and psychological evidence with regard to the inabilities and needs of the commissioning parents, the suitability of the surrogate, the origin of donor gametes, the interests of descendants and adopted children and particulars as to the fertilisation process that will be followed.

with the court.⁴⁶⁶ Tolmay J further emphasises that the court has an obligation to protect and advance the best interests of children, and that all role players have an obligation to ensure that all relevant information that may impact on the court's discretion to grant or dismiss the application for confirmation, is set out in the affidavit.⁴⁶⁷ It is unfortunate that the Children's Act does not provide guidelines regarding supporting documentation that the parties should submit to the court in support of a surrogacy agreement confirmation in the form of regulations promulgated in terms of the Act.⁴⁶⁸

Parties to a surrogacy agreement must realise that section 292 determines that no surrogacy agreement shall be valid unless all the requirements set out in the section are complied with. One aspect that this section does not address is the risk of the surrogate leaving the Republic after the conclusion of the surrogacy agreement. Regulations should provide guidance as to how this risk should be addressed or resolved. A surrogate who is not biologically related to the commissioned child can then flee with the commissioning parents' biologically related child and there is no provision safeguarding the interests of the commissioned child or of the surrogate mother discuss her intention to leave the Republic with the commissioning parents first. It would appear that the commissioning parents' only remedy would be relying on the contractual provisions governing a breach of contract (which would occur if the surrogate mother failed to honour the agreement to deliver the child).

This discussion identified some of the legal gaps regarding surrogate motherhood agreements. Not only is clarity required as to exactly what type of information and supporting documentation should be included in an application for the confirmation of a surrogate agreement, but risk of the surrogate mother leaving the Republic without

⁴⁶⁶ Ex parte CJD and others at par 12. See further *In re confirmation of three surrogate agreements* at para 16.

⁴⁶⁷ 2018 (3) SA 197 (GP) at par 13. The court concluded at par.15 that parties in surrogacy agreements should set out the following in the founding affidavit: (a) If and how the applicants will function as a family unit and whether they are comfortable with society regarding them as such; (b) Whether they are living together or not, and if not, why this state of affairs will not impact on the interests of the child and them functioning as a family unit.

⁴⁶⁸ Nicholson and Bauling 2013 *De Jure* 517.

delivering the child should also be clarified (e.g whether this is governed by the contract between the parties only).

The next section will focus on the legal consequence of an invalid surrogacy agreement for all parties involved, especially for the commissioned child.

3.4 Section 293: Consent of the commissioning mother- and surrogate mother's husband, wife or partner⁴⁶⁹

Where a commissioning parent is married or in a permanent relationship, he or she must obtain the written consent of his or her husband, wife or partner before a surrogate agreement is concluded, as the husband, wife or partner must be a party to the agreement, before a court may confirm a surrogacy agreement.⁴⁷⁰ The same consent requirement is also applicable should the surrogate be married or in a permanent relationship.⁴⁷¹ It is possible for the court to confirm the agreement in circumstances where the husband or partner of the surrogate (who is not the genetic parent of the child) unreasonably withholds his or her consent.⁴⁷²

Section 293(1) envisions a situation where a commissioning parent can be a single person, or a single person in a life-partnership (not legally married but in a committed relationship).⁴⁷³ The case of *Ex parte CJD and others* demonstrates that although the court must give a wide interpretation of what a "permanent relationship" may entail, it still has to determine whether the nature of the permanent relationship can be regarded as supportive for the raising of a family.⁴⁷⁴ In *Ex parte CJD and others*, Tolmay J

⁴⁶⁹ Sec 293 of the Act. In re confirmation of three surrogate agreements at para 15. Ex parte CJD and others at par 17. Bosman-Sadie and Corrie Practical approach to the new Children's Act 294. Nöthling Slabbert 2012 SAJBL 28. AB and another v Minister of Social Development at par 39. SALC Project 65 at par 8.2.11.

 ⁴⁷⁰ Children's Act sec 293(1). Report of the Ad Hoc Committee 60. Slabbert and Roodt "South Africa" 328. SALRC Project 140 84. Sloth-Nielsen "Surrogacy in South Africa" 187. The writer points out that it was clearly the intention of the legislator to prevent South Africa from becoming an international surrogacy destination. Oluwaseyi and Oladimeji 2021 *Obiter* 32.

⁴⁷¹ Sec 293(2).

⁴⁷² Sec 293(3). Report of the Ad Hoc Committee 58. Nöthling Slabbert 2012 SAJBL 30. Florescu, S and Sloth-Nielsen, J "Visions on surrogacy – From North to South: The approach of the Netherlands and South Africa to the issue of surrogacy and the child's right to know his origin" 2017 International Survey of Family Law 2017 248.

⁴⁷³ *Ex parte CJD and others* at par 18.

⁴⁷⁴ *Ex parte CJD and others* at par 18. Nicholson and Bauling 2013 *De Jure* 521. The writers indicate that the concept of a permanent relationship is vague and for the sake of clarity, a regulation to

expresses the view that a reading of section 293 appears to favour an interpretation that, since a partner in a permanent relationship not only gives written consent to the surrogate agreement but also becomes a party to the application for the confirmation of the agreement, he or she acquires parental responsibilities and rights when the order of confirmation is made by the court.⁴⁷⁵ As alluded to already, both the surrogate and the commissioning parent must obtain the written consent of his or her spouse or permanent partner before the court will confirm the surrogate agreement.⁴⁷⁶ In circumstances where a surrogate is going through a divorce and the divorce is not yet finalised, the surrogate's spouse must still provide his consent.⁴⁷⁷ This section refers specifically to the words "married" and "permanent relationship" and it is thus important to determine the scope of the meaning of these words to enable the parties to comply with the consent requirement. The term "marriage" is defined as a marriage recognised in terms of South African law or customary law or a marriage concluded in accordance with a system of religious law which is subject to specified procedures.⁴⁷⁸ Heterosexual civil marriages and civil unions of the same-sex have been recognised in South Africa with the enactment of the Marriage Act⁴⁷⁹ (for heterosexual civil marriages) and the Civil Union Act⁴⁸⁰ (with regard to same-sex unions).

The reference to a "permanent relationship", however, remains vague and should be clarified. There is also no definition in the Children's Act that explains the meaning of a permanent relationship. The term suggests a relationship which holds a degree of permanence.⁴⁸¹ There seems to be no indication of who should make the decision of whether the relationship is "permanent" enough to warrant the consent of the partner

the Act should stipulate what exactly the parties should prove in this regard. They further point out that guidelines to determine the permanence of the relationship should be provided for in a regulation.

⁴⁷⁵ Ex parte CJD and others at par 19.

⁴⁷⁶ Sec 293(1) and (2).

⁴⁷⁷ LexisNexis *Family Law service* at par W4. The writer argues that the support and consent of all the parties involved in a surrogacy agreement is essential to the success of the entire surrogacy process.

⁴⁷⁸ Sec 1 of the Children's Act. Further, any reference to a husband, wife, widower, widow, divorced person, married person or spouse must be construed accordingly.

⁴⁷⁹ 25 of 1961.

⁴⁸⁰ 17 of 2006. Civil union is defined in sec 1 as the voluntary union of two persons who are both 18 years of age or older, which is solemnised and registered by way of either a marriage or a civil partnership, in accordance with the procedures prescribed in this Act, to the exclusion, while it lasts, of all others. Sec 1 further defines civil union partner as a spouse in a marriage or a partner in a civil partnership, as the case may be, concluded in terms of this Act.

⁴⁸¹ LexisNexis *Family Law service* at par W4.

of the surrogate or the commissioning parent.⁴⁸² The only assumption that can be made is that the court, dealing with the confirmation application, will have the discretion to decide if the parties to the surrogacy agreement are in a permanent relationship or not. The need for, and importance of, the scope and quality of the information that should be provided may not only be gleaned from the requirements set out in section 295, but also appear from the judgments of *Ex parte CJD and others*⁴⁸³ and *Ex parte WH and others*.⁴⁸⁴ However, despite the guidance offered by these judgments, it is submitted that regulations promulgated in terms of chapter 19 of the Children's Act should provide more detail on the issue of parties in a permanent relationship to ensure that the appropriate parties to the surrogacy agreement provide the requisite consent.

Louw argues that it is not clear from the provisions of the Children's Act whether the consent provided by the spouse or partner would also include consent to the artificial fertilisation of the surrogate⁴⁸⁵, but I would argue that it does. The artificial fertilisation of the surrogate mother clearly is part and parcel of the surrogate agreement as it is impossible to have the one without the other. If the spouse or partner of the surrogate mother does not give his (or her) consent to the artificial fertilisation of his or her wife, parental responsibility will vest exclusively in the surrogate mother as the woman who gave birth to the child.⁴⁸⁶ Section 293(3) appears to have created an exception to the requirement of consent by the surrogate's spouse or partner in that the court may confirm the agreement without such person's consent, should he or she unreasonably withholds consent.⁴⁸⁷

The Children's Act requirement that a genetic link should exist between the commissioning parent(s) and the commissioned child, will be discussed next.

3.5 Section 294: The genetic link requirement⁴⁸⁸

⁴⁸² Davel and Skelton *Commentary* 11.

⁴⁸³ 2018 (3) SA 197 (GP). ⁴⁸⁴ 2011 (4) All SA 630 (CN)

⁴⁸⁴ 2011 (4) All SA 630 (GNP).

⁴⁸⁵ Davel and Skelton *Commentary* 13.

⁴⁸⁶ Sections 40(1), 40(2) and 297(2) of the Act.

⁴⁸⁷ Sec 293(3) of the Children's Act 38 of 2005 Nöthling Slabbert 2012 SAJBL 29.

⁴⁸⁸ Sec 294 of the Act. In re confirmation of three surrogate agreements at para 15. Ex parte CJD and others at par 294. Bosman-Sadie and Corrie Practical approach to the new Children's Act 294. AB and another v Minister of Social Development at par 45.

For a child to have a genetic link with one or both of his or her parents, the gametes of one or both parents must be used for conception. Section 294 determines that at least one of the commissioning parents' gametes must be used in the conception of the child or where the commissioning parent is a single person, that person's gamete.⁴⁸⁹

The use of donor gametes is a controversial issue in the regulation of surrogacy. The requirement of a genetic link mirrors the SALC's and *Ad Hoc* Committee's view that donor gametes should not be permitted where it is possible to use the gametes of both the commissioning parents.⁴⁹⁰ The gametes of the surrogate mother (partial surrogacy) may be used for the conception of the child to be born in terms of the agreement, provided that the gametes of at least one of the commissioning parents (or parent if single) is used in the conception process to enable the commissioning parents (or parent) to be genetically linked to the child.⁴⁹¹ Thus, the exclusive use of donor gametes is prohibited and even in a situation where a single commissioning parent is, or both commissioning parents are infertile, as to allow the latter as commissioning

⁴⁸⁹ Sec 294. Report of the Ad Hoc Committee 56 and 59. AB and another v Minister of Social Development at par 45 and 46 Thaldar 2018 De Rebus 28. SALRC Project 140 85 and 87. Sloth-Nielson, J and van Heerden, B "The constitutional family developments in South African child and family law 2003-2013" 2014 International Journal of Law, Policy and the Family 28(1) 114. Meyerson, D "Surrogacy, Geneticism and Equality: The case of AB Minister of Social Development" 2019 Constitutional Court Review 9 318. Heaton 2015 THRHR 29.

⁴⁹⁰ SALC Project 65 at par 8.2.6. The SALC was convinced that it is desirable, to promote the bond between the child and the commissioning parents, to use the gametes of at least one of the commissioning parents and that it is in the best interests of the child to do so. SALRC Project 140 87. The Ad Hoc Committee Report recommended the retention of the genetic link requirement (that the gametes of at least one of the commission parents be used towards conception or in the case of a single person, that person's gametes). It is explained that the reasoning for this recommendation is the following: "In the instance where both the male and female gametes used in the creation of the embryo are donor gametes, it would result in a similar situation to adoption, as the child or children would not be genetically linked to the commissioning parent or parents. This would obviate the need for surrogacy as the couple could adopt a child. This type of surrogacy was not preferred by most commentators. It was felt that in both partial and full surrogacy it should be a pre-condition that the child or children should always be genetically linked to the commissioning parent or parents."

⁴⁹¹ Thaldar 2018 *De Rebus* 28. Boniface, A.E "The genetic link requirement for surrogacy: A family cannot be defined by genetic lineage" 2017 *Journal of South African Law* 2017(1) 190. Nicholson 2013 *SAJHR* 501. Florescu, S and Sloth-Nielsen, J "Visions on surrogacy – From North to South: The approach of the Netherlands and South Africa to the issue of surrogacy and the child's right to know his origin" 2017 *International Survey of Family Law* 2017 248. At 256 the writers point out that there can be situations where a child has access to information about his story of origin only in respect of one biological parent. If the genetic link requirement were dispensed with, there would be situations where the child born of surrogacy would have absolutely no identifying information about the circumstances of his birth except for the limited information available as in terms of sec 41 of the Act after he or she reaches the age of eighteen years.

parents would make the surrogacy process a 'commissioned adoption',⁴⁹² or as Meyerson explains, a situation tantamount to 'double-donor' surrogacy.⁴⁹³ For her, this form of surrogacy allows the commissioning parents to be involved in selecting both the gametes and the surrogate and it also give them the chance to participate in the surrogate's pregnancy.⁴⁹⁴ These features of double-donor surrogacy serve to encourage the development of an emotionally significant bond with the child.⁴⁹⁵ Despite Meyerson's optimism, I believe that the benefits do not negate the risks associated with double-donor surrogacy and that an ordinary adoption should be the indicated route. With no biological link with the commissioned child, double-donor surrogacy leaves the commissioning parents with only the contract as their only legal recourse, should something go wrong.

The discussion above supports the conclusion that surrogacy agreements in South Africa are intended to assist the infertile while at the same time preventing the surrogate from being exploited.⁴⁹⁶ However, Van Niekerk argues that the real target group that surrogacy should assist (those who are infertile) are rendered helpless by the genetic-link requirement in section 294.⁴⁹⁷ This provision is also regarded as harsh and discriminatory by practising reproductive specialists who often have to assist parents that both suffer from infertility.⁴⁹⁸ Van Niekerk views it as unjust to expect these couples to follow the route of an ordinary adoption, as there may be long waiting lists

⁴⁹² Nöthling Slabbert 2012 SAJBL 33-34. Thaldar 2018 De Rebus 28. SALRC Project 140 88. Louw 2013 THRHR 575. Thus, where none of the parties is genetically related to the child to be born. Meyerson, D "Surrogacy, Geneticism and Equality: The case of AB Minister of Social Development" 2019 Constitutional Court Review 9 318. Van Niekerk 2015 PELJ 408. The author states that the presence or the absence of a genetic link on the part of the commissioning parents should be immaterial. The courts are described as the guardians of what could potentially be an exploitative process and they are required to vet any applicant(s) who are able to provide a genetic link. Thus, it can be presumed that they are equally equipped to do so in instances where no genetic link exists. The courts would be able to root out improper motives before surrogacy agreements are sanctioned.

⁴⁹³ Meyerson, D "Surrogacy, Geneticism and Equality: The case of AB Minister of Social Development" 2019 *Constitutional Court Review* 9 324.

⁴⁹⁴ Meyerson, D "Surrogacy, Geneticism and Equality: The case of AB Minister of Social Development" 2019 *Constitutional Court Review* 9 324.

⁴⁹⁵ Meyerson, D "Surrogacy, Geneticism and Equality: The case of AB Minister of Social Development" 2019 *Constitutional Court Review* 9 324.

⁴⁹⁶ Van Niekerk 2015 *PELJ* 419. The primary beneficiaries are the childless commissioning parent(s).

⁴⁹⁷ Van Niekerk 2015 *PELJ* 419.

⁴⁹⁸ SALRC Project 140 88. Louw 2013 *THRHR* 571. Van Niekerk 2015 *PELJ* 403. Unfortunately, the reality is that in instances where a single commissioning parent or both of the commissioning parents are infertile, they are excluded from using surrogacy as an option.

for a new-born baby or the person or couple may perhaps be too old to qualify for adoption.⁴⁹⁹ These obstacles, in my view, are not serious enough to be convincing, as some couples may want to circumvent the requirements relating to adoption in order to conceive a child through surrogacy. What is true, however, is Van Niekerk's argument that biology or genetics provides no guarantee for the welfare of the child to be born. I do not agree with her view that the absence of a genetic link may in fact provide a better guarantee of the child's welfare.⁵⁰⁰

The question rightly arises whether the genetic link-requirement may be said to infringe upon an infertile person's right to make decisions regarding reproduction as in terms of section 12(2)(a) of the Constitution.⁵⁰¹ Access to donor gametes is fairly simple and no person should be denied the right to make decisions regarding their reproduction.⁵⁰² Regulations to chapter 19, if enacted in future, should also provide clarity on this issue to assist the High Court to exercise its discretion on good cause shown.⁵⁰³ From a justice and fairness perspective, one may ask whether it is fair to exclude a person who is both pregnancy and conception infertile from using surrogacy as a way to have a child. The answer to this question would undoubtedly be "no".

During a webinar, hosted by the University of KwaZulu-Natal in March 2022, it was suggested that a case-by-case approach should be followed in determining if a pregnancy and conception infertile woman is allowed to use surrogacy to have a child.⁵⁰⁴ The judgment in *AB and another v Minister of Social Development*⁵⁰⁵ serves as a good example of a 'good cause shown' scenario. In this case, the commissioning mother turned to surrogacy as a last resort after fourteen failed IVF treatments. Her situation illustrates that she has made many attempts to use her own ova, without any

⁴⁹⁹ Davel and Skelton *Commentary* 15. SALRC Project 140 88. Meyerson, D "Surrogacy, Geneticism and Equality: The case of AB Minister of Social Development" 2019 *Constitutional Court Review* 9 324.

⁵⁰⁰ Van Niekerk 2015 *PELJ* 421. She further argues that what should be important is the commissioning parent(s) suitability to parent, which can be gathered from, amongst other evidence, their intention to parent.

⁵⁰¹ SALRC Project 140 88. Including his or her right to dignity and privacy. *AB and another v Minister of Social Development* par 213.

⁵⁰² Van Niekerk 2015 *PELJ* 420.

⁵⁰³ Van Niekerk 2015 *PELJ* 421.

⁵⁰⁴ Webinar hosted by University of KwaZulu-Natal, organised by Prof Donrich Thaldar, on 2 March 2022. The theme of the webinar was: Is it time to reconsider the genetic link requirement?

⁵⁰⁵ 2017 (3) SA 570 (CC).

success. The constitutionality of section 294 has been brought to the fore in a recent judgment, discussed in detail in chapter 4 of this thesis.

The next section will examine the legal requirements for a court to confirm a surrogate agreement in more detail.

3.6 Section 295: Confirmation of the surrogacy agreement by a High Court⁵⁰⁶

Section 295 directs that the court may not confirm a surrogate agreement unless the commissioning parent(s) are permanently unable to give birth to a child.⁵⁰⁷ In addition, the commissioning parent(s) must be competent to conclude the agreement;⁵⁰⁸ they must be suitable persons to accept the parenthood of the child that is to be conceived;⁵⁰⁹ and they must understand and accept the legal consequences of the

⁵⁰⁶ Sec 295. In re confirmation of three surrogate agreements at para 15 and 27. Ex parte SA and others 2014 JDR 2616 (GP) at para 11. Ex parte WH and others at par 38 and 69. Ex parte MS and others; In re: Confirmation of surrogate mother agreement at par 19 Bosman-Sadie and Corrie Practical approach to the new Children's Act 294-295. Nöthling Slabbert 2012 SAJBL 30. AB and another v Minister of Social Development at par 40.

⁵⁰⁷ Sec 295(a). Report of the Ad Hoc Committee 19, 56 and 59. Surrogacy should only be exercised by the commissioning parent(s) as a last resort. Louw, A "Ex Parte MS 2014 JDR 1012: Surrogate motherhood agreements, condonation of non-compliance with confirmation requirements and the best interests of the child" 2014 De Jure 47(1) 111. SALRC Project 140 85. Jordaan DW "Surrogate Motherhood in illness that does not cause infertility" 2016 SAMJ 684. The writer refers to this requirement as the 'threshold requirement' for surrogate motherhood. He further states that the persons who qualify in terms of this requirement include male same-sex couples and single men (who are biologically unable to give birth) and heterosexual couples or single women where the woman is medically unable to carry a pregnancy to term. Sloth-Nielsen "Surrogacy in South Africa" 194. It is confirmed through case law that medical supporting evidence of the pregnancy or conception infertility of the commissioning parents must be provided unless the commissioning parent is single of a same-sex male couple Florescu, S and Sloth-Nielsen, J "Visions on surrogacy - From North to South: The approach of the Netherlands and South Africa to the issue of surrogacy and the child's right to know his origin" 2017 International Survey of Family Law 2017 247. Both situations of pregnancy infertile and conception infertile would meet the requirements of sec 295(a). Van Niekerk 2015 PELJ 418. The author argues that the mere fact that a commissioning parent is infertile is not a guarantee that he or she will make a good parent to a prospective child.

⁵⁰⁸ Sec 295(b)(i). Report of the Ad Hoc Committee 59. Mills 2010 *Stell LR* 435.

Sec 295(b)(ii). Report of the Ad Hoc Committee 19, 20 and 59. At 20 the commissioning parent(s) should be financially secure and be able to provide a healthy family environment. The Committee recommended that this should be confirmed through the screening process. However, as a result of the principle of non-discrimination and the difference of opinion in our communities as to what constitute a healthy environment, it was argued that the only criteria that should prevent people from becoming commissioning parent(s) should be medial ones. Further, unmarried, divorced or widowed persons who are not involved in a homosexual relationship should qualify. Nicholson and Bauling 2013 *De Jure* 528. The writers point out that possibly the most important and yet uncertain requirements are those regarding the suitability of the commissioning parents and the surrogate who has to act in these respective capacities. At 529 they state that it is necessary for the court to have a wide array of information at its disposal in assessing the suitability of the commissioning parents as their parenting of the intended child will go beyond their contractual

agreement, as well as their rights and obligations in terms of the agreement.⁵¹⁰ The surrogate mother, on the other hand, must be competent to conclude the agreement;⁵¹¹ she must be a suitable person to act as a surrogate mother in all respects;⁵¹² she must also understand and accept the legal consequences of the agreement and the Act and her rights and obligations in terms thereof;⁵¹³ she may not use surrogacy as a source of income;⁵¹⁴ she may only enter into a surrogacy agreement for altruistic reasons and not for commercial purposes;⁵¹⁵ she must have had at least one pregnancy and a viable delivery;⁵¹⁶ and finally, she must have a living child of her own.⁵¹⁷

statement of intentions. Some aspects which need to be evaluated during the psychosocial assessment of the commissioning parents is suggested to include the presence of existing psychological conditions, the reason for their desire to have a child, the length and stability of their relationship and whether or not the parent not genetically linked to the child will display a jealous tendency which might cause conflict in the relationship or adversely affect the child. "The inevitable result of surrogacy and adoption is that prospective parents are held to a higher standard of care than natural parents, but this is unfortunately an unavoidable by-product of the quest to protect the child." LexisNexis *Family Law service* at par W7.

⁵¹⁰ Sec 295(b)(iii Report of the Ad Hoc Committee 61.

⁵¹¹ Sec 295(c)(i). Report of the Ad Hoc Committee 57. It was submitted that any competent woman, regardless of her sexual orientation or marital status, should be allowed to act as surrogate.

⁵¹² Sec 295(c)(ii). Report of the Ad Hoc Committee 57. Nicholson and Bauling 2013 De Jure 529. The courts need more information on the surrogate mother before she can be considered to act in this capacity. It is indicated that a thorough medical report regarding her health and physical suitability to bear a child is necessary. At 530 the writers state that the Children's Act is thus severely lacking as none of the recommendations made by the courts or other writers, and which seem to be crucial to the effective and successful implementation of the surrogate agreement, have been regulated. Reference was also made to recommendations made by the SALC (as it was known then). Since 17 January 2003 the SALC changed to the South African Law Reform Commission (SALRC). A strict six-month screening process and counselling for all parties before and after the conclusion and implementation of the agreement was recommended. It is stated that this provision aimed to ensure the suitable social and psychological backgrounds of all the parties involved. It is recommended by the writers that the promulgation of regulations to the Children's Act is still necessary to address these omissions and when, or if, these are drafted, care should be taken to make sure that provision is made for details pertaining to screening panels, timeframes and aspects to be monitored. See further the discussion of Ex parte WH and others in par 3.7.2. Thaldar, DW "Criteria for assessing the suitability of intended surrogate mothers in South Africa: Reflections on Ex Parte KAF II" 2019 South African Journal of Bioethics and Law 12(2) 61. The writer points out that the section does not provide further clarity regarding what exactly is meant by a 'suitable' person. Report of the Ad Hoc Committee 16. The Committee recommended that the surrogate should be psychologically and physically fit and proper person with moderate social habits.

⁵¹³ Sec 295(c)(iii). Report of the Ad Hoc Committee 57.

⁵¹⁴ Sec 295(c)(iv). Report of the Ad Hoc Committee 18 and 58. The surrogate should be financially secure so as to eradicate the possibility for her to use the agreement as a source of income.

⁵¹⁵ Sec 295(c)(v). Report of the Ad Hoc Committee 23 and 58. At 23 the Committee pointed out that the main arguments in this respect was that commercial surrogacy is degrading to the surrogate as she is dehumanised to being a mere "incubator" and pregnancy is cheapened if it is entered into for financial gain, as there should be certain things in life that money cannot buy. Nicholson 2013 SAJHR 502. Louw 2013 THRHR 564.

⁵¹⁶ Sec 295(c)(vi). Louw 2013 *THRHR* 570.

⁵¹⁷ Sec 295(c)(vii). Oluwaseyi and Oladimeji 2021 *Obiter* 36. The authors explain that this requirement reduces the risk of a refusal to hand over the commissioned baby which can lead to legal disputes that affect such children psychologically. Report of the Ad Hoc Committee 18 and 58.

The court may also not confirm the agreement if the application does not include adequate provisions for the contact, care, upbringing and general welfare of the child that is to be born in a stable home environment, including the child's position in the event of the death of the commissioning parents or one of them, or their divorce or separation before the birth of the child.⁵¹⁸ The court must have regard to the personal circumstances and family situations of all the parties concerned, but above all the interests of the child that is to be born.⁵¹⁹ The issue of sexual intercourse between the surrogate and her husband or partner during her pregnancy needs to be considered

⁵¹⁸ Sec 295(d). Report of the Ad Hoc Committee 50. Nicholson and Bauling 2013 *De Jure* 522. The writers point out that our courts have not as yet stipulated what should be incorporated in the 'care' clauses in the agreement in order to be deemed sufficient. The further uncertainty is regarding what constitutes a stable home or by whom the existence thereof should be determined. They further point out that the suitability of the commissioning parents is intrinsically linked to the care-requirement. It is thus clear that a regulation to the Act regarding the above would be helpful to both the parties to the application as well as the court hearing the matter. Slabbert and Roodt "South Africa" 332. SALRC Project 140 85. *Ex Parte SA and others* (case no 82202/2014). At par 15 the Court pointed out that proof was given of the appointed guardians for the child/children in the event of death of the commissioning parents as well as evidence in respect of the financial welfare of the child/children in case of death of one or both of the commissioning parents or in case of divorce or separation before the child/children is born Mills 2010 *Stell LR* 435. Sloth-Nielsen "Surrogacy in South Africa" 190. Oluwaseyi and Oladimeji 2021 *Obiter* 32.

⁵¹⁹ Sec 295(e). Nicholson and Bauling 2013 De Jure 531. It was mentioned by the writers that the parties may voluntarily include other types of clauses (incidentalia) in their surrogate agreement that is not set out in the Act. These incidentalia should be lawful and in accordance with the prevailing boni mores. Examples of clauses are: social disease testing on all the parties; assurance provided by the commissioning parents that they will accept parentage of the child regardless of the physical or mental defects or disabilities; clarification of the situation where such disabilities are due to actions of the surrogate mother; clarification regarding a duty of care where the surrogate mother is advised by a medical professional that it is better to terminate the pregnancy and she chooses not to do so; specifications regarding the number of embryos to be transferred at one time; etcetera. At 532 the writers further state that the fact that the surrogate mother should observe a sensible lifestyle, diet and standard of hygiene throughout the pregnancy is also viewed as a contractual stipulation which should be required by legislation. It is however still unclear how non-compliance with the requirements of such a personal nature could be enforced. It is important that the financial arrangements between the parties is set out clearly in order to protect all the parties to the agreement. It is recommended that the financial clauses should specify all the aspects that will be compensated for, as claims above and beyond what is stipulated might not be contractually enforceable. At 533 it is further stated that stipulations regarding the breach of contract by any of the parties should be addressed as well and highlighted in the incidentalia. It is thus better for the parties to the agreement to act preventatively and include as much information in the agreements as possible as this will help the court in its attempt to balance the true intention of the parties with what ch 19 of the Children's Act prescribe. Slabbert and Roodt "South Africa" 332. Rautenbach, I.M "Overview of constitutional court judgments on the bill of rights - 2016" 2017 Journal of South African Law 2017(2) 363. The writer argues that we should never lose sight of the reason why sec 28(2) of the Constitution refers to the 'paramount' importance of children's interests in every matter. It is explained that the fundamental reasons are that children are not capable of defending their own interests and that they more than often become innocent victims of endeavours by adults who exercise their 'autonomy' to assert their adult rights. Oluwaseyi and Oladimeji 2021 Obiter 32.

by the parties to avoid any problems which may arise as a result of the paternity of the commissioned child.⁵²⁰

The SALC (as it was known then) was of the opinion that, given the extent of the risk factors inherent in surrogacy, it should be accessible only as a last option.⁵²¹ The SALC observed that the majority of problematic cases in surrogacy arose as a result of insufficient screening of the parties involved.⁵²² Importantly, the social and psychological background of the parties should be meticulously scrutinised to serve the best interests of the child to be born.⁵²³ Years ago, this requirement was emphasised by the Ad Hoc Committee who recommended proper screening for physical and psychological suitability of all the parties to the agreement and presentation of these results to the court in support of the application for the confirmation of the surrogate agreement.⁵²⁴

3.6.1 "The commissioning parent or parents are not able to give birth to a child and that the condition is permanent and irreversible" (s 295(a))

At the time, the SALC recommended that a surrogate agreement should only be allowed in circumstances where it is proved that the commissioning wife is unable to give birth to a child because of a medical condition.⁵²⁵ This view is the result of the fact that artificial fertilisation at that time was available to a couple under the then (and now repealed) Human Tissue Act.⁵²⁶

⁵²⁰ Report of the Ad Hoc Committee 34.

⁵²¹ SALC Project 65 at par 8.2.1. Report of the Ad Hoc Committee 59.

⁵²² SALC Project 65 at par 8.2.3. At par 8.2.9 the SALC submitted that the surrogate should be in good health with no mental defect. Further, the psychological impact of the surrogacy on herself, her family and her immediate social circle is a factor that has to be taken into account. It is important to determine if she has already given birth or has completed her family as this may have an effect on her relinquishing the child.

⁵²³ SALC Project 65 at par 8.2.3. Further, the surrogate and the person whose gametes are being used should be subjected to strict medical selection.

⁵²⁴ Report of the Ad Hoc Committee 45. The reports should also be made available to each party. At 62 the Committee recommended a continuous process of counselling before and after the conclusion and implementation of the agreement. At 63 the Committee recommended a screening of all the parties to the agreement six months before the conclusion of the agreement. Further, compulsory HIV testing of all the parties involved for a period of 12 months before the artificial fertilisation of the surrogate.

⁵²⁵ SALC Project 65 at par 8.2.4.

⁵²⁶ 65 of 1983. SALC Project 65 at par 8.2.4.

Section 295(a) retains this original requirement, which has been extended to include one or both commissioning parents. Up to date, there is no complete list detailing what the possible cause of the inability to give birth could be, just as long as the condition is permanent and irreversible.⁵²⁷

An important question arising from section 295(a) is whether all three requirements mentioned in the provision should be present contemporaneously.⁵²⁸ In the case of a gay commissioning couple, all three the requirements (permanent medical condition, irreversible medical condition and unable to give birth to a child) may be easily met and the requirements of the Act thus complied with.⁵²⁹ The question is more complex where a heterosexual commissioning mother suffers from a serious medical condition which could be exacerbated by a pregnancy or where she is unlikely to survive a pregnancy.⁵³⁰ Despite her infertility, a commissioning mother may still be able to carry and give birth to a child by reason of the fact that her uterus is intact.⁵³¹

Florescu and Sloth-Nielsen argue that once the threshold in section 295(a) is established, a couple would meet the requirement of section 294(a).⁵³² However, the couple would still be required to provide a gamete for the purposes of fertilisation.⁵³³ A narrow interpretation of section 295(a) would clearly lead to an insensible result, as it would force a woman who wishes to have her own biologically related child, to undergo a pregnancy which might result in significant and/or life-threatening medical harm to herself.⁵³⁴

⁵²⁷ Davel and Skelton *Commentary* 18. One can assume that a medical practitioner would need to testify or a medical report would need to be filed to show the cause the wife's inability. Nicholson 2013 *SAJHR* 501. Heaton 2015 *THRHR* 29.

⁵²⁸ LexisNexis *Family Law service* at par W6.

⁵²⁹ LexisNexis *Family Law service* at par W6.

⁵³⁰ LexisNexis *Family Law service* at par W6.

⁵³¹ LexisNexis *Family Law service* at par W6.

⁵³² Florescu, S and Sloth-Nielsen, J "Visions on surrogacy – From North to South: The approach of the Netherlands and South Africa to the issue of surrogacy and the child's right to know his origin" 2017 International Survey of Family Law 2017 248.

⁵³³ Florescu, S and Sloth-Nielsen, J "Visions on surrogacy – From North to South: The approach of the Netherlands and South Africa to the issue of surrogacy and the child's right to know his origin" 2017 International Survey of Family Law 2017 248.

⁵³⁴ APP and another v NKP 2021 JDR 1650 (WCC)/ (17962/2020) [2021] ZAWCHC 69 (11 March 2021) at par 24.

3.6.2 "The commissioning parent or parents — (i) are in terms of this Act competent to enter into the agreement; (ii) are in all respects suitable persons to accept the parenthood of the child that is to be conceived; and (iii) understand and accept the legal consequences of the agreement and this Act and their rights and obligations in terms thereof; are in all respects suitable persons to accept the parenthood of the child that is to be conceived" (s 295(b))

The existence of a family unit and common household is regarded as an important aspect in surrogacy applications, as it is desirable that a family share a community of life.⁵³⁵ There is acknowledgment and respect for the many varied permutations of what a family would constitute of in South Africa's constitutional dispensation. The court will look at the circumstances of each case.⁵³⁶ In the judgment of *Ex parte WH and others*,⁵³⁷ the court cautions against the setting of unreasonably high standards that are not justifiable for people who select surrogacy as an option to have a child, since that would contravene the spirit of the Constitution and the Equality Act.⁵³⁸ It is an objective test that needs to be applied when determining if the commissioning parent(s) is or are suitable or not.⁵³⁹

Section 295 provides no indication as to whom must inform the parties about the legal consequences of the agreement and the court should *mero motu* be able to satisfy itself that the parties understand and accept the legal consequences of the agreement. Section 295(b)(iii) determines that a court may not confirm a surrogate agreement without the commissioning parent(s) understanding and accepting the legal consequences of the agreement and this Act, as well as their rights and obligations in terms of the said agreement and Act. This provision can be seen as vital insofar as the aim of eliminating the most obvious risks inherent in surrogate arrangements is concerned. Taking into account the importance for the parties to understand the legal consequences of the arrangement, it is probably more imperative that the surrogate (and her spouse or partner) accept and understand the legal consequences because

⁵³⁵ Ex parte CJD and others at par 22 - 24.

⁵³⁶ Ex parte CJD and others at par 26.

⁵³⁷ 2011 (6) SA 514 (GNP) at para 54.

⁵³⁸ 4 of 2000. *Ex parte WH and others* at par 70.

⁵³⁹ *Ex parte WH and others* at par 70.

she is the one who will be exposed to all the risks inherent in the conclusion of the agreement, including that she is the person subjecting herself to the physical risks of artificial fertilisation and pregnancy.

3.6.3 "The surrogate mother— is in terms of this Act competent to enter into the agreement; (ii) is in all respects a suitable person to act as surrogate mother; (iii) understands and accepts the legal consequences of the agreement and this Act and her rights and obligations in terms thereof" (s 295(c))

The surrogate needs to understand that she would need to hand the child over to the commissioning parent(s) upon the birth of the child, as it is not her child, even in circumstances where she is genetically related to the child. This provision justifies why psychological screening of the surrogate is very important. It is necessary to make sure that the surrogate fully understand what will be expected of her and to what she actually agrees in the surrogate agreement.

3.6.4 "The surrogate mother— (iv) not using surrogacy as a source of income;
(v) has entered into the agreement for altruistic reasons and not for commercial purposes; (vi) has a documented history of at least one pregnancy and viable delivery; and (vii) has a living child of her own" (s 295(c))

Sections 295(c)(iv) and 295(c)(v) determine that a court may not confirm a surrogate agreement if the surrogate is using surrogacy as a form of livelihood, and where she has entered into the agreement for commercial purposes instead of for altruistic purposes.⁵⁴⁰ One of the strongest concerns with regard to surrogacy is that it can lead to baby-selling if the surrogate stands to gain financially from the arrangement.⁵⁴¹ By inclusion of this section, the Children's Act aims to prevent the situation where the

⁵⁴⁰ Sec 295(c)(iv) and (v). The court may not confirm an agreement which provides for commercial surrogacy which means that the agreement is inevitably invalid.

⁵⁴¹ SALC Project 65 at par 2.5.1 and 8.2.8. Louw 2013 *THRHR* 570. The writer states that if the surrogate mother's child is no longer alive, that her motives for becoming a surrogate mother should be investigated, but should this automatically make her unsuitable?

surrogate mother is motivated by pure financial gain.⁵⁴² This was also the view of the SALC at the time, when it submitted that this could be seen as a form of exploitation if an economically disadvantaged woman is used to carry a child for a more affluent member of the community.⁵⁴³ This requirement appears to limit the possible abuse of economically disadvantaged women and preventing problems from arising in the event of the commissioning parent(s) refusing to pay the agreed amount or to take the child.⁵⁴⁴

Section 295(c)(vii) requires that the surrogate mother must have a child of her own.⁵⁴⁵ This requirement of already having an existing child could be seen as both an advantage and disadvantage.⁵⁴⁶ The rationale for this provision appears from the SALC's report that stated that when a surrogate gives birth to her own child, she would be able to appreciate the risks related with pregnancy and the implications of surrendering a child upon birth.⁵⁴⁷ The other side of this, however, is that the child(ren) of the surrogate could be traumatised upon surrendering the child, fearing that they too will be given away.⁵⁴⁸ Despite the latter concern, it remains a fact that having previously been pregnant and giving birth to a child(ren), will at least provide a reasonable measure of physical suitability, if not also the desired psychological disposition.⁵⁴⁹ The reasons for this requirement is arguably linked to the possibility that a surrogate who does not have a child of her own may be more inclined to bond with the commissioned child, making her more reluctant to surrender the child born as a result of such an agreement.⁵⁵⁰

⁵⁴² SALC Project 65 at par 2.4.5. *Ex parte KAF and others (I)* (14341/17) [2017] ZAGPJHC 227 (10 August 2017) at par 18. Bonthuys and Broeders 2013 *SALJ* 490. The writers argue that the emphatic prohibition on commercial surrogacy is expressed as a prohibition on payment for the surrogate beyond the allowed reimbursement for specified categories of expenses and loss of earnings.

⁵⁴³ SALC Project 65 at par 2.4.5.

⁵⁴⁴ SALC Project 65 at par 2.4.5.

⁵⁴⁵ LexisNexis *Family Law service* at par W9. Clarke states that details of the surrogate's obstetric history, her pregnancies, deliveries and any complications experienced during a pregnancy must be set out in the application. Clarke argues that the surrogate must at least have one living child of her own. She will not qualify as a surrogate if her child is deceased or if she adopted a child.

⁵⁴⁶ Davel and Skelton *Commentary* 22.

⁵⁴⁷ SALC Project 65 at par 2.6.1.

⁵⁴⁸ SALC Project 65 at par 2.6.1.

⁵⁴⁹ SALC Project 65 at par 8.2.9.

⁵⁵⁰ SALC Project 65 at par 8.2.9.

3.6.5 "The agreement includes adequate provisions for the contact, care, upbringing and general welfare of the child that is to be born in a stable home environment, including the child's position in the event of the death of the commissioning parents or one of them, or their divorce or separation before the birth of the child" (s 295(d))

Section 295(d) determines that a court may not confirm a surrogate agreement if the agreement does not include satisfactory provisions for the contact, care, upbringing, and general welfare of the child that is to be born in a stable home environment, including what will happen with the child in the event of the death, divorce or separation before the birth of the child. The parties thus have to consider the care of the child in the event of a change in the commissioning parties' circumstances caused by death, separation or divorce before the commissioned child is born.⁵⁵¹ The parties are thus obliged to reach consensus on the care of the child to be born and the word "including" seems to allow the parties to make provision for matters not regulated by the Act.⁵⁵² One such provision may relate to the appointment by the commissioning parents of suitable guardians for the child to be born in the event of death.⁵⁵³ Clark argues that it is advisable for the commissioning parents to take out life insurance for the benefit of the child to be born as a result of the agreement.⁵⁵⁴ It is critical to remember, as Dhai and McQuoid-Mason advise, that any unfair clauses in the surrogacy agreement may fall foul of the Consumer Protection Act.⁵⁵⁵

⁵⁵¹ Nöthling Slabbert 2012 SAJBL 28.

⁵⁵² Davel and Skelton *Commentary* 21. It is further explained that other important aspects that can be included is for example, the health and insurance policies to be maintained by all parties throughout the agreement; the financial responsibility of the parties requiring funds to be placed in trust to cover all anticipated expenses; specific arrangements regarding the child or children, including custody in the case of divorce or death of the commission parents and visitation rights, if any of the surrogate mother, a provision requiring that social disease testing, incl. HIV, be performed on all parties as well as a provision that the proposed surrogate mother be medically examined and declared suitable, etcetera.

⁵⁵³ LexisNexis *Family Law service* at par W10. The commissioning parents should make provision in their will for the appointment of a guardian(s) for the child. She advises that details on the relationship of the guardian(s) to the commissioning parents must be explained and the guardians must be suitable person(s) to accept the guardianship of the child to be born and the person(s) must be willing and able to act in such a capacity. Letter(s) of acceptance of the guardianship should be attached to the affidavits in the application to court.

⁵⁵⁴ LexisNexis *Family Law service* at par W10.

⁵⁵⁵ 68 of 2008. Dhai and McQuoid-Mason *Bioethics, Human Rights and Health Law* 146. The authors made specific reference to sec 49.

3.6.6 "In general, having regard to the personal circumstances and family situations of all the parties concerned, but above all the interests of the child that is to be born, the agreement should be confirmed" (s 295(e))

Lastly, section 295(e) determines that a court may not confirm a surrogacy agreement unless the court has regard to the personal circumstances and family situations of all the parties to the agreement, and especially, the best interests of the child that is to be born.⁵⁵⁶ This provision places the responsibility on the court to make sure that the confirmation of the agreement will be in the best interests of the commissioned child to be born.⁵⁵⁷ This was confirmed by the Constitutional Court in *AB and another v Minister of Social Development*, when it emphasised that the court must on every possible occasion make a decision whether or not to confirm a surrogate agreement, and engage with the value judgment of whether the confirmation would serve the best interests of the commissioned child to be born.⁵⁵⁸ The best interests yardstick requires a flexible enquiry which must be determined on the facts of each particular case.⁵⁵⁹

One conclusion that may be drawn from this provision is that the Act seems to require strict adherence to narrowly defined requirements, yet leaves a back door of discretion open if, in the court's opinion, the confirmation would not be in the resultant child's best interest.⁵⁶⁰ For Rautenbach, the significance of section 295(e) regarding the different interests that it serves may be summarised as follows:

An interesting and outstanding feature of section 295(e) of the Children's Act is that it provides protection to the best interests of children yet to be conceived when infertile

⁵⁵⁶ Sec 295(e). Heaton 2015 *THRHR* 30.

⁵⁵⁷ Nöthling Slabbert 2012 SAJBL 28.

⁵⁵⁸ AB and another v Minister of Social Development at par 192.

⁵⁵⁹ AB and another v Minister of Social Development at par 193.

⁵⁶⁰ Louw 2013 *THRHR* 572. The writer states that while investing the court with such residual discretionary powers is not uncommon in legislation, it may be problematic in the context of confirmation proceedings. Since it is so difficult to determine what would be in the resultant child's best interests, it is not entirely clear on what grounds the court would be able to exercise its residuary discretion. Unless perhaps it is argued that the refusal may be justified on the basis that the parties have not in the agreement sufficiently catered for e eventualities that are reasonably foreseeable as far as the child is concerned. She further states that the vagueness of sec 295(e) in which the discretionary powers of the court are outlined also makes it difficult to ascertain how (in)flexible the Children's Act was meant to be as far as the regulation and confirmation of surrogate agreements are concerned. The ambivalence found in this regard in the Children's Act would seem to be reflected the approach which was adopted by the court in the WH case.

people take steps to alleviate the negative psychological effects of their infertility by concluding surrogacy agreements.⁵⁶¹

Part of the court's responsibility is to consider all the personal circumstances of the parties concerned.⁵⁶² Thaldar argues that there is a *lacuna* in the South African law on surrogacy in that it lacks objective criteria for evaluating the suitability of a surrogate mother.⁵⁶³ He posits that in the absence of clear, objective criteria, the different psychologists and the different judges will all be working with their own tacit, subjective criteria and this is clearly not in the interests of justice.⁵⁶⁴

A recent case confirms Thaldar's concern. In a 2017 confirmation application in *Ex parte KAF and others (I)*,⁵⁶⁵ the court expressed two pertinent concerns: (a) the objectivity of the relevant health care professionals; and (b) the possible commercialisation of surrogacy.⁵⁶⁶ The court cautions that there is a need for medicolegal experts involved in the surrogacy process to be impartial and objective regarding the facts at hand in their attempt to assist the court.⁵⁶⁷ These professionals owe their allegiance to the court and not to the parties to the surrogacy arrangements, the court concludes that it is desirable that medico-legal professionals, who are involved in the process, function independently.⁵⁶⁹

⁵⁶¹ Rautenbach 2017 *TSAR* 363.

⁵⁶² In the judgment of *Ex parte WH and others* the court emphasised that the court as upper guardian of all minors is not simply a rubber stamp validating the private arrangements between the parties. At 531D the court concluded that the affidavit supporting the application for confirmation should contain all the factors as set out in the Act together with documentary proof where applicable. Nicholson and Bauling 2013 *De Jure* 520-521. The writers give the example of documentary proof regarding the sterility of the commissioning parent/parents where such proof may take the form of a letter from a specialist medical practitioner stating the exact cause of the irreversibility of the sterility. The writers further stated that the current lack of regulations pertaining to ch 19 hinders the process of compiling an application for the confirmation of a surrogate agreement as it still remains unclear what constitutes sufficient evidence of compliance with the requirements set out in the Act.

⁵⁶³ Thaldar 2018 *SAJBL* 35. He explains that a court gradually, on a case-by-case basis as required by the facts of a particular matter, applies the general principles of our law to the requirements stipulated by the Act. Metaphorically, the Act created the legal hoops and the court then determines how one is to jump through the hoops. In the matter of *Ex parte KAF* the court determined how not to jump through the hoops.

⁵⁶⁴ Thaldar 2018 *SAJBL* 36.

⁵⁶⁵ 2017 ZAGPJHC 227. (Hereinafter referred to as *Ex parte KAF and others (I)*). This case is discussed in detail in par 3.7.7. Thaldar 2018 *SAJBL* 36.

⁵⁶⁶ Thaldar 2018 *SAJBL* 36.

⁵⁶⁷ Ex parte KAF and others (I) at par 27.

⁵⁶⁸ Ex parte KAF and others (1) at par 27.

⁵⁶⁹ Ex parte KAF and others (I) at par 28.

In Ex parte KAF, the first confirmation of a surrogate agreement application was dismissed by the court.⁵⁷⁰ In the second confirmation application in the same matter,⁵⁷¹ the court discusses section 292(1)(e), read together with section 295 of the Act, focussing specifically on the suitability of the intended surrogate mother and the criteria to be applied in assessing her suitability.⁵⁷² The court alludes to the dangers of commercialisation of surrogacy and states that the effort to achieve a balance between the rights and interests of all the parties to a surrogate arrangement is a delicate one.⁵⁷³ The court also points out that considering the interests of the child to be born, it is important to subject an application to strict judicial scrutiny, especially since the court is the upper guardian of all minor children.⁵⁷⁴ Reference was made in Ex parte KAF and others cases to the earlier cases of Ex parte WH and others⁵⁷⁵ and Re Confirmation of three Surrogate Motherhood Agreements,⁵⁷⁶ where criticism that the Act fails to establish objective criteria regarding the supporting documents that should be placed before the court in support of a confirmation application, was also raised.⁵⁷⁷ The Ex parte KAF and others cases reiterates that more clarity is needed regarding the requirements for assessing the suitability of the surrogate mother.⁵⁷⁸

The need for regulations to provide clarity as to the type of information to be provided to court to enable the court to make an informed decision in respect of confirming a

⁵⁷⁰ *Ex parte KAF and others (I)* at par 29-30. Thaldar "Criteria for assessing the suitability of intended surrogate mothers in South Africa: Reflections on *Ex Parte KAF II*" 2019 *SAJBL* 61.

⁵⁷¹ Ex parte KAF and others (II) 2019 (2) SA 510 (GJ). (Hereinafter referred to as Ex parte KAF and others (II)) at par 1. This case is discussed in detail in par 3.7.7.

⁵⁷² Ex parte KAF and others (II) at par 1 and 11.

⁵⁷³ Ex parte KAF and others (II) at par 7. Thaldar 2019 SAJBL 64. Thaldar referred to the experts (used in the application) and that they suggested that the intended surrogate mother should be financially stable in that she or her family unit must have a reliable source of income and be living within their means.

⁵⁷⁴ Ex parte KAF and others (II) at par 16.

⁵⁷⁵ See par 3.7.2 below for full discussion of case.

⁵⁷⁶ See par 3.7.1 below for full discussion of case.

⁵⁷⁷ Ex parte KAF and others (II) at par 17. The Court stated that it is evident that given the discretionary powers conferred and the likely different circumstances where the court's confirmation is sought, each court may construe the parameters for the exercise of discretion differently from the next.

⁵⁷⁸ Ex parte KAF and others (II) at par 20. Thus, the court said, there is a need to develop further the guidelines and requirements as was set out in Ex parte WH and others. (See par 3.7.2 below). Thaldar 2019 SAJBL 64-65. The writer states that the main issue with the judgment that detracts from its practical utility is that many criteria are vaguely formulated to the point of being merely factors for consideration and thus lacking any clear standard that must be complied with. He concluded by saying that "by adding factors for consideration without indicating exactly how such factors should impact on the overall assessment of suitability, the court is not solving the problem of uncertainty regarding how to assess suitability. In fact, the court is creating additional loci of uncertainty.

surrogate agreement is long overdue. The regulations should also provide clarity on what is required in respect of the "care" of the commissioned child. Since a stable home environment has different meanings for different people and different cultures, a consensus description of a "stable home environment" will put the parties in a position to provide the court with the necessary information. Guidance is furthermore needed regarding the evidence necessary for complying with section 295(a)'s requirements. For example, does the court require various medical reports or will one report by a specific expert suffice? The need for clarity also relates to the assessment of the suitability of the parties to the surrogacy agreement.

The next discussion will turn to section 296 of the Act. From a medical perspective, surrogacy starts when the surrogate mother is artificially fertilised.

3.7 Section 296: Artificial fertilisation of the surrogate mother⁵⁷⁹

Because the surrogate will carry the commissioned child to term, and this child must be genetically linked to the commissioning parent(s), the surrogate must be artificially fertilised. Thus, even in circumstances where the surrogate's own ovum will be used, she still has to participate in the process of artificial fertilisation.⁵⁸⁰ Section 296 prohibits the surrogate to be artificially fertilised before the surrogate agreement is confirmed by the court,⁵⁸¹ which may not take after the lapse of eighteen months from the date of the confirmation of the said agreement by the court.⁵⁸²

⁵⁷⁹ The Act. In re confirmation of three surrogate agreements at para 15. Ex parte MS and others; In re: Confirmation of surrogate mother agreement at par 20. Bosman-Sadie and Corrie Practical approach to the new Children's Act 296. Nöthling Slabbert 2012 SAJBL 28. AB and another v Minister of Social Development at par 41.

⁵⁸⁰ Partial or traditional surrogacy is explained in par 2.11 above.

Sec 296(1)(a). SALC Project 65 at par 8.3.2. Louw 2014 *De Jure* 116. Louw points out that health professionals acting in contravention of the prohibition remain open to criminal prosecution provides no assurance of future compliance with the requirement. Nöthling Slabbert 2012 *SAJBL* 28. SALRC Project 140 85. Thaldar 2019 *SAJBL* 61. *AB and another v Minister of Social Development* at par 41.

Sec 296(1)(b). Nöthling Slabbert 2012 SAJBL 28. Slabbert and Roodt "South Africa" 332. Louw 2013 THRHR 570. The writer asks the question of why the artificial fertilisation of the surrogate may not take place later than 18 months after the confirmation of the surrogate agreement as it is a well-known fact that artificial fertilisation procedures are time-consuming and multiple attempts may be needed to conceive. Further, by reasoning that a time limit would avoid circumstances changing too much after the confirmation of the agreement also does not hold sway as circumstances, such as death or separation of the commissioning parents, can change at any time. Sloth-Nielsen "Surrogacy in South Africa" 192.

The artificial fertilisation of a surrogate in the execution of the said agreement must furthermore be performed in accordance with the provisions of the NHA⁵⁸³ and Regulations.⁵⁸⁴

It is important to bear in mind what the consequences will be if the surrogate is artificially fertilised without the court confirming the agreement first. The purpose of the above provision seems to protect the parties to the agreement, and especially the surrogate and/or her partner or spouse.⁵⁸⁵ Although legal precedent exists where the court confirmed the surrogate agreement after artificial fertilisation of the surrogate, it was done to protect the best interests of the children to be born. It is possible that faced with another scenario, the court may refuse to confirm the agreement 'after the fact'.⁵⁸⁶ Should the agreement not be confirmed, and as a consequence be invalid, any child born as a result of the invalid surrogate agreement will be deemed to be the child of the surrogate for all purposes.⁵⁸⁷ As mentioned earlier, the child born will thus not be seen as that of the commissioning parents and they would then have to adopt the child.⁵⁸⁸

As was cautioned by the SALC earlier, it is of fundamental importance that the best interests of the commissioned child be considered *before* conception.⁵⁸⁹ Situations where a court has to make a decision after conception should be prevented at all costs.⁵⁹⁰

⁵⁸³ 61 of 2003. Sec 2. Louw 2014 *De Jure* 116. Louw further points out that neither the NHA nor the Regulations specifically address the artificial fertilisation of a surrogate mother in the execution of a surrogate agreement. He argues that the health professional can therefore simply deny any knowledge of the surrogacy agreement at the time of performing the procedure, thereby escaping prosecution.

⁵⁸⁴ Regulations relating to the artificial fertilisation of persons published under Government Notice R175 in Government Gazette 35099 of 2 March 2012.

⁵⁸⁵ Sec 292 and sec 297(2).

⁵⁸⁶ Judgments will be discussed below.

⁵⁸⁷ Sec 297(2). Nicholson and Bauling 2013 *De Jure* 523. Heaton 2015 *THRHR* 35. The author explains that the wording of sec 296 suggests that the legislature envisaged that the section would apply only to confirmed surrogate agreements.

⁵⁸⁸ Sec 299(d). Nicholson and Bauling 2013 *De Jure* 523. It can be a complicated process should the commissioning parent(s) be biologically linked to the child as there are problems associated with adopting one's own child. Thus, when the parentage of a child comes into question, it could have disastrous effects on the child. Heaton 2015 *THRHR* 35 and 38. The commissioning parent(s)' names would not appear on the birth certificate as the surrogate, and if she is married, her husband or civil partner would be registered as the parents of the child born as a result of the invalid surrogate agreement.

⁵⁸⁹ SALC Project 65 at par 8.2.14.

⁵⁹⁰ SALC Project 65 at par 8.2.14.

Section 296 further prohibits the artificial fertilisation of the proposed surrogate after a lapse of eighteen months after said confirmation by the court.⁵⁹¹ This restriction may be seen as a way of preventing the circumstances of the parties involved from changing so drastically over the time that the agreement no longer reflects the real circumstances and/or intentions of the said parties, in spite of contractual provision for some changes in the agreement, for instance death or divorce of the commissioning parents.⁵⁹² However, conception may not always be achieved within this time frame, which will mean that the parties would need to approach the court for a new confirmation of the surrogacy agreement.⁵⁹³ The pregnancy must be established within the stated eighteen months but the surrogate does not have to give birth to the child within the period.⁵⁹⁴

Clarke argues that the last-mentioned provision does not clarify whether the commissioning parents are permitted to cryopreserve their gametes for future use in instances where surrogacy is medically indicated.⁵⁹⁵ She argues that an interpretation of section 303(2) of the Children's Act which provides that "no person may in any way

(https://www.glowm.com/section_view/heading/gamete-and-embryo-

⁵⁹¹ Sec 296(1)(b). Nicholson and Bauling 2013 *De Jure* 523. The writers point out that parties are ignoring the requirement that the court confirmation of the surrogate agreement must precede the artificial fertilisation of the surrogate and that the law is thus being flouted, either out of ignorance or disrespect and this may be to the detriment of the parties or the child.

⁵⁹² Davel and Skelton *Commentary* 23.

⁵⁹³ Nicholson and Bauling 2013 *De Jure* 522. The writers point out that the process of artificial fertilisation can be time consuming and it is not always successful on the first or second attempt which may necessitate the drafting and confirmation of a further surrogate agreement. This can lead to further costs. LexisNexis *Family Law service* at par W11. Clarke argues that the parties may apply to Court for an extension of the time period by showing good cause why an extended period of time is required for further artificial fertilisation treatments of the surrogate. She states that the extension is usually required for medical reasons especially in instances where the commissioning mother has utilised her own eggs without success and will require donor eggs to be fertilised with her partner's sperm/gametes for the sake of improving their chances of conception.

⁵⁹⁴ Lexis *Family Law service* at par W11.

⁵⁹⁵ LexisNexis *Family Law service* at par W11. Reg 1 of Regulations 2021. Freezing or cryopreservation is defined as freezing or cryopreserving genetic material including ova, sperm, embryos, ovarian tissue or stem cells by an authorised institution. Further, it is the ability to freeze and thaw with retention of viability—provides flexibility in human infertility therapy when gametes or embryos are handled *in vitro* because frozen tissue can be stored indefinitely in liquid nitrogen at -196°C. Gamete and embryo cryopreservation accessed from the Global Library of Women Medicine

cryopreservation/item/365#:~:text=Cryopreservation%E2%80%94the%20ability%20to%20freez e,nitrogen%20at%20%2D196%C2%B0C) (accessed on 14 February 2021). She further argues that cryopreservation of embryos or gametes is often medically indicated, for example, prior to chemotherapy treatment or where the intended parent is placed on medication which may adversely affect their gametes.

for or with a view to compensation make known that any person is or might possibly be willing to enter into a surrogate motherhood agreement", would imply that even rendering assistance in artificial fertilisation without the authorisation of the court would be prohibited.⁵⁹⁶ This conclusion is strengthened by the fact that the definition of artificial fertilisation includes *in vitro* fertilisation in terms of the NHA. Thus, this seems to indicate that no fertilisation is permitted to take place of the egg and the sperm; however, the prohibition does not extend to the extraction of a woman's eggs and the subsequent cryopreservation thereof in anticipation of being used in a future surrogate agreement.⁵⁹⁷

This discussion on the application of section 296 also points to the need for clarity in regulations regarding the issue of cryopreservation of embryos or gametes, especially in circumstances where the parties decide to extract gametes before lodging an application for confirmation of a surrogate motherhood agreement. Looking at the Children's Act more closely, there is no prohibition in chapter 19 regarding the extracting of gametes. In specific cases, it is conceivable that this may be the only way for the commissioning parents to use their gametes when required. An example would be where one or both the commissioning parents must undergo medical treatment that carries the risk of affecting the viability of the gametes. It is my submission that in these instances, gametes should be withdrawn and frozen before treatment of the person commences. In these instances, the commissioned child will still have a genetic link to the commissioning parents.

The discussion will next turn to the consequences of the surrogate motherhood agreement for the relevant parties thereto.

3.8 Section 297: The consequence of a surrogate motherhood agreement on the status of the child⁵⁹⁸

⁵⁹⁶ LexisNexis *Family Law service* at par W11. Act 61 of 2003. The definition of *in vitro* fertilisation is explained in par 1.2.5 above.

⁵⁹⁷ LexisNexis *Family Law service* at par W11.

⁵⁹⁸ The Act. Ex parte WH and others at par 58. Ex parte MS and others; In re: Confirmation of surrogate mother agreement at par 22. Bosman-Sadie and Corrie Practical approach to the new Children's Act 296-297. Nöthling Slabbert 2012 SAJBL 28. AB and another v Minister of Social Development at par 42.

This section determines what the *effect* of a valid surrogate agreement is, as well as the effect on the legal status of the commissioned child.⁵⁹⁹ Should the agreement be valid, any child born of a surrogate in terms of the agreement is for all purposes legally the child of the commissioning parent(s) from the moment of the birth of the child concerned;⁶⁰⁰ the surrogate is obliged to hand the child over to the commissioning parent(s) as soon as it is reasonably possible after the birth;⁶⁰¹ the surrogate or her husband, partner or relatives has no rights of parenthood or care of the child;⁶⁰² the surrogate or her husband, partner or relatives has/have no right of contact with the child unless it is specifically provided for in the surrogate agreement may not be terminated after the artificial fertilisation of the surrogate has taken place;⁶⁰⁴ and the child will not have a claim for maintenance or of succession against the surrogate mother, her husband or partner or any of their relatives.⁶⁰⁵

Section 297(2) determines further that any surrogate agreement that does not comply with the provisions of the Act is invalid and the child born as a result of any action taken in execution of such an arrangement, is for all purposes deemed to be the child of the woman that gave birth to that child.⁶⁰⁶ This section must be read in conjunction with and subject to the provisions of section 298, which provides for the dissolution of a surrogate agreement, and section 299, which deals with the consequences of

⁵⁹⁹ Louw 2014 *De Jure* 111. Florescu, S and Sloth-Nielsen, J "Visions on surrogacy – From North to South: The approach of the Netherlands and South Africa to the issue of surrogacy and the child's right to know his origin" 2017 *International Survey of Family Law* 2017 248.

⁶⁰⁰ Sec 297(1)(a). Mills 2010 Stell LR 435. Sloth-Nielsen "Surrogacy in South Africa" 191.

⁶⁰¹ Sec 297(1)(b). SALC Project 65 at par 8.3.10. The child should be handed over to the commissioning parents as soon as possible despite any disputes regarding compensation between the parties. The commission submitted that handing over of the child should have no connection with the payment of the compensation Report of the Ad Hoc Committee 33. Nöthling Slabbert 2012 *SAJBL* 31. LexisNexis *Family Law service* at par W12. Provision may be made in the agreement for the post birth protocols which are to be followed between the parties. Examples would be, who may be present at the birth and may the surrogate hold the child after birth.

⁶⁰² Sec 297(1)(c). Mills 2010 *Stell LR* 436.

⁶⁰³ Sec 297(1)(d). Nicholson, S and Nicholson, C "I used to have two parents and now I have three? When science (fiction) and the law meet: Unexpected complications" 2016 *Medicine and Law* 35 433. Oluwaseyi and Oladimeji 2021 *Obiter* 31.

⁶⁰⁴ Sec 297(1)(e). Sloth-Nielsen "Surrogacy in South Africa" 191.

⁶⁰⁵ Sec 297(1)(f).

⁶⁰⁶ Sloth-Nielsen "Surrogacy in South Africa" 192. Florescu, S and Sloth-Nielsen, J "Visions on surrogacy – From North to South: The approach of the Netherlands and South Africa to the issue of surrogacy and the child's right to know his origin" 2017 *International Survey of Family Law* 2017 248.

termination of a surrogate agreement.⁶⁰⁷ The full effect of the surrogate agreement depends on whether the parties have intended a full or a partial surrogacy agreement.⁶⁰⁸

Section 297 in essence provides that a valid surrogate agreement will automatically confer full parental responsibilities on the commissioning parent(s) from the moment of birth.⁶⁰⁹ Contrary to the common-law rule, identifying the mother as the woman who gave birth to the child, encapsulated in the maxim *mater sempter certa est*, the surrogate and her spouse or partner do not obtain any rights of parenthood and are thus obliged to hand over the child as soon as it is reasonably possible after birth.⁶¹⁰ This provision is aligned with the SALC's recommendation that a child born as a result of a legal surrogate agreement should, from the moment of birth, for all purposes have the legal status of a legitimate child of the commissioning parent(s). Similarly, the commissioning parent(s) should have the legal status of the lawful parent(s) of the said child.⁶¹¹

Regarding section 297(e), which determines that the effect of a valid surrogate agreement is that the agreement may not be terminated after the surrogate was artificially fertilised, Louw argues that apart from a partial surrogate who may terminate the agreement in terms of section 298 and the right of any surrogate to terminate the agreement by terminating her pregnancy in terms of the Choice on Termination of

⁶⁰⁷ Davel and Skelton *Commentary* 25. Sections 298 and 299 determines that a partial surrogate mother may withdraw from a fully enforceable surrogate agreement. The surrogate who is not genetically linked to the child (full surrogacy) does not have this option. Slabbert and Roodt "South Africa" 329. Thus, if the surrogate mother is genetically related to the child, she has the right to cancel the agreement and keep the child but if the surrogate mother is only carrying the child for the commissioning parents, she does not have the same right.

⁶⁰⁸ Sec 298(1) and sec 299 of the Children's Act. Davel and Skelton *Commentary* 25. Louw 2013 *THRHR* 571. The writer states that it is fallacious to assume that a partial surrogate who is genetically linked to the child may find it more difficult to give up her baby compared to a full surrogate who is not so related. She argues that the Children's Act differentiates between these types of surrogate mothers based on their biological connection or absence thereof. She further argues that biology alone cannot explain why parental rights upon termination of the agreement should vest in the surrogate mother and her husband or partner, if she has any, but if she is single, she shares parental rights with the commissioning father.

⁶⁰⁹ Sec 297(1)(a). LexisNexis *Family Law service* at par W12. The High Court order is declaratory in nature and it declares the commissioning parent(s) to be the parent(s) of the child prior to conception taking place. Thus, the commissioning parent(s) does no longer need to adopt the child born of the agreement.

⁶¹⁰ Sec 297(b). Nöthling Slabbert 2012 SAJBL 31.

⁶¹¹ SALC Project 65 at par 8.2.12. The surrogate should have no lawful claim to or legal obligation towards the commissioned child.

Pregnancy Act,⁶¹² the surrogate agreement may not be terminated *after* the surrogate has been artificially fertilised.⁶¹³ It is clear from the construction of the sections that the commissioning parent(s) do not have a right to terminate the agreement after the surrogate was artificially fertilised. The surrogate, on the other hand, has the right to terminate the agreement as will be discussed in section 298 below.⁶¹⁴

The effect of section 297(2) is that an invalid surrogate agreement will sever the rights of the commissioning parent(s), who will not be regarded as the parent(s) of the child, whereas the surrogate who gives birth to the child would become the legal parent.⁶¹⁵ The surrogate and her husband or partner need to understand the consequences should she refuse to hand over the commissioned child. Although the valid agreement gives full responsibilities and rights to the commissioning parent(s), the surrogate's actions after the birth of the child may change the consequences as set out in the agreement. The commissioning parent(s) will thus have no rights in respect of the commissioned child where the surrogate decides to keep the commissioned child after birth. The termination of a surrogacy agreement will next be considered.

3.9 Section 298: The dissolution of a surrogate agreement⁶¹⁶

Section 298 determines that a surrogate, who is also a genetic parent of the child, may terminate the surrogate agreement at any time *within a period of sixty days after* the birth of the child *and by the filing of a written notice with the court*.⁶¹⁷ The confirmation

⁶¹² 92 of 1996. Sec 300 of the Children's Act.

⁶¹³ Davel and Skelton *Commentary* 26. Sloth-Nielsen "Surrogacy in South Africa" 191.

⁶¹⁴ Nöthling Slabbert 2012 *SAJBL* 31.

⁶¹⁵ Sec 297(2).

⁶¹⁶ The Act. Bosman-Sadie and Corrie *Practical approach to the new Children's Act* 297. *AB and another v Minister of Social Development* at par 43.

⁶¹⁷ Sec 298(1). Nöthling Slabbert 2012 SAJBL 31. Slabbert and Roodt "South Africa" 329. SALRC Project 140 85. Nicholson, S and Nicholson, C "I used to have two parents and now I have three? When science (fiction) and the law meet: Unexpected complications" 2016 Medicine and Law 35 434. The writer states that it is suggested that this cooling off period would, in light of the implications of epigenetics (as discussed in ch 5 below), be available to all surrogates, irrespective of whether or not their gametes are used as all surrogate mothers are, as a result of the epigenome, genetically linked to the child. Florescu, S and Sloth-Nielsen, J "Visions on surrogacy – From North to South: The approach of the Netherlands and South Africa to the issue of surrogacy and the child's right to know his origin" 2017 International Survey of Family Law 2017 248. Oluwaseyi and Oladimeji 2021 Obiter 22. At 32 the authors explain that the rights of the commissioning parents to the child are suspended until the surrogate makes a decision either to renege or abide by the terms of the surrogacy agreement. The authors further explain that the distinction between full and partial surrogacy is made because of the surrogate mother's rights

of the termination of the agreement must be confirmed by the court and the court may issue any other appropriate order if it is in the best interest of the child, after notice to the other parties to the agreement, and after a hearing, upon the finding that the surrogate mother has willingly terminated the agreement and that she understands the consequences of the termination.⁶¹⁸ Should the surrogate exercise her rights of termination, she will not incur any liability to the commissioning parent(s), except for compensating them for any payments made by them in terms of section 301.⁶¹⁹

With the introduction of chapter 19 of the Children's Act, the legislator adopted the 'delayed direct-parentage' model in the case of a partial surrogacy.⁶²⁰ In terms of this model, the effects of the agreement are postponed until sixty days after the birth of the child, unless the surrogate decides not to surrender the child as agreed upon.⁶²¹ It can be explained that the acquisition of parental rights and responsibilities by the commissioning parent(s) is/are therefore suspended for a 'cooling-off' period within which time a surrogate, who is genetically related to the child, has the right to terminate the surrogacy agreement and keep the child.⁶²² A surrogate who is genetically related to the child at any time until sixty days after the birth of the child, whereas the commissioning parent(s) and a full surrogate may only terminate the agreement before the artificial fertilisation of the surrogate.⁶²³ Thus, it would appear that the full surrogate has no right to terminate the agreement as there is no 'cooling-off period' for her and the child will be the child of the

to dignity, privacy and autonomy are violated by being forced to give up the baby contrary to her wishes.

⁶¹⁸ Sec 298(2). Nöthling Slabbert 2012 SAJBL 31. Sloth-Nielsen "Surrogacy in South Africa" 191. Sloth-Nielson, J and van Heerden, B "The constitutional family developments in South African child and family law 2003-2013" 2014 International Journal of Law, Policy and the Family 28(1) 115. Heaton 2015 THRHR 33.

⁶¹⁹ Sec 298(3). Nöthling Slabbert 2012 *SAJBL* 31. Slabbert and Roodt "South Africa" 329. SALRC Project 140 85.

⁶²⁰ Report of the Ad Hoc Committee 8. Direct parentage is defined as a child or children born as a result of a surrogate agreement are considered the legitimate child or children of the commissioning parent or parents immediately at birth. Davel and Skelton *Commentary* 29. Louw 2013 *THRHR* 575.

⁶²¹ Sec 298(1).

⁶²² Nöthling Slabbert 2012 *SAJBL* 29. LexisNexis *Family Law service* at par W13. Clarke argues that the traditional surrogacy agreements should be discouraged because of the potential risks inherent to all the parties may prove to be problematic. Thus, she argues, traditional surrogacy agreements should only be resorted to as a last resort and only in exceptional circumstances due to the extensive availability of anonymous egg donors to infertile commissioning parent(s).

⁶²³ Davel and Skelton *Commentary* 30. Nöthling Slabbert 2012 *SAJBL* 31. The writers point out that the specific enforcement of a surrogate agreement against the will of the surrogate mother may also violate her rights to dignity, privacy and bodily autonomy, including the child's right to dignity.

commissioning parent(s).⁶²⁴ The surrogate must give notification of the termination through the filing of a written notice with the court.⁶²⁵ The procedure for the termination is sufficiently outlined in the Act and it is imperative that the parties understand the consequences of a termination, especially the effect thereof on the status of the commissioned child and all the parties concerned.

Section 298(3), which directs that the surrogate must compensate the commissioning parent(s) for any payments made by them in terms of section 301, can be seen as a form of protection of the commissioning parent(s) in that the surrogate must understand the financial consequences of a decision to terminate the agreement in terms of section 298(1). There are clearly consequences for all the parties to the agreement should the agreement be terminated. The effect of termination will next be examined.

3.10 Section 299: The consequence of termination of the surrogate agreement⁶²⁶

Two scenarios regarding the termination of the agreement arise, e.g termination after the child is born and before the child is born. Where the agreement is terminated after the child is born, the parental rights established in terms of section 297 are terminated and will vest in the surrogate, her husband or partner, if any, or if she has none, the commissioning father.⁶²⁷

Should the agreement be terminated before the child is born, the child will legally be the child of the surrogate, her husband or partner, if any, or if she has none, the commissioning father, from the moment of the child's birth.⁶²⁸ Where the agreement is terminated, the surrogate and her husband or partner, if any, or if she has none, the commissioning father, is obliged to accept the obligation of parenthood.⁶²⁹ Subject to sections 299(a) and 299(b), upon termination of the agreement either before or after the child is born, the commissioning parent(s) will have no rights of parenthood and

⁶²⁷ Sec 299(a). Sloth-Nielsen "Surrogacy in South Africa" 192.

 ⁶²⁴ Sloth-Nielson, J and van Heerden, B "The constitutional family developments in South African child and family law 2003-2013" 2014 *International Journal of Law, Policy and the Family* 28(1) 115. The authors mention that if needs be, the child will be the child of the commissioning parents under the compulsion of the court.

⁶²⁵ Sec 298(1).

⁶²⁶ The Act. Bosman-Sadie and Corrie *Practical approach to the new Children's Act* 298.

⁶²⁸ Sec 299(b). Sloth-Nielsen "Surrogacy in South Africa" 192.

⁶²⁹ Sec 299(c).

can only obtain such rights through adoption.⁶³⁰ Furthermore, section 299(e) provides that, also subject to sections 299(a) and 299(b), that the child will have no claim for maintenance or of succession against the commissioning parent(s) or any of their relatives in the event that the surrogate agreement was terminated.⁶³¹

The surrogate's right to terminate the pregnancy will be canvassed next.

3.11 Section 300: Termination of the pregnancy by the surrogate mother⁶³²

A surrogate agreement is terminated when the surrogate terminates her pregnancy in terms of the Choice on Termination of Pregnancy Act.⁶³³ The decision to terminate the pregnancy lies with the surrogate, but she must inform the commissioning parent(s) of her decision and consult with them before her pregnancy is terminated.⁶³⁴ By exercising her right, the surrogate incurs no liability to the commissioning parent(s) except for compensation for any payments made by the commissioning parent(s) in terms of section 301 where the decision to terminate is for any other reason than on medical grounds.⁶³⁵

Section 300 permits a surrogate to terminate her pregnancy in terms of the Choice of Termination of Pregnancy Act, even after an undertaking in terms of the surrogate agreement, to carry and give birth to the child on behalf of the commissioning parent(s).⁶³⁶ Louw argues that a right to terminate the pregnancy necessarily includes the right to refuse to terminate the pregnancy and that is even at the risk of giving birth

⁶³⁰ Sec 299(d). Sloth-Nielsen "Surrogacy in South Africa" 192.

⁶³¹ Sec 299(e).

⁶³² The Act. Bosman-Sadie and Corrie Practical approach to the new Children's Act 298.

⁶³³ 92 of 1996. Sec 300(1). SALC Project 65 at par 8.3.4. The SALC submitted that provisions of the Abortion and Sterilisation Act 2 of 1975 should be incorporated in surrogacy legislation. Florescu, S and Sloth-Nielsen, J "Visions on surrogacy – From North to South: The approach of the Netherlands and South Africa to the issue of surrogacy and the child's right to know his origin" 2017 International Survey of Family Law 2017 248. Heaton 2015 THRHR 33.

⁶³⁴ Sec 300(2). LexisNexis *Family Law service* at par W14.

⁶³⁵ Sec 300(3).

⁶³⁶ Sec 300(2). Report of the Ad Hoc Committee 32. The Committee submitted that the surrogate's rights should not be considered in isolation. Her rights should be weighed up against the rights of the commissioning parent(s). At 75 the Committee recommended that the issue of an abortion should be handled by the parties in their terms of contract as it is a complex issue. This, however, should be done within the confines of the Constitution and the Choice on Termination of Pregnancy Act 92 of 1996.

to a child with severe physical or mental defects.⁶³⁷ One may argue that it would be in the best interests of all the parties concerned, especially the commissioned child, to make provision for circumstances where the child might be born with severe physical or mental defects, if detected during the surrogate's pregnancy.⁶³⁸ The Ad Hoc committee was adamant that there should be an agreement by the commissioning parent(s) that the commissioned child would be accepted in spite of any mental or physical handicap.⁶³⁹ The SALC recommended that the commissioning parent(s) should be consulted in the event that a medical risk regarding the foetus is detected, as they may wish to continue to have the child, despite the risk of disability or a handicap.⁶⁴⁰ This recommendation by the SALC is rational, as the commissioning parent(s) will at least have a chance to make an informed decision instead of agreeing to accept a child with mental or physical handicaps without having the opportunity to make an informed decision.

Section 300(3) also protects the commissioning parent(s) in the event that the surrogate terminates the pregnancy for a reason other than medical grounds, for example, social reasons.⁶⁴¹ In circumstances where the pregnancy is potentially life threatening to the surrogate or the foetus, the surrogate will not need to compensate the commissioning parent(s) for any payments made by them in respect of the agreement.⁶⁴²

Parties should be advised to agree in advance on what should happen when a termination of pregnancy is necessary. This will create some degree of certainty in preventing a termination without a medical basis. The agreement may not require a surrogate mother not to exercise her fundamental right to decide whether or not to continue with the pregnancy. In all instances the final decision on termination still rests with the surrogate.⁶⁴³

⁶³⁷ Davel and Skelton *Commentary* 33.

⁶³⁸ Report of the Ad Hoc Committee 47. The Committee submitted that provision should be made for the possibility that the commissioned child being born abnormal as no person can be guaranteed a healthy child.

⁶³⁹ Report of the Ad Hoc Committee 47.

⁶⁴⁰ SALC Project 65 at par 8.3.4.

⁶⁴¹ LexisNexis *Family Law service* at par W14.

LexisNexis *Family Law service* at par W14.

⁶⁴³ LexisNexis *Family Law service* at par W14. Clarke recommends that the psychological assessment on the parties should fully deal with this aspect and the different scenarios discussed

The form of payments that are allowed to be made in terms of a surrogacy agreement will be discussed in the next section, as the Act permits only a limited number of expenses.

3.12 Section 301: Prohibition of payments in respect of surrogacy⁶⁴⁴

Section 301 expressly states that no person may give or promise to give to any person, or receive from any person a reward or compensation in cash or in any kind in respect of a surrogate agreement, subject to subsections (2) and (3).⁶⁴⁵ The only promise or agreement for the payment of compensation to a surrogate, or any other person in connection with a surrogate agreement or the execution of such an agreement which is enforceable, is for the compensation for expenses that relate directly to the artificial fertilisation and pregnancy of the surrogate, the birth of the child; the confirmation of the surrogate agreement;⁶⁴⁶ loss of earnings suffered by the surrogate as a result of the surrogate agreement;⁶⁴⁷ or for the insurance to cover the surrogate for anything that may lead to death or disability as a result of the pregnancy.⁶⁴⁸ Reasonable compensation may be paid to a person who renders a *bona fide* professional legal or medical service with a view to the confirmation of a surrogate agreement in terms of section 295 or in the execution of such an agreement.⁶⁴⁹

in which a termination would or would not be undertaken by the surrogate. SALC Project 65 at par 8.3.4. The final choice of termination of pregnancy should remain with the woman who is carrying the child.

⁶⁴⁴ The Act. In re confirmation of three surrogate agreements at par 15 and 22. Ex parte HPP and others; Ex parte DME and others par 27 and 28. Ex parte WH and others at par 40 and 65. Bosman-Sadie and Corrie Practical approach to the new Children's Act 299. AB and Another v Minister of Social Development at par 44.

⁶⁴⁵ Sec 301(1). SALC Project 65 par 8.4.2. Report of the Ad Hoc Committee 61 and 74. It was submitted that a criminal ban on compensation of the surrogate would be impractical. SALRC Project 140 85.

⁶⁴⁶ Sec 301(2)(a). Report of the Ad Hoc Committee 25 and 62. SALRC Project 140 86.

⁶⁴⁷ Sec 301(2)(b). SALRC Project 140 86.

⁶⁴⁸ Sec 301(2)(c). Report of the Ad Hoc Committee 49. The commissioning parent(s) should also be insured accordingly. SALRC Project 140 86.

⁶⁴⁹ Sec 301(3). Nöthling Slabbert 2012 *SAJBL* 32. The writer explain that medical or legal professionals providing a *bona fide* professional service in respect of the surrogate agreement, for example, drawing op of the contract, and in its execution, for example, the artificial fertilisation of the surrogate mother, are entitled to reasonable compensation for their different services. SALRC Project 140 86. Bonthuys and Broeders 2013 *SALJ* 490.

Section 301 provides for limited exceptions to the above instances, underscoring the principle that commercial surrogacy is unlawful.⁶⁵⁰ Should a person contravene this provision, he or she is committing a criminal offence and if found guilty, will be liable to pay a fine or to imprisonment for a period not exceeding ten years, or both a fine and imprisonment.⁶⁵¹

Furthermore, any agreement or a promise between the parties to pay compensation in connection with a surrogate agreement is unenforceable.⁶⁵² Clarke sees the purpose of this section to prevent the abuse of persons charging a fee in relation to the surrogacy process, who are not professional persons required for the successful outcome of the surrogacy process.⁶⁵³ Should facilitation agreements be valid, it would open the floodgates and lead to commercial surrogacy and the abuse of vulnerable persons, which is exactly what section 301 was intended to prevent.⁶⁵⁴

A surrogacy facilitator's services are also covered by the prohibition in section 301 as these services are not directly linked to the allowed lawful expenses relating to surrogacy agreements.⁶⁵⁵ Louw argues that the pre-authorisation requirement in terms of the Act makes the surrogacy model adopted in South Africa far more inflexible,

Ex parte HPP and others; Ex parte DME and others par.27. Nöthling Slabbert 2012 SAJBL 32. Ex Parte KAF and others 2019 (2) SA 510 (GJ) at par 7. The court stated that it is to prevent the potential exploitation of commissioning parents from likely financial damage that could result on the one hand, as well as the potential exploitation and commodification of would be surrogates as well as the rights of the child to be born. It is also likely that some surrogate mothers may be desperate enough to enter into these contracts for the limited financial benefit that they may receive. Bonthuys and Broeders 2013 SALJ 490. Louw 2013 THRHR 580. The writer argues that South Africa prohibits commercial surrogacy for fear that it would lead to the commodification of babies because it could amount to baby selling if the surrogate stands to gain financially from the agreement. SALC Project 65 at par 8.2.8. The SALC submitted that paying a surrogate is seen as trading in children. It would be fair though for the surrogate to be reimbursed for actual expenses incurred in respect of the pregnancy.

⁶⁵¹ Sec 305(1)(b). SALRC Project 140 86. Florescu, S and Sloth-Nielsen, J "Visions on surrogacy – From North to South: The approach of the Netherlands and South Africa to the issue of surrogacy and the child's right to know his origin" 2017 *International Survey of Family Law* 2017 247.

⁶⁵² Sec 301(2). LexisNexis *Family Law service* at par E207.

⁶⁵³ LexisNexis *Family Law service* at par W8. Thus, no agency, surrogate coordinator/facilitator or person may charge an introductory fee for the purpose of introducing a surrogate to potential commissioning parent(s). Agencies and persons are also not allowed to charge a fee which is disguised as an introductory fee.

⁶⁵⁴ LexisNexis *Family Law service* at par W8. Report of the Ad Hoc Committee 26. Agencies and brokers that provide services in respect of surrogacy agreements should be outlawed to prevent any abuse of these arrangements.

⁶⁵⁵ LexisNexis *Family Law service* at par W8.

especially if compared to the broader and more flexible approach in the UK.⁶⁵⁶ She furthermore submits that the court does not have a discretion to authorise payments other than those listed in section 301 and if the court cannot confirm the agreement, the agreement will be invalid and unenforceable.⁶⁵⁷

3.13 Section 302: The identity of the parties to a surrogate agreement⁶⁵⁸

It is only with the written consent of the parties to court proceedings relating to a surrogate agreement, that their identities may be published.⁶⁵⁹ No person may publish any facts that reveal the identity of a commissioned child.⁶⁶⁰

The purpose of this section is the protection of the identities of the parties to the agreement, as well as the commissioned child. Should a person fail to comply with the above prohibition, such person will be committing a punishable offence and he or she would be liable to a fine or to imprisonment or to both a fine and such imprisonment.⁶⁶¹ The Ad Hoc Committee also emphasised that strict confidentiality should be maintained, as the practice of surrogacy should not be promoted.⁶⁶² It was also submitted that sensational publicity must be discouraged at all costs as it is important to protect the dignity of the child.⁶⁶³ The section also protects the identity of the surrogate mother to be disclosed to the commissioned child.⁶⁶⁴

Clarke, on the contrary, argues that there is no practical basis for prohibiting the disclosure of the identity in a gestational surrogacy agreement as the surrogate is only the carrier of the child.⁶⁶⁵ She regards section 297(1) of the Act as contradicting section

⁶⁵⁶ Louw 2013 *THRHR* 583. The writer explains that in the UK, the approach would allow intending parents to be granted legal parentage even where reasonable expenses have been exceeded.

⁶⁵⁷ Louw 2013 *THRHR* 583 - 584. At 585 the writer argues that payments should be regulated in South Africa in terms of a set minimum and maximum fee, alternatively, the approval of payments not expressly sanctioned by the act, could be placed in the discretion of the court.

⁶⁵⁸ The Act. Bosman-Sadie and Corrie Practical approach to the new Children's Act 299. Ex Parte SA and others (case no 82202/2014) at par 17.

⁶⁵⁹ Sec 302(1). Report of the Ad Hoc Committee 74.

⁶⁶⁰ Sec 302(2). SALC Project 65 at par 8.3.5. It is in the interests of the child that no facts that could reveal his or her identity should be published.

⁶⁶¹ Section 305(1)(*b*) read with s 305(6). Imprisonment is for a period not exceeding ten years. LexisNexis *Family Law service* at par E208 and par E209. Nöthling Slabbert 2012 *SAJBL* 28.

⁶⁶² Report of the Ad Hoc Committee 48.

⁶⁶³ Report of the Ad Hoc Committee 48.

⁶⁶⁴ LexisNexis *Family Law service* at par W15.

⁶⁶⁵ LexisNexis *Family Law service* at par W15.

41 of the Act, as section 41 determines that the identity of the surrogate may not be disclosed to the child born, but section 297(1) determines that the surrogate or her husband may have contact with the child by agreement between the parties.⁶⁶⁶ Thus, should the agreement stipulate that the surrogate may have contact with the commissioned child, the identity of the surrogate will need to be known to the child.⁶⁶⁷ It was pointed out by the Ad Hoc Committee that notions regarding anonymity may not have a place in surrogacy arrangements where mutual support between the parties involved becomes essential, so that the surrogate may develop trust in the commissioning parent(s) to enable her to hand over the commissioned child after birth.⁶⁶⁸

3.14 Section 303: Certain acts are prohibited⁶⁶⁹

A person may only artificially fertilise a woman in the execution of a surrogate agreement or render assistance in such artificial fertilisation, if such person is authorised by a court in terms of the provisions of this Act.⁶⁷⁰ Moreover, no person may advertise the services of a surrogate with a view to compensation.⁶⁷¹

This section prohibits any person from artificially fertilising a proposed surrogate in the execution of a surrogate agreement or rendering assistance in such artificial fertilisation, unless the process was first authorised by a court.⁶⁷² The person who

⁶⁶⁶ LexisNexis *Family Law service* at par W15.

⁶⁶⁷ LexisNexis *Family Law service* at par W15.

⁶⁶⁸ Report of the Ad Hoc Committee 49.

⁶⁶⁹ The Act. In re confirmation of three surrogate agreements at par 15 Ex parte HPP and others; Ex parte DME and others par 30. Ex parte MS and others; In re: Confirmation of surrogate mother agreement at par 21. Bosman-Sadie and Corrie Practical approach to the new Children's Act 299. AB and another v Minister of Social Development at par 44.

⁶⁷⁰ Sec 303(1). SALC Project 65 at par 8.3.6. Illegal assistance to establish a surrogate pregnancy is punishable. The SALC was not in favour of criminalising any aspect of surrogacy but realised that it was necessary to serve as a deterrent. Nöthling Slabbert 2012 *SAJBL* 28.

⁶⁷¹ Sec 303(2). SALC Project 65 at par 8.3.8. It was submitted that a ban on advertising may also contribute towards the restriction of commercialism. Report of the Ad Hoc Committee 28 and 74. Nöthling Slabbert 2012 SAJBL 33. It is pointed out that it is thus illegal to broker surrogacy arrangements on a commercial basis. The writers explain that a woman is free to offer her services to enter into a surrogate agreement that complies with the provisions of the Act and in terms of which she will only be compensated for reasonable expenses provided for in the Act. It is common in South Africa to advertise for egg donors and they offer to pay between R 5,000.00 and R 6,000.00 per donation.

⁶⁷² Report of the Ad Hoc Committee 74. Nicholson and Bauling 2013 *De Jure* 523. Sloth-Nielsen "Surrogacy in South Africa" 191.

contravenes the provisions of this section is guilty of an offence and he or she will be liable to a fine or imprisonment or both a fine and imprisonment.⁶⁷³ The SALC believed illegal surrogacy could be restricted in as far as medical practitioners will be discouraged from becoming involved in illegal surrogacy.⁶⁷⁴

As some of the provisions discussed in this chapter were the focus of litigation in recent times, it is necessary to turn to these cases next, and also to determine if the judgments provide clarity and guidance in respect of the legal gaps identified in these provisions above.

3.15 Judgments with a focus on chapter 19 of the Children's Act

At the time of writing of this thesis, no regulations have yet been promulgated in terms of the Children's Act relating to surrogacy.⁶⁷⁵ The discussion below will turn to recent case law in respect of chapter 19 of the Children's Act.

3.15.1 In re confirmation of three surrogate agreements⁶⁷⁶

Three applications were brought in terms of section 292 of the Children's Act⁶⁷⁷ for the confirmation of surrogate motherhood agreements.⁶⁷⁸ The first application was brought by a 43-year old single male, with the surrogate married (to the third applicant, a female).⁶⁷⁹ The second application was brought by a married couple (heterosexual), with the surrogate a divorced female.⁶⁸⁰ The third application was brought by a married couple (heterosexual), with the surrogate married to a male.⁶⁸¹ The applicants all

⁶⁷³ Sec 305(1)(b) and sec 305(6). Imprisonment for a period of not more than ten years.

⁶⁷⁴ SALC Project 65 at par 8.3.6.

⁶⁷⁵ Nöthling Slabbert 2012 *SAJBL* 29. The Deputy Judge President of the South Gauteng High Court has issued a practice directive dealing with surrogate agreement applications in view of the shortcoming of the Act.

⁶⁷⁶ 2011 (6) SĂ 22 (GSJ). This was the first unreported judgment from the South Gauteng High Court. Slabbert and Roodt "South Africa" 333.

⁶⁷⁷ 38 of 2005. Hereinafter "the Act".

⁶⁷⁸ In re confirmation of three surrogate agreements at par 1.

⁶⁷⁹ In re confirmation of three surrogate agreements at par 14.

⁶⁸⁰ In re confirmation of three surrogate agreements at par 14.

⁶⁸¹ In re confirmation of three surrogate agreements at par 14.

qualified as commissioning parents and surrogate mothers pursuant to the provisions of the Act.⁶⁸²

Wepener J and Victor J reiterate that applications for the confirmation of surrogacy agreements have significant implications for all the applicants concerned, not to mention those for the children to be born.⁶⁸³ The court also alludes to section 28 of the Constitution of South Africa, 1996, which provides an omnibus of the rights afforded to children in terms of the Constitution, as well as the relevance of article 20 of the African Charter on the Rights and Welfare of the Child, to which South Africa is a signatory.⁶⁸⁴ In view of the court's duty to regard children's' interests as paramount,⁶⁸⁵ judges are duty bound to ensure that the interests of the child, once born, are best served by the contents of the agreement, which the court is requested to confirm.⁶⁸⁶ As the upper guardian of minors, the court may require detailed information regarding: (a) who the commissioning parents are; (b) what their financial position is; (c) what support systems, if any, they have in place; (d) what their living conditions are; and (e) how the child will be taken care of.⁶⁸⁷

In the above application, the court refers to the good practice which is found in adoptions where expert assessment reports from social workers are required, and in practice, a police clearance is obtained to demonstrate the suitability of the adoptive parents.⁶⁸⁸ The court suggests that these could be applied to the commissioning parents with good results. An expert report could also address the suitability of the surrogate mother.⁶⁸⁹ Applicants must supply proper and full details regarding

⁶⁸² In re confirmation of three surrogate agreements at par 14.

⁶⁸³ In re confirmation of three surrogate agreements at par 5.

⁶⁸⁴ In re confirmation of three surrogate agreements at par 16.

⁶⁸⁵ In re confirmation of three surrogate agreements at par 16 Nöthling Slabbert 2012 SAJBL 29. The writers pointed out that the judgment emphasises that the court, in confirming the agreement, is not merely 'rubber stamping' the agreement. The court, in considering all the facts on which the application is based, will regard the interests of the child to be born of paramount importance. Slabbert and Roodt "South Africa" 333.

⁶⁸⁶ In re confirmation of three surrogate agreements at par 16. Louw 2013 THRHR 569. The writer argues that the impact of applying the best interest standard to a resultant child, rather than to an existing child when confirming a surrogate agreement, was never considered by the court in either this judgment or *Ex parte WH and others* 2011 (6) SA 514 GNP). She further argues that while the best interests of the child is expressly dealt with under a separate heading in the *Ex parte WH and others* judgment, the court does not seem to appreciate that the best interests of an existing child cannot be determined in the same way as the best interests of a resultant child.

⁶⁸⁷ In re confirmation of three surrogate agreements at par 17.

⁶⁸⁸ In re confirmation of three surrogate agreements at par 17.

⁶⁸⁹ In re confirmation of three surrogate agreements at par 17.

themselves in matters where the interests of children are paramount, otherwise the court would not be able to determine whether the commissioning parents are indeed fit and proper to be entrusted with full parental responsibilities.⁶⁹⁰

The court concludes that for it to perform its duty pursuant to the Act and the Constitution, complete and full compliance with all the provisions of the Act is required, as well as compliance with the requirements raised in the judgment.⁶⁹¹ This judgment is helpful in that the court provides some guidance as to what information is needed to decide in respect of a confirmation of a surrogate agreement.

3.15.2 Ex parte WH and others⁶⁹²

The facts of the above case are briefly the following: The first and second applicants, the commissioning parents, are two males married to one another.⁶⁹³ They brought an application for the confirmation of a surrogate agreement.⁶⁹⁴ They are Dutch and Danish citizens respectively, both domiciled in South Africa with the intention to stay in South Africa permanently.⁶⁹⁵ The couple was introduced to the potential surrogate mother by an agency known as Baby-2-Mom.⁶⁹⁶ The parties compiled a list of estimated costs regarding payments related to the surrogacy agreement and the execution thereof.⁶⁹⁷ No details, however, were provided regarding the specifics in respect of the costs.⁶⁹⁸ The court concludes that a detailed list of surrogacy expenses with sufficient specificity should be provided to minimise the possibility of abuse.⁶⁹⁹ The court furthermore considers the application with the view to establish guidelines on how similar applications brought in the future should be dealt with.⁷⁰⁰ The Bar, Law

⁶⁹⁰ In re confirmation of three surrogate agreements at par 24.

⁶⁹¹ In re confirmation of three surrogate agreements at par 26. Pillay, R and Zaal, FN "Surrogate motherhood confirmation hearings: The advent of a fundamentally flawed process" 2013 South African Law Journal 130(3) 478.

⁶⁹² 2011 (4) All SA 630 (GNP).

⁶⁹³ *Ex parte WH and others* at par 15. Jordaan DW "Surrogate Motherhood in illness that does not cause infertility" 2016 *SAMJ* 685. Pillay and Zaal 2013 *SALJ* 478.

⁶⁹⁴ Ex parte WH and others at par 14. Louw 2013 THRHR 566.

⁶⁹⁵ Ex parte WH and others at par 15.

⁶⁹⁶ Ex parte WH and others at par 18 Pillay and Zaal 2013 SALJ 478. Louw 2013 THRHR 566.

⁶⁹⁷ Ex parte WH and others at par 28.

⁶⁹⁸ *Ex parte WH and others* at par 29.

Ex parte WH and others at par 29 Nöthling Slabbert 2012 SAJBL 29. Pillay and Zaal 2013 SALJ 479.

⁷⁰⁰ Ex parte WH and others at par 9.

Society and Centre for Child Law were invited to make submissions regarding the following aspects: (a) the approach that should be followed in the case where the reproductive genetic material used is not that of the parties; (b) the approach, if any, that should be followed when same-sex couples apply for a surrogacy agreement confirmation; (c) the appropriate steps that should be followed and factors to be considered to determine the best interests of the child.⁷⁰¹

The court states that it should in all instances where an agency is involved in the surrogacy arrangement, be fully appraised of: (a) all the facts and circumstances relating to the *modus operandi* of the said agency, (b) the relationship between the agency and the commissioning parents, as well as (c) between the agency and the surrogate.⁷⁰² Tolmay and Kollapen JJ allude to pertinent constitutional and legal issues which may arise out of surrogacy applications, namely those following from (a) surrogacy and same sex relationships; (b) the best interests of the child; (c) the surrogate and the risk of commercial surrogacy; and (d) a suitable parent.⁷⁰³

The court emphasises that same-sex couples should be treated in the same way as heterosexual couples as far as their appropriateness to act as commissioning parents were concerned.⁷⁰⁴ The court also points out that the mothering of a child is a function that very often does not have anything to do with the gender of a parent.⁷⁰⁵ It further states that many children grow up without a father or a mother and courts should safeguard that it does not try and create a utopia for commissioned children that is far removed from the social reality of society.⁷⁰⁶

⁷⁰¹ Ex parte WH and others at par 10.

Ex parte WH and others at par 30. Nöthling Slabbert 2012 SAJBL 29. Pillay and Zaal 2013 SALJ 480.

⁷⁰³ *Ex parte WH* at par 54 to par 70.

⁷⁰⁴ Revision Service (RS) 4, 2012 ch 19 – p15 *Background to Children's Act* 38 of 2005.

⁷⁰⁵ Ex parte WH at par 54.

Ex parte WH at par 54. At par 55 the court referred to sec 292(1)(c) and stated that it is evident that the legislature had contemplated that a single person may also be a commissioning parent and this appears to be in line with the prohibition of non-discrimination found in sec 9 of the Constitution. Bonthuys and Broeders 2013 SALJ 488. The writers argue that the court can be found lacking in the application of substantive equality to the facts of the case. The court seems to require formal equality between same-sex and opposite-sex commissioning parents in that they should be treated in exactly the same manner. The court warned that different tests should not apply to same-sex and opposite-sex commissioning parents. But, on the other hand, the realisation that requiring maternal influences from a gay couple would affectively preclude them from becoming surrogate parents indicates an awareness of the different contexts that is applied to gay couples who wish to become parents. This, the writers argue, would require substantive,

Regarding the best interests of the child, the court refers to section 28(2) of the Constitution and section 7 of the Children's Act, both underscoring the importance of a child's best interests in every matter concerning a child.⁷⁰⁷ The court briefly considers the legal situation relating to surrogate motherhood, prior to the promulgation of the Children's Act, which provided that commissioning parent(s) could only become the legal parent(s) of the child through adoption.⁷⁰⁸ Unfortunately, under this dispensation, in instances where the surrogate changed her mind and refused to consent to the adoption of the child, she could do so irrespective of the genetic origin of the child.⁷⁰⁹ This was undoubtedly an untenable situation, which signalled uncertainty regarding the parents of the child, which could impact negatively on the child.⁷¹⁰ Such a scenario was clearly not in the best interests of the child.

In casu, the court refers to section 297(1)(b) of the Act which requires the surrogate mother to hand over the child as soon as it is reasonably possible after birth.⁷¹¹ The court emphasises that the surrogate, her partner and relatives do not have any right to parenthood or care.⁷¹² Furthermore, reference is made to section 298(2) in terms of which the court must terminate the confirmation of the agreement (after notice to the parties and a hearing) upon finding that the surrogate has voluntarily terminated the agreement and that she understands the effect of the termination.⁷¹³ A court may grant any other appropriate order if it is in the child's best interests.⁷¹⁴

rather than formal, equality between same-sex and opposite-sex commissioning couples. The court should question and transcend heteronormative legal and social assumptions.

Ex parte WH and others at par 56.

Ex parte WH and others at par 57.

Ex parte WH and others at par 57.

Ex parte WH and others at par 57.

⁷¹¹ *Ex parte WH and others* at par 58.

⁷¹² *Ex parte WH and others* at par 58.

⁷¹³ Ex parte WH and others par 60.

⁷¹⁴ Ex parte WH and others at par 60. It is stated that the court will be in a position to ensure that the best interests of the child is protected on the termination of the surrogate agreement. At par 63 the court stated importantly: "Thus when a court considers the question of the best interests of the child care should be taken that the rights of the commissioning parents in terms of the Bill of Rights and the Promotion of Equality and Prevention of Unfair Discrimination Act 4 of 2000 are not violated by unnecessary invasion of the privacy of commissioning parents or by setting the bar too high for parents whose only option is to have a child by way of surrogacy. This will entail a value judgment by the court taking into consideration the circumstances of the particular case." Pillay and Zaal 2013 SALJ 480. Louw 2013 THRHR 568.

Regarding the issue of a surrogate and the risk of commercial surrogacy, the court cites section 301 of the Act that prohibits payment to any person other than those specifically set out in the Act, which would include any facilitation fee to any person who introduced the surrogate to the commissioning parent(s) or any compensation of any nature other than those that the Act provides for.⁷¹⁵ The court advises that the affidavit should state that no such fee was paid to anyone.⁷¹⁶

Regarding section 295(b)(ii), in discussing the issue of a suitable parent, ⁷¹⁷ the court observes that courts should knowingly guard against personal perceptions influencing any decision on the suitability of a person to either accept parenthood or to act as a surrogate during the exercise of their discretion.⁷¹⁸ The court should, however, have regard to the personal and character details of a commissioning parent for instance, details of previous criminal convictions, particularly those relating to violent crimes or crimes of a sexual nature.⁷¹⁹ Importantly, when a court needs to decide on the suitability of a parent, an objective test should be applied (bearing in mind the

⁷¹⁵ *Ex parte WH and others* at par 65. Nicholson and Bauling 2013 *De Jure* 527.

⁷¹⁶ Ex parte WH and others at par 65. The court further stated at paras 66 and 67 what information is required to be set out in the affidavit. In the event that an agency was involved, the full particulars regarding the agency should be set out and an affidavit by the agency should also be filed with the following information: "(a) the business of the agency;(b) whether any form of payment is paid to or by the agency in regard of any aspect of the surrogacy: (c) what exactly the agency's involvement was regarding the (i) introduction of the surrogate mother, (ii) how the information regarding the surrogate mother was obtained by the agency, and (d) whether the surrogate mother received any compensation at all from the agency or the commissioning parents." Further, the court stated that: "Full particulars should be set out in the founding affidavit on how the commissioning parents came to know the surrogate mother and why she is willing to act as a surrogate to them. The surrogate mother's background as well as her financial position should be investigated and set out in the affidavit. Furthermore, a comprehensive report by a psychologist is essential to assess the suitability of the surrogate mother. This should deal in particular with her background, psychological profile and the effect that the surrogacy and the giving up of the baby will have on her. Full medical reports should also be obtained regarding her physical condition to indicate whether surrogacy poses any dangers for her and/or the child. In our view, the medical report should deal with the HIIV status of the mother, as well as any disease that could be transferred from her to the child in order to protect the child and to allow the court to exercise its discretion properly in confirming the agreement." At par 68 the court further stated that in their view, the application should also state where the gametes will come from without the parties revealing the identity of the donor. Nöthling Slabbert 2012 SAJBL 28. Nicholson and Bauling 2013 De Jure 527. Slabbert and Roodt "South Africa" 334 - 335. The writers pointed out that a detailed and specific list of surrogacy expenses should be provided to minimise the possibility of abuse. Bonthuys and Broeders 2013 SALJ 490-491. Sloth-Nielsen "Surrogacy in South Africa" 193.

⁷¹⁷ Ex parte WH and others at par 69. Pillay and Zaal 2013 SALJ 480.

⁷¹⁸ Ex parte WH and others at par 69.

Ex parte WH and others at par 69. The court pointed out that this information should be disclosed and fully set out as well as the circumstances surrounding them. Nöthling Slabbert 2012 SAJBL 30.

constitutional principles referred to) which would include an enquiry into the ability of the parents to care for the child both emotionally and financially and to provide an environment for the harmonious growth and development of the child.⁷²⁰

The vital role that the court plays in the confirmation of the agreement, is emphasised in this judgment.⁷²¹ On the one hand, the court is enjoined to advance the spirit and objectives of the Act without creating or placing additional obstacles in the path of the litigants who seek relief, but on the other hand, as the upper guardian of all minor children, it cannot simply be a rubber stamp validating the private arrangements between contracting parties.⁷²² The court must ensure that both the formal and substantive requirements of the Act are complied with and as it is dependent on the information placed before it by the applicants, the utmost good faith is expected and required from them in an application.⁷²³

Ex parte WH and others at par 70. Nicholson and Bauling 2013 De Jure 529.

⁷²¹ Ex parte WH and others at par 72.

⁷²² Ex parte WH and others at par 72. Nicholson and Bauling 2013 De Jure 520. Pillay and Zaal 2013 SALJ 481. At 484 the writers state that the resultant disparity between what courts is supposed to be doing and what the courts are actually able to do, does much to explain the dissonance between what the court directed and what it did in the case of WH. The unfortunate consequence is a real danger of inappropriate confirmations and it must be concluded that the chapter 19 confirmation process is fundamentally flawed. At 485 the writers recommended that a patient review panel should be appointed and this panel should prepare a psychological and medical report, together with a recommendation in respect of the confirmation of the surrogacy agreement to court. Louw 2013 THRHR 567.

⁷²³ Ex parte WH and others at par 73. At par 76 the court stated that the court hearing such an application and in the exercise of its judicial discretion may request any additional information from the parties or any other institution to assist the court in the determination of the said application. At par 77 the court set out the following information which the affidavit should contain: "77.1 all factors set out in the Act together with documentary proof where applicable. The affidavit should also contain the information referred to in paragraphs [67] (Full particulars should be set out in the founding affidavit on how the commissioning parents came to know the surrogate mother and why she is willing to act as a surrogate to them. The surrogate mother's background as well as her financial position should be investigated and set out in the affidavit. Furthermore, a comprehensive report by a psychologist is essential to assess the suitability of the surrogate mother. This should deal in particular with her background, psychological profile and the effect that the surrogacy and the giving up of the baby will have on her. Full medical reports should also be obtained regarding her physical condition to indicate whether surrogacy poses any dangers for her and/or the child. In our view, the medical report should deal with the HIV status of the mother, as well as any disease that could be transferred from her to the child in order to protect the child and to allow the court to exercise its discretion properly in confirming the agreement.) and [74] (In satisfying itself that the peremptory requirements of the Act have been met, the court must be placed in possession of sufficient information to support any of the conclusions that the applicants contend for. Where an applicant seeks to draw certain conclusions with regard to matters which may include the financial, emotional or general suitability as a parent, there should be facts to support such conclusions that a court can interrogate. Ultimately, the court must be satisfied that the conclusions arrived at are supported by the facts. Accordingly, vague and generic allegations in this regard that fall short of supporting a conclusion may well render an application defective.) hereof; 77.2 whether there have been any previous applications for surrogacy, the division in which the application was brought,

This judgment emphasises the importance of complying with the provisions of the Act, and also the importance of providing an as detailed as possible list of surrogacy expenses to curb possible abuse and the possibility of commercial surrogacy. The court's approach and insistence on specific information, especially where an agency is involved in the surrogacy process, the manner in which the agency operates and the relationship between the parties and the agency, is strongly commended. Also significant is the court's guidance that same-sex couples should be treated the same as heterosexual couples in respect of the appropriateness to act as commissioning parents.

The prohibitions in section 301 in respect of payments allowed in a surrogacy agreement include the payment of a facilitation fee to a person introducing a surrogate and commissioning parent(s). The affidavit accompanying the application should specifically state that no such a payment was made. The personal and character details of the commissioning parent(s), especially with regards to past criminal convictions, are very important. The suitability of a parent is an objective test which includes the ability of the commissioning parent(s) to provide in the emotional and financial needs of the commissioned child, as well as a harmonious environment for the growth and development of the said child. This judgement is invaluable with regard to the court's

whether such an application was granted and/or refused. If it was refused the reasons for the refusal should be set out; 77.3 a report by a clinical psychologist in respect of the commissioning parents and a separate report in respect of the surrogate and her partner; 77.4 a medical report regarding the surrogate mother which must include the details referred to in paragraph [67] in this judgment; 77.5 details and proof of payment of any compensation for services rendered, either to the surrogate herself or to the intermediary, the donor, the clinic or any third party involved in the process; 77.6 all agreements between the surrogate and any intermediary or any other person who is involved in the process; 77.7 full particulars, if any agency was involved, any payment to such agency as well as an affidavit by that agency containing the information referred to in paragraphs [65]-[66] of this judgment; 77.8 whether any of the commissioning parents have been charged with or convicted with a violent crime or a crime of sexual nature, as envisaged in paragraph [69] of this judgment." The court further gave the following guidelines regarding the enrolment of the matter to enable the protection of the identities of the parties at par 78: "78.1 any party who seeks to bring an application will cause same to be issued by the Registrar in the ordinary course; 78.2 the court file must thereafter immediately be brought to the office of the Deputy Judge President, together with a letter explaining the facts and that the application is brought in terms of section 295 of the Children's Act 38 of 2005 and requesting a date for hearing. In the event that there exists any urgency in the hearing of the matter that must be set out in the letter as well; 78.3 the Deputy Judge President will then give further directions as to how this matter shall be heard in due course, including the allocation of the judge for the hearing of the matter; 78.4 any consideration as to hearing in camera must be addressed to the judge allocated to hear the matter once the parties are notified of the relevant date of the hearing." Further see Nöthling Slabbert 2012 SAJBL 30.

approach in considering the financial aspects relating to a surrogate agreement and the involvement of a surrogacy agency. Finally, the court's discussion of the factors which the court will consider determining what will be in the best interests of the commissioned child, may be regarded as a yardstick for further applications.

3.15.3 CM v NG⁷²⁴

This judgment concerns an application in terms of sections 23 and 24 of the Children's Act, which regulate applications for contact, care and guardianship of a child.⁷²⁵ Section 23 provides that any person having an interest in the care, well-being or development of a child may apply to court for an order that he or she may have care and/or contact in respect of the child, whereas section 24 provides for the same regarding guardianship of a child. The parties in this case, although involved in a same sex relationship for several years, were not married.⁷²⁶ A child conceived through artificial fertilisation was born on 29 October 2008.⁷²⁷ The respondent in this case was the biological mother of the minor child, whereas the applicant had no biological connection with the child.⁷²⁸ The parties' relationship ended in November 2010.⁷²⁹ On 12 April 2011, the respondent advised the applicant that she wanted to stop the applicant's contact with the minor child as she believed that the contact was not in the child's best interest.⁷³⁰ The applicant brought an urgent application seeking an order to have full parental rights and responsibilities in respect of the minor child.⁷³¹

⁷²⁴ 2012 (3) All SA 104 (WCC).

⁷²⁵ 38 of 2005. *CM v NG* at par 1.

⁷²⁶ *CM v NG* at par 1.

⁷²⁷ *CM v NG* at par 2.

⁷²⁸ CM v NG at par 2. Because the child was born in London, the respondent was recorded as the parent as she gave birth to the child. Sloth-Nielson, J and van Heerden, B "The constitutional family developments in South African child and family law 2003-2013" 2014 International Journal of Law, Policy and the Family 28(1) 115.

⁷²⁹ *CM v NG* at par 3.

⁷³⁰ *CM v NG* at par 3.

⁷³¹ In terms of sec 18(2), (3), (4) and (5) of the Children's Act. *CM v NG* at par 1. The facts of the matter are set out in par 2 and 3. The parties applied in December 2006 to the Cape Fertility Clinic for the Respondent to undergo artificial insemination. The applicant underwent a similar process in August 2007. Respondent is the biological mother of the minor child and the applicant has no biological bond with the child. After the parties separated in November 2010, the applicant continued to have contact with the minor child. The parties did agree that the child's primary residence shall be with the respondent. The parties indicated that the fertilisation problem was as a result of 'lesbian couple'. It is further clear that the respondent signed forms in March 2009 consenting to the artificial insemination of the applicant.

The court observes that if the parties *in casu* were in a heterosexual relationship, the male person in the applicant's position would have been recognised as the father figure whose bond with the child would have been viewed as being that of a parent.⁷³² There is thus no reason why the applicant should not be treated in the same way.⁷³³ The court points out that it is evident that it was the intention of the parties to have the child together and that applicant played the role of a parent to the minor child.⁷³⁴ Turning to section 18(2) of the Act, the court stated that both "care" and "contact" are components of "parental rights and responsibilities" that are relevant to this aspect.⁷³⁵ Gangen AJ states that if one has regard to the best interest of the child standard, the specific facts of the matter in each instance will determine what are in the child's best interest.⁷³⁶ It could well be possible that not all the aspects of "care" and "contact" as set out in the definition may be relevant to a particular set of facts.⁷³⁷ Thus, the court retains the discretion to delineate the specific aspects of care and contact to be allocated to each party.⁷³⁸

Gangen AJ concludes that the applicant has no automatic parental rights and responsibilities in terms of sections 19 to 21 of the Act as she has no biological link to the minor child.⁷³⁹ Furthermore, the applicant does not acquire automatic parental rights and responsibilities in terms of section 40 (which deals with children conceived by artificial fertilisation) as she did not enter into a marriage with the respondent.⁷⁴⁰ Thus, in absence of an agreement in terms of section 22, the applicant's recourse is to apply to the court in terms of sections 23 and 24 of the Act.⁷⁴¹ The best interests of the child, the relationship between the applicant and the child and any other relevant person and the child, and any other factor which the court considers should be taken

⁷³⁸ *CM v NG* at par 34.

⁷³² *CM v NG* at par 21.

⁷³³ CM v NG at par 21.

⁷³⁴ CM v NG at par 22. Sloth-Nielson, J and van Heerden, B "The constitutional family developments in South African child and family law 2003-2013" 2014 International Journal of Law, Policy and the Family 28(1) 115. The authors pointed out that the social role which the applicant played as a parent, despite the complete absence of any biological or genetic link with the child, came to the fore as the determinant of the applicant's status as holder of parental responsibilities and rights.

⁷³⁵ *CM v NG* at par 32.

⁷³⁶ *CM v NG* at par 34.

⁷³⁷ CM v NG at par 34.

⁷³⁹ *CM v NG* at par 59.

⁷⁴⁰ *CM v NG* at par 60. This section makes reference to spouses.

⁷⁴¹ *CM v NG* at par 61. At par 62 the court explained that it has to consider the application against the criteria set out in the Act in sections 23(2) and 24(2).

into account, are common to both sections.⁷⁴² The court holds that section 23(2) goes further in that it specifies the degree of commitment that the applicant has shown towards the child and the extent to which the applicant has contributed towards expenses in connection with the birth and maintenance of the child.⁷⁴³ The court concludes that the applicant is entitled to parental rights and responsibilities as set out in section 18, as it would be in child's best interests to have a relationship with two parents.⁷⁴⁴

3.15.4 Ex parte SA and others⁷⁴⁵

This case concerns an application for the confirmation of a surrogate agreement by the applicants which was entered into on 28 October 2014.⁷⁴⁶ Manamela AJ remarks that each matter depends on its own circumstances, especially when dealing with issues such as surrogacy, which is very personal in nature.⁷⁴⁷ Specific mention was made to the prohibition of payments in respect of surrogacy, which requires very special attention due to its implications beyond the contractual and/or personal relationships of the said applicants.⁷⁴⁸ The court points out that commercial surrogacy is a real possibility in present times and that the court is assigned with a critical responsibility in this regard.⁷⁴⁹

In this matter also, as was the case in the matter of *CM v NG* above, the court observes that each matter is dependent on its own circumstances, especially when dealing with highly personal matters, such as surrogacy. Personal circumstances of the different parties involved in the agreement will always be different from the parties in the next application for confirmation of an agreement. The significance of this judgment is that it emphasises that a careful case-by-case consideration of all relevant circumstances is required to determine if the agreement will be in commissioned child's best interest.

- ⁷⁴⁶ Ex parte SA and others at par 1.
- ⁷⁴⁷ Ex parte SA and others at par 3.

⁷⁴² *CM v NG* at par 63.

⁷⁴³ *CM v NG* at par 64.

⁷⁴⁴ *CM v NG* at par 72. ⁷⁴⁵ 2014 JDR 2616 (GP

 ⁷⁴⁵ 2014 JDR 2616 (GP).
 ⁷⁴⁶ Ex parts SA and other

⁷⁴⁸ Ex parte SA and others at par 16.

⁷⁴⁹ *Ex parte SA and others* at par 16.

3.15.5 Ex parte HPP and others; Ex parte DME and others⁷⁵⁰

These cases refer to applications brought for the confirmation of surrogate agreements in terms of section 292(1)(e), read with section 295 of the Children's Act.⁷⁵¹ In these cases, Ms Strydom acted as a surrogacy coordinator, providing surrogacy facilitation services for which she charged a fee of R 5,000.00.⁷⁵² The court was called upon to interpret section 301 of the Children's Act in such a way that Ms Strydom's constitutional right to follow the occupation or trade of her choice is protected.⁷⁵³ The salient issues identified in these applications were:

(a) Whether the surrogacy facilitation agreements constitute a transgression of s 301 of the Children's Act; and

(b) Whether the court could confirm the surrogate motherhood agreements if it is found that the agreements between Ms Strydom and the applicants were unlawful.⁷⁵⁴

Aside from the issue above, both the applicants had complied with the requirements pertaining to surrogate agreements as provided for in the Act.⁷⁵⁵ They also complied with the guidelines provided by the court for surrogacy agreements.⁷⁵⁶ *In casu*, the court emphasises that commercial surrogacy is unlawful in South Africa and the payments are limited to only those specifically set out in the Act.⁷⁵⁷ Tolmay J remarks that the courts must apply the law within the existing legislative framework, as the foundational principle of the relevant framework is inextricably linked to the objective to protect persons involved in surrogacy arrangements.⁷⁵⁹ The court observes that the commonly held view seems to be that the potential for abuse far outweighs any possible advantage in surrogacy arrangements.⁷⁵⁹ Section 303(2) of the Act states that no person may in any way, for or with a view to compensation, make known that any person is or may possibly be willing to enter into a surrogate agreement.⁷⁶⁰ Tolmay J concludes that this provision makes the payment of any introductory fee pertaining to

⁷⁵⁰ 2017 (4) SA 528 (GP); [2017] 2 All SA 171 (GP).

⁷⁵¹ Ex parte HPP and others; Ex parte DME and others at par.1

⁷⁵² *Ex parte HPP and others; Ex parte DME and others* at par.2 and 8. At par 6. Ms Strydom is a surrogacy consultant through Destiny Babies and she advertised her services online.

⁷⁵³ Ex parte HPP and others; Ex parte DME and others at par 12.

Ex parte HPP and others; Ex parte DME and others at par 3.
 Ex parte HPP and others: Ex parte DME and others at par 4.

 ⁷⁵⁵ Ex parte HPP and others; Ex parte DME and others at part 4.
 ⁷⁵⁶ Ex parte HPP and others: Ex parte DME and others at part 4.

Ex parte HPP and others; Ex parte DME and others at part 4.
 Ex parte HPP and others: Ex parte DME and others at part 24

 ⁷⁵⁷ Ex parte HPP and others; Ex parte DME and others at par 24.
 ⁷⁵⁸ Ex parte HPP and others; Ex parte DME and others at par 24.

Ex parte HPP and others; Ex parte DME and others at par 24 and par 26.
 Ex parte HPP and others; Ex parte DME and others at par 26.

⁷⁵⁹ Ex parte HPP and others; Ex parte DME and others at par 26.

⁷⁶⁰ Sec 303(2) of the Children's Act. Nicholson and Bauling 2013 *De Jure* 528.

the surrogate unlawful, which also explained why Ms Strydom categorically stated that she had not received any introductory fee.⁷⁶¹ The court confirms that any surrogate agreement that does not comply with the provisions of the Act will be invalid and any child born as a result of any action taken in execution of such an agreement would, for all purposes, be deemed to be the child of the woman who gave birth to that child.⁷⁶² The payment of a facilitation fee or any compensation of any nature other than those that the Act makes provision for is expressly prohibited and the affidavit should state that no such payment was made.⁷⁶³ There was thus a duty on the applicants in the second application to inform the court in the affidavit about the payments made to Ms Strydom.⁷⁶⁴ The attorney in a surrogacy application should file an affidavit confirming that as far as he or she could ascertain, no payments were made to anyone apart from those provided for in the Act.⁷⁶⁵

Analysing section 301, the court refers to the two distinct categories of lawful expenses that are catered for.⁷⁶⁶ Firstly, the costs directly related to (a) artificial fertilisation and pregnancy; (b) the birth of the child; and (c) the confirmation of the surrogate agreement;⁷⁶⁷ and secondly, the *bona fide* professional legal and medical expenses.⁷⁶⁸ The expenses in the first category must be directly related to the processes referred to. It is important to note that the expenses listed are limited to those necessary to ensure that the process of surrogacy and the subsequent pregnancy are successfully completed and the surrogate agreement confirmed.⁷⁶⁹ The court admits that these expenses will obviously include costs that are necessary to ensure that the fertilisation

⁷⁶¹ Ex parte HPP and others; Ex parte DME and others at par 24 and par 30.

⁷⁶² Ex parte HPP and others; Ex parte DME and others at par 31.

⁷⁶³ Ex parte HPP and others; Ex parte DME and others at par 33.

⁷⁶⁴ Ex parte HPP and others; Ex parte DME and others at par 33.

⁷⁶⁵ Ex parte HPP and others; Ex parte DME and others at par 33. The court referred to the case of Ex parte WH and others 2011 (6) SA 541 GP in par 32, 35 and 37. The court set out guidelines that the parties need to comply with apart from the statutory requirements in the Act. Tolmay J stated the purpose of the directives in *Ex parte WH* is clearly to enable the court to obtain all relevant information pertaining to compliance with the Act and to determine whether there was any contravention of the Act. Tolmay J concluded that the same requirements should apply to whoever introduces the surrogate mother and whoever provides related services.

⁷⁶⁶ Ex parte HPP and others; Ex parte DME and others at par 43. At par 42 The court explained that: "A court should when interpreting legislation follow a purposive and contextual approach. In doing so a court is enjoined to provide a broad and generous reading in determining the ambit of constitutionally enshrined rights."

⁷⁶⁷ Ex parte HPP and others; Ex parte DME and others at par 43.

⁷⁶⁸ Ex parte HPP and others; Ex parte DME and others at par 43.

⁷⁶⁹ Ex parte HPP and others; Ex parte DME and others at par 45.

process can be completed as well as costs that are directly linked to the pregnancy.⁷⁷⁰ This will include, *inter alia*, the costs of the embryologist, which, although essential for the fertilisation process, may not necessarily qualify as a medical expense.⁷⁷¹

With regards to Ms Strydom, the court held that the services rendered by her are not directly linked to either the fertilisation process or the subsequent pregnancy, as they are not essential to ensure either the fertilisation or the pregnancy.⁷⁷² Neither can it be argued that her costs are directly related to the birth.⁷⁷³ With regard to costs relating to the confirmation of the surrogacy agreement, the court limits these to costs attendant to the confirmation of the agreement, which will include costs to comply with the court's requirements.⁷⁷⁴ These will include those due to psychologists, social workers or any costs that may be occasioned by the requirements set by the court and which will ensure that the surrogacy agreement is confirmed.⁷⁷⁵ The services of Ms Strydom were regarded as not falling into any of the categories stated above, nor directly related to the confirmation of the surrogate agreement⁷⁷⁶ or those provided for in section 301.⁷⁷⁷ The court reiterates that the purpose of the prohibition in section 301 is to prevent commercial surrogacy.⁷⁷⁸ The limitation's purpose, as stated already, is to prevent commercial surrogacy, which is ultimately enacted to protect the public interest.⁷⁷⁹ Thus, important issues of public policy arise which justify a limitation of Ms Strydom's right to ask for payment for her services.⁷⁸⁰ The reason for regulating surrogacy is

⁷⁷⁰ Ex parte HPP and others; Ex parte DME and others at par 45.

Ex parte HPP and others; Ex parte DME and others at par 45.

Ex parte HPP and others; Ex parte DME and others at par 45.

⁷⁷³ Ex parte HPP and others; Ex parte DME and others at par 45.

Ex parte HPP and others; Ex parte DME and others at par 46

Ex parte HPP and others; Ex parte DME and others at par 46.

⁷⁷⁶ Ex parte HPP and others; Ex parte DME and others at par 46.

Ex parte HPP and others; Ex parte DME and others at par 47. At par 48, the court looked at sec 22 of the Constitution in the context of what sec 301 is trying to accomplish and while taking into account the provisions of sec 36 of the Constitution. Sec 22 provides that the practice and trade may be regulated by law. The court stated at par 51: "As far as the alleged limitation of Ms Strydom's right to exercise her chosen profession is concerned, it would not seem to me that her choice to exercise her chosen profession is limited. What is limited is her right to ask for payment of expenses which fall foul of the provisions of sec 301." The court further said that the purpose of the limitation is for a greater good and as such it is justifiable. The court explained at par 52 that there is a very real danger in allowing surrogacy facilitation agreements as literally anyone could give himself out as a surrogacy coordinator which will open the floodgates and with no control, this could and probably would lead to commercial surrogacy and the abuse of vulnerable people.

Ex parte HPP and others; Ex parte DME and others at par 51.

⁷⁷⁹ Ex parte HPP and others; Ex parte DME and others at par 51.

⁷⁸⁰ Ex parte HPP and others; Ex parte DME and others at par 51.

ultimately to protect the public against unprincipled people who may exploit vulnerable participants.⁷⁸¹ The limitations enacted in chapter 19 of the Act are therefore reasonable and justified in an open and democratic society in accordance with the section 36 of the Constitution.⁷⁸² Because the expenses Ms Strydom had claimed fell afoul of the provisions of section 301 of the Act, the agreements between her and the commissioning parents were accordingly declared unlawful and unenforceable.⁷⁸³

The pertinent question asked by the court was whether it should refuse to confirm a surrogate agreement which, although compliant with the Act's requirements, is tainted by the unlawful collateral agreement (surrogacy facilitation agreement) between the applicants and a third party (Ms Strydom).⁷⁸⁴ The court expresses concern that there is a real danger that courts may unwittingly facilitate and encourage illegal agreements if a surrogate agreement emanating from the illegal facilitation agreement is confirmed.⁷⁸⁵ The court explains as follows:

Consequently, when deciding whether one should confirm the surrogate motherhood agreement, it is important to keep in mind why the legislature has created a strict regulatory framework within which only certain limited expenses are allowed. The unlawfulness of commercial surrogacy sits at the heart of the limitations provided for in the Children's Act. This entails that one must establish whether the unlawful contract has tainted the lawful contract to such an extent that the lawful contract cannot be endorsed. From the aforesaid, it would seem that the appropriate approach would be that the court has a discretion, which discretion must be exercised keeping in mind the purpose of the legislative framework and the ban on commercial surrogacy.⁷⁸⁶ I am of the view that a court should be sensitive to the fact that if courts proceed to declare surrogacy agreements valid, despite the fact that they are tainted by an unlawful surrogacy facilitation agreement, it might actually negate the whole purpose of sec 301, and commercial surrogacy will have sneaked in through the back door. Court can't be seen to condone commercial surrogacy directly or indirectly in the light of the existing legislative framework. In my view the invalidity of the subsequent surrogate motherhood agreements may in many instances be the unfortunate result of entering into an unlawful surrogacy facilitation agreement.⁷⁸⁷

However, despite the declaration of unlawfulness of the facilitation agreement, Tolmay J rightly decides that in the circumstances of these two applications, she is of the view that she should exercise her discretion and confirm the surrogacy agreements.⁷⁸⁸

⁷⁸¹ Ex parte HPP and others; Ex parte DME and others at par 51.

⁷⁸² Ex parte HPP and others; Ex parte DME and others at par 51.

⁷⁸³ Ex parte HPP and others; Ex parte DME and others at par 53.

⁷⁸⁴ Ex parte HPP and others; Ex parte DME and others at par 55.

⁷⁸⁵ Ex parte HPP and others; Ex parte DME and others at par 66.

⁷⁸⁶ Ex parte HPP and others; Ex parte DME and others at par 67.

⁷⁸⁷ Ex parte HPP and others; Ex parte DME and others at par 67-68.

⁷⁸⁸ Ex parte HPP and others; Ex parte DME and others at par 71.

It is important for courts to apply the law within the existing legislative framework so as to protect the parties to a surrogacy agreement and the commissioned child. The lesson to be noted from this judgment is that it confirms that any introductory fee in respect of a surrogate is unlawful.

The court's guidance that parties should specify in the affidavit if any payments other than those provided for in the Act was made and or received, is commendable, as the parties must take the court into their confidence.

3.15.6 Ex parte CJD and others⁷⁸⁹

In this application for the confirmation of a surrogacy agreement, the commissioning parents (CJD and HN, a homosexual couple in a relationship of ten years at the time), did not live together because the one party (HN), does not want his sexual orientation to become public.⁷⁹⁰ The plan was that CJD's sperm, together with the gametes of an unknown egg donor, would be used during the fertilisation process.⁷⁹¹

For Tolmay J, two questions that the court had to consider were: Firstly, the issue of how the fact that the parties do not live together impact on the best interests of the commissioned child, and secondly, how HN's refusal to have his sexual orientation made public, impact on the best interests of the commissioned child.⁷⁹² It would seem that HN had reservations about the impact that having a child with his partner may have on himself and his career.⁷⁹³

It is trite that the court as upper guardian of children has a duty to consider whether the concerns raised above may negatively affect the best interests of the child to be born.⁷⁹⁴ As emphasised elsewhere in this chapter, all aspects that may be pertinent to

⁷⁸⁹ 2018 (3) SA 197 (GP).

⁷⁹⁰ Ex parte CJD and others at par 2.

⁷⁹¹ Ex parte CJD and others at par 3.

⁷⁹² Ex parte CJD and others at par 4. Sloth-Nielsen "Surrogacy in South Africa" 194.

⁷⁹³ Ex parte CJD and others at par 6.

⁷⁹⁴ Ex parte CJD and others at par 10. The court further pointed out that no one can judge a gay person who, as a result of persisting public prejudice is reluctant to reveal his/her sexual orientation. The court further said that it is a sad indictment against society that HN is placed in this position, but the court must always place the rights of the children and their interests first, even in circumstances where the rights of the prospective parents may be compromised.

the question of whether a surrogacy agreement should be confirmed, should be set out in the affidavit.⁷⁹⁵ Parties should "play open cards with the court", as Tolmay J explains:

The interests of children are at stake and it is essential that applicants play open cards with the court. All relevant information must be set out in the affidavit itself and properly dealt with by the parties under oath. The fact that issues of relevance are set out in the reports attached to the application is not sufficient at all. It goes without saying that the best interests of the child to be born from a surrogacy agreement must be considered by the court, in order for a court to make an informed decision on whether the surrogacy agreement should be confirmed.⁷⁹⁶

She advises that parties to surrogacy agreements should set out the following in the founding affidavit to assist the court in exercising its discretion properly:⁷⁹⁷

(a) If and how the applicants will function as a family unit and whether they are comfortable with society regarding them as such;

(b) whether they are living together or not, and if not, why this state of affairs will not impact on the interests of the child and them functioning as a family unit.

The court next considers section 293(1) of the Act, which provides that where a commissioning parent is married or in a permanent relationship, he or she must obtain the written consent of his or her husband, wife or partner, and also that the husband, wife or partner become a party to the agreement before a court may confirm a surrogacy agreement.⁷⁹⁸ Although the court agrees that a wide interpretation of this provision is preferred, a court still needs to determine whether the manner in which the permanent relationship is structured is sufficiently supportive for the purpose of raising a family.⁷⁹⁹ Tolmay J views an interpretation of section 293 to support the conclusion that because a partner in a permanent relationship consents to the surrogacy agreement and becomes a party to the application for the confirmation of the agreement, he or she acquires parental responsibilities and rights when the order of confirmation is made.⁸⁰⁰

⁷⁹⁵ *Ex parte CJD and others* at par 11.

⁷⁹⁶ Ex parte CJD and others at par 12. The court further, at par 14, states that it is not only the court's obligation to protect and advance the best interests of children, but the obligation of all role players to ensure that all relevant information, that may impact on a court's discretion to either grant or dismiss the application is set out in the affidavit.

⁷⁹⁷ Ex parte CJD and others at par 15.

⁷⁹⁸ Ex parte CJD and others at par 17.

⁷⁹⁹ *Ex parte CJD and others* at par 18. Sloth-Nielsen "Surrogacy in South Africa" 195.

⁸⁰⁰ *Ex parte CJD and others* at par 19. The court further referred to sec 22 of the Act and stated that the interpretation is in line with sec 22 of the Act which opened the door for 'any person having an interests in the care, well-being and development of the child' to obtain parental rights and obligations. He explained that someone who co-signs the surrogacy agreement may be such a person.

The court turns to section 231 of the Act which deals with adoption next.⁸⁰¹ The categories of persons eligible to adopt a child in terms of section 231(1) of the Children's Act suggests that for persons to jointly adopt a child, a permanent household or common household run by such persons should exist.⁸⁰² The existence of a family unit and common household is similarly an important aspect in surrogacy applications.⁸⁰³ As one may have expected, the court did not confirm the agreement in this application, yet emphasised that South Africa's constitutional dispensation acknowledges and respects the many varied permutations of what a family would constitute.⁸⁰⁴

I concur with the court's reasoning and judgment not to confirm the said agreement. It is important for a child to have a stable home environment and to be able to have a relationship with both parents in circumstances where there will be two commissioning parents. It will not be in a child's best interests to have a parent who is not comfortable with the public knowing that he is homosexual, as there is a possibility that the parent will not want to be seen in public with the child, which could be detrimental to the child's emotional well-being. The further importance of this judgment is the guidance in respect of the information that the parties must include in their affidavits and the type of attached expert reports that will not suffice.

3.15.7 Ex parte KAF and others (I)⁸⁰⁵ and Ex parte KAF and others (II)⁸⁰⁶

Both these cases dealt with confirmation applications in which the assessment and reporting on the suitability of the surrogate was the focus. Both the Gauteng High Court judgments will be discussed to enable understanding of the different views and arguments advanced in the two judgments.

Ex parte CJD and others at par 21 and 22. Section 231(1): "A child may be adopted (a) jointly by
 (i) a husband and wife; (ii) partners in a permanent domestic life-partnership; or (iii) other persons sharing a common household and forming a permanent family unit; (b) by a widower, widow, divorced or unmarried person; (c) by a married person whose spouse is the parent of the child or by a person whose permanent domestic life-partner is the parent of the child; (d) by the biological father of a child born out of wedlock; or by the foster parent of the child."

⁸⁰² Ex parte CJD and others at par 22.

⁸⁰³ Ex parte CJD and others at par 22.

⁸⁰⁴ Ex parte CJD and others at par 26.

⁸⁰⁵ (14341/17) [2017] ZAGPJHC 227 (10 August 2017) (hereinafter "*Ex parte KAF and others (I)*")

⁸⁰⁶ 2019 (2) SA 510 (GJ) (Hereinafter "*Ex parte KAF and others (II)*").

In the application of *Ex parte KAF and others (I)*,⁸⁰⁷ the applicants sought the confirmation of a surrogate agreement in terms of section 295 of the Children's Act.⁸⁰⁸ The commissioning parents have been married for eleven years and as a result of the commissioning mother suffering from a congenital malformation of the uterus, they were unable to naturally conceive a child together.⁸⁰⁹ They underwent five *in-vitro* procedures which were unsuccessful because the commissioning mother was unable to conceive and carry a pregnancy to term.⁸¹⁰ Medfem Clinic, which has an internal surrogacy programme, introduced the surrogate mother to the commissioning parents.⁸¹¹ The court was particularly concerned with three issues that arose in this application, namely, (a) whether the surrogate is using surrogacy as a means of income, (b) whether there are problems regarding the surrogate's well-being, and (c) whether a risk of commercialisation of the surrogate arrangement exists.⁸¹²

Unfortunately, there was insufficient evidence before the court to satisfy the court that the surrogate is not using the agreement as a means of income.⁸¹³ From the facts, it appears that the surrogate was twenty years old and a stay-at-home mother for her two children.⁸¹⁴ The court found that the surrogate and her partner did not take the court into their confidence as they did not provide the court with a full disclosure of their financial circumstances.⁸¹⁵ Further, a breakdown of expenses covered by the proposed allowance to the surrogate was not stated in the papers.⁸¹⁶

Rightly so, the court questions the surrogate's psychological well-being, as this is one of the critical factors that the court must be satisfied of.⁸¹⁷ The court accordingly rejects

⁸⁰⁹ Ex parte KAF and others (I) at par 4.

⁸⁰⁷ (14341/17) [2017] ZAGPJHC 227.

⁸⁰⁸ 38 of 2005.

⁸¹⁰ Ex parte KAF and others (I) at par 4.

⁸¹¹ Ex parte KAF and others (I) at par 6.

⁸¹² Ex parte KAF and others (1) at par 9.

⁸¹³ Ex parte KAF and others (I) at par 10.

⁸¹⁴ Ex parte KAF and others (I) at par 11.

⁸¹⁵ Ex parte KAF and others (I) at par 14.

⁸¹⁶ Ex parte KAF and others (I) at par16.

Ex parte KAF and others (I) at par 19. At par 21 the court said that the surrogate's personal circumstances do not paint a picture of a person who was able to make decisions in her best interests as teenager. The assessment report did not give the court an objective analysis of the surrogate's psychological well-being in the light of the disconcerting facts. Par 11 sets out her personal circumstances. In short, the surrogate was thirteen years old when she met her partner and their first child was born when she was sixteen years old. The surrogate dropped out of school and the second child was born when she was eighteen years old.

the assessment report that was done with regard to the surrogate, as the surrogate's well-being was not evident or clear from the report that was presented, which brings her suitability to act as a surrogate into question.⁸¹⁸ The court concluded further that the nature of the relationship between the medical professionals involved and Medfem was not disclosed to the court.⁸¹⁹ Accordingly, Modiba J held that the parties have failed to satisfy the court that the surrogate agreement met the requirements as set out in the Children's Act and the application was thus as a result thereof dismissed.⁸²⁰

*Ex parte KAF and others (II)*⁸²¹ was the second application brought by the parties for the confirmation of a surrogate agreement after their first confirmation application was dismissed by the court previously.⁸²² Siwendu J states that it is a trite legal position that it is in the court's discretion to confirm a surrogate agreement or not.⁸²³ The application is subjected to strict judicial scrutiny in light of the fact that the court is the upper guardian of all minor children and the interests of the commissioned child has to be taken into account.⁸²⁴ The court states that it is necessary to scrutinise the value chain and the relationship between the parties to ensure that the agreement is lawful and that it meets the requirements under section 301.⁸²⁵

Regarding the interpretation of section 295 of the Act, which concerns the confirmation of the surrogacy agreement by a High Court, Siwendu J remarks that the relevant section should not be read in isolation but in the context of the objects of chapter 19 of

⁸¹⁸ Ex parte KAF and others (I) at par 23.

⁸¹⁹ Ex parte KAF and others (I) at par 26. At par 25 the court mentioned the risk of commercialisation of the surrogacy process that arose from the papers with specific reference to the internal surrogacy programme of Medfem Clinic. It appeared that the same medical specialist who conducted the medical assessment of the parties involved will be the one who administer the artificial fertilisation treatment on the surrogate. At par 26 the court pointed out that no information was provided regarding the payment, if any, of the complementary medical professional services to the surrogacy matching service of Medfem Clinic.

⁸²⁰ Ex parte KAF and others (I) at par 30.

⁸²¹ 2019 (2) SA 510 (GJ).

⁸²² Ex parte KAF and others (II) at par 1. See Ex parte KAF and others (I) (14341/17) [2017] ZAGPJHC 227 (10 August 2017). At par 20 the court points out that Ex parte KAF and others (I) illustrates that the requirements for assessing the suitability of the surrogate to act as such remains unclear. The court further pointed out that there is thus a need to develop further the guidelines and the requirements as is set out in Ex parte WH.

⁸²³ Ex parte KAF and others (II) at par 16.

⁸²⁴ *Ex parte KAF and others (II)* at par 16. At par 17 the court further explained that it is evident that given the discretionary powers conferred on the court, and the likely different circumstances where the court's confirmation is sought, each court may construe the parameters for the exercise of the discretion differently from the next.

⁸²⁵ Ex parte KAF and others (II) at par 16.

the Children's Act and other relevant provisions, particularly, taking into account the interplay of sections 293(1), 297(1) and 301(1) (discussed above) in the assessment and the decision of the court.⁸²⁶ This assessment is an objective one, although the decision is made with regard to the subjective circumstances of the applicants, which will differ from case to case.⁸²⁷ The following guidelines were set out by the court *in casu*:⁸²⁸

- (a) The personal clinical assessment of the prospective surrogate mother and her surrounding circumstances must be supported by other relevant information, where necessary, and include information on whether the surrogate mother:⁸²⁹
 - (i) is physically and medically fit to carry the gamete and in turn the unborn child to full term;⁸³⁰
 - (ii) has an agreement with the commissioning parents regarding selective reduction and the risks pertaining thereto;⁸³¹
 - (iii) is of sound mind enjoys good mental health, and does not suffer from any personality disorder, severe psychiatric illness, or has a history of selfharming behaviour;⁸³²

⁸²⁶ Ex parte KAF and others (II) at para 25. The court explained that its primary purpose of the assessment under this section is thus to safeguard the health of the commissioned child, the fulfilment of the surrogate agreement and to prevent the potential commercial exploitation of the commissioned parents and the surrogate mother in equal measure.

⁸²⁷ Ex parte KAF and others (II) at par 26. The structure of the family unit has evolved over the years and in view of the varying circumstances, there cannot be an exhaustive closed checklist and each case requires that it be judged on its peculiar facts.

⁸²⁸ Ex parte KAF and others (II) at par 27-29. Thaldar 2019 SAJBL 62. The writer states that it is clear that the court's intent was that these paragraphs contain the criteria for assessing the suitability of an intended surrogate mother. The writer interprets para 11 of the judgment as implicitly adding an additional criterion for assessing the suitability of an intended surrogate mother as the court did not state that this list of criteria is exhaustive.

Ex parte KAF and others (II) at par 27.

⁸³⁰ Ex parte KAF and others (II) at par 27.1.

⁸³¹ Ex parte KAF and others (II) at par 27.2. Thaldar 2019 SAJBL 62. The surrogate must understand that the commissioning parents may not wish to have multiple children and they may want the surrogate to undergo 'selective reduction'. Thaldar stated that the assessing clinical psychologist should be familiar with both the potential psychological sequelae of selective reduction as well as the potential medical sequelae.

⁸³² Ex parte KAF and others (II) at par 27.3. Thaldar 2019 SAJBL 63. The proposed surrogate mother must not suffer from any personality disorder or severe psychiatric illness or have a history of any self-harming behaviour.

- (iv) does not have a history of substance abuse, including drugs and/or alcohol and addiction, likely to have similar effects as those referred to in (iii) above.⁸³³
- (b) The emotional welfare, emotional needs and resources available to the surrogate must be considered to determine the likely effects on the commissioned child as well as the fulfilment of the agreement.⁸³⁴ Thus, having regards to section 293(1) of the Act which requires the written consent of the spouse if the surrogate is married or involved in a permanent relationship, a report is required relating to the:⁸³⁵
 - (i) surrogate's need for emotional resources, if any;⁸³⁶
 - (ii) existing emotional resources;⁸³⁷
 - (iii) quality and stability of the existing emotional support structure; and⁸³⁸
 - (iv) whether the surrounding relationships are conducive for the fulfilment of the surrogacy agreement and may result in termination of the agreement after artificial fertilisation or a breach of the agreement.⁸³⁹
- (c) In terms of sections 297(1)(a) and 297(1)(c) of the Children's Act, as discussed above, the surrogate will not have any rights of parenthood or care of the commissioned child or contact with the child and neither will her husband, partner or relatives.⁸⁴⁰ As the surrogate must thus understand the nature of the surrogacy relationship, the nature of surrogate motherhood and that the child to

⁸³³ *Ex parte KAF and others (II)* at par 27.4. Thaldar 2019 *SAJBL* 64. Thaldar pointed out that the court was silent on the social drinking of alcohol and smoking during pregnancy although the experts (in their joint opinion to the court) suggested that if an intended surrogate smokes, she must refrain from smoking for at least three months prior to conception, submit a blood test result to indicate that she stopped smoking, undertake to refrain from smoking for the duration of pregnancy and be free of any medication that she may use to stop smoking unless authorised by the attending gynaecologist.

⁸³⁴ Ex parte KAF and others (II) at par 28.

⁸³⁵ Ex parte KAF and others (II) at par 28.

⁸³⁶ Ex parte KAF and others (II) at par 28.1.

⁸³⁷ Ex parte KAF and others (II) at par 28.2.

⁸³⁸ Ex parte KAF and others (II) at par 28.3.

⁸³⁹ Ex parte KAF and others (II) at par 28.4. Thaldar 2019 SAJBL 63.

⁸⁴⁰ Ex parte KAF and others (II) at par 29.

be born will legally not be her child, but the child of the commissioning parents, a report on the following issues is required:⁸⁴¹

- (i) the psycho-social support structure of the surrogate;⁸⁴²
- (ii) the understanding and influence of the husband, partner, relatives or extended family in the decision;⁸⁴³
- (iii) the understanding that the child to be born will belong to the commissioning parents;⁸⁴⁴
- (iv) how handing the baby over to the commissioning parents will affect her;⁸⁴⁵
- (v) that the psychosocial support structure is not likely to result in the termination of the agreement after fertilisation or in a breach;⁸⁴⁶ and
- (vi) whether she is emotionally available for her own child/ children, including her readiness to discuss the surrogate pregnancy with her child/ children, depending on their ages and levels of comprehension. ⁸⁴⁷

Siwendu J concludes by stating that a court must, as far as possible, provide a 'check' on a potential intrusion or encroachment on the fundamental rights of the commissioning parents and that of the surrogate and offer equal weight to their protection.⁸⁴⁸

Thaldar maintains that the very purpose of the pre-approval mechanism for surrogate motherhood is to establish legal certainty for all the parties involved in the agreement, including the child to be born.⁸⁴⁹ The pre-approval mechanism depends heavily on the

⁸⁴¹ *Ex parte KAF and others (II)* at par 29.

⁸⁴² Ex parte KAF and others (II) at par 29.1.

⁸⁴³ Ex parte KAF and others (II) at par 29.2. Thaldar 2019 SAJBL 63.

Ex parte KAF and others (II) at par 29.3. Thaldar 2019 SAJBL 62.

⁸⁴⁵ Ex parte KAF and others (II) at par 29.4.

⁸⁴⁶ Ex parte KAF and others (II) at par 29.5. Thaldar 2019 SAJBL 63.

⁸⁴⁷ *Ex parte KAF and others (II)* at par 29.6. Thaldar 2019 *SAJBL* 64. Thaldar pointed out that this includes the readiness to discuss the surrogate pregnancy with her child/children, depending on their ages and levels of comprehension.

⁸⁴⁸ Ex parte KAF and others (II) at par 33.

⁸⁴⁹ Thaldar 2019 *SAJBL* 61.

input from clinical psychologists.⁸⁵⁰ The reason why the psychological assessment of the proposed surrogate mother is so important, is that because the commissioning parents are motivated by the desire to have a child, it is generally unusual for a woman to offer to gestate someone else's child without any financial benefit.⁸⁵¹

Both the *Ex parte KAF* judgments are valuable in that they provide additional guidance in respect of what type of information is required for the purpose of applications for the confirmation of surrogate agreements. Lessons to be learned from these judgments speak to the requirements of the objectivity of the medical practitioners who perform the assessments on the parties, as well as the significance of the assessments to provide clarity on the proposed surrogate's mental and physical health. Moreover, parties to a surrogate agreement must provide adequate information in respect of their financial positions, especially the proposed surrogate mother. It is important that the court is able to discern if acting as a surrogate is for reasons other than generating an income, as the court has to be satisfied that the agreement is not for commercial gain.

3.15.8 Ex parte MS and others; In re: Confirmation of surrogate mother agreement⁸⁵²

This judgment, briefly referred to under paragraph 2 in this chapter, follows an application for the confirmation of a surrogate agreement *after* the surrogate was already artificially fertilised.⁸⁵³ The surrogate was thirty-three weeks pregnant at the time that the application was brought.⁸⁵⁴ Her pregnancy came about as a result of

⁸⁵⁰ Thaldar 2019 *SAJBL* 61.

⁸⁵¹ Thaldar 2019 *SAJBL* 61. At 62 the writer criticises the court in that it focussed on the budget or the out-of-pocket expenses for which the commissioning parents would have to reimburse the surrogate instead of including altruistic motivation in its list of criteria for assessing the potential surrogate mother. Thaldar stated that the apparent inference is that the actual agreement regarding reimbursable expenses is the only relevant consideration on which altruism will be judged and that an intended surrogate mother's motivation for engaging in surrogacy is not legally relevant. At 63 Thaldar pointed out that there is an important lesson to be learned from the *Ex parte KAF and others (II)* judgment and that is that motivation as expressed by an intended surrogate during consultation with the assessing clinical psychologist must be aligned with the reality of the financial provisions of the surrogacy agreement.

⁸⁵² 2014 (2) All SA 312 (GNP).

⁸⁵³ Ex parte MS and others; In re: Confirmation of surrogate mother agreement at par 1. Louw 2014 De Jure 110.

⁸⁵⁴ Ex parte MS and others; In re: Confirmation of surrogate mother agreement at par 4. Louw 2014 De Jure 111.

fertilisation using her ova and the commissioning father's sperm.⁸⁵⁵ The parties had a verbal agreement in terms whereof the surrogate agreed to act as surrogate mother for the commissioning parents.⁸⁵⁶

The court referred to sections 292, 296, 303(1), and 297(1) and (2), which detail the eligibility requirements for those wishing to enter into lawful surrogacy arrangements.⁸⁵⁷ Parties thus do not have *carte blanche* to enter into surrogacy agreements.⁸⁵⁸

Regarding the competence of a court to confirm a surrogate agreement postfertilisation, the court explains that the Act does not stipulate what the consequences of non-compliance with the Act's provisions will be on the validity of a written agreement subsequently entered into between the parties.⁸⁵⁹ The court refers to the wellestablished common law principle that an agreement to commit an unlawful act is not enforceable and this include acts that are unlawful in terms of a statute.⁸⁶⁰ *In casu*, the unlawful act is the artificial fertilisation of the surrogate prior to the written agreement being confirmed by the court.⁸⁶¹

A surrogacy agreement aims to protect the best interests of the commissioned child and his or her right to family and parental care.⁸⁶² The regulation of the rights and obligations of the respective parties in the agreement also advances further constitutional rights of the parties, which include the right to dignity, the right to make decisions concerning reproduction and the surrogate mother's right to security in and

⁸⁵⁵ Ex parte MS and others; In re: Confirmation of surrogate mother agreement at par 5.

⁸⁵⁶ Ex parte MS and others; In re: Confirmation of surrogate mother agreement at par 5. Baase 2019 PELJ 5. The writer explains that the non-compliance by the parties was both administrative (parties failed to adhere to the provisions of ch 19 that regulate the confirmation of a surrogate agreement) and material (the non-compliance resulted in the artificial fertilisation of the surrogate mother) in nature.

⁸⁵⁷ Ex parte MS and others; In re: Confirmation of surrogate mother agreement at par 18 to 23.

⁸⁵⁸ Ex parte MS and others; In re: Confirmation of surrogate mother agreement at par 24.

⁸⁵⁹ Ex parte MS and others; In re: Confirmation of surrogate mother agreement at par 30. Louw 2014 De Jure 112.

⁸⁶⁰ Ex parte MS and others; In re: Confirmation of surrogate mother agreement at par 31. Baase 2019 PELJ 6.

⁸⁶¹ *Ex parte MS and others; In re: Confirmation of surrogate mother agreement* at par 32. "However, it was the very absence of a written agreement duly confirmed and authorised by the court that rendered the artificial fertilisation unlawful."

⁸⁶² Ex parte MS and others; In re: Confirmation of surrogate mother agreement at par 34.

control over her body.⁸⁶³ Section 295 appears to cover both the situation where a child has not yet been conceived at the time that confirmation of a surrogacy agreement is sought and the situation where a child has already been conceived, but is not yet born at the time.⁸⁶⁴ Neither section 292, nor section 295 requires the court to be satisfied that the surrogate has not yet undergone the process of artificial fertilisation and that she is not already pregnant as a result thereof.⁸⁶⁵

The Act expressly prohibits the artificial fertilisation of a surrogate before the agreement is confirmed by the court.⁸⁶⁶ It is an offence to artificially fertilise a woman in the execution of a surrogacy agreement without the authorisation by a court for such fertilisation.⁸⁶⁷ Interestingly, the Act does not make it unlawful for the parties involved in the prohibited act to proceed afterwards by entering into a valid surrogacy agreement and to apply to court to confirm the agreement.⁸⁶⁸ The question arising is whether the prohibition regarding the surrogate mother's artificial fertilisation without the court's authorisation should have a legal effect on the validity of the surrogacy agreement, or prevent the court from confirming the said agreement.⁸⁶⁹

The court further argues that to interpret the Act as precluding the court from sanctioning a surrogacy agreement after artificial fertilisation has already taken place, would undermine the constitutional rights of the parties involved and would be contrary to the broad objective of the Act.⁸⁷⁰ Should it have been that the agreement was invalid,

⁸⁶³ Ex parte MS and others; In re: Confirmation of surrogate mother agreement at par 35.

⁸⁶⁴ Ex parte MS and others; In re: Confirmation of surrogate mother agreement at par 41. Baase 2019 PELJ 8.

⁸⁶⁵ Ex parte MS and others; In re: Confirmation of surrogate mother agreement at par 42. Louw 2014 De Jure 113. Baase 2019 PELJ 7.

⁸⁶⁶ *Ex parte MS and others; In re: Confirmation of surrogate mother agreement* at par 45. Sec 296 of the Act.

⁸⁶⁷ *Ex parte MS and others; In re: Confirmation of surrogate mother agreement* at par 45. Sec 303 of the Act.

⁸⁶⁸ Ex parte MS and others; In re: Confirmation of surrogate mother agreement at par 46. Louw 2014 De Jure 115. Baase 2019 PELJ 8.

⁸⁶⁹ *Ex parte MS and others; In re: Confirmation of surrogate mother agreement* at par 48. "The Act prescribes an express criminal penalty for the commission of a prohibited act of artificial fertilisation; it does not provide for invalidity of the surrogacy agreement as penalty." Louw 2014 De Jure 114-115. Louw argues that the care and welfare of a child before its birth is exclusively in the hands of the surrogate mother. Sec 295(e) gives the court a discretion to confirm the agreement if, in general, having regard to the considerations as set out in the section, the court is satisfied that the agreement should be confirmed. She further argues that the considerations in sec 295 refer to anticipated circumstances and family situations of all the parties concerned and the best interests of the child, specifically after the child is born.

⁸⁷⁰ Ex parte MS and others; In re: Confirmation of surrogate mother agreement at par 50.

the result would have been that once the child is born, he or she would be deemed for all purposes to be the legal child of the surrogate.⁸⁷¹ In the court's opinion, it would also infringe on the dignity of the commissioning parents who would then be denied the opportunity to experience a family of their own.⁸⁷² Moreover, it would encroach on their right to make reproductive choices, as the commissioning parents would then be required to adopt the child born of the surrogate.⁸⁷³ It would also violate the surrogate's full parental rights and responsibilities in respect of the child, which could be an infringement of her constitutional right to make her own decisions regarding reproduction.⁸⁷⁴ The rights and interests of the "sleeping partner" in the surrogacy relationship, meaning the unborn child, should demand the most protection. The court explains its reasoning as follows:⁸⁷⁵

[U]ncertainty regarding the parents could impact negatively on the child. He or she is denied the family life that was planned for him or her. The child will have to rely instead on the parental care of the surrogate mother, who, by virtue of the agreement, has deliberately chosen not to take responsibility for another child.⁸⁷⁶

Such an effect should never follow without a court considering all the facts at hand and making a determination as to what is in the best interests of the child.⁸⁷⁷ It would be patently contrary to section 29(2) of the Constitution to hold that a court has no discretion to confirm a surrogacy agreement in circumstances like this where the confirmation is sought post-fertilisation.⁸⁷⁸ The court must thus retain the discretion to do so if it is satisfied that this is in the best interests of the child to be born.⁸⁷⁹ It goes

⁸⁷¹ Ex parte MS and others; In re: Confirmation of surrogate mother agreement at par 50.

⁸⁷² Ex parte MS and others; In re: Confirmation of surrogate mother agreement at par 51. The court further stated: "The rights of individuals to bear and raise children are broadly recognised and supported by the State through various measures, including the provision of financial assistance, social and other support services. It encompasses the right to have one's own child with whom the parents share a genetic link, the right to adopt a child under certain circumstances, and, more recently, in recognition of the physical and medical difficulties people may experience in seeking to have a child of their own, the right to have a child through a surrogacy arrangement." Baase 2019 *PELJ* 9.

⁸⁷³ Ex parte MS and others; In re: Confirmation of surrogate mother agreement at par 51. Louw 2014 De Jure 115. Baase 2019 PELJ 9.

⁸⁷⁴ Ex parte MS and others; In re: Confirmation of surrogate mother agreement at par 52. Baase 2019 PELJ 9.

Ex parte MS and others; In re: Confirmation of surrogate mother agreement at par 53. "Section 28(1)(b) of the Constitution guarantees to every child the right to family or parental care. In addition, section 29(2) specifies that: "A child's best interests are of paramount importance in every matter concerning the child."

⁸⁷⁶ Ex parte MS and others; In re: Confirmation of surrogate mother agreement at par 54. Louw 2014 De Jure 115.

⁸⁷⁷ Ex parte MS and others; In re: Confirmation of surrogate mother agreement at par 55.

⁸⁷⁸ Ex parte MS and others; In re: Confirmation of surrogate mother agreement at par 55.

⁸⁷⁹ Ex parte MS and others; In re: Confirmation of surrogate mother agreement at par 55.

without saying that the discretion to confirm such agreements should only be exercised in exceptional circumstances and when the best interests of the child demand the said confirmation.⁸⁸⁰

Regarding the approach which the courts should adopt in confirmation applications post-fertilisation, a court will always be guided by the particular facts of each case before it.⁸⁸¹ The parties would need to place sufficient facts before the court, in addition to the required information, to explain why confirmation is sought at a late stage and why the confirmation is warranted, notwithstanding their being in breach of the general scheme.⁸⁸² The court would have to be satisfied that the application is not aimed at, or will not have the effect of, permitting the parties to circumvent the objectives of the regulatory scheme.⁸⁸³

It is important to note that the window period for a post-fertilisation confirmation of a surrogacy agreement exists only during the period before the child is born as the purpose of chapter 19 is to establish certainty regarding the legal and parental status

Ex parte MS and others; In re: Confirmation of surrogate mother agreement at par 57. Louw 2014 De Jure 116. Louw argues that what the court is saying here does little to assuage the effect of its judgment on the enforcement of the Act (in par 51-54 of his judgment) as discussed above. Louw poses the question of when it would not be in the best interests of a child to confirm a surrogacy agreement once the child has been conceived in light of the negative effect of imposing parenthood on the surrogate mother. At 117 Louw observes that sec 295(e) could be used to prove the existence of the court's residuary discretion. She further argues that the court has the power to confirm a surrogacy agreement if it is satisfied that the agreement should be confirmed. The personal circumstances and the family situation in the MS case could easily have justified a post-fertilisation confirmation of the agreement especially taking into account the two previously failed surrogacy agreements. For Louw, the judgment of the court could have been justified based on the exceptional circumstances and not, as indicated by the court, on the implied unconstitutionality of the requirement as a whole.

⁸⁸¹ Ex parte MS and others; In re: Confirmation of surrogate mother agreement at par 59. Louw 2014 De Jure 118. Louw observes that the MS case should be used as another example of hard cases making bad law. She argues that while the applicants may have been accommodated not entirely without good cause, the arguments justifying the intervention of the court have undermined the value of the legislation in question and created uncertainty in the fragile context of surrogacy where it can be ill afforded.

⁸⁸² Ex parte MS and others; In re: Confirmation of surrogate mother agreement at par 62. Baase 2019 PELJ 10.

⁸⁸³ Ex parte MS and others; In re: Confirmation of surrogate mother agreement at par 63.

of, and in relation to, the child that will be born.⁸⁸⁴ Parental rights and responsibilities in respect of a child are established and these take effect at birth.⁸⁸⁵

Finally, in ordering confirmation of this specific agreement, the court cautions that it is mindful of the need for courts to facilitate a balance between ensuring compliance with the regulatory scheme established under chapter 19 of the Act, on the one hand, and not setting the bar too high for parents whose only option is to have a child by way of surrogacy, on the other.⁸⁸⁶

This judgment points to the importance of complying with the requirements in respect of a court order allowing artificial fertilisation of the surrogate as the court might not confirm an agreement post-fertilisation. A discussion of chapter 19 of the Act would not be complete without reference to the punishable offences relating to the regulation of surrogate motherhood.

3.16 Criminal offences in terms of chapter 19 of the Children's Act⁸⁸⁷

Ex parte MS and others; In re: Confirmation of surrogate mother agreement at par 67. At par 68 the court makes reference of the definition of a child. In terms of the Act, a child is a person under the age of 18 years and in terms of the common law, a person is someone who is born. At par 71 the court stated that a court will not blithely reach the conclusion that a surrogacy agreement is in the best interests of the child to be born. They therefore need adequate time to consider the application and to make a properly considered judicial decision in this regard. The applicant should thus understand that if they leave the application for the confirmation too late, that they run the risk that a decision will not be made before the child is born. It is thus important that postfertilisation confirmation applications should be made timeously and as soon as practically possible in the circumstances. Baase M "The Ratification of Inadequate Surrogate Motherhood Agreements and the Best Interest of the Child" 2019 PELJ 10.

⁸⁸⁵ Ex parte MS and others; In re: Confirmation of surrogate mother agreement at par 68.

⁸⁸⁶ Ex parte MS and others; In re: Confirmation of surrogate mother agreement at par 76. Baase M "The Ratification of Inadequate Surrogate Motherhood Agreements and the Best Interest of the Child" 2019 PELJ 10-12. The writer criticises the judgment by stating that the Court's interpretation and application of ch 19 and other provisions contained in the Constitution raises concern. It is pointed out by the writer that the deviation from the protocol by the applicants was not in the interest of justice but due to personal preference and financial implications. Furthermore, the commissioning parents' behaviour de-emphasises the imperative purpose of the surrogate agreement and it undermines the complex inter-relational nature of such agreements by merely taking their own rights and interests into account. Baas points out that while sec 292 and sec 295 may not have required that the surrogate mother may not be artificially fertilised prior to the court confirming the agreement, sec 296 and sec 303 prohibit such an act. Sec 297(2) makes provision for parties' non-compliance. The non-compliance does affect the legal status of the child although the affect is not permanent, and it can be altered after the child is born. At p 16-17 the writer explains that in surrogacy matters there is no guarantee that the child will be born alive, and the surrogate still reserves the right to terminate the pregnancy before the child is born or she can terminate the agreement after the child has been born.

⁸⁸⁷ 38 of 2005.

The Act provides for several criminal offences relating to breaches of provisions in chapter 19. Section 301 specifically prohibits a reward or compensation in respect of a surrogate agreement.⁸⁸⁸ Promises or agreements to pay compensation in respect of a surrogate agreement are unenforceable except for the exceptional circumstances set out in section 301(2) of the Act.⁸⁸⁹

No person is permitted to publish or disclose the identity of parties to court proceedings in respect of a surrogate agreement without the written consent of those parties.⁸⁹⁰ An offence will be committed by any person publishing information that reveals the identity of a person that was born of a surrogate agreement.⁸⁹¹ Any person who artificially fertilise a woman pursuant to a surrogate agreement, but without a court order authorising the person to do so, will commit an offence.⁸⁹²

It is also an offence for a person to publish the fact that a person is willing or might be willing to enter into a surrogate agreement.⁸⁹³ A person found guilty of committing any of the above offences, will be liable, upon conviction, to pay a fine or may be imprisoned for a maximum period of ten years. Section 305(7) however determines that a second or subsequent conviction may cause imprisonment of a maximum of twenty years or a fine, or both.⁸⁹⁴

3.17 The medical threshold requirement and permanent illness

The medical threshold requirement in section 295(a) requires that a court may not confirm a surrogate agreement unless the commissioning parent(s) is or are permanently unable to give birth to a child.⁸⁹⁵

⁸⁸⁸ Sec 301.

⁸⁸⁹ Sec 301(2).

⁸⁹⁰ Sec 305(6) read with sec 305(1)(b) and sec 302(1).

⁸⁹¹ Sec 305(6) read with sec 305(1)(b) and sec 302(2).

⁸⁹² Sec 305(6) read with sec 305(1)(b) and sec 303(1). The same applies where the person assists someone in artificially fertilising a woman without the necessary court order.

⁸⁹³ Sec 305(6) read with sec 305(1)(b) and sec 303(2).

⁸⁹⁴ Sec 305(7).

⁸⁹⁵ Sec 295(a). Louw 2014 *De Jure* 111. SALRC Project 140 85. Jordaan, DW "Surrogate Motherhood in illness that does not cause infertility" 2016 *South African Medical Journal* 106(7) 684. The writer refers to this requirement as the 'threshold requirement' for surrogate motherhood. He further states that the persons who qualify in terms of this requirement include male same-sex couples and single men (who are biologically unable to give birth) and

Sloth-Nielsen, in analysing this section, distinguishes between conception infertile (when a person is unable to contribute a gamete for the purposes of conception through artificial fertilisation) and pregnancy infertile (when a person is permanently and irreversibly unable to carry a pregnancy to term).⁸⁹⁶ Jordaan explains that a narrow interpretation of the threshold requirement focuses only on a person's inherent ability to carry a successful pregnancy and excludes considering the medical *sequelae* of pregnancy.⁸⁹⁷ A broader interpretation on the other hand, would consider the medical *sequelae* of pregnancy as integral to the person's ability to give birth to a child.⁸⁹⁸ The legislative intent of the requirement was to exclude surrogacy for convenience and restrict surrogacy to a reproductive remedy as a last resort to the persons.⁸⁹⁹

A case in point is the unreported judgment of *Ex Parte LS*, where the court dealt with an application where the commissioning mother was fertile but suffering from a permanent illness that would render pregnancy a significant risk to her health and that of the prospective baby's health *in utero*.⁹⁰⁰ The applicants (commissioning parents) in this case applied to have their surrogacy agreement confirmed by the court.⁹⁰¹ Persons qualifying in terms of the threshold requirement would include male same-sex couples, single men and heterosexual couples or single women medically unable to carry a pregnancy to term.⁹⁰² In *Ex Parte LS*, the court granted the application for the confirmation of the surrogate agreement, but did not provide reasons.⁹⁰³ It is possible to conclude from the case that not all three requirements set out in section 295(a) need to be present contemporaneously and that it will suffice if an intended mother has a serious and irreversible medical condition which will be exacerbated by a pregnancy.⁹⁰⁴

heterosexual couples or single women where the woman is medically unable to carry a pregnancy to term.

⁸⁹⁶ Sloth-Nielsen "Surrogacy in South Africa" 187. Both these situations will meet the requirements of sec 295(a).

⁸⁹⁷ Jordaan 2016 *SAMJ* 684.

⁸⁹⁸ Jordaan 2016 *SAMJ* 684.

⁸⁹⁹ Jordaan 2016 *SAMJ* 684. The writer states that the wording 'not able to give birth' was thus designed to give effect to the legislative intent.

⁹⁰⁰ Jordaan 2016 *SAMJ* 684. Unreported case of the High Court of South Africa, Gauteng Local Division, Johannesburg, Case no 2015/24392. LexisNexis *Family Law service* at par W6. The commissioning mother had a permanent and irreversible medical condition known as cystic fibrosis.

⁹⁰¹ Jordaan 2016 *SAMJ* 684. They were a heterosexual couple.

⁹⁰² Jordaan 2016 *SAMJ* 684.

⁹⁰³ Jordaan 2016 *SAMJ* 685. The matter thus cannot be used to establish a precedent regarding the interpretation of the threshold requirement.

⁹⁰⁴ LexisNexis *Family Law service* at par W6.

It could well be that the commissioning parent(s) are able to give birth to a child but serious health risks may arise for both or either the mother and the foetus as a result of permanent illness of the mother.⁹⁰⁵ Fundamental rights issues may come into play, notably those relating to human dignity, the right to freedom and security of the person and the best interests of the child.⁹⁰⁶

Jordaan regards the objective of the threshold prerequisite as striving to balance the right of a person to start a family with the government's purpose of avoiding surrogacy agreements for convenience and of reserving surrogacy as reproductive means of last resort.⁹⁰⁷ There are, however, cases where there is a significant health risk for the commissioning mother and the child, surrogacy cannot be seen as a matter of convenience as it is a reproductive means of last resort.⁹⁰⁸ The right to freedom of security of a person clearly includes the right not to be treated in an inhuman or demeaning way. To expect a commissioning mother (with a permanent illness) to become pregnant to enable her to start a family and thereby to potentially sacrifice her health as well as the child's, would be inhuman and demeaning.⁹⁰⁹ Here, the best interests of the child clearly demand that a surrogate be used instead of the commissioning mother.⁹¹⁰

Jordaan argues that the human rights dimension requires a broad interpretation of the threshold prerequisite.⁹¹¹ Further, a narrow interpretation of the threshold prerequisite clearly ignores the consequences of pregnancy and is hence untenable.⁹¹² The commissioning mother is considerably compromised as she remains childless, or she becomes pregnant and accepts the risk of a significant health risk to herself or to her prospective child.⁹¹³ Jordaan concedes that pregnancy in general carries health risks,

⁹⁰⁵ Jordaan 2016 *SAMJ* 685.

⁹⁰⁶ Jordaan 2016 *SAMJ* 685.

⁹⁰⁷ Jordaan 2016 SAMJ 685.

⁹⁰⁸ Jordaan 2016 SAMJ 685.

⁹⁰⁹ Jordaan 2016 *SAMJ* 685.

⁹¹⁰ Jordaan 2016 *SAMJ* 685. The Constitution further provides that the best interests of the child is paramount in all matters pertaining to the child.

⁹¹¹ Jordaan 2016 *SAMJ* 685.

⁹¹² Jordaan 2016 *SAMJ* 685.

⁹¹³ Jordaan 2016 *SAMJ* 685. Jordaan concludes by stating that this is an inhuman choice that the law cannot force on any person.

not to mention a range of permanent illnesses that may increase a woman's health risk during pregnancy.⁹¹⁴

In conclusion, it is important to evaluate each case on its own merits as not all health risks would satisfy the threshold prerequisite.⁹¹⁵ Courts should rule that the threshold prerequisite is fulfilled and allow surrogacy as a reproductive means of last resort *only* if expert medical evidence shows that pregnancy by the commissioning mother would entail a significant health risk to her or the child and she is effectively unable to give birth to a child.⁹¹⁶ As Clarke rightly maintains, the intention of the legislature could never have been to limit surrogacy to only be available to male parent(s) or to intended mothers who had hysterectomies, or in instances where the intended mother was born without an uterus.⁹¹⁷ She further argues that the legislature clearly intends for surrogacy to have a broader purpose to assist persons with a serious medical condition that is both permanent and irreversible.⁹¹⁸ The fact that an intended mother's uterus is still present does not and should not disqualify her from surrogacy.⁹¹⁹

A recent Western Cape High Court judgment on this exact issue provides more clarity on the threshold requirement, discussed below.

3.17.1 APP and another v NKP⁹²⁰

The first and second applicants in this case (concerning an application for the confirmation of a surrogate agreement) are married and have two children.⁹²¹ The third applicant is the proposed surrogate. Although the first applicant was able to have a

⁹¹⁴ Jordaan 2016 *SAMJ* 685.

⁹¹⁵ Jordaan 2016 *SAMJ* 685.

⁹¹⁶ Jordaan 2016 *SAMJ* 685. It must be as a result of a permanent illness.

⁹¹⁷ LexisNexis *Family Law service* at par W6.

⁹¹⁸ LexisNexis *Family Law service* at par W6.

⁹¹⁹ LexisNexis *Family Law service* at par W6.

⁹²⁰ 2021 JDR 1650 (WCC)/ (17962/2020) [2021] ZAWCHC 69 (11 March 2021).

⁹²¹ APP and another v NKP at par 3 and 4. The first child was born from a surrogate agreement and the second child was born from a natural pregnancy. At par 9 it is apparent that she only conceived after undergoing five IVF treatments and the pregnancy was complicated and difficult which became life-threatening. The first applicant suffered from hypertension and gestational diabetes during her pregnancy and she developed placenta previa. She was hospitalised from 32 weeks whereafter she underwent an emergency caesarean section which was complicated by significant blood loss for which she required a blood transfusion and a further four surgeries post birth brought on by infection.

natural pregnancy, a further pregnancy would endanger her life and that of the foetus, hence the first and second applicants' decision to have another child through surrogacy.⁹²² Because of the first applicant's medical problems, which are both permanent and irreversible, she is unable to give birth to a child.⁹²³ She is also forty-five years old.⁹²⁴ Her condition was described to the court as follows: She suffers from post-traumatic stress disorder and a recurrent depressive disorder for which she has been taking a range of psycholeptic medications in high doses for the past fifteen years.⁹²⁵ There is thus an increased risk of congenital abnormalities of the foetus should the first applicant continue to use these medication during her pregnancy.⁹²⁶ Moreover, she also suffers from hypothyroidism and diabetes for which she needs to take chronic medication.⁹²⁷ At one point, she stopped taking her medication, which resulted in an acute deterioration in her condition. Her psychological condition was described as permanent and irreversible.⁹²⁸

Bozalek J rightly observes that the use of the word 'condition' is not qualified or prefaced by anything to limit the meaning of the term to a physical medical condition only.⁹²⁹ Thus, the condition should include both physical and psychological conditions.⁹³⁰ The court remarks that a narrow interpretation of section 295(a) would be at odds with the apparent purpose of chapter 19 which are designed to afford an opportunity to persons who would not otherwise be able to have a child genetically related to them to do so by means of surrogacy.⁹³¹ The court interprets the term 'not

⁹²² APP and another v NKP at par 4. Another pregnancy will be life threatening to first applicant as a result of various medical and psychological conditions.

⁹²³ APP and another v NKP at par 7.

⁹²⁴ APP and another v NKP at par 8.

⁹²⁵ APP and another v NKP at par 8.

⁹²⁶ APP and another v NKP at par 8. The first applicant's obstetrician and gynaecologist stated in her report that some of the adverse pregnancy outcomes of the first applicant's medication is pre-term labour, low birth weight, poor neo-natal adaptation and increased risk of congenital abnormalities.

⁹²⁷ APP and another v NKP at par 8.

⁹²⁸ APP and another v NKP at par 9. The first applicant's age plays a role as she is more at risk of gestational diabetes recurring and early onset pre-eclampsia which is life-threatening to both the mother and the foetus and she is further at risk of the complication of a repeat placenta previa with possible morbid adherence. At par 10 a reproductive medicine specialist further confirmed that surrogacy is the only way for the first and second applicants to have a child genetically linked to at least one of them.

⁹²⁹ APP and another v NKP at par 21.

⁹³⁰ APP and another v NKP at par 21.

⁹³¹ APP and another v NKP at par 24. The court also considered the meaning of 'not able to give birth' and whether this entails absolute physical incapacity or something less.

able to give birth' as unable to give birth without a significant medical risk to the health of the life of the mother.⁹³² The court concludes that the first applicant did establish, on the basis of the risk to her own health and life by a further pregnancy, that she is not able to give birth to a child and that her condition is permanent and irreversible.⁹³³ The surrogacy agreement was subsequently confirmed by the court.⁹³⁴

This judgement is of great assistance as it paves the way courts should interpret section 295(a). Medical evidence in this case has convincingly demonstrated that psychological problems can also interfere with pregnancy and conception of a woman.

4. CONCLUSION

Chapter 19 has brought legal certainty in respect of the status of the commissioned child, as well as the rights that a child, conceived by artificial fertilisation, will have. Furthermore, there is clarity regarding parental rights and responsibilities in respect of the commissioned child. Case law has assisted to clarify the position of same-sex couples and/or spouses where chapter 19 did not specifically provide for such a scenario. A number of judgments have also offered guidance on the specific issues that should be put before the court when applications for the confirmation of agreements are submitted.

This chapter has alluded to the specific instances where regulations may complement the Act by providing more granularity. For example, the term "permanent relationship" must be defined to enable parties to know when the other the partner must provide written consent and become a party to the surrogacy agreement. The same applies to the reference "stable home environment".

⁹³² APP and another v NKP at par 25.

APP and another v NKP at par 30.
 APP and another v NKP at par 32.

APP and another v NKP at par 32.

Currently, each High Court has its own practice directives that attempt to provide guidance to legal practitioners where the Act is lacking regulations, but this is clearly not the most practical or conducive option at present.

The constitutional framework relevant to surrogate motherhood will be explored next. This chapter will also interrogate the constitutionality of some of the provisions in chapter 19 of the Children's Act.

CHAPTER 4

CONSTITUTIONAL FRAMEWORK RELEVANT TO SURROGATE MOTHERHOOD

1. INTRODUCTION

The Constitution of the Republic of South Africa, 1996, sets out the values underpinning South Africa's new constitutional democracy, and includes the Bill of Right as the cornerstone of the South African democracy,⁹³⁵ contained in Chapter 2 of the Constitution. The rights of each person are protected by the Bill of Rights, the democratic values of human dignity, equality and freedom.⁹³⁶ It is the duty of the State to protect, respect, promote and fulfil each of the rights contained in the Bill of Rights.⁹³⁷ Each right, however, is subject to certain limitations as provided for in section 36 in the Bill of Rights.⁹³⁸

The most pertinent human rights relevant to the regulation of surrogacy and artificial fertilisation are the right to equality⁹³⁹, the right to dignity⁹⁴⁰ and the right to bodily and psychological integrity which includes, among others, the right to make choices regarding reproduction.⁹⁴¹ These rights are crucial, especially in respect of the woman acting as a surrogate mother. The rights of all parties—the surrogate mother, the commissioning parents, including those of the commissioned child, must be respected throughout the surrogacy process.

As stated, the focus of this chapter will be on the constitutional provisions relevant to surrogacy and artificial fertilisation.⁹⁴² Modern international and national legal developments reflect an increased awareness of the need to recognise and protect the human rights of both women and children.⁹⁴³ This view is borne out by the protection of the rights of the parties to the surrogacy arrangement, supported by relevant case

- ⁹³⁵ Sec 7(1).
- 936 Sec 7(1).
- ⁹³⁷ Sec 7(2).
- ⁹³⁸ Sec 7(3).
- ⁹³⁹ Sec 9.
- ⁹⁴⁰ Sec 10.
- ⁹⁴¹ Sec 12(2)(b).

⁹⁴³ Nicholson 2013 *SAJHR* 502.

⁹⁴² Constitution of the Republic of South Africa, 1996 - Chapter 2: Bill of Rights.

law explored in the previous chapter. Although the Constitution does not directly protect a person's right to have children or to procreate, this right is indirectly protected, as will be discussed below.⁹⁴⁴

The discussion on human rights is necessary to gain a complete picture of the legal position regarding surrogacy and artificial fertilisation in South Africa. The discussion below will explore the relevant rights in more detail.

2. RIGHTS OF PARTIES TO SURROGATE MOTHERHOOD AND ARTIFICIAL FERTILISATION

2.1 The right to bodily and psychological integrity of a person

In terms of section 12(2), every person has the right to bodily and psychological integrity, which includes the right to make decisions regarding reproduction;⁹⁴⁵ the right to security in and control over their own body;⁹⁴⁶ and not to be subjected to medical or scientific experiments without their informed consent.⁹⁴⁷ In the case of *Christian Lawyers Association v Minister of Health and Others (Reproductive Health Alliance as Amicus Curiae)*, the court describes the right to individual self-determination that flows, *inter alia*, from section 12(2), as follows:

The fundamental right to individual self-determination itself lies at the very heart and base of the constitutional right to termination of pregnancy.⁹⁴⁸

Mojapelo J further states that the recognition of the right of every individual to selfdetermination has now become an imperative under the Constitution and particularly sections 12(2), 27(1)(a), 10 and 17 of the Bill of Rights.⁹⁴⁹

⁹⁴⁴ Nicholson 2013 *SAJHR* 502.

⁹⁴⁵ Sec 12(2)(a).

⁹⁴⁶ Sec 12(2)(b).

⁹⁴⁷ Sec 12(2)(c).

⁹⁴⁸ Christian Lawyers Association v Minister of Health and Others (Reproductive Health Alliance as *Amicus Curiae*) 2005 (1) SA 509 (T).

⁹⁴⁹ Christian Lawyers Association v Minister of Health and Others (Reproductive Health Alliance as *Amicus Curiae*) 2005 (1) SA 509 (T) at 518.

Psychological and physical integrity include the ideas of self-determination and autonomy.⁹⁵⁰ Referring to the case of *AB* and another *v* Minister of Social *Development*,⁹⁵¹ Rosenberg emphasises the court's observation that a child's proper emotional development is dependent upon the child's ability to maintain contact with his or her family.⁹⁵² She argues that section 12(2) protects both the physical and psychological well-being of a person.⁹⁵³ From this, it follows that the judgment in *AB* and another *v* Minister of Social Development may be interpreted to protect a child's right to know his or her origins, as the child's psychological development may be affected without such knowledge.⁹⁵⁴

Chapter 19 of the Children's Act invokes the constitutional right to make decisions concerning reproduction.⁹⁵⁵ The question arising is whether the state's legitimate interest in the regulation of surrogate agreements is adequately compelling to suggest that most aspects of chapter 19 will survive constitutional scrutiny. As stated above, the right to bodily and psychological integrity has a bearing on reproductive issues as it is specifically recognised that this right includes the right to make decisions concerning reproduction.⁹⁵⁶ The scope of the right to "make decisions concerning reproduction.⁹⁵⁶ The scope of the right to "make decisions concerning reproduction.⁹⁵⁶ The scope of the right to "make decisions concerning reproduction.⁹⁵⁶ The scope of the right to make decisions concerning reproduction.⁹⁵⁶ The scope of the right to make decisions concerning reproduction.⁹⁵⁶ The scope of the right to make decisions concerning reproduction.⁹⁵⁶ The scope of the right to that reproductive rights are protected under section 12(2)(a).⁹⁵⁷ However, with reference to the judgment in *AB and another v Minister of Social Development*, it may be argued that the latter

⁹⁵⁰ Rosenberg 2020 *TSAR* 724.

⁹⁵¹ 2017 (3) BCLR 267 (CC); 2017 (3) SA 570 (CC).

⁹⁵² Rosenberg 2020 *TSAR* 728.

⁹⁵³ Rosenberg 2020 *TSAR* 724. At 740 the author stated that it in the instance of surrogacy agreements, there is a loss experienced by the child in that he or she is being separated from the gestational mother but despite this loss, the children have not shown any negative impact psychologically.

⁹⁵⁴ Rosenberg 2020 *TSAR* 724.

⁹⁵⁵ LexisNexis *Bill of Rights Compendium* at par 3E28. Nicholson 2013 *SAJHR* 503. Van Niekerk 2015 *PELJ* 404. The author points out that making decisions regarding reproduction can be interpreted as deciding to have a child or not to have a child.

⁹⁵⁶ Carstens and Pearmain *Foundational Principles* 177. Van Niekerk 2015 *PELJ* 420. The author points out that this right has not been interpreted as referring only to choices in respect of contraception and the termination of pregnancy. It includes the right to make decisions about the manner which one reproduces, and this includes the right to decide to use a surrogate and donor gametes if necessary.

⁹⁵⁷ Van Niekerk 2017 PELJ 14. Van Niekerk 2015 PELJ 405. The author states that sec 12(2)(a) presumably suggests (in the absence to the contrary) that every person can make decisions regarding reproduction and that is regardless of whether the person is fertile or not. Further, the content of the right to decide includes the right to make use of natural conception or conception by assisted means. Importantly, she argues that this entails the decision whether to use one's own gametes or those of a gamete donor and whether recourse will be had to a surrogate mother or not.

judgment implies that whilst section 12(2)(a) protects the right to reproduce noncoitally, it applies only to parties who are physically involved in the reproductive process.⁹⁵⁸ Looking at the reference in section 12 to both bodily and psychological integrity as a tacit acknowledgement that both the body and the psyche are protected,⁹⁵⁹ Van Niekerk concludes that the provision in section 12(2)(a) is not flexible enough to accommodate advancements in medical technology and that it should be amended or at least interpreted more broadly in the future.⁹⁶⁰ As alluded to earlier in this thesis, section 294 of the Children's Act does not permit infertile persons unable to provide a genetic link to the commissioned child to enter into a surrogacy arrangement, and by doing so, arguably denies these persons their right under section 12(2)(a) of the Constitution.⁹⁶¹

2.2 The right to human dignity

Section 10 of the Constitution provides every person a right to an inherent dignity and the right to have their dignity respected and protected.⁹⁶² Dignity may be described as the state or quality of being worthy of honour or respect⁹⁶³

and also:

⁹⁵⁸ Van Niekerk 2017 *PELJ* 15. Presumably either by providing genetic material or by carrying the child. The author stated that this effectively excludes individuals who cannot meet these criteria. Thus, the person's own body must be engaged in the act of reproduction.

⁹⁵⁹ Van Niekerk 2017 *PELJ* 16. The author pointed out that the provision in sec 12(2)(b) makes specific reference to "security in and control over their body" in comparison to sec 12(2)(a). Deciding on reproduction may involve the assistance of others in the reproductive process and the need of a person to provide his or her own gametes would be unnecessary in these circumstances. She explains that if the interpretation of the majority in the *AB and another v Minister of Social Development* case (the need to provide one's own gametes) is followed, then all forms of noncoital reproduction are not protected by sec 12(2)(a).

⁹⁶⁰ Van Niekerk 2017 PELJ 16. At 18 the author stated that section 294 of the Children's Act violates the rights of individuals to exercise their reproductive choice and to do so freely. Further, the limitations of these rights are not reasonable and justifiable and she agrees with the minority judgment in AB and another v Minister of Social Development. AB should at least have been awarded the same opportunity as other individuals to exercise her reproductive autonomy. She further explains that sec 12(2)(a) does not guarantee the right to give effect to a decision but merely to make one and the Constitutional Court failed AB in this respect.

⁹⁶¹ Van Niekerk 2015 *PER/PELJ* 406. The denial of the right thus amounts to a limitation of the said the right. At 412 the author stated that the infringement of sec 12(2)(a) is neither reasonable nor justifiable.

⁹⁶² Nicholson 2013 SAJHR 503.

⁹⁶³ <u>https://www.vocabulary.com/dictionary/dignity</u> (accessed on 31 January 2023).

The importance and value that a person has, that makes other people respect them or makes them respect themselves. 964

Ackerman J, in the judgment of *National Coalition for Gay and Lesbian Equality and Another v Minister of Justice and others*,⁹⁶⁵ reaffirms the view expressed earlier in *S v Makwanyane*⁹⁶⁶ that the right to dignity is the cornerstone of the South African Constitution. The importance of this right is emphasised in the role accorded to it in section 36 of the Constitution.⁹⁶⁷ The Court further explains this right by stating:

Dignity is a difficult concept to capture in precise terms. At its least, it is clear that the constitutional protection of dignity requires us to acknowledge the value and worth of all individuals as members of our society.⁹⁶⁸

Sachs J explains the centrality to the concept of dignity by reference to other constitutional rights:⁹⁶⁹

It will be noted that the *motif* which links and unites equality and privacy, and which, indeed, runs right through the protections offered by the Bill of Rights, is dignity.⁹⁷⁰

For O'Regan J, the value of dignity in our Constitutional framework cannot be doubted.⁹⁷¹ The court furthermore observes that the Constitution asserts dignity to contradict our past and to inform the future, to invest in our democracy respect for the intrinsic worth of all human beings.⁹⁷² It can be said that human dignity informs constitutional adjudication and interpretation at a range of levels, most possibly all other rights.⁹⁷³ Not only is dignity a value that is fundamental to our Constitution, it is

⁹⁶⁴ The definition may be accessed from <u>https://dictionary.cambridge.org/dictionary/english/dignity</u> (accessed on 4 January 2023).

⁹⁶⁵ 1999(1) SA 6 (CC).

⁹⁶⁶ 1995 (6) BCLR 665; 1995 (3) SA 391; [1996] 2 CHRLD 164; 1995 (2) SACR 1 at par 144. The Court held that the rights to life and dignity were the most important of all human rights and the source of all the other personal rights detailed in Chapter 3 of the Interim Constitution.

⁹⁶⁷ National Coalition for Gay and Lesbian Equality and Another v Minister of Justice and others at par 28.

⁹⁶⁸ National Coalition for Gay and Lesbian Equality and Another v Minister of Justice and others at par 28.

⁹⁶⁹ National Coalition for Gay and Lesbian Equality and Another v Minister of Justice and others at par 120.

⁹⁷⁰ National Coalition for Gay and Lesbian Equality and Another v Minister of Justice and others at par 120.

⁹⁷¹ Dawood and Another v Minister of Home Affairs and others; Shalabi and Another Minister of Home Affairs and others: V Thomas and Another v Minister of Home Affairs and others 2000 (3) SA 936 (CC) at par 35. 972

Home Affairs Dawood and Another v Minister of and others: Shalabi and Another v Minister of Home Affairs and others: Thomas and Another v Minister of Home Affairs and others at par 35.

⁹⁷³ Dawood and Another Minister of Home Affairs and others: V of Shalabi and Another v Minister Home Affairs and others; Thomas and Another v Minister of Home Affairs and others at par 35.

also a justiciable and enforceable right that must be respected and protected.⁹⁷⁴ When the value of human dignity is offended, the primary constitutional breach occasioned may be of a more specific right in many cases, for example, the right to bodily integrity or the right to equality.⁹⁷⁵

The prohibition of commercial surrogacy protects the surrogate's right to dignity.⁹⁷⁶ Section 294 of the Children's Act, on the other hand, has the effect of treating persons who are able to provide genetic material differently from those who are unable to.⁹⁷⁷ This differentiation is based on personal traits inherently attached to the infertile person.⁹⁷⁸ Van Niekerk thus argues that these persons' right to human dignity is violated by this section, as it ignores and/or marginalises the persons incapable of providing genetic material legally required in terms of a surrogacy agreement and in doing so, offends their sense of self-worth.⁹⁷⁹ Legislation that unjustifiably and unfairly prevents certain persons from achieving their goal to have children is tantamount to a violation of their right to have their dignity respected and protected.⁹⁸⁰ Although the ostensible goal of section 294 is to prevent child trade, the commodification of babies and to promote a bond between the parents and the child (the latter in the child's best interests), there are other less restrictive means to achieve the same goal.⁹⁸¹ Van Niekerk hence contends that the limitation of the right to dignity is therefore not justified and it is unconstitutional.⁹⁸²

A further aspect of dignity is the question whether the lack of knowledge of a commissioned child regarding its biological origins, born as a result of donor sperm as part of a surrogacy arrangement, may violate such child's right to human dignity.⁹⁸³ For Rosenberg, answering this question depends on a subjective investigation into

974	Dawood Shalabi	and and	Another Another	V V	Minister Minister	of of	Home Home	Affairs Affairs	and and	others; others;
	Thomas a	nd Anoth	ers at par 3	35.						
975	Dawood	and	Another	v	Minister	of	Home	Affairs	and	others;
	Shalabi	and	Another	v	Minister	of	Home	Affairs	and	others;
	Thomas and Another v Minister of Home Affairs and others at par 35.									
976	Boothuws and Broeders 2013 SAL 1490									

 ⁹⁷⁶ Bonthuys and Broeders 2013 SALJ 490.
 ⁹⁷⁷ Van Nickerk 2015 *BEL* 1412

- ⁹⁸⁰ Van Niekerk 2015 *PELJ* 413.
- ⁹⁸¹ Van Niekerk 2015 *PELJ* 413.
- ⁹⁸² Van Niekerk 2015 *PELJ* 413. ⁹⁸³ Pasaphora 2020 *TSAP* 732
- ⁹⁸³ Rosenberg 2020 *TSAR* 732.

⁹⁷⁷ Van Niekerk 2015 *PELJ* 412. ⁹⁷⁸ Van Niekerk 2015 *PEL J* 412.

 ⁹⁷⁸ Van Niekerk 2015 *PELJ* 412.
 ⁹⁷⁹ Van Niekerk 2015 *PEL J* 412-47

⁹⁷⁹ Van Niekerk 2015 *PELJ* 412-413.

everyone's own experiences.⁹⁸⁴ Importantly, dignity protects the right to knowledge of one's origins as dignity is at the core of a person's psychological well-being and a lack of knowledge of one's origins could possibly have an impact on one's psychological state.⁹⁸⁵ In the case of the *National Coalition for Gay and Lesbian Equality and Another v Minister of Justice and others*,⁹⁸⁶ a situation-sensitive human rights approach is adopted, which requires a focus on the lived experiences of persons through the lens of equality:

One of the great gains achieved by following a situation-sensitive human rights approach is that analysis focuses not on abstract categories, but on the lives as lived and the injuries as experienced by different groups in our society. The manner in which discrimination is experienced on grounds of race or sex or religion or disability varies considerably - there is difference in difference. The commonality that unites them all is the injury to dignity imposed upon people as a consequence of their belonging to certain groups. Dignity in the context of equality has to be understood in this light. The focus on dignity results in emphasis being placed simultaneously on context, impact and the point of view of the affected persons. Such focus is in fact the guarantor of substantive as opposed to formal equality.⁹⁸⁷

2.3 The right to privacy

Section 14 of the Constitution provides that every person has the right to privacy, which includes the right to not have their person, home or property searched⁹⁸⁸, their possessions seized⁹⁸⁹ or the privacy of their communications infringed.⁹⁹⁰ The Constitutional court describes privacy in the *National Coalition* case as follows:

Privacy recognises that we all have a right to a sphere of private intimacy and autonomy which allows us to establish and nurture human relationships without interference from the outside community.⁹⁹¹

The right to privacy can be defined as the right of a person to be left alone or the right of a person to live his or her life as he or she wants.⁹⁹² For the Constitutional Court, private facts refer to those matters that the disclosure of which will cause mental distress and injury to anyone possessed of ordinary feelings and intelligence in the

⁹⁸⁴ Rosenberg 2020 *TSAR* 732.

⁹⁸⁵ Rosenberg 2020 *TSAR* 732.

⁹⁸⁶ 1999(1) SA 6 (CC).

 ⁹⁸⁷ National Coalition for Gay and Lesbian Equality and Another v Minister of Justice and others at par 126.
 ⁹⁸⁸ Soc 14(a) and 14(b)

⁹⁸⁸ Sec 14(a) and 14(b).

⁹⁸⁹ Sec 14(c).

⁹⁹⁰ Sec 14(d).

⁹⁹¹ National Coalition for Gay and Lesbian Equality and Another v Minister of Justice and others at par 32.

⁹⁹² NM and others v Smith and others (Freedom of expression institute as Amicus Curiae) 2007 (5) SA 250 (CC) at par 32-33.

same circumstances and in respect of which there is a will to keep those matters private.⁹⁹³ In another seminal constitutional case that dealt with the right to privacy, *NM* and others v Smith and others (Freedom of expression institute as Amicus Curiae), the court remarks as follows on the interdependence between the rights to privacy and dignity:

Underlying our Constitution is a recognition that, although as human beings we live in a community and are in a real sense both constituted by and constitutive of that community, we are nevertheless entitled to a personal sphere from which we may and do exclude that community. In that personal sphere, we establish and foster intimate human relationships and live our daily lives. This sphere in which to pursue our own ends and interests in our own ways, although often mundane, is intensely important to what makes human life meaningful. The right to privacy recognises the importance of protecting the sphere of our personal daily lives from the public. In so doing, it highlights the inter-relationship between privacy, liberty and dignity as the key constitutional rights which construct our understanding of what it means to be a human being. All these rights are therefore interdependent and mutually reinforcing. We value privacy for this reason at least - that the constitutional conception of being a human being asserts and seeks to foster the possibility of human beings choosing how to live their lives within the overall framework of a broader community. The protection of this autonomy, which flows from our recognition of individual human worth, presupposes personal space within which to live this life.⁹⁹⁴

It is important that we do not deny the right to privacy its importance in the new constitutional order at the cost of protecting the right to equality.⁹⁹⁵ Sachs J stated that while recognising the unique value of each person, the Constitution does not presuppose that a holder of rights is an isolated, lonely and abstract figure possessing a disembodied and socially disconnected self.⁹⁹⁶ The Constitution acknowledges that people live in their bodies, their communities, their cultures, their places and their times.⁹⁹⁷ The court stated that respect for one's personal privacy does not require disrespect for social standards.⁹⁹⁸

Sachs J acknowledges that the law may continue to proscribe what is acceptable and what is unacceptable and that includes sexual expression of a person. Even in the sanctum of a person's home, the court may, within justifiable limits, penalise what is

NM and others v Smith and others (Freedom of expression institute as Amicus Curiae) at par 34.
 NM and others v Smith and others (Freedom of expression institute as Amicus Curiae) at par 130

 <sup>- 131.
 &</sup>lt;sup>995</sup> National Coalition for Cay and Lesbian Equality and Another y Minister of Justice and others at

National Coalition for Gay and Lesbian Equality and Another v Minister of Justice and others at par 32.
 Notional Coalition for Cay and Leabian Equality and Another v Minister of Justice and others at National Coalition for Cay and Leabian Equality and Another v Minister of Justice and others at par 32.

⁹⁹⁶ National Coalition for Gay and Lesbian Equality and Another v Minister of Justice and others at par 117.

⁹⁹⁷ National Coalition for Gay and Lesbian Equality and Another v Minister of Justice and others at par 117.

⁹⁹⁸ National Coalition for Gay and Lesbian Equality and Another v Minister of Justice and others at par 119.

damaging and control what is offensive.⁹⁹⁹ Thus, for present purposes, the limits that are established may not offend the Constitution.¹⁰⁰⁰ Although the matter of *National Coalition for Gay and Lesbian Equality and Another v Minister of Justice and others* dealt with the unconstitutionality of the common-law offence of sodomy and certain other legal provisions, the statement above by Sachs J applies equally to other contexts, such as those of surrogacy and artificial fertilisation. The constitutionality of section 294 of the Children's Act will be explored in more detail below.

Respecting a single infertile woman's right to privacy by allowing her to enter into a surrogacy agreement is clearly not disrespecting of social standards or the *boni mores*. It constitutes a rightful affirmation of her right to dignity, self-determination and privacy, by allowing her to make her own decisions regarding reproduction without unjustifiable interference, and to have her own family in circumstances where she is unable to have her own child.

A further aspect of the right to privacy in the context of surrogacy relates to the right to know one's biological origins. A commissioning and adoptive parent (in the case of a child) have the right to decide to inform the child of his or her biological background. Although this right is not explicitly acknowledged in the Constitution, the concept of identity has been closely related to the right to privacy, based on the notion that it is necessary to have one's own autonomous identity.¹⁰⁰¹

With reference to the interpretation of the right to privacy, the rule that no right is to be considered absolute, implies that at the start of interpretation, each right is always already limited by every other right accruing to another citizen.¹⁰⁰² Thus, looking at privacy, it would mean that it is only the inner sanctum of a person which is shielded from erosion by conflicting rights of the community.¹⁰⁰³ Further, privacy is recognised in the truly personal realm, but as a person moves into communal relations and

⁹⁹⁹ National Coalition for Gay and Lesbian Equality and Another v Minister of Justice and others at par 119.

¹⁰⁰⁰ National Coalition for Gay and Lesbian Equality and Another v Minister of Justice and others at par 119.

¹⁰⁰¹ Bernstein and others v Bester and others NNO 1996 (2) SA 751 (CC) at par 65.

¹⁰⁰² Bernstein and others v Bester and others NNO at par 67.

¹⁰⁰³ Bernstein and others v Bester and others NNO at par 67. The Court gave the examples of inner sanctum of a person as his/her family life, sexual preference and home environment.

activities, for example business and social interaction, the scope of a person's personal space shrinks accordingly.¹⁰⁰⁴ Ackermann J explains this in the *Bernstein* case as follows:

A very high level of protection is given to the individual's intimate personal sphere of life and the maintenance of its basic preconditions and there is a final untouchable sphere of human freedom that is beyond interference from any public authority. So much so that, in regard to this most intimate core of privacy, no justifiable limitation thereof can take place. But this most intimate core is narrowly construed. This inviolable core is left behind once an individual enters into relationships with persons outside this closest intimate sphere; the individual's activities then acquire a social dimension and the right of privacy in this context becomes subject to limitation.¹⁰⁰⁵

In the *NM* and others *v* Smith and others (Freedom of expression institute as Amicus *Curiae*) case, referring to access to private medical information, the Constitutional court determined that the protection of privacy raises in every individual an expectation that he or she will not be interfered with.¹⁰⁰⁶ There must thus be an insistent social need for that expectation to be violated and the person's rights to privacy meddled with.¹⁰⁰⁷

The interpretation of the right to privacy in section 14 of the Constitution in the context of surrogacy means that a person acting as a surrogate's right to privacy includes the right not to have her identity or the detail of the surrogate agreement disclosed and published without her consent, as well as those of the commissioning parents. Section 302 of the Children's Act further endorses the right to privacy by providing protection to the parties involved in a surrogacy agreement, including the right of the commissioned child by determining that the identity of the child may not be revealed.¹⁰⁰⁸ The proper protection of one's privacy depends in a significant way on it being respected by others.¹⁰⁰⁹ Every person's (as a party to a surrogate agreement) right to privacy be respected by the community as a whole and especially by third parties involved in (and during) the surrogacy process having access to the parties'

¹⁰⁰⁴ Bernstein and others v Bester and others NNO at par 67.

¹⁰⁰⁵ Bernstein and others v Bester and others NNO at par 77. NM and others v Smith and others (Freedom of expression institute as Amicus Curiae) at par 34.

¹⁰⁰⁶ NM and others v Smith and others (Freedom of expression institute as Amicus Curiae) at par 45.

¹⁰⁰⁷ *NM and others v Smith and others (Freedom of expression institute as Amicus Curiae)* at par 45. The court found that there was no such compelling public interest in the case.

¹⁰⁰⁸ See par 3.5.11 for a discussion on sec 302.

¹⁰⁰⁹ *NM* and others v Smith and others (Freedom of expression institute as Amicus Curiae) at par 132.

information. Each party must provide consent to make the facts regarding the surrogacy agreement known.

2.4 The right to equality

The Constitution provides in section 9(1) that "everyone is equal before the law and has the right to equal protection and benefit of the law", whereas section 9(4) prohibits the unfair (direct and indirect) discrimination on any of the grounds stated in section 9(3) of the Constitution.¹⁰¹⁰

In the case of the National Coalition for Gay and Lesbian Equality and Another v Minister of Justice and others, the Constitutional court describes the purpose of equality jurisprudence as follows:

At the heart of equality jurisprudence is the rescuing of people from a caste-like status and putting an end to their being treated as lesser human beings because they belong to a particular group. The indignity and subordinate status may flow from institutionally imposed exclusion from the mainstream of society or else from powerlessness within the mainstream; they may also be derived from the location of difference as a problematic form of deviance in the disadvantaged group itself, as happens in the case of the disabled.¹⁰¹¹

¹⁰¹⁰ Nicholson 2013 SAJHR 503. Nicholson pointed out that the section unambiguously sets out what is meant by equality. Sloth-Nielsen "Surrogacy in South Africa" 186. The writer points out that this clause prohibits discrimination on a wide range of bases which includes gender, marital status and sexual orientation. Van der Merwe v Road Accident Fund and another (Women's Legal Centre Trust as Amicus Curiae) 2006 (4) SA 230 (CC) at par 49. "It is so that laws rarely prescribe the same treatment for everyone. Yet it bears repetition that when a law elects to make differentiation between people or classes of people it will fall foul of the constitutional standard of equality if it is shown that the differentiation does not have a legitimate purpose or a rational relationship to the purpose advanced to validate it. Absent the pre-condition of a rational connection the impugned law infringes, at the outset, the right to equal protection and benefit of the law under s 9(1) of the Constitution. This is so because the legislative scheme confers benefits or imposes burdens unevenly and without a rational criterion or basis. That would be an arbitrary differentiation which neither promotes public good nor advances a legitimate public object. In this sense, the impugned law would be inconsistent with the equality norm that the Constitution imposes, inasmuch as it breaches the 'rational differentiation' standard set by s 9(1) thereof." Hoffmann v South African Airways 2001 (1) SA 1 (CC) at par 27. The Court pointed out that "At the heart of the prohibition of unfair discrimination is the recognition that under our Constitution all human beings, regardless of their position in society, must be accorded equal dignity. That dignity is impaired when a person is unfairly discriminated against. The determining factor regarding the unfairness of the discrimination is its impact on the person discriminated against. Relevant considerations in this regard include the position of the victim of the discrimination in society, the purpose sought to be achieved by the discrimination, the extent to which the rights or interests of the victim of the discrimination have been affected and whether the discrimination has impaired the human dignity of the victim."

¹⁰¹¹ National Coalition for Gay and Lesbian Equality and Another v Minister of Justice and others at par 129.

The World Health Organisation defines infertility as a disability.¹⁰¹² South Africa's legal understanding of disability is very narrow, unfortunately, and infertility is also not specifically listed as one of the prohibited grounds of unfair discrimination in section 9(3) of the Constitution, despite "pregnancy" and "disability" being mentioned under this section. Infertility is also not listed as a ground on which a claim of unfair discrimination may be lodged under the Employment Equity Act.¹⁰¹³ Infertility should be regarded as a form of disability under section 9(3), especially if read together with the other grounds of "pregnancy" and "disability". The impossibility or permanent incapacity to fall pregnant or to conceive constitutes an impairment that affects functioning to the detriment of those affected.

Had infertility been considered a disability under South African law, it would, for example, be expected of employers to reasonably accommodate infertile employees, which would also include the obligation to provide appropriate leave for infertile employees to comply with the requirements of the surrogacy agreement. It is submitted that infertility is, however, a disability that may be alleviated through the advancements in medical technologies, including surrogacy and artificial fertilisation. There should be no distinction between "conception infertile" and "pregnancy infertile" persons, as both should be regarded as forms of disability that pose physical and psychological consequences for the infertile persons.

Some of the grounds listed in section 9(3) of the Constitution, such as gender, marital status and sexual orientation are relevant in the context of surrogacy.¹⁰¹⁴ Meyerson argues that section 294 of the Children's Act (which denies infertile persons unable to provide a genetic link to the commissioned child access to surrogacy) infringes the right to equality because the distinction it draws between fertile and infertile persons is not rationally connected to the legitimate goal of protecting the best interests of children, as well as because it successfully serves an illegitimate goal, one of forcibly imposing a contested bio-normative conception of the family on people who reasonably

¹⁰¹² WHO/International Classification of Functioning, Disability and Health (ICF) (2018); accessed from: <u>https://www.who.int/classifications/international-classification-of-functioning-disability-and-health</u> (date of access: 19 July 2022).

¹⁰¹³ 55 of 1998.

¹⁰¹⁴ Sloth-Nielsen "Surrogacy in South Africa" 186.

disagree.¹⁰¹⁵ Persons who are unable to provide the genetic material, as is a mandatory requirement in terms of section 294 of the Children's Act, find themselves in this position because of a biological abnormality or inability.¹⁰¹⁶ Thus, section 294 differentiates between the persons who are able to provide genetic material for surrogacy and those who are unable to which constitutes discrimination.¹⁰¹⁷ Section 294 may be said to unfairly discriminates against a particular group of persons, which constitutes unfair discrimination.¹⁰¹⁸ Although the purpose behind section 294 might be honourable, the nature of the infringement suffered by a particular group of infertile persons serves only to reinforce their feelings of inadequacy at being unable to procreate.¹⁰¹⁹ This then has the result of furthermore impairing their human dignity in an equally serious manner.¹⁰²⁰

Turning again to the case of *National Coalition v Minister of Justice*,¹⁰²¹ Sachs J explains that equality recognises difference in people in that it means equal concern and respect across difference.¹⁰²² Equality does not presuppose that difference be eliminated or suppressed.¹⁰²³ Thus, by respecting human rights, the 'self' is affirmed instead of denied.¹⁰²⁴ Equality does not imply a levelling or homogenisation of behaviour but rather an acknowledgment and acceptance of difference.¹⁰²⁵ The court concludes by stating that:

At the very least, it affirms that difference should not be the basis for exclusion, marginalisation, stigma and punishment. At best, it celebrates the vitality that difference brings to any society.¹⁰²⁶

 ¹⁰¹⁵ Meyerson, D "Surrogacy, Geneticism and Equality: The case of AB Minister of Social Development" 2019 *Constitutional Court Review* 9 341.
 ¹⁰¹⁶ Van Nickork 2015 *PEL* 1409

¹⁰¹⁶ Van Niekerk 2015 *PELJ* 409.

¹⁰¹⁷ Van Niekerk 2015 *PELJ* 409-410.

¹⁰¹⁸ Van Niekerk 2015 *PELJ* 411.

¹⁰¹⁹ Van Niekerk 2015 *PELJ* 411.

¹⁰²⁰ Van Niekerk 2015 *PELJ* 411.

¹⁰²¹ 1999(1) SA 6 (CC).

¹⁰²² National Coalition for Gay and Lesbian Equality and Another v Minister of Justice and others at par 132.

¹⁰²³ National Coalition for Gay and Lesbian Equality and Another v Minister of Justice and others at par 132.

¹⁰²⁴ National Coalition for Gay and Lesbian Equality and Another v Minister of Justice and others at par 132.

¹⁰²⁵ National Coalition for Gay and Lesbian Equality and Another v Minister of Justice and others at par 132.

¹⁰²⁶ National Coalition for Gay and Lesbian Equality and Another v Minister of Justice and others at par 132.

The judgment emphasises that the function of the court is to recognise the sphere which each person inhabits and not to force one person into the sphere of the other.¹⁰²⁷ The law will legitimately acknowledge a diversity of strongly held opinions on matters of great public controversy as long as there is no prejudice to the fundamental rights of any person or group.¹⁰²⁸ Further, it is the function of the Constitution and of the law to counter unfair discrimination against a minority:¹⁰²⁹

The hallmark of an open and democratic society is its capacity to accommodate and manage difference of intensely-held world views and lifestyles in a reasonable and fair manner. The objective of the Constitution is to allow different concepts about the nature of human existence to inhabit the same public realm, and to do so in a manner that is not mutually destructive and that at the same time enables government to function in a way that shows equal concern and respect for all.¹⁰³⁰

A minority of single women are "pregnancy- and conception infertile". The example of *AB* and another *v* Minister of Social Development is a case in point, as *AB* found herself in that position because she was unable to provide a gamete and maintain a pregnancy. It would have been impossible for her to have a child that is genetically linked to her.

Sachs J, in the judgment above, underlines the importance of both the Constitution and the law to protect a minority from unfair discrimination. I submit that to remedy this situation and obviate unfair discrimination, an amendment to section 294 of the Children's Act is necessary; one that does not provide the requirement of a genetic link between the commissioned child and the commissioning parents. Further, it is submitted that the construction of section 9 of the Constitution supports the conclusion that surrogacy must be treated equally to any other form of conception. Part of this assertion flows from the logical inference that surrogacy agreements must be drafted in a manner that ensures that commissioning mothers are treated equally and fairly in all respects. Thus, commissioning mothers should be treated no differently from other

¹⁰²⁷ Minister of Home Affairs and another v Fourie and another (Doctors for Life International and others Amici Curiae); Lesbian and Gay Equality project and others v Minister of Home Affairs and others 2006 (1) SA 524 (CC) at par 94.

¹⁰²⁸ Minister of Home Affairs and another v Fourie and another (Doctors for Life International and others Amici Curiae); Lesbian and Gay Equality project and others v Minister of Home Affairs and others at par 94.

¹⁰²⁹ Minister of Home Affairs and another v Fourie and another (Doctors for Life International and others Amici Curiae); Lesbian and Gay Equality project and others v Minister of Home Affairs and others at par 94. "The test, whether majoritarian or minoritarian positions are involved, must always be whether the measure under scrutiny promotes or retards the achievement of human dignity, equality and freedom."

¹⁰³⁰ *Minister v Fourie* at par 95.

mothers who are able to conceive naturally without artificial assistance. The same conclusion would apply to children born as a result of surrogacy agreements, compared to naturally conceived children born from biological mothers. That is why all the parties', including the commissioned child's, rights must be protected.

2.5 The right to reproductive health care

Section 27(1) of the Constitution determines that every person has the right to have access to health care services, including reproductive health care.¹⁰³¹ Reproductive rights have traditionally focussed on the rights of a person to access contraceptives and sterilisation procedures, together with termination of a pregnancy as per the time frames stipulated by legislation.¹⁰³² As a result of the increase in the use of assisted reproductive technologies, the focus has turned to the rights of a person to reproduce non-coitally, including the scope of their reproductive freedom.¹⁰³³ Van Niekerk describes reproductive rights as follows, explaining that reproductive rights may be inferred from other rights already in existence:

Reproductive rights embrace certain human rights that are already recognised in national laws, international human rights documents and other consensus documents. These rights rest on the recognition of the basic right of all couples and individuals to decide freely and responsibly the number, spacing and timing of their children and to have the information and means to do so, and the right to attain the highest standard of sexual and reproductive health. It also includes their right to make decisions concerning reproduction free from discrimination, coercion and violence, as expressed in human rights documents.¹⁰³⁴

Carstens and Pearmain argue that there are at least two possible reasons for the express inclusion of reproductive health care in section 27, depending on whether the reference to reproductive health care was intended to widen the concept of health care services to include services which would not ordinarily be regarded as such or whether it is a subset of health care services that was intentionally highlighted to emphasise health care for women in the area of reproduction.¹⁰³⁵ An example of health care services which would not be regarded as such generally, is the artificial fertilisation of

¹⁰³¹ Sec 27(1)(a).

¹⁰³² Van Niekerk 2017 *PELJ* 2.

¹⁰³³ Van Niekerk 2017 *PELJ* 2.

¹⁰³⁴ Van Niekerk 2017 *PELJ* 9. The existing rights include the right to make decisions regarding reproduction, the right to non-discrimination and equal treatment, the right to information and the right to the highest standard of health.

¹⁰³⁵ Carstens and Pearmain *Foundational Principles* 176.

a fertile healthy woman with the sperm of a donor because her husband is infertile.¹⁰³⁶ This procedure is clearly not designed to cure or remedy the husband's infertility and it cannot really be seen as a health care service to the wife as there is nothing wrong with her.¹⁰³⁷ Thus, in human rights terms, it is a medical procedure, which is assisting the couple to exercise their reproductive rights.¹⁰³⁸ The writers rightfully state that reproductive rights are not gender-specific and that these rights are of fundamental importance to both women and men.¹⁰³⁹

In the Constitutional Court judgment of *Soobramoney v Minister of Health (KwaZulu-Natal)*,¹⁰⁴⁰ the focus was on section 27(3) of the Constitution, 1996, and the right not to be refused medical treatment.¹⁰⁴¹ For Chaskalson J, it was apparent from the provisions in section 27 that the right of access to health care services is dependent upon the resources that are available for such purpose.¹⁰⁴² The corresponding right, however, is limited as a result of lack of resources.¹⁰⁴³ Emergency medical treatment may not be denied to a person and the section requires that necessary and available remedial treatment be given to a person to prevent harm.¹⁰⁴⁴

The appellant in this matter, who suffered from chronic renal failure, was dependent on kidney dialysis two to three times a week in order to stay alive.¹⁰⁴⁵ Chaskalson J correctly observes that section 27(3) (which provides for the right not to be refused emergency medical treatment) did not apply to the facts as the appellant's ongoing renal failure was as a result of the deterioration of incurable renal function.¹⁰⁴⁶ Sections 27(1) and 27(2) entitle a person to have access to health care services provided by the

¹⁰³⁶ Carstens and Pearmain *Foundational Principles* 176.

¹⁰³⁷ Carstens and Pearmain *Foundational Principles* 176.

¹⁰³⁸ Carstens and Pearmain *Foundational Principles* 176.

¹⁰³⁹ Carstens and Pearmain *Foundational Principles* 178.

¹⁰⁴⁰ (CCT32/97) [1997] ZACC 17; 1998 (1) SA 765 (CC); 1997 (12) BCLR 1696. (Hereinafter "Soobramoney v Minister of Health").

¹⁰⁴¹ Soobramoney v Minister of Health at par 7. At par 38 Madala J explained that the provision of sec 27(3) envisages a dramatic, sudden situation or event which is of a passing nature in terms of time and "there is some suddenness and at times even an element of unexpectedness in the concept of "emergency medical treatment".

¹⁰⁴² Soobramoney v Minister of Health at par 11. The court also referred to sec 26 which dealt with the right to housing.

¹⁰⁴³ Soobramoney v Minister of Health at par 11.

¹⁰⁴⁴ Soobramoney v Minister of Health at par 20.

¹⁰⁴⁵ Soobramoney v Minister of Health at par 21.

¹⁰⁴⁶ Soobramoney v Minister of Health at par 21.

state and which are within the state's available resources.¹⁰⁴⁷ The state however has to manage its limited resources in order to address the aspect of access to housing, food and water, employment opportunities and social security, all aspects of the right to human life.¹⁰⁴⁸ Madala J emphasises that the state has a strong interest in protecting and preserving its citizens life and health and thus must do all in its power to protect and preserve life.^{*1049} In addition, the court explains as follows:

The Constitution is forward-looking and guarantees to every citizen fundamental rights in such a manner that the ordinary person-in-the-street, who is aware of these guarantees, immediately claims them without further ado and assumes that every right so guaranteed is available to him or her on demand. Some rights in the Constitution are the ideal and something to be strived for. They amount to a promise, in some cases, and an indication of what a democratic society aiming to salvage lost dignity, freedom and equality should embark upon. They are values which the Constitution seeks to provide, nurture and protect for a future South Africa.¹⁰⁵⁰

The guarantees set out in the Constitution are not absolute and may be limited in one way or another.¹⁰⁵¹ Unfortunately, one of the limiting factors in the fulfilment of the Constitution's guarantees is that of limited or scarce resources.¹⁰⁵² Sachs J describes the dilemma as follows:

The inescapable fact is that if governments were unable to confer any benefit on any person unless it conferred an identical benefit on all, the only viable option would be to confer no benefit on anybody.¹⁰⁵³

The decisions involving the rationing of health care resources are likely to place treatment for infertility somewhere at the bottom of the hierarchy of health care services that must be provided.¹⁰⁵⁴ It is for this reason that assisted reproduction techniques such as artificial fertilisation and *in vitro* fertilisation are not often considered part of the standard package of health care services in either the public or the private health sectors, although they are readily available in the private health sector.¹⁰⁵⁵

The judgment of the Constitutional Court in the matter of the *Minister of Health and* others *v Treatment Action Campaign and others*¹⁰⁵⁶ provides some explanation in that

¹⁰⁴⁷ Soobramoney v Minister of Health at par 22.

¹⁰⁴⁸ Soobramoney v Minister of Health at par 31.

¹⁰⁴⁹ Soobramoney v Minister of Health at par 39.

¹⁰⁵⁰ Soobramoney v Minister of Health at par 42.

¹⁰⁵¹ Soobramoney v Minister of Health at par 43.

¹⁰⁵² Soobramoney v Minister of Health at par 43.

¹⁰⁵³ Soobramoney v Minister of Health at par 53.

¹⁰⁵⁴ Carstens and Pearmain Foundational Principles 180.

¹⁰⁵⁵ Carstens and Pearmain *Foundational Principles* 180.

¹⁰⁵⁶ (No 2) 2002 (5) SA 721 (CC).

the court made it clear that the socio-economic rights in the Constitution should not be interpreted as entitling everyone to demand that the minimum care be provided to them.¹⁰⁵⁷ The state is not obliged to go beyond the available resources or to realise these rights immediately.¹⁰⁵⁸ This is succinctly explained by the court in the case of the *Minister of Health and others v Treatment Action Campaign and others:*

There is, accordingly, a distinction between the self-standing rights in ss 26(1) and 27(1), to which everyone is entitled, and which in terms of s 7(2) of the Constitution '(*t*)he State must respect, protect, promote and fulfil', and the independent obligations imposed on the State by ss 26(2) and 27(2). This minimum core might not be easy to define, but includes at least the minimum decencies of life consistent with human dignity. No one should be condemned to a life below the basic level of dignified human existence.¹⁰⁵⁹

It would be impossible for the state to provide everyone access even to a "core" package of services immediately.¹⁰⁶⁰ Thus, as per the *Minister of Health and others v Treatment Action Campaign and others* judgment, all that can be expected from the state is that the state acts reasonably to provide access to the socio-economic rights identified in both sections 26 and 27, *on a progressive basis*.¹⁰⁶¹ It was also held in this judgment that section 27(1) of the Constitution does not give rise to a self-standing and independent positive right enforceable regardless of the considerations mentioned in section 27(2).¹⁰⁶² It is important to read sections 27(1) and 27(2) together, as such a reading would define the scope of the positive rights that every person has, together with the corresponding obligations on the state to respect, protect, promote and fulfil such rights.¹⁰⁶³

The rights conferred by sections 26(1) and 27(1) are for a person to have access to the services that the state is obliged to provide in terms of sections 26(2) and 27(2).¹⁰⁶⁴ It has long been recognised that public health services in South Africa are over extended and that the state already faces massive demands in respect of access to education, land, housing, health care, food, water and social security.¹⁰⁶⁵ However, despite this, the state is obliged to take reasonable and legislative measures to achieve

¹⁰⁵⁷ Minister of Health and others v Treatment Action Campaign and others at par 34.

¹⁰⁵⁸ Minister of Health and others v Treatment Action Campaign and others at par 32.

¹⁰⁵⁹ Minister of Health and others v Treatment Action Campaign and others at par 28.

¹⁰⁶⁰ Minister of Health and others v Treatment Action Campaign and others at par 35.

¹⁰⁶¹ Minister of Health and others v Treatment Action Campaign and others at par 35.

¹⁰⁶² Minister of Health and others v Treatment Action Campaign and others at par 39.

¹⁰⁶³ Minister of Health and others v Treatment Action Campaign and others at par 39.

¹⁰⁶⁴ Minister of Health and others v Treatment Action Campaign and others at par 39.

¹⁰⁶⁵ Minister of Health and others v Treatment Action Campaign and others at par 93-94.

the progressive realisation of each of these socio-economic rights as they are entrenched in the Constitution.¹⁰⁶⁶ It is important to understand that the measures taken by the state only need to be *within its available resources*.¹⁰⁶⁷

The judgments referred to in this section support the conclusion that reproductive health care would not fall within the ambit of the 'core' services, as it is not a type of treatment that can save a life, despite its ability to enable new life because of medical technology. The submission by Carstens and Pearmain that fertility treatment will be at the bottom of the hierarchy of priorities, is correct. The many competing demands on the state's health care resources are already completely inadequate to provide in the needs of those who are seriously ill or continuously dependent on the state to provide them with the necessary treatment.

2.6 Children's rights and the protection of children

Artificial fertilisation concerns not only the desire to conceive a child, but more importantly the conception of a new person whose interests need to be protected:

In participating in the process of procreation, the doctor is performing an act which is not morally neutral. Even if the principal objective of DI is to satisfy the desire for a child, thought has to be given to the child who is the end result; most would hold that the child's interests should be considered paramount – and that this holds throughout the spectrum of assisted reproduction.¹⁰⁶⁸

Section 28 of the Constitution protects the rights of children in different subsections.¹⁰⁶⁹ The most relevant rights regarding children for the purpose of this chapter are the right of children to: a name and a nationality from birth;¹⁰⁷⁰ to family care or parental care, or to appropriate alternative care when removed from the family environment;¹⁰⁷¹ to basic nutrition, shelter, basic health care services and social

¹⁰⁶⁶ Minister of Health and others v Treatment Action Campaign and others at par 94.

¹⁰⁶⁷ Minister of Health and others v Treatment Action Campaign and others at par 94.

¹⁰⁶⁸ Mason, McCall Smith and Laurie *Law and Medical Ethics* 78.

 ¹⁰⁶⁹ The Constitution. A child is a person under the age of 18 years. *Ex parte Applications for Confirmation of Surrogate Motherhood Agreements* [2014 JOL 32134 (GSJ)] at para 16.
 ¹⁰⁷⁰ Sec 28(1)(a).

¹⁰⁷¹ Sec 28(1)(b). Du Toit and another v Minister of Welfare and population development and others (Lesbian and gay equality project as *Amicus Curiae*) 2003 (2) SA 198 (CC) (hereinafter "*Du Toit v Minister*") at par 18. It was recognised that many children are not brought up by their biological parents.

services;¹⁰⁷² and to be protected from maltreatment, neglect, abuse or degradation.¹⁰⁷³ Most importantly, a child's best interests are of paramount importance in every matter concerning the child.¹⁰⁷⁴

The Constitutional Court has emphasised the right of a child to proper parental care.¹⁰⁷⁵ The obligation to ensure that a child is properly cared for is an obligation imposed by the Constitution on the child's parents. The inverse of this is an obligation placed on the state to create the necessary environment for parents to do so.¹⁰⁷⁶ In the case of *S v M*, Sachs J, with reference to South Africa's status as a State party to the United Nations Convention on the Rights of the Child (UNCRC) and the four primary principles of the UNCRC (survival, development, participation and protection) that guide all policy in South Africa relating to children, observes as follows:

What unites these principles, and lies at the heart of s 28, I believe, is the right of a child to be a child and enjoy special care. Every child has his or her own dignity. If a child is to be constitutionally imagined as an individual with a distinctive personality, and not merely as a miniature adult waiting to reach full size, he or she cannot be treated as a mere extension of his or her parents, umbilically destined to sink or swim with them.¹⁰⁷⁷

It is difficult to pinpoint or identify the best interests of a child that will be born as a result of a surrogacy agreement because such an exercise would need to consider the competing rights and obligations of all the parties to the agreement, including a range of ethical and moral issues and choices relating to unborn children.¹⁰⁷⁸ Decisions that have consequences for children should be well considered by all the parties to ensure that decisions made do not adversely affect the child's health and well-being.¹⁰⁷⁹

Differing views on how to unpack the 'best interest' standard regarding children exist. Sachs J alludes to this by lamenting the fact that the very extent of the paramountcy

¹⁰⁷² Sec 28(1)(c).

¹⁰⁷³ Sec 28(1)(d).

¹⁰⁷⁴ Sec 28(2). Bannatyne v Bannatyne (Commission for gender equality, as Amicus Curiae) 2003 (2) SA 363 (CC) at par 17 and 24. Sec 28(2) enjoins a court to give paramountcy to the best interests of the child in every matter concerning the child; Du Toit v Minister at par 20. The Court stated that both international law and the domestic law of many countries have affirmed the paramountcy of the best interests of the child.

¹⁰⁷⁵ Bannatyne v Bannatyne at par 24.

¹⁰⁷⁶ Bannatyne v Bannatyne at par 24.

¹⁰⁷⁷ S v M at par 17 – 18.

¹⁰⁷⁸ Oluwaseyi and Oladimeji 2021 *Obiter* 27.

¹⁰⁷⁹ Oluwaseyi and Oladimeji 2021 *Obiter* 27. The authors point out the importance of the intending (commissioning) parents to be appropriately examined so that their willingness and commitment to safeguarding their children's rights are confirmed. Thus, failure to make investigations concerning the background of the commissioning parents undermines the best interests principle.

principle creates the risk of appearing to promise everything in general, but actually delivering little in particular.¹⁰⁸⁰ The concept of the best interests is often criticised for being fundamentally indeterminate and for providing little guidance to those who are given the task to apply it.¹⁰⁸¹ Some authors argue that the best interests of a child are protected by the prohibition of commercial surrogacy.¹⁰⁸² For Nicholson, legislative provisions that prohibit certain types of surrogacy infringe upon an infertile person's reproductive rights, the right to dignity and the right to privacy.¹⁰⁸³ She further argues that it may also violate the surrogate's right to an improved life, her right to freedom and control of her body and her right to choose a trade, occupation or profession.¹⁰⁸⁴ The Constitutional Court in the case of $S v M^{1085}$ cautions that the list of factors to be considered the most important regarding a child's best interests may be endless, yet should not be seen as a problem, as a truly "principled child-centered approach"

situation of the child in question:

Furthermore '(t)he list of factors competing for the core of best interests [of the child] is almost endless and will depend on each particular factual situation'. Viewed in this light, indeterminacy of outcome is not a weakness. A truly principled child-centred approach requires a close and individualised examination of the precise real-life situation of the particular child involved. To apply a predetermined formula for the sake of certainty, irrespective of the circumstances, would in fact be contrary to the best interests of the child concerned.¹⁰⁸⁶

Louw, on the other hand, maintains that the child-centred approach is not possible as it is generally agreed that the determination of an existing child's best interests cannot be determined in the abstract or in advance.¹⁰⁸⁷ She further asserts that the only

¹⁰⁸⁰ S *v M* at par 23.

¹⁰⁸¹ S v M at par 23.

¹⁰⁸² Bonthuys and Broeders 2013 *SALJ* 490.

¹⁰⁸³ Nicholson 2013 SAJHR 503.

¹⁰⁸⁴ Nicholson 2013 *SAJHR* 503. Nicholson pointed out that fully informed adults should be able to make any arrangement regarding their bodies and reproduction that they wish, provided that their decision do not harm the children born as a result of the arrangement. She argues that the practice may lead to desperate women entering into surrogacy agreements for very little financial consideration and rendering them vulnerable to the type of exploitation that the statute seeks to avoid. The writer argues that the pervasive poverty that prevails in South Africa makes for an environment which compromises the ability of destitute women to forego an opportunity to make some money, exposing them to exploitation by the wealthy. This could compromise the surrogate's dignity and, in extreme cases, amount to her engaging in a form of forced labour or slavery. On the other hand, if carrying a child for another is an altruistic labour of love, there should be no objection to the surrogate's actions and they should be encouraged.

¹⁰⁸⁵ 2008 (3) SA 232 (CC).

¹⁰⁸⁶ S v M at par 24.

¹⁰⁸⁷ Louw 2013 *THRHR* 573. The checklist of factors set out in the Children's Act to provide guidance in such cases cannot be applied.

practical way in which to determine the best interests of a resultant child (the child to be born), is to ensure that the surrogate mother and the commissioning parent(s) are suitable persons to assume their respective roles, and that all reasonably foreseeable eventualities have been provided for in the agreement.¹⁰⁸⁸ Thus, the test to determine the best interests of the child to be born is also child-centred, but in an adapted way.¹⁰⁸⁹ For Louw, courts should not deny people their right to make procreative choices based on what the legislator or the court presumes would be in the best interests of the child born as a result of such choices.¹⁰⁹⁰

Sloth-Nielson views the language of section 28 of the Constitution as comprehensive and emphatic,¹⁰⁹¹ and that statutes dealing with children must be interpreted (and the common law developed) in a way that favours the protection and advancement of children's interests.¹⁰⁹² The best interests of a child is a constitutional right, a rule of procedure and a principle.¹⁰⁹³ Since section 28(2) of the Constitution mandates a court to ensure that the best interests of a child is protected, it is imperative that the court must have sufficient information at its disposal to enable it to understand the impact of its decision on the child(ren).¹⁰⁹⁴ Thus, according to Sloth-Nielson, the 'best interests' safeguards children's rights against arguments based purely on legal technicalities and which do not take into account the children's individual circumstances.¹⁰⁹⁵ She concludes that although the courts have repeatedly stated that "best interest" is not a trump over all rights, "it remains a right which is accorded a degree of privilege in the balancing of rights".¹⁰⁹⁶

¹⁰⁸⁸ Louw 2013 *THRHR* 573. The writer indicates that consideration must be given to, for example, what is to happen in the case of the death, divorce or separation of the commissioning couple or where the child is born with disabilities.

¹⁰⁸⁹ Louw 2013 *THRHR* 573. The writer concludes by stating that the court should endeavour to ensure that when the child is eventually born, his or her interests were protected as far as was reasonably possible at the time of confirmation of the agreement given the information available to the court.

¹⁰⁹⁰ Louw 2013 *THRHR* 573.

¹⁰⁹¹ Sloth-Nielsen, J "Children's rights Jurisprudence in South Africa – a 20 year retrospective" 2019 *De Jure Law Journal* 52 510.

¹⁰⁹² Sloth-Nielsen 2019 *De Jure* 510-511. The author states that this has continued to occur and she gave the example of where the genetic link requirement was uphold for the confirmation of a valid surrogacy agreement on the ground of children's rights, and more specifically the rights of the unborn child.

¹⁰⁹³ Sloth-Nielsen 2019 *De Jure* 516.

¹⁰⁹⁴ Sloth-Nielsen 2019 *De Jure* 516.

¹⁰⁹⁵ Sloth-Nielsen 2019 *De Jure* 516.

¹⁰⁹⁶ Sloth-Nielsen 2019 *De Jure* 517.

The paramountcy principle relating to the best-interest standard raises a concern to which Sachs J alludes in S v M. If the paramountcy principle is spread too thin, it risks being converted from an effective instrument of child protection into an empty rhetorical phrase of weak application which means that the objective of section 28(2) will be defeated and not promoted.¹⁰⁹⁷ The 'how to' apply the paramountcy principle in a meaningful way, without unduly destroying other valuable and constitutionally protected interests, can be seen as a challenge.¹⁰⁹⁸

Turning now to the right to a legal identity in the context of section 28(1)(a), Rosenberg reminds us that the right to a legal identity is an extension of the right to human dignity and that the right to an identity has important psychological and emotional substance in that a name connects a child to his or her family.¹⁰⁹⁹ Thus, for Rosenberg, the lack of knowledge of one's origins does not necessarily deprive one of a legal identity as legal identity is not necessarily biological identity.¹¹⁰⁰ The court in *AB and another v Minister of Social Development*¹¹⁰¹ had to determine what is in the best interests of the child in terms of section 28(2) of the Constitution.¹¹⁰² The court asserts that section 294 of the Children's Act protects the child by ensuring that a genetic link exists when that child is conceived.¹¹⁰³ Rosenberg regards this as a recognition by the court that the child's right to knowledge of his or her genetic origins is in the child's best interests.¹¹⁰⁴ She further contends that the right in section 28(2) of the Constitution stands separately from the other rights in section 28(1) and thus serves as a standard against which all law and conduct affecting children should be tested.¹¹⁰⁵

2.7 Miscellaneous constitutional arguments relating to artificial fertilisation and surrogate agreements

¹⁰⁹⁷ *S v M* at par 25.

¹⁰⁹⁸ *S v M* at par 25.

¹⁰⁹⁹ Rosenberg 2020 *TSAR* 733.

¹¹⁰⁰ Rosenberg 2020 *TSAR* 724.

¹¹⁰¹ (CCT155/15) [2016] ZACC 43; 2017 (3) BCLR 267 (CC); 2017 (3) SA 570 (CC).

¹¹⁰² The case will be discussed in detail below.

¹¹⁰³ AB and another v Minister of Social Development at par 294.

¹¹⁰⁴ Rosenberg 2020 *TSAR* 735. At 736 the author stated that if it is not proven that not knowing one's genetic origins to cause psychological harm to a child, it is submitted that there is no infringement of the best interests of the child in not knowing his or her genetic origins.

¹¹⁰⁵ Rosenberg 2020 *TSAR* 735.

Carstens and Pearmain aver that, constitutionally speaking, a single woman who wants to have a child without being involved in a sexual relationship, has a constitutional right to artificial fertilisation, provided that she can afford the procedure and that the donation of the sperm is in conformity with legal provisions concerning the control of human tissue and gametes.¹¹⁰⁶ Thus, the right to reproduce attaches to individuals and not only to pairs of individuals, and is based on the constitutional right to freedom and security of the person, which is a very personal and private right in terms of its exercise.¹¹⁰⁷

If one considers the definition of what a surrogate mother is, thinking about surrogate motherhood in constitutional terms, may be construed as the exercise of the right to freedom and security of the person for the benefit of a third party.¹¹⁰⁸ Carstens and Pearmain raise the interesting question whether a surrogacy agreement would be valid to the extent that surrogate motherhood constitutes a waiver of the surrogate's right to freedom and security of the person (this depends on the terms of the agreement as she may agree to give up some of her rights to make decisions concerning her pregnancy).¹¹⁰⁹ It was submitted by the writers that the question of whether a person can agree to waive, or more appropriately, limit their constitutional rights and freedoms, depends very much on the values promoted by the Constitution and public policy.¹¹¹⁰ It further depends on the nature and extent of the limitation and the nature of the right.¹¹¹¹ The authors furthermore conclude that:

In the final analysis surrogate motherhood is about the deployment of existing technologies in a manner which impacts upon social values and public policy rather than the utilisation of developing technologies to artificially create a human being.¹¹¹²

Surrogate motherhood is also not about reproductive care but rather the extent of reproductive rights.¹¹¹³ The same authors argue that surrogate motherhood is best situated within the context of section 12 of the Constitution.¹¹¹⁴ It is further argued that,

¹¹⁰⁶ Carstens and Pearmain *Foundational Principles* 182.

¹¹⁰⁷ Carstens and Pearmain Foundational Principles 182.

¹¹⁰⁸ Carstens and Pearmain *Foundational Principles* 182.

¹¹⁰⁹ Carstens and Pearmain *Foundational Principles* 183.

¹¹¹⁰ Carstens and Pearmain *Foundational Principles* 183. This was a general question posed by the writers relating to the possibility of waiver of a person's constitutional rights rather than being related to surrogate motherhood specifically.

¹¹¹¹ Carstens and Pearmain Foundational Principles 183.

¹¹¹² Carstens and Pearmain *Foundational Principles* 184.

¹¹¹³ Carstens and Pearmain *Foundational Principles* 185.

¹¹¹⁴ Carstens and Pearmain *Foundational Principles* 185.

as a legal topic, surrogate motherhood does not fall within the scope of reproductive health care services, since a pregnant mother, as a rights holder, is in any event entitled to reproductive care in terms of the Constitution and whether she is a surrogate or not is irrelevant.¹¹¹⁵

Assisted reproductive techniques used in surrogacy, such as AID and *in vitro* fertilisation, for example, are also more suited to discussions of section 12-rights than section 27-rights. Should they fall within the ambit of section 12-rights, it follows logically that they must in principle be included within the scope of the rights of access to reproductive health care, bearing in mind the internal limitation in section 27(2) of the Constitution which refers to the progressive realisation of rights within the scope of available state resources.¹¹¹⁶

2.8 The constitutionality of selected provisions in chapter 19 of the Children's Act

It has been argued that the state's legitimate interest in the regulation of surrogate agreements is adequately compelling to suggest that most aspects of chapter 19 will survive constitutional scrutiny.¹¹¹⁷ Two possible exceptions are section 294, which is the genetic link requirement, and section 297(1)(b) which makes provision for the surrogate mother to hand over the child to the commissioning parent(s) as soon as reasonably possible.¹¹¹⁸

The genetic link requirement, as alluded to earlier in the thesis, refers to the use of the gametes of only the commissioning parent(s) for the artificial fertilisation of the surrogate mother.¹¹¹⁹ Thus, for a valid surrogate agreement, the gametes can be from either one or both of the commissioning parents and in the case of a single person, from that person.¹¹²⁰ The result of this provision is that if the commissioning parents, or in the case of a single person, that person, are biologically or medically unable to use their gametes, they would not be able to enter into a valid surrogate agreement.¹¹²¹

¹¹¹⁵ Carstens and Pearmain *Foundational Principles* 185.

¹¹¹⁶ Carstens and Pearmain *Foundational Principles* 185.

¹¹¹⁷ LexisNexis *Bill of Rights Compendium* at par 3E28.

¹¹¹⁸ LexisNexis *Bill of Rights Compendium* at par 3E28.

¹¹¹⁹ LexisNexis *Bill of Rights Compendium* at par 3E28.

¹¹²⁰ LexisNexis *Bill of Rights Compendium* at par 3E28.

¹¹²¹ LexisNexis *Bill of Rights Compendium* at par 3E28.

The constitutionality of section 294 was challenged in the High Court case of *AB Surrogacy Advisory Group v Minister of Social Development*¹¹²² on the ground that it unjustifiably infringes upon the constitutional rights to equality, dignity, reproductive autonomy, privacy and access to health care of persons.¹¹²³ Section 294 was further challenged in the matter of *KB v Minister of Social Development*¹¹²⁴ on the ground that the provision does not provide for a genetic link between siblings, but only requires a genetic link with one or both of the parents.

Section 40 of the Children's Act was recently challenged in the High Court case of *EJ* and others v Haupt NO^{1125} on the basis that sections 40(1) and 40(3)(b) unfairly discriminate against a same-sex female couple.¹¹²⁶ Section 40 was furthermore successfully challenged in the unreported judgment of *VVJ v Minister of Social Development*¹¹²⁷ on the ground that it excludes all unmarried life partners.

Although section 297 provides that the surrogate mother must hand the child over to the commissioning parent(s) as soon as reasonably possible, it has been argued that it is unlikely that a court would be able to enforce such an obligation without risking violating the right to dignity of the child and the right to privacy of the surrogate mother.¹¹²⁸

The next section will explore, in more detail, the Constitutional Court judgment in the matter of *AB Surrogacy Advisory Group v Minister of Social Development.*

2.8.1 AB and another v Minister of Social Development¹¹²⁹

¹¹²² Case no 40658/2013.

¹¹²³ AB and another v Minister of Social Development at par 12. LexisNexis Bill of Rights Compendium at par 3E28.

¹¹²⁴ Mpumalanga High Court, Mbombela case no 966/2022. (At the time of writing the thesis the judgment was not yet available.)

¹¹²⁵ 2022 (1) SA 514 (GP).

¹¹²⁶ EJ and others v Haupt NO at par 42.

¹¹²⁷ Unreported case, North Gauteng High Court, Pretoria, case no 27706/2021.

¹¹²⁸ LexisNexis *Bill of Rights Compendium* at par 3E28.

¹¹²⁹ 2017 (3) SA 570 CC; (2017 (3) BCLR 267; [2016] ZACC 43). High Court case reference [2015] ZAGPPHC 580; [2015] 4 All SA 24 (GP); 2015 (10) BCLR 1228 (GP); 2016 (2) SA 27 (GP) (12 August 2015).

AB, the first applicant ("AB"), an adult female, wished to have a child of her own by entering into a surrogacy agreement.¹¹³⁰ She underwent 18 unsuccessful in vitro fertilisation cycles (IVF).¹¹³¹ She is permanently and irreversibly infertile in two different ways. Firstly, she is unable to contribute her own gametes for the conception and artificial fertilisation process, and secondly, she is unable to carry a pregnancy to term.¹¹³² Thus, the only way for AB to proceed with surrogacy was to use both donor ova and donor sperm.¹¹³³ She was, however, made aware of the requirement in section 294 of the Children's Act that, as a single woman, she would need to provide her own gamete for a valid surrogacy agreement to be concluded.¹¹³⁴ Thus, as a result of the fact that she was unable to donate a gamete, she was also unable to enter into a valid surrogacy agreement.¹¹³⁵ As a result of her inability to conclude a surrogacy agreement, she brought an application in the High Court of South Africa, Gauteng Division, Pretoria, seeking an order declaring section 294 of the Children's Act inconsistent with the Constitution and thus invalid.¹¹³⁶ AB's constitutional challenge (together with that of the Surrogacy Advisory Group) was based on the argument that section 294 violates the rule of law, as well as the rights to equality, human dignity, reproductive autonomy, privacy and the right of access to health care services of the surrogate mother.¹¹³⁷ The Minister, as respondent, opposed the application on the following grounds: (a) the prospective child has the right to know its genetic origins; (b) South Africa's adoption process already caters for AB's need to have a child; (c) and to allow a single infertile person to create a child with no genetic link to her would result in the creation of a designer child, which would not be in the public interest; and, finally, that (d) commercial surrogacy is prevented by section 294.1138

¹¹³⁰ AB and another v Minister of Social Development at par 5.

¹¹³¹ At par 8. Term is discussed in ch 1 at par 1.2.5. *AB and another v Minister of Social Development* at par 242. 16 of the IVF cycles were done with embryos that had no genetic link to her and 14 used both male and female anonymous donor gametes. SALRC Project 140 89.

¹¹³² AB and another v Minister of Social Development at par 5, 8 and par 18. Conception infertile and pregnancy infertile is discussed in par 2.1. Rautenbach 2017 TSAR 352.

¹¹³³ AB and another v Minister of Social Development at par 9.

¹¹³⁴ AB and another v Minister of Social Development par 9, par 10 and par 242. Thaldar 2018 De Rebus 28.

¹¹³⁵ At par 10.

¹¹³⁶ AB and another v Minister of Social Development at par 11 Thaldar 2018 De Rebus 29. Boniface 2017 TSAR 193.

¹¹³⁷ AB and another v Minister of Social Development at par 12. Rautenbach 2017 TSAR 352.

¹¹³⁸ AB and another v Minister of Social Development par 12. The Surrogacy Group sought an order declaring sec 294 inconsistent with the Constitution and thus invalid as it asserted that none of the grounds mentioned by the Minister offered sufficient justification for the retention of the offending provision.

The High Court hearing this matter concluded that section 294 of the Children's Act unjustifiably violates AB's rights to equality, human dignity, reproductive autonomy, privacy and access to health care,¹¹³⁹ and declared section 294 constitutionally invalid.¹¹⁴⁰ The matter was thereafter heard by the Constitutional Court on 1 March 2016. Considering the importance of the Constitutional Court judgment and the different aspects dealt with by the court, the judgment will be discussed in some detail. The minority judgment by Khampepe J (with Cameron J, Froneman J and Madlanga J)¹¹⁴¹ is discussed first, followed by the majority judgment by Nkabinde J (with Mogoeng CJ, Moseneke DCJ, Bosielo AJ, Jafta J, Mhlanta J and Zondo J).

2.8.1.1 The minority judgment

The minority decision identified the pertinent issues as the following:¹¹⁴² (a) the historical and legislative framework of surrogacy and the effect and purpose of section 294 placed within that framework; (b) whether section 294 (on proper interpretation) limits AB's rights to psychological integrity, human dignity, equality, privacy and access to reproductive health care;¹¹⁴³ and (c) if so, whether it has been shown that section 294 constitutes a reasonable and justifiable limitation of AB's rights; and (d) if not, what the appropriate remedy would be.

Looking at the historical legislative framework, the court observes that surrogacy was not expressly regulated by any legislation in South Africa before the enactment of chapter 19 of the Children's Act,¹¹⁴⁴ despite the fact that surrogacy is not new.¹¹⁴⁵

¹¹³⁹ AB and another v Minister of Social Development par 15. SALRC Project 140 90.

AB and another v Minister of Social Development par 15. SALRC Project 140 89. Florescu, S and Sloth-Nielsen, J "Visions on surrogacy – From North to South: The approach of the Netherlands and South Africa to the issue of surrogacy and the child's right to know his origin" 2017 International Survey of Family Law 2017 249.

At par 4 Khampepe J pointed out that at the heart of this matter lies the question of the extent to which the state may regulate the reproductive opportunities available to those who are unable to have children of their own because they are conception and pregnancy infertile.

¹¹⁴² AB and another v Minister of Social Development par 33.

¹¹⁴³ Florescu, S and Sloth-Nielsen, J "Visions on surrogacy – From North to South: The approach of the Netherlands and South Africa to the issue of surrogacy and the child's right to know his origin" 2017 International Survey of Family Law 2017 250.

¹¹⁴⁴ AB and another v Minister of Social Development par 34.

¹¹⁴⁵ AB and another v Minister of Social Development par 35. The court gave the example of the story of Abram, Sarai and Hagar where Hagar, as Abram and Sarai's servant, was used to bear Sarai's children as Sarai was unable to bear her own child. The other example is of Jacob and his wives Rachel and Leah. Rachel could not bear her own children and she told Jacob to sleep with Bilhah,

Khampepe J, considering whether section 294 limits constitutional rights, begins by examining the constitutional values as interpretative aids through which the meaning of the constitutional rights may be better understood.¹¹⁴⁶

With reference to the value of freedom, the court states that one of the constitutional values in section 1 of the Constitution includes the advancement of human rights and freedoms.¹¹⁴⁷ To be autonomous is to be socially and politically connected rather than be an agent of unfettered individual choice.¹¹⁴⁸ Autonomy is a necessary but socially embedded part of the value of freedom.¹¹⁴⁹ The value of freedom is animated by the recognition of each person's distinctive aptitude to understand and act on their own desires and beliefs.¹¹⁵⁰ By exercising our capacity to assess our own socially-rooted situations, we define our nature, give meaning and coherence to our lives and take responsibility for the kind of people that we are:¹¹⁵¹

Our Constitution actively seeks to free the potential of each person; a goal which can only be achieved through a deep respect for the choices each of us makes.¹¹⁵²

Discussing the right of freedom and security of a person, Khampepe J explains that the freedom protected by the right is not coextensive with autonomy.¹¹⁵³ She further explains that the court has adopted a purposive and contextual approach to interpreting the Constitution and in doing so, the court is enjoined to provide a broad and generous reading in determining the ambit of constitutionally protected rights.¹¹⁵⁴ Section 12(1) provides a general right to freedom grounded in bodily security whereas sections 12(2) and 12(2)(a) provide for a new, freestanding and definitionally-

the servant, to bear children for her. SALRC Project 140 80 and 177. Oluwaseyi and Oladimeji 2021 *Obiter* 20. Iannacci, B.R "Why New York should legalize surrogacy: A comparison of surrogacy legislation in other states with current proposed surrogacy legislation in New York" 2018 *Touro Law Review* 34(4) 1244.

¹¹⁴⁶ AB and another v Minister of Social Development at par 49.

¹¹⁴⁷ AB and another v Minister of Social Development at par 50.

AB and another v Minister of Social Development at par 51.

¹¹⁴⁹ AB and another v Minister of Social Development at par 51.

¹¹⁵⁰ AB and another v Minister of Social Development at par 52. "The value recognises the inherent worth of our capacity to assess our own socially-rooted situations and make decisions on this basis."

¹¹⁵¹ AB and another v Minister of Social Development at par 52.

¹¹⁵² AB and another v Minister of Social Development at par 52.

¹¹⁵³ AB and another v Minister of Social Development at par 53.

¹¹⁵⁴ AB and another v Minister of Social Development at par 54. Further, because of the inclusion of sec 36 limitation clause, rights should not be interpreted in a miserly fashion.

proscribed freedom right.¹¹⁵⁵ Section 12 is consequently an amalgam of freedom and security rights, brought together because both sets protect persons' ability to lead their lives without being subject to certain constitutionally prohibited impediments.¹¹⁵⁶ Section 12(2)(a) should be given as broad a reading similar to other rights, despite it forming part of the general entitlement to freedom and security of the person.¹¹⁵⁷

Whereas section 12(1) protects specific physical freedoms along with a residual right to freedom and by contrast, section 12(2) is a freestanding freedom right that should be interpreted broadly on its own terms.¹¹⁵⁸ A close connection exists between the freedoms protected by the Constitution and integrity, as we are actively turning away from indifference and moving towards respect, empathy and compassion as a result of the Constitution.¹¹⁵⁹ The section 12(2) protection is grounded in these ideals.¹¹⁶⁰ The right to protection of bodily and psychological integrity is especially important for women who may, for example, decide to terminate a pregnancy in appropriate circumstances.¹¹⁶¹ Section 12(2)(c) protects an individual against medical or scientific experiments without informed consent and this proposes that section 12(2) should be interpreted generously to cover all instances where the bodily or psychological integrity of a person is damaged or compromised.¹¹⁶² Thus, the emphasis in section 12(2) is on whether a law or a conduct deprives a person of freedom or security, broadly understood.¹¹⁶³

¹¹⁵⁵ AB and another v Minister of Social Development at par 59. Sec 12(2) introduces a new freedom right akin to those enumerated elsewhere in the Bill of Rights.

¹¹⁵⁶ \overrightarrow{AB} and another v Minister of Social Development at par $\overrightarrow{61}$.

¹¹⁵⁷ AB and another v Minister of Social Development at par 62.

¹¹⁵⁸ AB and another v Minister of Social Development at par 63. At par 64 it is further stated that sec 12(1) principally provides procedural and substantive protection for any deprivation of physical liberty.

¹¹⁵⁹ AB and another v Minister of Social Development at par 65.

¹¹⁶⁰ AB and another v Minister of Social Development at par 66. When we interpret the provisions of the Constitution, it is incumbent on us to enhance the integrity of those who seek to rely on it.

¹¹⁶¹ AB and another v Minister of Social Development at par 66.

¹¹⁶² AB and another v Minister of Social Development at par 66.

¹¹⁶³ AB and another v Minister of Social Development at par 66. At 67 Khampepe J states that the change in language illustrates a shift in emphasis towards one which acknowledges the multifaceted lives people may choose to live by providing for a more expansive range of bodily and psychological protections. Further, the Constitution enjoins us to develop a new understanding of 'freedom and security of the person' that demonstrates respect and attentiveness to the decisions of others. The inclusion of sec 12(2) is one facet of this new approach.

Section 12(2)(a) protects the right to make decisions concerning reproduction.¹¹⁶⁴ A person relying on this right only needs to show that his or her inability to make the decision which is the result of a law or conduct, has at least caused psychological harm.¹¹⁶⁵ Section 12(2)(b) protects *security in* and *control ove*r one's body, which, although often mutually supporting, remain independently enforceable.¹¹⁶⁶ The decision concerning reproduction and the possible physical implications of the choices are crucial to a person's wellbeing.¹¹⁶⁷ AB is a single woman who was unable to donate a gamete of her own.¹¹⁶⁸ Since the process of surrogacy has no physical implications for her, the court acknowledged that the constitutional challenge of section 294 is grounded in section 12(2)(a).¹¹⁶⁹

The result of the section 294 prohibition for the court to confirm a surrogate agreementwhere there is no genetic link between one of the commissioning parents and the childis that the prospective parent, who is both conception and pregnancy infertile and who is also a single person, is precluded from considering surrogacy as a possibility to have his/her own child.¹¹⁷⁰ Khampepe J pointed out that the capacity to decide is an implication of what is physically possible in the world and that it has nothing to do with any positive act by the state.¹¹⁷¹ In circumstances where the state forecloses a reproductive option of a person where this essentially impacts upon a person's ability to make his/her own reproductive decisions, this would be protected by section 12(2)(a) of the Constitution.¹¹⁷² The effect of section 294 is that it removes one option of reproduction that would have been available to a person and this limits the affected

¹¹⁶⁴ AB and another v Minister of Social Development at par 70.

¹¹⁶⁵ AB and another v Minister of Social Development at par 70.

¹¹⁶⁶ AB and another v Minister of Social Development at par 70.

AB and another v Minister of Social Development at par 70.

¹¹⁶⁸ AB and another v Minister of Social Development at par 71 and 72.

¹¹⁶⁹ AB and another v Minister of Social Development at par 71 and 72. The court stated that it falls to be determined whether sec 294 objectively prevents a person from making a decision regarding reproduction in a manner which is detrimental to their psychological integrity in terms of sec 12(2)(a) and the inquiry encompasses of three parts, namely (a) Does the impugned law or conduct prevent or inhibit a person or group of persons from making a decision?; (b) If the answer is yes, does the decision concern reproduction?; (c) If the answer to (b) is yes, does preventing or inhibiting the decision detrimentally affect the psychological integrity of the person or persons concerned?

¹¹⁷⁰ AB and another v Minister of Social Development at par 73.

¹¹⁷¹ AB and another v Minister of Social Development at par 74.

¹¹⁷² AB and another v Minister of Social Development at par 74.

person's ability to make his/her own decisions.¹¹⁷³ The fact that adoption remains an alternative does not remedy this limitation.

A plain reading of the word "reproduction", according to the court, suggests that it incorporates all matters to do with the process of producing new individuals of the same species by some form of generation.¹¹⁷⁴ This process can occur when a surrogate is bearing a child for a single commissioning parent who cannot donate a gamete.¹¹⁷⁵ The decision itself does not result in physical implications for the affected person but it does have a reproductive outcome in the form of the conception and the birth of the child.¹¹⁷⁶ The court concludes that this is a reproductive act.¹¹⁷⁷

Furthermore, the effect of section 294 on an aggrieved person's psychological integrity must be seen within the context of their infertility,¹¹⁷⁸ which is as much a social as well as a physical condition.¹¹⁷⁹ The state should avoid standing in the way of decisions that a person takes to mitigate the socio-psychological harm of this condition.¹¹⁸⁰ This includes reproductive decisions on how to have a child using the modern reproductive technologies.¹¹⁸¹ As chapter 19 unambiguously recognises that because infertility puts people in a position that is harmful to their psychological integrity, they should be allowed to pursue surrogacy as a means of having a child of their own.¹¹⁸² Section 294,

¹¹⁷³ AB and another v Minister of Social Development at par 74. The court answers the first leg of the section 12(2)(a) test in the affirmative.

¹¹⁷⁴ AB and another v Minister of Social Development at par 75. The court further stated that it is alluring in this regard to restrict matters regarding reproduction to a person's own physical reproductive capacities and to do so would not be in line with the generous approach to rights adopted by the court.

¹¹⁷⁵ AB and another v Minister of Social Development at par 75.

¹¹⁷⁶ AB and another v Minister of Social Development at par 75.

¹¹⁷⁷ AB and another v Minister of Social Development at par 75. As the decision is one concerning reproduction the answer to the second question is also in the affirmative.

¹¹⁷⁸ AB and another v Minister of Social Development at par 83. At par 84 the court quoted the following: "[t]he emotional and psychological devastation wrought by the recognition or diagnosis of infertility cannot be overstated. Numerous studies have reported that the inability to reproduce takes a severe toll on both men and women." The court stated that the psychological harm is especially damaging for women.

¹¹⁷⁹ AB and another v Minister of Social Development at par 86.

¹¹⁸⁰ AB and another v Minister of Social Development at par 86.

¹¹⁸¹ AB and another v Minister of Social Development at par 86. At par 87: "Understood in this manner, infertility affects the psychological integrity of a person by placing them in a socially precarious situation. Being able to choose to have a child is almost universally accepted as central to "identity and meaning in life". Stripping a person of this choice has far-reaching personal and social ramifications. Infertility is thus harmful partly because it removes the ability to elect to have a child; a decision almost universally considered important."

¹¹⁸² AB and another v Minister of Social Development at par 91. Rautenbach 2017 TSAR 361.

in the view of the court, unfortunately prevents a segment of this same class of people from accessing surrogacy to their psychological detriment.¹¹⁸³ The section also imposes and compounds psychological harm for a vulnerable constituency and it does so by limiting their ability to make reproductive decisions.¹¹⁸⁴ This limitation is seen as a violation of section 12(2)(a) as a result of the impact which section 294 has on the psychological integrity of conception and pregnancy infertile people:¹¹⁸⁵

Section 294 takes the option of surrogacy away, and thus becomes the cause of continuing psychological trauma: if the provision did not exist, then a conception and pregnancy infertile person could choose to use surrogacy in order to have a child.¹¹⁸⁶

One cannot defend a law which is prejudicial on the basis that other forms of family life are open to an affected person as it is not appropriate for courts to interfere in this decision-making process of an individual.¹¹⁸⁷ Section 294 hence objectively limits the right to psychological integrity by preventing AB and others from making decisions concerning reproduction in terms of section 12(2)(a) of the Constitution.¹¹⁸⁸

The applicants' challenge on the ground of the right to equality rests on two arguments:¹¹⁸⁹ Firstly, the argument is that section 294 differentiates between persons who are capable of contributing a gamete to the conception of a child and those who are not.¹¹⁹⁰ This can be seen as differentiation on the basis of conception infertility.¹¹⁹¹ Secondly, section 294 differentiates between persons who are unable to contribute a gamete to the conception of a child and intend to use IVF, and those who are unable

AB and another v Minister of Social Development at par 91. Further see at par 110. The removal of the choice limits the ability of those persons to construct their reproductive lives in a manner that adheres to their own conscience and convictions and prevents society from enjoying what may be a beneficial new form of family life.

¹¹⁸⁴ AB and another v Minister of Social Development at par 91.

¹¹⁸⁵ AB and another v Minister of Social Development at par 91 - 92. The court explained that sec 294 prevents a conception and pregnancy infertile person from moderating the harmful consequences of their infertility by using surrogacy to have a child of their own. At par 93. Thus, the section creates a legal barrier between a conception and pregnancy infertile person and the use of surrogacy. At par 94 the court stated that because sec 294 has the effect of preventing a person from making a decision concerning reproduction, it constitutes a rights infringement.

¹¹⁸⁶ AB and another v Minister of Social Development at par 93.

¹¹⁸⁷ AB and another v Minister of Social Development at par 96.

¹¹⁸⁸ AB and another v Minister of Social Development at par 97.

¹¹⁸⁹ AB and another v Minister of Social Development at par 98.

¹¹⁹⁰ AB and another v Minister of Social Development at par 98.

¹¹⁹¹ AB and another v Minister of Social Development at par 98. Meyerson, D "Surrogacy, Geneticism and Equality: The case of AB Minister of Social Development" 2019 Constitutional Court Review 9 318.

to contribute a gamete to the conception of a child and intend to use a surrogate.¹¹⁹² This can be seen as differentiation on the basis of pregnancy infertility.¹¹⁹³

The intended effect of section 294 is thus that those who are able to contribute a gamete are permitted to conceive a child using a surrogate while those who cannot, may not.¹¹⁹⁴ The differential treatment is further explained by the prohibition in section 294 of the use of both male and female donor gametes of the commissioning parent or the parents' choice in the conception process.¹¹⁹⁵ The surrogacy regime requires that the parent or parents of a child to be conceived donate a gamete, whilst the IVF regime does not.¹¹⁹⁶

In order to determine whether the law or conduct in question violates the right to equality, the court applied the multi-stage approach adopted in the case of *Harksen v Lane NO*.¹¹⁹⁷ The first stage of that process is to establish whether the impugned law or conduct differentiates between people or categories of people.¹¹⁹⁸ If differentiation is then established, it must next be determined whether the differentiation bears a rational connection to a legitimate government purpose.¹¹⁹⁹ If the law or conduct does not bear a rational connection to a legitimate government purpose, the conclusion is that it violates section 9(1) of the Constitution.¹²⁰⁰

The question that should be asked is whether the means that the government chose in this section are rationally connected to the purpose as opposed to being arbitrary or capricious.¹²⁰¹ The purpose of section 294 is primarily to ensure that the best interests

¹¹⁹² AB and another v Minister of Social Development at par 98. Meyerson, D "Surrogacy, Geneticism and Equality: The case of AB Minister of Social Development" 2019 Constitutional Court Review 9 319.

¹¹⁹³ AB and another v Minister of Social Development at par 98. Meyerson, D "Surrogacy, Geneticism and Equality: The case of AB Minister of Social Development" 2019 Constitutional Court Review 9 319.

¹¹⁹⁴ AB and another v Minister of Social Development at par 99.

¹¹⁹⁵ AB and another v Minister of Social Development at par 101.

¹¹⁹⁶ AB and another v Minister of Social Development at par 101. Boniface 2017 TSAR 194.

¹¹⁹⁷ AB and another v Minister of Social Development at par 102. Meyerson, D "Surrogacy, Geneticism and Equality: The case of AB Minister of Social Development" 2019 Constitutional Court Review 9 322.

¹¹⁹⁸ AB and another v Minister of Social Development at par 102.

¹¹⁹⁹ AB and another v Minister of Social Development at par 102.

¹²⁰⁰ AB and another v Minister of Social Development at par 103.

¹²⁰¹ AB and another v Minister of Social Development at par 104.

of the child to be born are protected.¹²⁰² The commissioning parent does not carry the child, as is the case with surrogacy, which is different where conception is realised through IVF treatment of one of the commissioning parents.¹²⁰³ In the court's view, the differentiation is not arbitrary or capricious and section 294 is therefore not constitutionally invalid on the basis of section 9(1).¹²⁰⁴

Next considered, was the issue whether the differentiation amounts to discrimination and if the differentiation is on a ground listed in section 9(3), it would by implication be discriminatory.¹²⁰⁵ Since conception infertility is not a listed ground under section 9(3),¹²⁰⁶ it must be determined if differentiation on the basis of conception infertility has the capacity to impair the fundamental dignity of those who are both conception and pregnancy infertile, or if it adversely affects them in a comparably serious manner.¹²⁰⁷ Differentiation to the level of discrimination is raised as those who are pregnancy infertile, but not conception infertile, can use surrogacy to ameliorate the psychological harms of infertility, however, those who are both conception and pregnancy infertile cannot.¹²⁰⁸ For the court, the harm to psychological integrity that is caused by infertility and which results in discriminatory treatment is strengthened by our dignity jurisprudence:¹²⁰⁹

Thus, our Constitution acknowledges that protecting and promoting diversity of thought and action is a requirement for human flourishing, and for community building. It is only by accepting that the opinions and decisions of each individual should be respected and encouraged that dignity is ensured. The right to dignity "requires us to acknowledge the value and worth of all individuals in a society.¹²¹⁰

AB and another v Minister of Social Development at par 104. Meyerson, D "Surrogacy, Geneticism and Equality: The case of AB Minister of Social Development" 2019 Constitutional Court Review 9 326.

¹²⁰³ AB and another v Minister of Social Development at par 104.

AB and another v Minister of Social Development at par 104. Meyerson, D "Surrogacy, Geneticism and Equality: The case of AB Minister of Social Development" 2019 Constitutional Court Review 9 336. With reference to section 294 and the no-double-donor requirement, Meyerson argues that the requirement bears no relationship to the purpose of promoting more loving and stable families (if this is what sec 294 is intended to achieve) as it fails to satisfy the rational connection test under section 9(1) of the Constitution.

¹²⁰⁵ AB and another v Minister of Social Development at par 105.

¹²⁰⁶ AB and another v Minister of Social Development at par 105. The court pointed out that determining whether there has been discrimination is an objective question, independent of the intentions of the legislator.

¹²⁰⁷ AB and another v Minister of Social Development at par 106.

¹²⁰⁸ AB and another v Minister of Social Development at par 106.

¹²⁰⁹ AB and another v Minister of Social Development at par 107.

¹²¹⁰ AB and another v Minister of Social Development at par 108.

Section 294 exists as a result of a biological condition and the suspected consequences thereof.¹²¹¹ For Khampepe J, the effect of section 294 is to assume without the support of evidence, that the state is in a better position to make reproductive decisions than the parent who will raise the child.¹²¹² This, in her view, is a flagrant violation of dignity, especially where the consequences are an increase in stigma, and an endorsement of homogeneity over difference.¹²¹³ Section 294 has the effect of disregarding the impact of new technologies on the ambit of rights.¹²¹⁴ The state has a negative duty to avoid standing in the way of advances in technology that can drastically alter someone's life for the better.¹²¹⁵

Returning to the question whether section 294 passes constitutional muster, a determination must be made whether the section violates constitutional rights, including the prohibition of unfair discrimination.¹²¹⁶ Thus, in establishing to what extent one kind of family life is privileged over another, is a factor to consider in determining whether section 294 is constitutionally valid.¹²¹⁷ Khampepe J finds that section 294 affects both the dignity of prospective parents, the families with adopted children and our society as a whole.¹²¹⁸ She further concludes that the second differentiation further demeans the dignity of the conception and pregnancy infertile by compelling them to accept that the law does not deem it necessary to police the implications of the choices of people who elect to use IVF, but does where surrogacy is used.¹²¹⁹ It is discriminatory to require a genetic link in only one of the two scenarios and if all children must have a genetic link to at least one of their parents in order to have a worthwhile

AB and another v Minister of Social Development at par 111. The section is targeted at prospective parents who are unable to contribute a gamete to a surrogate agreement.
1212
1212

¹²¹² AB and another v Minister of Social Development at par 112.

¹²¹³ AB and another v Minister of Social Development at par 112.

¹²¹⁴ AB and another v Minister of Social Development at par 113.

¹²¹⁵ AB and another v Minister of Social Development at par 113.

¹²¹⁶ AB and another v Minister of Social Development at par 116.

¹²¹⁷ AB and another v Minister of Social Development at par 116. At par 117. The court pointed out that alternative forms of family life are valued by constitutional values. There is no one correct version of the family against which others can be assessed as a result of the diversity that characterises our society. There is no definition of what an acceptable family entails as it would be presumptuous and arbitrary.

¹²¹⁸ AB and another v Minister of Social Development at par 120.

AB and another v Minister of Social Development at par 120. Persons who use IVF can actually decide to use two anonymous donor gametes but persons who choose to use surrogacy as a result of pregnancy infertility are prevented from doing the same. At par 121 the court pointed out that the only difference between the two scenarios is that in the case of the IVF treatment the one parent carries the child.

life, then surely this reasoning must also apply in the case of IVF.¹²²⁰ As a result of this, it can only be concluded that the differentiation is harmful to the affected person's dignity as potential users of surrogacy must live with the indignity of knowing that the law gives extra entitlements to those who are not pregnancy infertile.¹²²¹ Thus, section 294 harms the dignity of those persons who are both conception and pregnancy infertile as it fails to consider all people as worthy of our mutual concern and respect.¹²²²

The challenged law or conduct must also discriminate *unfairly* in order to violate section 9(3) of the Constitution.¹²²³ The impact of both the first and second differentiations on those persons who are both conception and pregnancy infertile is apparent, as they cannot choose to have a child through surrogacy.¹²²⁴ After finding that section 294 limits the rights to psychological integrity and equality, the court next had to consider whether the limitation is justifiable in terms of section 36(1) of the Constitution.¹²²⁵

The Children's Act clearly endorses the use of surrogacy as a way of reproduction but it is regulated with the best interests of children in mind.¹²²⁶ Surrogate agreements are not made available to everyone as it is only available to persons who are pregnancy infertile.¹²²⁷ It was conceded by the Minister (respondent in the matter) that section 294 does not in any way ensure the non-proliferation of commercial surrogacy in a manner that the provisions already in place for that purpose do not.¹²²⁸ It was held that alleging

¹²²⁰ AB and another v Minister of Social Development at par 122.

AB and another v Minister of Social Development at par 123.

AB and another v Minister of Social Development at par 124. SALRC Project 140 101. Meyerson, D "Surrogacy, Geneticism and Equality: The case of AB Minister of Social Development" 2019 Constitutional Court Review 9 324. It has a detrimental impact on conception infertile people by excluding them from access to surrogacy services.

AB and another v Minister of Social Development at par 125. Unfairness is then determined by focussing on the impact of the discrimination on the complainant and others in his or her situation.
 AB and another v Minister of Social Development at par 127.

¹²²⁵ AB and another v Minister of Social Development at par 129. The onus is on the Minister of Social Development to show that the limitation is justified.

AB and another v Minister of Social Development at par 141. At par 142 the court stated that ch 19 was passed to ensure that surrogacy is practiced in a manner that considers primarily the wellbeing of children born of a surrogate agreement. Rautenbach 2017 TSAR 358.

¹²²⁷ AB and another v Minister of Social Development at par 143.

AB and another v Minister of Social Development at par 147. The court stated that the removal of sec 294 from the Children's Act will make no difference to the proliferation of commercial surrogacy as a practice. Other sections in ch 19 prevent commercial surrogacy and not sec 294. No evidence was placed before the court to show that sec 294 is aimed at restricting commercial surrogacy. SALRC Project 140 101.

that the purpose of section 294 is aimed at the prevention of creating designer children, seems to be contrived.1229

Responding to the argument that one of the purposes of section 294 is to ensure that children born as a result of a surrogate agreement may know their genetic origins, the court rightfully concludes that if this is true, it would then allow some children (e.g those born of surrogacy) to know, partially or fully, their genetic origins, but would at the same time mean that children born of double-donor IVF are completely barred from knowing their genetic origins.¹²³⁰ Sections 41(2) and 294 of the Children's Act thus seem to be contradictory instead of complementary, which is problematic for the following reasons:¹²³¹ Firstly, it suggests that there are two different best interests of the child standards, one for children born as a result of surrogacy and another for children born from double-donor IVF.¹²³² Secondly, it suggests that genetic origins matter less if one is born from double-donor IVF, presumably because there is a gestational link between mother and child even if there is no genetic tie.¹²³³ Gestational link has no bearing on genetic identity.¹²³⁴ The main purpose of section 294, as proposed by the Minister, is simply to ensure that surrogacy is only used in order to have a child that is genetically related to a commissioning parent.¹²³⁵

Khampepe J, however, advances the argument that the purpose of section 294 is preventing the circumvention of the adoption process. This purpose is in line with a plain reading of the section, as well as its legislative context.¹²³⁶ Section 294 may consequently be seen as a regulative provision that seeks to limit the ambit of surrogacy agreements that can be lawfully pursued.¹²³⁷ By creating this legal barrier. the section infringes upon protected rights and as such, the purpose of the limitation is

¹²²⁹ AB and another v Minister of Social Development at par 152. SALRC Project 140 101.

¹²³⁰ AB and another v Minister of Social Development at par 163. Rautenbach 2017 TSAR 358 and 362. The writer argues that in times to come, the right to know one's origin might become so important that the constitutionality of sec 41(2) could come under scrutiny.

¹²³¹ AB and another v Minister of Social Development at par 163-164.

¹²³² AB and another v Minister of Social Development at par 164. The court stated that this is impermissible and inconsistent with a plain understanding of the phrase "best interests". 1233

AB and another v Minister of Social Development at par 164. 1234

AB and another v Minister of Social Development at par 164. 1235

AB and another v Minister of Social Development at par 168.

¹²³⁶ AB and another v Minister of Social Development at par 171. At par 173 the court stated that the chief goal of sec 294 is to prevent a child from being born as a result of a surrogate agreement without being genetically related to at least one of the commissioning parents.

¹²³⁷ AB and another v Minister of Social Development at par 172.

conterminous with the purpose of the provision.¹²³⁸ The purpose of section 294 and thus also the purpose of the rights limitation, is to discourage the use of surrogacy where other adequately similar avenues are available.¹²³⁹ Further, the importance of the purpose of section 294 depends on whether adopting a child is suitably similar to having a genetically unrelated child by way of surrogacy to render the limitation of AB's rights reasonable and justifiable.¹²⁴⁰ The court concludes that it is misguided to premise the limitation of rights on this purpose, as adoption and surrogacy are essentially different.¹²⁴¹

The court found it necessary to address the value judgment of whether being unable to determine one's genetic parents results in such extensive a harm as to rationalise the view that it is better never to have been born.¹²⁴² In terms of the Children's Act, this is a determination that the High Court is mandated to do by virtue of section 295(e) of the Act when the court has to confirm a surrogate agreement.¹²⁴³ Thus, the court must engage with the value judgment of whether it would be in the best interests of the child to be born when deciding to confirm the agreement.¹²⁴⁴ The facts of each particular case will determine what is in the child's best interests, as this is a flexible inquiry.¹²⁴⁵ Section 294 enables a commissioned child to know his or her genetic origins above any of her other interests, including life itself.¹²⁴⁶ However, in discussing this, the court observes that section 294 contradicts section 7(1) of the Children's Act.¹²⁴⁷ Section 294 privileges one factor out of the factors set out in section 7, to the exclusion of all other factors by making it possible that a child brought into a loving and stable family environment which would enable the child's physical, intellectual and emotional, social

¹²³⁸ AB and another v Minister of Social Development at par 172. Meyerson, D "Surrogacy, Geneticism and Equality: The case of AB Minister of Social Development" 2019 Constitutional Court Review 9 322.

¹²³⁹ AB and another v Minister of Social Development at par 173.

¹²⁴⁰ AB and another v Minister of Social Development at par 175. SALRC Project 140 102.

¹²⁴¹ AB and another v Minister of Social Development at par 185. Thaldar 2018 De Rebus 29.

¹²⁴² AB and another v Minister of Social Development at par 191.

¹²⁴³ AB and another v Minister of Social Development at par 192.

¹²⁴⁴ AB and another v Minister of Social Development at par 192.

¹²⁴⁵ AB and another v Minister of Social Development at par 193.

¹²⁴⁶ AB and another v Minister of Social Development at par 194. It can be inferred from sec 294 superseding sec 295(e) that it will never be in any child's best interests to be born of a surrogate agreement if he/she will not be genetically related to a commissioning parent.

¹²⁴⁷ AB and Another v Minister of Social Development at par 195.

and cultural development, would be prevented from being born purely because she could never know the identity of her genetic parents.¹²⁴⁸

Section 294 results in an absolute block to the use of surrogacy as a means of reproduction in cases where a gamete is not provided by at least one commissioning parent.¹²⁴⁹ Such limitation of rights is far-reaching as there is no comparable alternative to double-donor surrogacy for the people who are unable to provide a gamete in order to have a child.¹²⁵⁰ For the court, the ultimate and quintessential question that lies at the heart of the matter is whether section 294, as it stands, serves a purpose which is so fundamental as to outweigh and justify the corresponding limitations of the rights in question.¹²⁵¹ The court justifiably concludes that it is of the view that section 294 is an extensive and unjustifiable intrusion into a central part of the lives of those persons who are both conception and pregnancy infertile, which violates the rights to psychological integrity and equality in an unjustifiable manner.¹²⁵² Thus, section 172(1(a) of the Constitution obliges the court to declare that section 294 is inconsistent with the Constitution and invalid.¹²⁵³

The central reason for the court's finding that section 294 is constitutionally invalid, is the effect that it has on the people who are both conception- and pregnancy infertile.¹²⁵⁴ The scheme of chapter 19 will be upset should section 294 be struck.¹²⁵⁵

¹²⁴⁸ AB and another v Minister of Social Development at par 195-196. At par 196 the court refers to the 2007 judgment of AD v DW and said "...child law is an area that abhors maximalist legal propositions that preclude or diminish the possibilities of looking at and evaluating the specific circumstances of the case". The court found that sec 294 has this maximalist effect. Thaldar 2018 De Rebus 29. It was pointed out that the psychological expert opinion during the hearing was that all agreed that a child's psychological wellbeing is not negatively affected by being donorconceived.

¹²⁴⁹ AB and another v Minister of Social Development at par 208.

¹²⁵⁰ AB and another v Minister of Social Development at par 209.

¹²⁵¹ AB and another v Minister of Social Development at par 213. The court referred to the Bhulwana case where it stated: "the court places the purpose, effects and importance of the infringing legislation on one side of the scales and the nature and effect of the infringement caused by the legislation on the other. The more substantial the inroad into fundamental rights, the more persuasive the grounds of justification must be."

¹²⁵² AB and another v Minister of Social Development at par 213. Rautenbach 2017 TSAR 352.

¹²⁵³ AB and another v Minister of Social Development at par 214.

¹²⁵⁴ AB and another v Minister of Social Development at par 223.

¹²⁵⁵ AB and another v Minister of Social Development at par 223. It is explained that sec 295(a) requires that would-be commissioning parents be pregnancy infertile. Sec 294 creates a further hierarchy which, as explained, obliges the use of the gametes of both commissioning parents where possible. Should this not be possible for biological, medical or other valid reasons, sec 294 permits the use of the gamete of one of the two commissioning parents and if the commissioning parent is single, the gamete of that person should be used.

The minority judgment in this case was that the order of the High Court was confirmed, the declaration of invalidity was suspended for 18 months and the Minister was to bear the applicant's costs.¹²⁵⁶ Because of the fundamental constitutional issues that emerge from this case, the majority judgment by Nkabinde J with Mogoeng CJ, Moseneke DCJ, Bosielo AJ, Jaftha J, Mhlantla J and Zondo J concurring, also needs to be discussed in more detail.

2.8.1.2 The majority judgment

The majority judgments acknowledges that this case touches on delicate issues cutting across cultures for both genders,¹²⁵⁷ which at its core, concerns the power of the state to regulate the assistive reproductive opportunities available to those who are conception and pregnancy infertile, and to have children of their own.¹²⁵⁸ The issues trigger sympathy for those persons who are conception and pregnancy infertile.¹²⁵⁹

With reference to the minority judgment above, Nkabinde J stated that she is in agreement with the remarks regarding the effects of a woman's inability to have a child of her own and with the exposition of the background facts as well as its conclusion regarding the costs.¹²⁶⁰ However, she is not in agreement with the declaration of invalidity of section 294 as she believes that none of the implicated rights are violated.¹²⁶¹ It was against the legal historical background (as discussed in chapter 2

¹²⁵⁶ AB and another v Minister of Social Development at par 236. SALRC Project 140 101.

¹²⁵⁷ AB and another v Minister of Social Development at par 237. The issues are infertility and the inability to conceive a child or to produce a gamete, in order to meet the legal requirement to enter into a surrogate motherhood agreement.

¹²⁵⁸ AB and another v Minister of Social Development at par 237. The issues further raise complicated legal and ethical questions that definitely have an impact on many people who are unable to give birth to their own children.

¹²⁵⁹ AB and another v Minister of Social Development at par 237.

¹²⁶⁰ AB and another v Minister of Social Development at par 240.

AB and another v Minister of Social Development at par 240. Thaldar 2018 De Rebus 29. The writer concludes that the majority of the Constitutional Court simply rejected all of the psychological evidence out of hand, without any critical engagement. And this effectively means that the majority judges are of the opinion that when considering the psychology of donor-conceived children, that they cannot be assisted by the opinions of qualified psychologists who have studied the wellbeing of donor-conceived children over many years. The rule of law demands that judgments be based on evidence properly before the court and by simply disregarding evidence out of hand without even considering it, and then by replacing it with the judges' own conceptions, are antithetical to the rule of law. The objective facts were placed before the court but were then unabashedly ignored by the court.

of this thesis) and the constitutional provisions and the legislative scheme, that the issues before the court need to be considered.¹²⁶²

The majority judgment next turns to the applicants' argument in the High Court that the genetic link requirement assigns special value to genetic lineage in the context of establishing a family through surrogacy.¹²⁶³ The applicants contended that the genetic link criterion in section 294 impacted on both the following: (a) the rights of all members of the class of persons by prohibiting them from electing to use donor gametes for the conception of their child-to-be¹²⁶⁴ and (b) on the right of the sub-class by prohibiting them from using surrogacy.¹²⁶⁵ It was further submitted by the applicants that the prospective parents' right to human dignity and reproductive autonomy as guaranteed by section 12(2)(a) of the Constitution was violated.¹²⁶⁶ The respondent, on the other hand, argued that the constitutional right of a child (as guaranteed in section 28(2) of the Constitution) would be compromised if the genetic link requirement is to be removed.1267

The court, approaching the matter by analysing the legal conception of what constitutes a family,¹²⁶⁸ held that the genetic link requirement is irrational in terms of section 9(1) of the Constitution and that it infringes AB's or the sub-class's right.¹²⁶⁹

¹²⁶² The history is broadly discussed in ch 2 above. Nkabinde J's comments are also included in the abovementioned discussion. The constitutional provisions and legislative scheme are set out and discussed above in ch 3.

¹²⁶³ AB and another v Minister of Social Development at par 258. The central challenge in the court was thus based on the infringement of the right to equality.

¹²⁶⁴ AB and another v Minister of Social Development at par 259. It was submitted that the genetic link requirement violated AB's right to equality as it causes a certain category of persons to be treated differently. They cannot enjoy equal protection and benefit of the law in terms of sec 9(1) of the Constitution. The applicants contented that social marginalisation of infertile people is perpetuated and the negative consequence of infertility is strengthened.

¹²⁶⁵ AB and another v Minister of Social Development at par 258 and 260. The applicants relied on sec 9(3) and sec 27(1) of the Constitution. It was said that the genetic link requirement imposed on the Subclass is particularly "noxious" and it constitutes unfair discrimination. It was further said that surrogacy is a form of reproductive health care and thus, the genetic link requirement violates AB's right to reproductive health care.

¹²⁶⁶ AB and another v Minister of Social Development at par 261. Thus, the prospective parent's right to choose whether to use her own gametes or through IVF make use of donor gametes was allegedly infringed. Autonomy is a core element of human dignity as argued by the applicants. 1267

AB and another v Minister of Social Development at par 262.

¹²⁶⁸ AB and another v Minister of Social Development at par 263. The court stated that the Legislature should take cognisance of the advances in fertility and reproductive technology and the obligation to redefine the traditional view of the family.

¹²⁶⁹ AB and another v Minister of Social Development at par 263. The infringement is (i) by unfairly discriminating on the basis of infertility whereas under the IVF regulations parents are free to use double-donor gametes in terms of sec 9(3) of the Constitution; (ii) to her human dignity in terms

The court accepts that the procedures of IVF and surrogacy are fundamentally different but concludes that it does not offer a justification for the fact that in law a differentiation is drawn between the two procedures and allowing infertile people to become parents.¹²⁷⁰ Also, the court is not convinced that there is no persuasive evidence before the court that information relating to the child's genetic origin is necessarily in the best interests of the child.¹²⁷¹

The majority judgment also emphasises that the issue before the court is the validity of section 294 of the Children's Act and not about whether the genetic link requirement in section 294 has relevance to the legal conception of a family.¹²⁷² The issue to be determined is whether the order of the High Court should be confirmed and in deciding the question, regard must be had to the text of the impugned provision to determine its legislative objectives.¹²⁷³ Thus, the court has to consider whether: (a) the impugned legislation is irrational in terms of section 9(1) of the Constitution; (b) AB's implicated rights to equality, dignity, bodily integrity (including the right to make decisions concerning reproduction), access to reproductive health care and privacy are limited by the genetic link requirement in terms of section 294, and if so; (c) whether the limitation of the rights is justifiable in terms of section 36(1) of the Constitution.¹²⁷⁴

of sec 10 of the Constitution; (iii) to her reproductive autonomy in terms of sec 12(2)(a) of the Constitution; (iv) to her access to health care services in terms of sec 27 of the Constitution. It is to be noted that it was held that the genetic link requirement infringes human dignity and the right to bodily and psychological integrity, which includes the right to make decisions concerning reproduction. The Court's remarks were based on the fact (at par 268) that gamete donor selection and double-donor selection is recognised as a legal right in the context of IVF. SALRC Project 140 92. Meyerson, D "Surrogacy, Geneticism and Equality: The case of AB Minister of Social Development" 2019 *Constitutional Court Review* 9 341. Sec 294 is infringing sec 9(3) in that it unfairly discriminates against commissioning parents who would like to enter into a surrogacy agreement but is unable to contribute genetic material.

¹²⁷⁰ AB and another v Minister of Social Development at par 266.

¹²⁷¹ AB and another v Minister of Social Development at par 267. As per sec 28 of the Constitution.

¹²⁷² AB and another v Minister of Social Development at par 273. SALRC Project 140 91.

¹²⁷³ AB and another v Minister of Social Development at par 273. The court held that that entails an interpretive process limited to what the text of the impugned provision is reasonably capable of meaning.

AB and another v Minister of Social Development at par 273. The court pointed out at 274 that in determining whether the declaration of invalidity should be confirmed, the starting point is to delineate the correct approach to statutory interpretation. Thus, words in legislation must be given their ordinary meaning unless doing so would result in absurdity and statutes must be interpreted purposively, regarding the context of the statute as a whole.

Section 294 favours infertile commissioning parent(s) as it disqualifies the fertile commissioning parent(s) who can conceive without the assistance of surrogacy.¹²⁷⁵ Thus, the prohibition in this challenged provision relates to the conclusion of a surrogacy agreement in the case where the gametes of the commissioning parent(s) are not used.¹²⁷⁶

For the majority of the judges, a consideration of the regulatory scheme in chapter 19 in the context of the Children's Act as a whole is necessary.¹²⁷⁷ Although the Children's Act seeks to protect other rights in the Constitution, its main objective is to give effect to the constitutional rights of children.¹²⁷⁸ The importance of the best interests of the commissioned child is confirmed in section 295(e), read with section 9 of the Act.¹²⁷⁹ In deciding if section 294 is irrational,¹²⁸⁰ the court has to determine whether there is a rational connection between the means chosen and the objective sought to be achieved, as a mere differentiation does not render a legislative measure irrational.¹²⁸¹

¹²⁷⁵ AB and another v Minister of Social Development at par 277.

¹²⁷⁶ AB and another v Minister of Social Development at par 278. Thus, the commissioning parents will be disqualified if both are unable to contribute gametes for procreation and if a single commissioning parent cannot contribute gametes for that purpose, that person will also be disqualified.

¹²⁷⁷ AB and another v Minister of Social Development at par 279. SALRC Project 140 92.

¹²⁷⁸ AB and another v Minister of Social Development at par 279. In terms of sec 2 of the Children's Act. The legislative scheme under Ch 19 and, especially sec 294, also protects the child by ensuring that a genetic link exists when that child is conceived. Florescu, S and Sloth-Nielsen, J "Visions on surrogacy – From North to South: The approach of the Netherlands and South Africa to the issue of surrogacy and the child's right to know his origin" 2017 International Survey of Family Law 2017 251.

¹²⁷⁹ AB and another v Minister of Social Development at par 280. Sec 295(e) determines that the personal circumstances and family situations of all the parties concerned must be considered when confirming the agreement. Sec 28(2) of the Constitution declares the paramountcy of the best interests of the child in every matter concerning the child. It is to be noted at 281 that the court stated sec 28(2), like the other rights in the Bill of Rights, is subject to limitations that are reasonable and justifiable in compliance with sec 36 of the Constitution and it thus follows that children's rights do not trump other rights.

¹²⁸⁰ AB and another v Minister of Social Development at par 283. It is determined by answering the following questions: (a) when is the differentiation permissible and (b) under what circumstances is the differentiation a limitation of the equality right in section 9(1) and thus unconstitutional? The court held that rationality is an incident of the rule of law and when enacting laws, the legislator is constrained to act rationally and not capriciously or arbitrarily.

AB and another v Minister of Social Development at par 284. It was stated that legislation or provisions within legislation becomes invalid for being inconsistent on its own terms with the Constitution. At par 285. The court pointed out that the differentiation must be arbitrary or must manifest "naked preferences" that serve no legitimate governmental purpose for it to render the measure irrational. Florescu, S and Sloth-Nielsen, J "Visions on surrogacy – From North to South: The approach of the Netherlands and South Africa to the issue of surrogacy and the child's right to know his origin" 2017 International Survey of Family Law 2017 251.

obvious differences between IVF and surrogacy.¹²⁸² A statutory provision cannot be measured against regulations under different legislation to decide whether it is rational or consistent with the Constitution.¹²⁸³

The requirement of donor gamete(s) within the context of surrogacy serves a rational purpose of creating a bond between the child and the commissioning parent(s).¹²⁸⁴ The court argues that the creation of a bond is designed to protect the best interests of the child-to-be born so that the child has a genetic link with its parent(s) and therefore, a rational connection exists.¹²⁸⁵ Although AB is disqualified from concluding a surrogate agreement as a result of her biological, medical or other reasons, she is not left without a legal option in that she could bring herself within the ambit of section 294 by entering into a partnership relationship with someone whose gamete may be used for the conception of the child as intended in the agreement.¹²⁸⁶ The legislative measure

¹²⁸² AB and another v Minister of Social Development at par 286. SALRC Project 140 93.

AB and another v Minister of Social Development at par 286. It is when the regulatory measure does not serve a legitimate government purpose that it can fall foul of sec 9(1) of the Constitution. Sloth-Nielsen "Surrogacy in South Africa" 189. Florescu, S and Sloth-Nielsen, J "Visions on surrogacy – From North to South: The approach of the Netherlands and South Africa to the issue of surrogacy and the child's right to know his origin" 2017 International Survey of Family Law 2017 252.

AB and another v Minister of Social Development at par 287. Florescu, S and Sloth-Nielsen, J "Visions on surrogacy – From North to South: The approach of the Netherlands and South Africa to the issue of surrogacy and the child's right to know his origin" 2017 International Survey of Family Law 2017 252.

AB and another v Minister of Social Development at par 287. The court further pointed out at par 288 that the disqualification of AB or of other people similarly placed is rational in that it safeguards the genetic origin of the child as contemplated in the surrogacy agreement for the child's best interests. SALRC Project 140 104. Rautenbach 2017 TSAR 358. Sloth-Nielsen "Surrogacy in South Africa" 189.

¹²⁸⁶ AB and another v Minister of Social Development at par 288. At par 289 the court explains the difference between IVF and surrogacy. With IVF the "host mother" may not necessarily be the genetic mother of the child but she holds a gestational link to the child as she is carrying the child. With surrogacy, a genetic link is created between the child-to-be and the commissioning parent(s). At par 290 the court said that the applicants did not dispute that the clarity of origin may be important to a self-identity and self-respect of the child SALRC Project 140 104. It is explained that "origin" of a child relates to how the child was conceived and by whom and who gave birth to the child. Clarity, on the other hand, could mean that the child is entitled to know his or her genetic origins not only as it relates to the commissioning parent, but also of the donor and the surrogate mother. Rautenbach 2017 TSAR 358. Sloth-Nielsen "Surrogacy in South Africa" 189. The writer states that the suggestion that the infertile applicant find a fertile partner for the purposes of conception with non-donor gametes does not seem judicious and it smacks of a different form of baby shopping. She points out that the finding of the Constitutional Court lead to a South African Law Commission investigation into the rights of the children to know their biological origins more generally. An Issue paper "The right to know one's own biological origins" was released in 2017. Florescu, S and Sloth-Nielsen, J "Visions on surrogacy – From North to South: The approach of the Netherlands and South Africa to the issue of surrogacy and the child's right to know his origin" 2017 International Survey of Family Law 2017 252. Meyerson, D "Surrogacy, Geneticism and Equality: The case of AB Minister of Social Development" 2019 Constitutional Court Review 9 323 and 326. Meyerson correctly argues that for AB to enter into

chosen by the legislature in section 294 is rationally related to the public good sought to be achieved by government.¹²⁸⁷ The importance of the genetic link requirement is further confirmed in the adage "*ngwana ga se wa ga ka otla ke wa ga katsola*" (a child belongs not to the one who provides but to the one who gives birth to the child) and therefore clarity regarding the origin of a child is important to the self-identity and self-respect of the child.¹²⁸⁸ According to the court, the rationality challenge must fail, as

a surrogacy agreement, she would have needed to find an intimate partner with whom she wanted to share her life; her feelings would have had to be shared and the partner would have had to be both fertile and willing to enter into a surrogacy agreement. Further correctly stated is that it is difficult to see how these variables can be said to have been under AB's control, such that her failure to be in a relationship with a person with all these characteristics can be ascribed to her personal preferences. The author gave the example of a married couple who cannot contribute a gamete for conception and thus, they are disqualified by sec 294 to enter into a surrogacy agreement. Their only option would be to enter into separate surrogacy agreements if they were to dissolve their relationship and find a fertile partner. Would anyone take the view that if the couple refuse to avail themselves of this option that they can only blame themselves? She concludes this argument by stating that even radically neoliberal views, which are willing to ascribe many forms of disadvantage to people's choices, would not be so hyper-voluntarist. People who do not suffer from conception infertility are not required to find or change partners in order to have a child through the surrogacy process.

¹²⁸⁷ AB and another v Minister of Social Development at par 292. We cannot therefore interfere with the lawfully chosen measure on the ground that the legislature should have taken other considerations into account or that it should have considered a different decision that is preferable. The purpose of the enquiry is to determine whether the means selected are rationally related to the objective sought to be achieved Florescu, S and Sloth-Nielsen, J "Visions on surrogacy – From North to South: The approach of the Netherlands and South Africa to the issue of surrogacy and the child's right to know his origin" 2017 International Survey of Family Law 2017 252.

¹²⁸⁸ AB and another v Minister of Social Development at par 293. SALRC Project 140 95 and 105. Florescu, S and Sloth-Nielsen, J "Visions on surrogacy - From North to South: The approach of the Netherlands and South Africa to the issue of surrogacy and the child's right to know his origin" 2017 International Survey of Family Law 2017 253. Meyerson, D "Surrogacy, Geneticism and Equality: The case of AB Minister of Social Development" 2019 Constitutional Court Review 9 318. At 330 the author explains why she disagrees with the majority's finding in this regard by using the example of where the commissioning parents have only contributed one gamete: Here, she points out, the law does not impose a duty on the commissioning parents or anyone else to inform the child about the fact that a donor gamete was used in his or her conception. Further, there is no provision that requires this to be recorded on the child's birth certificate. Importantly, even if the commissioning parents inform the child that a donor gamete was used in his or her conception, sec 41(2) of the Children's Act would prevent the child from finding out the identity of the donor. The author states that in light of these, two scenarios can arise (the child knows that he or she does not know one of her genetic parents or the child falsely believes that he or she is related to both of his or her genetic parents) and in neither of the scenarios will the child have the knowledge of his or her origins required to satisfy his or her hypothesised identity needs. At 331 Meyerson argues that it is difficult to see how the purpose of sec 294 can be to serve children's interests in constructing their self-identity around knowledge of their genetic parents. If it was that the legislature had really sought to protect these putative interests, it surely would not have allowed commissioning parents to contribute only one gamete for the conception while at the same time preventing the child from learning the identity of the donor of the other gamete (used for his or her conception) and allowing the existence of the donor to be kept a secret. Thus, making it possible for children born as a result of a surrogacy agreement to be deceived about their genetic origins and to construct a spurious sense of identity build on an illusion. At 332 the author states that the legislature's purpose in enacting sec 294 could thus not have been to serve the children's identity needs because there are other restrictions and permissions which is

there is a rational nexus between the purpose of the legislative scheme, including section 294, that provides a framework within which persons are able to have children and become parents in circumstances where they would otherwise not have been.¹²⁸⁹ The right to equality provides a tool to achieve substantive equality, which, unlike formal equality that presumes that all people are equal, tolerates difference.¹²⁹⁰ A two-stage enquiry regarding the categories of discrimination is important and the court refers to the following extract from *Harksen v Lane NO*:¹²⁹¹

Where discrimination results in treating persons differently in a way which impairs their fundamental dignity as human beings, it will clearly be a breach of section 8(2). Other forms of differentiation, which in some other way affect persons adversely in a comparably serious manner, may well constitute a breach of section 8(2) as well.¹²⁹²

The question whether there has been differentiation on a specified or an unspecified ground must be answered objectively.¹²⁹³ Differentiation will amount to discrimination if the challenged provision authorises unequal treatment of people based on certain attributes and characteristics attaching to them.¹²⁹⁴ Hence, equality will mean nothing if it does not recognise a person's equal worth as a human being.¹²⁹⁵

AB is disqualified from using surrogacy as a result of biological, medical or other reasons as contemplated in section 294.¹²⁹⁶ Section 294 neither creates nor compounds infertility.¹²⁹⁷ Thus, the challenged provision does not disqualify

contained in the Children's Act that undermine the construction as a matter of statutory interpretation.

¹²⁸⁹ AB and another v Minister of Social Development at par 294. Meyerson, D "Surrogacy, Geneticism and Equality: The case of AB Minister of Social Development" 2019 Constitutional Court Review 9 328.

¹²⁹⁰ AB and another v Minister of Social Development at par 296.

¹²⁹¹ 1998 (1) SA 300 (CC).

AB and another v Minister of Social Development at par 297. "There will be discrimination on an unspecified ground if it is based on attributes or characteristics which have the potential to impair the fundamental dignity of persons as human beings or affect them adversely in a comparably serious manner."

¹²⁹³ AB and another v Minister of Social Development at par 297.

AB and another v Minister of Social Development at par 298. Sec 294 regulates the conclusion of a valid surrogate agreement and it does not confer a right to conclude an agreement. SALRC Project 140 96.

¹²⁹⁵ AB and another v Minister of Social Development at par 299.

¹²⁹⁶ AB and another v Minister of Social Development at par 299.

AB and another v Minister of Social Development at par 301. Rautenbach 2017 TSAR 359 - 360. The writer states that the question to be answered is what the concrete effect of the exclusion of the applicant from concluding a surrogacy agreement was on the interests of the applicant. Meyerson, D "Surrogacy, Geneticism and Equality: The case of AB Minister of Social Development" 2019 Constitutional Court Review 9 322. Meyerson explains that although sec 294 as a legal barrier is reasonable, it is still a legal barrier. Further, it is likewise the law, not their

commissioning parent(s) because they are infertile, it affords the infertile commissioning parent(s) the opportunity to have children of their own by contributing gametes for the conception of the child intended in the surrogate agreement.¹²⁹⁸

The majority decision concludes that the challenged provision does not unfairly discriminate against AB or members of the sub-class.¹²⁹⁹ The differentiation can only be unfair if the differentiation results in AB or the sub-class being treated differently in a way which impairs their fundamental dignity as human beings or which affects them adversely in a comparably serious manner.¹³⁰⁰

The next question to be answered is whether section 294 limits AB's right to reproductive autonomy. The court fails to see how section 294 constitutes a limitation of AB's rights to dignity and reproductive autonomy (as guaranteed by section 12(2)(a) of the Constitution).¹³⁰¹ For the court, section 12(2)'s inclusion of a reproductive autonomy right in section 12(2)(a) is combined with the right to security in and control over a person's body in section 12(2)(b) and to be free from medical experimentation without informed consent in section 12(2)(c).¹³⁰² The right to bodily integrity as set out in section 12(2) is held to be a universally accepted fundamental right.¹³⁰³ The right

medical condition, which prevents the conception infertile parents from having a child through surrogacy.

AB and another v Minister of Social Development at par 302. If the parent(s) cannot contribute a gamete, the parent still has available options afforded by the law: a single parent has the choice to conclude a permanent relationship with a fertile parent, thereby qualifying the parent for surrogacy. But, if the infertile commissioning parent(s) decide not to use the available legal options, they have to live with the choices they make. SALRC Project 140 96.

¹²⁹⁹ AB and another v Minister of Social Development at par 303. Meyerson, D "Surrogacy, Geneticism and Equality: The case of AB Minister of Social Development" 2019 Constitutional Court Review 9 321. The author argues that the majority's understanding of the right to equality under the Constitution is unattractively cramped. It failed to recognise sec 294's differential treatment of prospective parents who are conception infertile is unfair discrimination. Further, in being unwilling to interrogate the legitimacy of the state's purpose, the majority gave us an ungenerous version of the constitutional principle of equality in the service of a dubious legislative goal.

¹³⁰⁰ *AB and another v Minister of Social Development* at par 304. Meyerson, D "Surrogacy, Geneticism and Equality: The case of AB Minister of Social Development" 2019 *Constitutional Court Review* 9 318.

¹³⁰¹ AB and another v Minister of Social Development at par 306. Sec 12(2) recognises that each physical body is of equal worth and gives protection to the construction regarding their bodies. At 309 the Court stated that the primary purpose of the right was to ensure that the physical integrity of every person was protected. Rautenbach 2017 *TSAR* 360. The writer argues that all rights in the bill of rights empowers one to choose freely the way in which one performs the actions and protects and promotes the interests protected by the right and not every limitation of other rights amounts to violation of human dignity.

¹³⁰² AB and another v Minister of Social Development at par 310.

¹³⁰³ AB and another v Minister of Social Development at par 311.

relating to reproductive autonomy in section 12(2)(a) directly addresses the fact that many women do not enjoy security in and control over their own bodies.¹³⁰⁴ Thus, the focus is on the individual woman's own body and not a body of another woman.¹³⁰⁵ After considering comparable judgments in other jurisdictions which show that security of the person encompasses personal autonomy involving control over a person's bodily integrity,¹³⁰⁶ the majority concludes that the applicant's argument that the donor gametes decision entails a decision regarding AB's reproduction is misconceived.¹³⁰⁷ The court argues that although the donor gamete decision is an important exercise of a prospective parent's autonomy, it does not entail a decision regarding the commissioning parent's bodily integrity, but one regarding the body of a surrogate or "host" mother.¹³⁰⁸

The court's argument was that it can hardly be argued that section 294 is invalid because it is not in line with the constitutional value of self-autonomy, encapsulated in the maxim *pacta sunt servanda*,¹³⁰⁹ as section 294 does not prevent AB from regulating her own affairs.¹³¹⁰ Section 293 strengthens the view that the legislative scheme favours infertile commissioning parents.¹³¹¹ The decision allowing the creation of a child without a genetic link between the commissioning parent(s) and the commissioned child will not accord with the object of the legislation that also favours the commissioning parent(s).¹³¹² Thus, in the view of the majority, the challenge regarding section 294 based on section 12(2)(a) must fail.¹³¹³

¹³⁰⁴ AB and another v Minister of Social Development at par 313.

¹³⁰⁵ AB and another v Minister of Social Development at par 313. SALRC Project 140 99. Rautenbach 2017 TSAR 360. Meyerson, D "Surrogacy, Geneticism and Equality: The case of AB Minister of Social Development" 2019 Constitutional Court Review 9 319.

AB and another v Minister of Social Development at par 314.

¹³⁰⁷ AB and another v Minister of Social Development at par 314.

¹³⁰⁸ AB and another v Minister of Social Development at par 315. The court acknowledged the need to respect the autonomy of commissioning parents in relation to the choices they make for the purposes of concluding surrogacy agreements. Sec 12(2)(a) however does not give anyone the right to bodily integrity in respect of another person's body. SALRC Project 140 99.

¹³⁰⁹ *AB and another v Minister of Social Development* at par 316. The Children's Act regulates the right to conclude a surrogacy agreement.

¹³¹⁰ AB and another v Minister of Social Development at par 316. Meyerson, D "Surrogacy, Geneticism and Equality: The case of AB Minister of Social Development" 2019 Constitutional Court Review 9 319.

¹³¹¹ AB and another v Minister of Social Development at par 317.

¹³¹² AB and another v Minister of Social Development at par 317.

¹³¹³ AB and another v Minister of Social Development at par 318.

With reference to AB's right of access to health care, the court observes that section 27(1) of the Constitution does not give rise to a self-standing and independent positive right that is immediately enforceable.¹³¹⁴ Consequently, as per Nkabinde J, based on the facts of the case and on deliberations regarding the challenged provision in relation to the position of the infertile commissioning parent(s), the court is unable to conclude that the genetic link requirement prevents AB and members of the sub-class from enjoying the right to have access to health care services, including reproductive health care.¹³¹⁵ Hence, it cannot be said that the guaranteed rights in section 27(1) of the Constitution are limited by virtue of the genetic link requirement in section 294 of the Children's Act.¹³¹⁶

Moreover, also the applicants' challenge based on AB's right to privacy in terms of section 14 of the Constitution should fail, as this right is not limited by the genetic link requirement in section 294 of the Children's Act.¹³¹⁷ The right to make autonomous decisions in respect of intensely significant aspects of one's personal life falls within the ambit of the right to privacy but it does not suggest an independent right to autonomy.¹³¹⁸ Thus, the challenged provision does not limit AB's right to privacy.¹³¹⁹ The issue regarding limitation of rights consequently does not arise.¹³²⁰

Thus, as per the majority judgment, it was held that the order of constitutional invalidity in respect of section 294 of the Children's Act 39 of 2005 made by the High Court of South Africa, Gauteng Division, Pretoria, is not confirmed.¹³²¹

I agree with the contentions raised and argued by the minority judgment. Section 294 does distinguish between the different forms of infertility and consequently, unfair

¹³¹⁴ AB and another v Minister of Social Development at par 320. The court found it difficult to understand the applicants' constitutional challenge based on the right to have access to reproductive health care in terms of sec 27(1).

¹³¹⁵ AB and another v Minister of Social Development at par 322.

¹³¹⁶ AB and another v Minister of Social Development at par 322.

¹³¹⁷ AB and another v Minister of Social Development at par 322. Meyerson, D "Surrogacy, Geneticism and Equality: The case of AB Minister of Social Development" 2019 Constitutional Court Review 9 319.

¹³¹⁸ AB and another v Minister of Social Development at par 323. The court referred to S v Jordan 2002 (11) BCLR 1117 (CC). SALRC Project 140 101.

¹³¹⁹ AB and another v Minister of Social Development at par 323.

¹³²⁰ AB and another v Minister of Social Development at par 324.

¹³²¹ AB and another v Minister of Social Development at par 330. SALRC Project 140 101.

discrimination against women that are pregnancy infertile, especially compared to the IVF scenario, does exist. It is unacceptable to say that children will experience happier lives just because there is a genetic link with one or both of his/her parents. Allowing commissioning parent(s) to use surrogacy, although there will be no genetic link with the commissioned child, will not cause commercial surrogacy. It is important to look at each application objectively and decide on the facts and circumstances relevant to the specific applicants if there are grounds on which the court can allow the commissioning parent(s) to use surrogacy in circumstances where he/she/they cannot be genetically linked to the commissioned child. AB had no alternative than to use surrogacy as a means of having her own child and that is clear from the fact that she underwent 18 unsuccessful IVF treatments, which in my view, must have been emotionally and physically traumatic.

2.8.2 EJ and others v Haupt NO¹³²²

This judgment was discussed fully in chapter 3 above. The matter dealt with the applicants challenging section 40 of the Children's Act on the ground that the provisions of sections 40(1) and section 40(3)(b) discriminate against same-sex female couples. The court held that it does have a discretion to raise a constitutional issue *mero motu* and the discretion should be used sparingly and only in circumstances where the legislation cannot be read inclusively and where it is restrictive in its interpretation.¹³²³

Neukircher J found that no declaration of invalidity or unconstitutionality is required in this application.¹³²⁴ Both the spouses automatically acquired rights and responsibilities in respect of the child born from artificial fertilisation although the second applicant does not share a genetic link with the child.¹³²⁵ Same-sex female couples are clearly included in section 40(1) and the court could thus not find that the section is unconstitutional.¹³²⁶

¹³²² 2022 (1) SA 514 (GP).

¹³²³ *EJ and others v Haupt NO* at par 48.

EJ and others v Haupt NO at par 75.

¹³²⁵ *EJ and others v Haupt NO* at par 73.

¹³²⁶ *EJ and others v Haupt NO* at par 73 and 75.

2.8.3 KB and another v Minister of Social Development¹³²⁷

In the above matter, an application was brought in terms of High Court Rule 16(A)(a) in that the genetic link requirement in section 294 must be expanded to include that a genetic link between siblings will also satisfy this requirement.

The argument advanced by the applicants is that section 294, which makes no provision or does not require a genetic link between siblings, is irrational and that the provision should be amended to include "or where the genetic origin of the child is the same as that of any of her siblings". The applicants further argue that section 294 is unconstitutional on the ground that it offends against the principle of the paramountcy of the best interests of a child which is constitutionally protected, by denying surrogacy where the commissioned child will be genetically linked to his/her sibling.¹³²⁸ The section further unjustifiably limits the commissioned child's rights to dignity and equality.¹³²⁹ Further, the current provision impacts on the right to have a family with siblings that are genetically linked to him. The proposed amended section 294, according to the applicants, should read as follows:

No surrogate motherhood agreement is valid unless the conception of the child contemplated in the agreement is to be effected by the use of the gametes of both commissioning parents or, if that is not possible due to biological, medical or other valid reasons, the gamete of at least one of the commissioning parents or where the commissioning parent is a single person, the gamete of that person or where the genetic origin of the child is the same as that of any of her siblings.¹³³⁰

The Constitutional Court dismissed the application for direct access and an application was brought in the Mbombela High Court.¹³³¹ Sibuyi AJ concluded that the only question was whether the prohibition contained in section 294 of the Act, which has the effect of denying the minor child a genetically related sibling, passes constitutional

¹³²⁷ (966/2022) [2023] ZAMPMBHC 12 (20 February 2023).

¹³²⁸ KB and another v Minister of Social Development at par 28.

¹³²⁹ KB and another v Minister of Social Development at par 9. The applicants have a child that was born from them through the use of donor gametes. The first applicant however lost her uterus and their second baby (at six months) as a result of complications. The applicants have three embryos left which they want to use through a surrogate but in light of the fact that the embryo is as a result of donor gametes, the commissioned child will not be genetically linked to either of the applicants. The commissioned child will however be genetically linked to the first child born of the applicants. It was further argued by the applicants that it is important that the child is given a full biological sibling for in case of possible illness later in life, a biological sibling could be a potential match for bone marrow, tissue or organs should there be a need.

¹³³⁰ KB and another v Minister of Social Development at par 7.

¹³³¹ KB and another v Minister of Social Development at par 1.

muster.¹³³² The court made reference to the Constitutional Court judgement in *AB and another v Minister of Social Development* and concluded that the court cannot interfere with a lawfully chosen measure (in terms of section 294 of the Act) as there is factually and legally nothing new in this application to justify interference.¹³³³ The court reiterated that the purpose of section 294 is to protect the child by ensuring that a genetic link exists when the child is conceived.¹³³⁴ Section 294 of the Act, however, has nothing to do with the right of a minor child to have a sibling with the same genetic link.¹³³⁵

The court held that a determination of the constitutional validity must be made against the legislative history of chapter 19 of the Act.¹³³⁶ The court further concluded that the removal of the genetic link prerequisite from section 294 of the Act or the creation of an exception thereto would in principle constitute a fundamental departure from the lawfully chosen policy position.¹³³⁷ Thus, the principle of separation of powers will be violated should the court grant the relief sought by the applicants and further, the lawfully chosen measure by the legislature will be interfered with.¹³³⁸ The application failed as there was no finding that section 294 of the Act is unconstitutional and there could thus be no reading-in as requested by the applicants.¹³³⁹

2.8.4 VVJ and another v Minister of Social Development and another¹³⁴⁰

The applicants who have been in a lesbian permanent life partnership since January 2019, are considered as a permanent couple by their respective families, friends and the broader community where they live.¹³⁴¹ The applicants intend to use artificial fertilisation to conceive a child. The first applicant's ovum will be used and the second

¹³³² KB and another v Minister of Social Development at par 30.

¹³³³ *KB* and another *v* Minister of Social Development at par 40.

¹³³⁴ *KB and another v Minister of Social Development* at par 40. The court further pointed out that sec 294 is to ensure the need for a genetic link between a child and at least one of the parents so that there is clarity regarding the child's origin which is important to the self-identity and the self-respect of the child and to assist parent(s) who are conception or pregnancy infertile.

¹³³⁵ KB and another v Minister of Social Development at par 40.

¹³³⁶ KB and another v Minister of Social Development at par 41.

¹³³⁷ KB and another v Minister of Social Development at par 41.

¹³³⁸ KB and another v Minister of Social Development at par 41.

¹³³⁹ KB and another v Minister of Social Development at par 42 and 43.

¹³⁴⁰ Unreported judgment, North Gauteng High Court, Pretoria, case no 27706/2021.

¹³⁴¹ VVJ and another v Minister of Social Development and another at par 6.

applicant will be artificially fertilised with the first applicant's ovum and donor sperm.¹³⁴² The main concern of the applicants was that the current legislation determines that only the second applicant will establish rights, responsibilities, duties or obligations regarding the child(ren) born through this arrangement.¹³⁴³ The *amicus* argued that it is not whether the provisions unfairly discriminate against persons on the ground of sexual orientation, but rather whether the exclusion of unmarried persons in a committed relationship is constitutionally justifiable.¹³⁴⁴

Van Veenendaal AJ acknowledges that the Children's Act remains conservatively lagging in terms of artificial fertilisation and the subsequent recognition of partners as parents.¹³⁴⁵ For the court, one concern regarding a lifelong permanent relationship is the lack of formal indication of the intention to remain together.¹³⁴⁶ This remark is curious, as the formalisation of a relationship via marriage or a civil union is hardly a guarantee of the intention to remain together, if the high prevalence of divorce and separation are considered among married couples.

Van Veenendaal AJ concludes that section 40 of the Children's Act is indeed unconstitutional in that the provision unfairly discriminates on the basis of marital status in terms of its treatment of children born in or out of wedlock.¹³⁴⁷ Further, rights to equality and dignity of partners whose children were born as a result of artificial fertilisation, but whose relationship is not recognised, are violated by this section.¹³⁴⁸ The children's right to family and/or parental care is violated when they are born as a result of artificial fertilisation of unmarried parents.¹³⁴⁹ Also, the child's right to have

¹³⁴² VVJ and another v Minister of Social Development and another at par 9.

¹³⁴³ *VVJ and another v Minister of Social Development and another* at par 12. The applicants argued that sec 40 is unconstitutional in that it discriminates unfairly against the applicants. At par 1 and 2 it states that sec 40 does not include the words "or life partner" after the word "spouse" and "husband" wherever such words appear in the section.

¹³⁴⁴ VVJ and another v Minister of Social Development and another at par 16.

¹³⁴⁵ *VVJ and another v Minister of Social Development and another* at par 17. "However, the more murky side of recognizing contributing partners, whether as a nurturing parent or as a contributing donor of gametes, while in a committed relationship, albeit without a ceremony that constitutes some form of union or a registered contract, still presents a problem."

¹³⁴⁶ *VVJ and another v Minister of Social Development and another* at par 18. The court pointed out that unions, civil unions, marriages and customary marriages also break down and the only difference is that through litigation and extension of the law, the parties and the children involved therein are protected.

¹³⁴⁷ VVJ and another v Minister of Social Development and another at par 27.

¹³⁴⁸ *VVJ and another v Minister of Social Development and another* at par 28. Inroads are being made to their right to a family life.

¹³⁴⁹ VVJ and another v Minister of Social Development and another at par 29.

his/her best interests considered of paramount importance is violated.¹³⁵⁰ The rights to dignity and equality of the children born of such artificial fertilisation are furthermore impeded, together with the rights to dignity and equality of the parties to a lifelong permanent partnership.¹³⁵¹ Section 40 of the Children's Act is thus declared inconsistent with the Constitution to the extent that the said section does not include the words "or permanent life partner" after the word "spouse" and "husband" wherever such words appear in section 40, "or permanent life partners" after the word "spouse" after the word "spouses" wherever such word appears in section 40.¹³⁵²

There is no doubt that this judgment should be welcomed, as it will definitely bring about a positive change to the Children's Act in that persons involved in a permanent life partnership will be acknowledged as parents of a child born through artificial fertilisation.

2.8.5 Surrogacy Advisory Group v Minister of Health¹³⁵³

In this Gauteng High Court judgment, Du Plessis AJ declared certain provisions of the 2012 Regulations Relating to the Artificial Fertilisation of Persons and the 2012 Regulations Relating to the Use of Human Biological Material unconstitutional.¹³⁵⁴ These Regulations, promulgated in terms of the NHA,¹³⁵⁵ are regulations 7(j)(ii), 13 and 19 of Regulations Relating to the Artificial Fertilisation of Persons.¹³⁵⁶ The respondent argued that the application is premature as the Minister did publish draft regulations on 25 March 2021 which will address many of the issues that were raised by the applicant.¹³⁵⁷ It was argued by the applicants that their right to equality, privacy and bodily integrity is infringed by the said regulations.¹³⁵⁸ Regulation 7(j)(i) provides that a psychological evaluation of both the gamete donor and recipient is a requirement

¹³⁵⁰ VVJ and another v Minister of Social Development and another at par 30.

¹³⁵¹ VVJ and another v Minister of Social Development and another at par 33.

¹³⁵² *VVJ and another v Minister of Social Development and another* at par 36. The court referred section 40 to the parliament for reconsideration.

¹³⁵³ (50683/2020) [2022] ZAGPPHC 558 (19 July 2022).

¹³⁵⁴ Surrogacy Advisory Group v Minister of Health at par 1.

¹³⁵⁵ 61 of 2003. Surrogacy Advisory Group v Minister of Health at par 1.

¹³⁵⁶ Surrogacy Advisory Group v Minister of Health at par 45. Reg 7(j)(ii) is the psychological evaluation requirement; Reg 13 is the sex selection prohibition; and Reg 19 is the prohibition of disclosure of certain facts.

¹³⁵⁷ Surrogacy Advisory Group v Minister of Health at par 17-18.

¹³⁵⁸ Surrogacy Advisory Group v Minister of Health at par 38.

for the removal or withdrawal of gametes in circumstances where the parties are known to each other.¹³⁵⁹ The applicants contended that there is no rationale for the parties to undergo a psychological evaluation where the gamete donor is the husband or partner of the recipient.¹³⁶⁰ This requirement thus infringes on the right to equality (section 9), the right to privacy (section 14) and the right of access to health care services (section 27(1)(a)).¹³⁶¹ The applicants contended that the discrimination is based on disability in terms of section 9(3) of the Constitution.¹³⁶² The applicants further argued that their right to privacy is infringed in that a psychological evaluation means an interview will be conducted by a clinical psychologist regarding their personal issues relating to their decisions to start a family through the process of artificial fertilisation.¹³⁶³ It was lastly stated that the applicants' right of access to health care is infringed in that the prerequisite for a psychological evaluation creates a financial and an emotional obstacle to their access to artificial fertilisation health care services.¹³⁶⁴ Section 27(1)(a) of the Constitution places a negative duty on the state to not limit access to health care:¹³⁶⁵

I accept that conceiving children through artificial fertilisation is an invasive and stressful procedure. It might involve risks and disappointment, which can impact the individuals and their relationships. Ensuring that parents who conceive children through artificial insemination is psychologically prepared is thus a legitimate government purpose and is not irrational or arbitrary.¹³⁶⁶

Du Plessis AJ concluded that regulation 7(j)(i) infringes on the parties' right to equality on the ground that infertility is a disability and that the discrimination is presumed unfair.¹³⁶⁷ The respondent did not provide the court with information as to the policy or why it can be said that it is reasonable to limit the rights of the applicants (and people in the same circumstances).¹³⁶⁸

The decision of people in a relationship to conceive a child through artificial fertilisation is within the truly person "(sic)" realm. It is close to the core of privacy, the most protected end of the continuum. And while it might be good and advisable for people to ensure that they have phycological "(sic)" support through the process as it can be a roller coaster ride

¹³⁵⁹ Surrogacy Advisory Group v Minister of Health at par 53.

¹³⁶⁰ Surrogacy Advisory Group v Minister of Health at par 55.

¹³⁶¹ Surrogacy Advisory Group v Minister of Health at par 56.

¹³⁶² Surrogacy Advisory Group v Minister of Health at par 64.

¹³⁶³ Surrogacy Advisory Group v Minister of Health at par 65.

¹³⁶⁴ Surrogacy Advisory Group v Minister of Health at par 68.

¹³⁶⁵ Surrogacy Advisory Group v Minister of Health at par 68.

¹³⁶⁶ Surrogacy Advisory Group v Minister of Health at par 79.

¹³⁶⁷ Surrogacy Advisory Group v Minister of Health at par 80-81.

¹³⁶⁸ Surrogacy Advisory Group v Minister of Health at par 80.

of unbounded hope and unmet expectations, a legal requirement to this effect is an infringement of their privacy. $^{\rm 1369}$

The court further found that the regulation infringes on a person's right to privacy and it constitutes a limitation on health care.¹³⁷⁰ Regarding regulation 13, the court observed that sex selection should be understood as a part of reproductive autonomy and the decision if and how to have children.¹³⁷¹

For Du Plessis AJ, two competing rights concerned with women's rights exist, namely non-medical sex selection that may encourage or lead to sex discrimination against women, vis-à-vis a prohibition on non-medical sex selection which violates a woman's right to reproductive autonomy.¹³⁷² The court next considered the interests of society as well as the interests of the child.¹³⁷³ Regarding the interests of children, the court's argument is that the complexity of future children's traits are reduced to their sex:¹³⁷⁴

In the case of sex selection, the moral issue is not only an individual moral issue but an issue that can impact society as a whole. It asks whether we as a society should allow people to choose the sex of their child and live with the possible consequences of such a choice (e.g. sex rations).¹³⁷⁵

The court found that section 12(2)(a) of the Constitution does not reduce a women's reproductive choices to only when an embryo is inside her body and the prohibition in terms of regulation 13 is an infringement of this right.¹³⁷⁶ In respect of the right to privacy, the court found that the regulation does infringe the right.¹³⁷⁷ Although the state can limit the right to choose the sex of a child, the state did not provide reasons for this, and neither did it try to justify the limitation in terms of section 36 of the Constitution.¹³⁷⁸ The court concluded that the argument regarding an infringement on the right to equality must fail.¹³⁷⁹ Regulation 6 of the Regulations Relating to the use of

¹³⁶⁹ Surrogacy Advisory Group v Minister of Health at par 86.

¹³⁷⁰ Surrogacy Advisory Group v Minister of Health at par 87 and 89.

¹³⁷¹ Surrogacy Advisory Group v Minister of Health at par 97 and 112. At par 113 "Preimplantation and prenatal testing for selecting the sex of a child is prohibited except in the case of a [sic] serious sex linked or sex-limited genetic conditions."

¹³⁷² Surrogacy Advisory Group v Minister of Health at par 103.

¹³⁷³ Surrogacy Advisory Group v Minister of Health at par 105 – 108.

¹³⁷⁴ Surrogacy Advisory Group v Minister of Health at par 108.

¹³⁷⁵ Surrogacy Advisory Group v Minister of Health at par 153.

¹³⁷⁶ Surrogacy Advisory Group v Minister of Health at par 158-159.

¹³⁷⁷ Surrogacy Advisory Group v Minister of Health at par 165.

¹³⁷⁸ Surrogacy Advisory Group v Minister of Health at par 165.

¹³⁷⁹ Surrogacy Advisory Group v Minister of Health at par 166.

Human Biological Material, which also deals with pre-implantation and prenatal testing for sex selection, was also declared unconstitutional and invalid.¹³⁸⁰

With reference to regulation 19, the court pointed out that the regulation puts a blanket ban on the disclosure of personal information by any persons involved in the surrogacy arrangement, including information relating to the gamete donor.¹³⁸¹ The applicants argued that this regulation infringes on the right to privacy and the right to freedom of expression of the parties concerned.¹³⁸² The court found that, insofar as it bars the parties involved in artificial fertilisation from sharing their experiences with family and friends, the prohibition is unconstitutional.¹³⁸³

Freedom of expression serves two important functions. Firstly, it is vital for the establishment of a democratic society. But it is also, secondly, an essential aspect of what it is to be human. It empowers individuals, gives them agency, and helps with informed decision-making. It has been regarded as a *sine qua non* for a person's right to realise their potential as a human being, which is important for every individual's empowerment to autonomous self-development. ¹³⁸⁴

The court concluded that regulation 19 does in fact infringe on the right to privacy and the right to freedom of expression.¹³⁸⁵ The court's conclusion has to be supported, especially in respect of regulations 7(j)(i) and 19. The court's "reading in" in terms of regulation 7(j)(i) of "except where such donor and recipient are a couple that is married or in a permanent domestic life-partnership"¹³⁸⁶ is accurate, as the couple in question would have taken an informed decision before approaching a fertility clinic. Requiring these parties to undergo psychological evaluation before proceeding with the artificial fertilisation procedure is pointless and an infringing on the privacy of and autonomy of their personal lives.

In respect of regulation 19, the court found that a reading down of the provision is the best remedy.¹³⁸⁷

¹³⁸⁰ Surrogacy Advisory Group v Minister of Health at par 195.

¹³⁸¹ Surrogacy Advisory Group v Minister of Health at par 170. At par 169 "No person shall disclose the identity of any person who donated a gamete or received a gamete, or any matter related to the artificial fertilisation of such gametes, or reproduction resulting from such artificial fertilisation except where a law provides otherwise or a court so orders."

¹³⁸² Surrogacy Advisory Group v Minister of Health at par 171.

¹³⁸³ Surrogacy Advisory Group v Minister of Health at par 178.

¹³⁸⁴ Surrogacy Advisory Group v Minister of Health at par 180.

¹³⁸⁵ Surrogacy Advisory Group v Minister of Health at par 181.

¹³⁸⁶ Surrogacy Advisory Group v Minister of Health at par 201.

¹³⁸⁷ Surrogacy Advisory Group v Minister of Health at par 199.

Therefore, interpreting "no person" to exclude the persons who undergo, donate towards, or result from artificial fertilisation themselves, would save it from unconstitutionality.¹³⁸⁸

There can be no doubt that this is correct, as it is each person's choice when and with whom he or she would like to share his or her experience, especially where a couple is undergoing artificial fertilisation. Because it is not only a physical experience but also involves many sensitive emotions, a couple or individual could always share their/his/her experience during the process with friends and family to enable them to receive the necessary support when they most need it. Draft regulations in terms of the NHA relating to artificial fertilisation of persons were published in 2016, March 2021 and November 2021.¹³⁸⁹ To date, the only regulations in force are the 2012 Regulations Relating to Artificial Fertilisation of Persons.

The framework of international and regional human rights will be briefly explored next, as these instruments may have a direct bearing on the Constitution and the interpretation of the rights pertaining to the parties to a surrogate agreement, not to mention the paramountcy of the best interests of a child.

3. THE INTERNATIONAL AND REGIONAL HUMAN RIGHTS FRAMEWORK RELEVANT TO SURROGATE MOTHERHOOD AND ARTIFICIAL FERTILISATION

The section below provides a brief overview of the most important human rights instruments relevant to surrogacy and artificial fertilisation. Although South Africa has pledged commitments to a range of international human rights instruments, two instruments that stand out because of their direct relevance to children, are the United Nations Convention on the Rights of the Child and the African Charter on the Rights and Welfare of the Child (ACRWC). The discussion will first briefly explore the relevance of the Convention for the regulation of surrogacy in South Africa. The preamble of the Convention on the Rights of the Child aptly states that:

The child, by reason of his physical and mental immaturity, needs special safeguards and care, including appropriate legal protection, before as well as after birth.¹³⁹⁰

¹³⁸⁸ Surrogacy Advisory Group v Minister of Health at par 199.

¹³⁸⁹ The different draft regulations will be discussed in ch 5 of the thesis.

¹³⁹⁰ United Nations Convention on the Rights of a Child, Preamble.

3.1 The United Nations Convention on the Rights of the Child (UNCRC)¹³⁹¹

South Africa signed the United Convention on the Rights of the Child charter in 1993 and thereafter ratified it on 16 June 1995.¹³⁹² This is the world's most ratified treaty which became the first legally binding international convention to confirm that all children are human rights bearers.¹³⁹³ The UNCRC considers it important that a child should be fully prepared to live an individual life in society and be brought up in the spirit of the ideals proclaimed in the Charter of the United Nations, and more specifically, in the spirit of peace, dignity, tolerance, freedom, equality and solidarity.¹³⁹⁴ As the UNCRC was the first legally binding international convention that declared human rights for all children,¹³⁹⁵ its aim can be described as setting out certain rights of children and reiterating the vulnerability of children.¹³⁹⁶

Rights that are provided for in the UNCRC are civil, political, economic, social and cultural rights.¹³⁹⁷ Article 1 defines a child as every human being below the age of

^{Adopted and opened for signature, ratification and accession by General Assembly resolution 44/25 of 20 November 1989 entry into force 2 September 1990 and in accordance with article 49. See C and Others v Department of Health and Social Development, Gauteng and Others (CCT 55/11) [2012] ZACC 1; 2012 (2) SA 208 (CC); 2012 (4) BCLR 329 (CC) at par 25.}

¹³⁹² See C and Others v Department of Health and Social Development, Gauteng and Others (CCT 55/11) [2012] ZACC 1; 2012 (2) SA 208 (CC); 2012 (4) BCLR 329 (CC) at par 25. Fast facts-United Nations Convention on the Riaht of the Child found on https://www.parliament.gov.za/storage/app/media/Pages/2019/november/19-11-2019 30 Year Commemoration of the United Nations Convention on the Rights of the Child/docs/FAST_FACTS_UNCRC_draft_2_19_November_2019final.pdf (last visited Child/converted and converted and convert 30 September 2022). Labuschaigne M, Mahomed S, and Dhai, A. "Evolving capacity of children and their best interests in the context of health research in South Africa: An ethico-legal position" 2022 Developing World Bioethics 3-4.

¹³⁹³ Fast facts- United Nations Convention on the Right of the Child found on <u>https://www.parliament.gov.za/storage/app/media/Pages/2019/november/19-11-</u> <u>2019_30_Year_Commemoration_of_the_United_Nations_Convention_on_the_Rights_of_the</u> <u>Child/docs/FAST_FACTS_UNCRC_draft_2_19_November_2019final.pdf</u> (last visited 30 September 2022). Labuschaigne M, Mahomed S, and Dhai, A. "Evolving capacity of children and their best interests in the context of health research in South Africa: An ethico-legal position" 2022 Developing World Bioethics 3-4.

¹³⁹⁴ CRC preamble.

¹³⁹⁵ Labuschaigne M, Mahomed S, and Dhai, A. "Evolving capacity of children and their best interests in the context of health research in South Africa: An ethico-legal position" 2022 *Developing World Bioethics* 3-4.

¹³⁹⁶ Fast facts- United Nations Convention on the Right of the Child found on <u>https://www.parliament.gov.za/storage/app/media/Pages/2019/november/19-11-</u>2019 30 Year Commemoration of the United Nations Convention on the Rights of the <u>Child/docs/FAST_FACTS_UNCRC_draft_2_19_November_2019final.pdf</u> (last visited 30 September 2022).

¹³⁹⁷ Fast facts- United Nations Convention on the Right of the Child found on <u>https://www.parliament.gov.za/storage/app/media/Pages/2019/november/19-11-</u> 2019 30 Year Commemoration of the United Nations Convention on the Rights of the

eighteen years unless under the law applicable to the child, majority is attained earlier.¹³⁹⁸ A child's rights, as provided for in the Convention, must be respected and ensured without any form of discrimination.¹³⁹⁹ The rights apply to all children, regardless of the child's, his or her parents' or his or her legal guardian's race, colour, sex, language, religion, political or other opinion, national, ethnic or social origin, property, disability, birth or other status.¹⁴⁰⁰ The best interests of a child shall be a primary consideration in all actions taken regarding children.¹⁴⁰¹

Protection and care of a child must be ensured, which are essential for his or her wellbeing.¹⁴⁰² Importantly, the child has a right to be registered immediately after birth.¹⁴⁰³ He or she shall have a right to a name (from birth), a right to acquire a nationality and, as far as it is possible, the right to know and be cared for by his or her parents.¹⁴⁰⁴ Each child has the right to have his or her identity preserved and this includes nationality, name and family relations as recognised by law without unlawful interference.¹⁴⁰⁵ A child shall not be exposed to arbitrary or unlawful interference with his or her privacy, family, home or correspondence.¹⁴⁰⁶ Further, a child shall not be exposed to unlawful attacks on his or her honour and reputation.¹⁴⁰⁷ The child has a right to protection of the law against any such interference or attack.¹⁴⁰⁸ Every child has the right to a standard of living that is adequate for his or her physical, mental, spiritual, moral and social development.¹⁴⁰⁹

¹⁴⁰⁰ Article 2(1) of CRC.

<u>Child/docs/FAST_FACTS_UNCRC_draft_2_19_November_2019final.pdf</u> (last_visited_30 September 2022).

¹³⁹⁸ Article 1 of CRC.

¹³⁹⁹ Article 2(1) of CRC.

¹⁴⁰¹ Article 3(1) of CRC.

¹⁴⁰² Article 3(2) of CRC. The rights and duties of the child's parents, legal guardian or other person legally responsible for him or her, must be taken into account, and to this end, State Parties shall take all appropriate legislative measure.

¹⁴⁰³ Article 7(1) of CRC.

¹⁴⁰⁴ Article 7(1) of CRC. Article 7(2) determines that the State Parties shall ensure that these rights are implemented in accordance with their national law and their obligations under the relevant international instruments in this field. This is also in particular where the child would otherwise be stateless. Skosana 2017 *Obiter* 267. Skosana argues that article 7(1) explicitly recognises a child's right to know its genetic origin.

¹⁴⁰⁵ Article 8 of CRC.

¹⁴⁰⁶ Article 16(1) of CRC.

¹⁴⁰⁷ Article 16(1) of CRC.

¹⁴⁰⁸ Article 16(2) of CRC.

¹⁴⁰⁹ Article 27(1) of CRC. Article 27(2) provides that the parent(s) or other responsible person have the primary responsibility to secure, within their own financial means, the conditions of living necessary for the child's development.

These entrenched rights in the UNCRC resonate with the manner in which the courts have considered the best interests of children when faced with applications for the confirmation of surrogate motherhood agreements. It is especially the best interests of the child yardstick that has informed the relevant decisions by the courts. By carefully considering this yardstick, as well as the relevant rights of the children that would have been born following the conclusion of a surrogate agreement, the courts have tacitly given effect to the provisions of the UNCRC discussed above.

3.2 The African Charter on the Rights and Welfare of the Child (ACRWC)¹⁴¹⁰

The African Charter on the Rights and Welfare of the Child as the most important regional human rights instrument relevant to children, was adopted on 11 July 1990 and entered into force on 29 November 1999.¹⁴¹¹ South Africa signed the ACRWC on 10 October 1997 and thereafter ratified it on 7 January 2000.¹⁴¹² The ACRWC recognises that a child occupies a unique and privileged position in the African society and that for the full and harmonious development of his or her personality, the child should grow up in a family environment in an atmosphere of happiness, love and understanding and further, that due to the needs of a child's physical and mental development, he or she requires particular care with regard to health, physical, mental, moral and social development and he or she requires legal protection in conditions of freedom, dignity and security.¹⁴¹³

A child is defined as a human being below the age of eighteen years.¹⁴¹⁴ A child is entitled to enjoyment of the rights and freedoms that are recognised and guaranteed in the Charter.¹⁴¹⁵ The "child's or his or her parents' or legal guardians' race, ethnic group, colour, sex, language, religion, political or other opinion, national and social

OAU Doc.CAB/LEG/24.9/49 (1990), entered into force Nov. 29, 1999. See C and Others v Department of Health and Social Development, Gauteng and Others (CCT 55/11) [2012] ZACC 1; 2012 (2) SA 208 (CC); 2012 (4) BCLR 329 (CC) at par 25.

OAU Doc.CAB/LEG/24.9/49 (1990), entered into force Nov. 29, 1999. See C and Others v Department of Health and Social Development, Gauteng and Others (CCT 55/11) [2012] ZACC 1; 2012 (2) SA 208 (CC); 2012 (4) BCLR 329 (CC) at par 25.

¹⁴¹² OAU Doc.CAB/LEG/24.9/49 (1990), entered into force Nov. 29, 1999. See C and Others v Department of Health and Social Development, Gauteng and Others (CCT 55/11) [2012] ZACC 1; 2012 (2) SA 208 (CC); 2012 (4) BCLR 329 (CC) at par 25.

¹⁴¹³ ACRWC preamble.

¹⁴¹⁴ Article 1 of ACRWC.

¹⁴¹⁵ Article 3 of ACRWC.

origin, fortune, birth or other status" does not have a bearing on the child's rights.¹⁴¹⁶ The best interest of a child shall be the primary consideration in all actions undertaken regarding the child.¹⁴¹⁷

A child has the right to privacy. Article 10 provides that no child shall be subject to arbitrary or unlawful interference with his privacy, family home or correspondence.¹⁴¹⁸ A child must not be subjected to attacks upon his honour or reputation, although his or her parents or legal guardians shall have the right to exercise reasonable supervision over the conduct of the child.¹⁴¹⁹ The child has the right to protection of the law against any such interference or attack.¹⁴²⁰

A notable difference between the UNCRC and the ACRWC is that the UNCRC regards a child's best interests as 'a' primary consideration, whereas the ACRWC considers it 'the' primary consideration in all actions taken regarding children. Whilst the UNCRC regards the best interests of a child as an important consideration, the ACRWC considers it to be *the most important* consideration. In surrogacy matters, the commissioned child's best interests should be *the* primary consideration, especially because the *child* is the primary reason for the existence of the surrogacy process. The court must be satisfied of the nature and scope of the child's rights and future circumstances before the surrogacy process can be proceeded with. The use of the surrogacy process is built on the foundation of the child to be born. If the commissioning parent(s) did not have the need or the longing for a child, surrogacy should not be an option for him/her/them. The emphasis in the ACRWC on the child's best interests as *the* primary consideration rings true for surrogacy arrangements, including artificial fertilisation procedures.

A comparison of the protections afforded to children by the UNCRC, the ACRWC and the South African Constitution, shows that whilst the UNCRC protects a child against discrimination in article 2(1), the South African Constitution provides similar protection

¹⁴¹⁶ Article 3 of ACRWC.

¹⁴¹⁷ Article 4 of ACRWC.

¹⁴¹⁸ Article 10 of ACRWC.

¹⁴¹⁹ Article 10 of ACRWC.

in section 9(1) and the ACRWC protects the child against discrimination in article 26.3. The UNCRC protects the liberty of a child in article 37, which is similarly protected in the South African Constitution in section 12(2)(b) and article 5, respectively. The Constitution makes specific reference to bodily and psychological integrity. The ACRWC entrenches the liberty of the child in article 17.2.

The right to a family is guaranteed by the UNCRC in article 8 and in the SA Constitution in section 28(1)(b), whereas the ACRWC provides in article 19.1 that a child is entitled to the enjoyment of parental care and protection and shall, whenever possible, have the right to reside with his or her parents. Interestingly, it is only the ACRWC that refers to the responsibility of a child towards his or her family and society (article 31). A child's right to privacy is protected in terms of section 14 of the Constitution, article 8 of the UNCRC and article 10 of the ACRWC. Finally, the principle of the best interests of a child is protected in section 28(2) of the Constitution, article 3.1 of the UNCRC and article 4 of the ACRWC.

The protection of children's rights in South Africa is strengthened by the similar provisions mirrored in the above-mentioned instruments. It is hoped that this strong international and legal framework will continue to guide South African courts in the interpretation of the rights of children in surrogacy arrangements and artificial fertilisation.

4. CONCLUSION

This chapter's focus on the constitutionality of some of the provisions in the Children's Act that have a bearing on children in surrogacy arrangements, including the international and regional human rights backdrop, reveals several shortcomings in the Children's Act and NHA that require revision.

The constitutional challenges to section 294 of the Children's Act in both *AB* and another *v* Minister of Social Development and *KB* and another *v* Minister of Social Development point to unfair discrimination against the pregnancy -and conception infertile persons who are excluded from concluding lawful surrogacy arrangements.

Establishing a genetical link to the commissioned child does not mean that it is and will be in that child's best interests. Refusing to expand the genetic link requirement of section 294 to include a genetic link between siblings is equally problematical. The High Court, in the case of *KB v Minister of Social Development*, could not find that section 294 of the Act is unconstitutional. The court further held that section 294 has nothing to do with the right of a minor child to have a sibling with the same genetic link and reiterated that the purpose of the section is to give clarity regarding the origin of a child.

A constitutional challenge to section 40 of the Children's Act happened twice in the Gauteng High Court—in the matter of *EJ and others v Haupt NO* and *VVJ and another v Minister of Social Development and another*. In the matter of *EJ and others v Haupt N.O*, the court concluded that section 40(1) is not unconstitutional as same-sex couples are included in the provisions of the section, whereas in the last-mentioned matter, section 40 was declared unconstitutional to the extent that it does not provide for a permanent life partner. As I have argued earlier, it is imperative that the provisions of the Children's Act accommodate the different forms of family that exist in present times.

Some medico-legal aspects relating to the regulation of surrogacy will be discussed in the next chapter. The discussion will turn more closely to infertility and the biological ability to reproduce, the advances in biotechnology and the different artificial fertilisation procedures. The difference between surrogacy and adoption will also be considered, as well as relevant ethical, social and religious issues. The interesting impact of epigenetics on the topic will also be explored.

CHAPTER 5

MEDICO-LEGAL ASPECTS RELATING TO THE REGULATION OF SURROGATE MOTHERHOOD

Medicine has made extraordinary advances in responding to the desire of women (and their partners) to have a child. These advances, as it may be imagined, have not been free from moral and legal difficulties.¹⁴²¹

1. INTRODUCTION

Medical technological advances create moral, legal and ethical dilemmas that have a real impact on the society.¹⁴²² The advances in obstetrics and gynaecology have created unexpected and unprecedented public interest in certain aspects of human reproduction, together with dilemmas not encountered before.¹⁴²³ Furthermore, the medical advances in treating infertility and in providing children to previously childless people, have brought together with the joy and happiness of parenting, thorny challenges regarding society's deeply held notions of what it means 'to parent' a child.¹⁴²⁴

This chapter focuses on the salient medico-legal issues relating to surrogate motherhood, turning first to relevant medical and clinical aspects of assisted reproduction, their key role in surrogacy, as well as legal aspects related to the clinical aspects. Although the private-law consequences of surrogate motherhood are primarily regulated by chapter 19 of the Children's Act, the artificial fertilisation of the surrogate mother, which is a medical or clinical procedure, is governed by the NHA and relevant regulations, discussed in this chapter.

¹⁴²¹ Kennedy and Grubb *Medical Law* 758.

¹⁴²² Nicholson, S and Nicholson, C "I used to have two parents and now I have three? When science (fiction) and the law meet: Unexpected complications" 2016 *Medicine and Law* 35 423. Skosana 2017 *Obiter* 261. The author states that medical and technological advancements often create new legal minefields which require the law to develop at the same level and pace as the medical and/or technological advancements.

¹⁴²³ Dada and McQuoid-Mason Introduction to Medico-Legal Practice 65.

¹⁴²⁴ Dada and McQuoid-Mason Introduction to Medico-Legal Practice 65.

As alluded to elsewhere in this thesis, the acceptance of novel medical technology to facilitate reproduction hinges on societal beliefs of, and expectations regarding parenthood. Surrogacy is part of social changes that invoke a different view of the modern family.¹⁴²⁵ As Nicholson explains, the modern family is no longer characterised by the narrow traditional view of family as a married heterosexual couple with a child or children, as society recognises that families may take an entirely new form, encompassing, amongst others, same sex couples.¹⁴²⁶ The notion of the family has developed to embrace the right of all individuals to reproduce and to draw upon the medical technologies to assist them to reproduce.¹⁴²⁷ Different kinds of family life that are constitutionally protected have altered over time as a result of change in social practices and traditions.¹⁴²⁸ The introduction of IVF and the benefits associated with it, meant that for the law, advances in technology vis-à-vis the protection of rights associated with reproduction have been extended beyond its previous range.¹⁴²⁹ The protection of fundamental rights is impacted by developments in technology, even more so where the state has sanctioned the technological development.¹⁴³⁰

¹⁴²⁵ Nicholson, S and Nicholson, C "I used to have two parents and now I have three? When science (fiction) and the law meet: Unexpected complications" 2016 *Medicine and Law* 35 424. Dada and McQuoid-Mason *Introduction to Medico-Legal Practice* 65. The authors argue that because of the ethical implications raised by creating and manipulating human embryos, legal responses to reproductive technologies are especially difficult. Some of the ethical issues raised are the type of assisted reproduction technology available; the disposal of abandoned embryos; what it means to be a parent and who should become one; the use of foetal oocytes in assisted reproduction etcetera.

¹⁴²⁶ Nicholson, S and Nicholson, C "I used to have two parents and now I have three? When science (fiction) and the law meet: Unexpected complications" 2016 *Medicine and Law* 35 424. Sloth-Nielson, J and van Heerden, B "The constitutional family developments in South African child and family law 2003-2013" 2014 *International Journal of Law, Policy and the Family* 28(1) 115. The authors state that the legal recognition of surrogate families entails a further step towards the diminution of biological connections as being the sole basis for constituting a family.

¹⁴²⁷ Nicholson, S and Nicholson, C "I used to have two parents and now I have three? When science (fiction) and the law meet: Unexpected complications" 2016 *Medicine and Law* 35 424 Sloth-Nielson, J and van Heerden, B "The constitutional family developments in South African child and family law 2003-2013" 2014 *International Journal of Law, Policy and the Family* 28(1) 115. The authors further argues that once we accept that the law provides a framework for an alternative where biology dictates that the members of the family unit cannot themselves reproduce, the definition of 'a family' can no longer be pegged to biological, adoptive or kinship bonds. Thus, the prior diagnostic tools for the identification of 'family life' are no longer appropriate.

¹⁴²⁸ AB and Another v Minister of Social Development at par 115.

¹⁴²⁹ AB and Another v Minister of Social Development at par 113. At par 114. Technological advancements make surrogacy a viable option for those who are both conception and pregnancy infertile and who could only have children by means of adoption. Sec 294 thus take this right to make a reproductive decision away.

¹⁴³⁰ AB and Another v Minister of Social Development at par 114. At par 126 the court pointed out that while surrogacy to have children is an ancient phenomenon, modern reproductive technologies have only recently enabled it to occur in the manner now governed by ch 19.

Even though in surrogacy, parentage and parenthood are shared by many for a period, the law has remained stuck on the very traditional understanding of legal parentage. As I have argued in the previous chapter, this understanding of parenthood is primarily based on the genetic relationship between a parent and a child:¹⁴³¹

The requirement for a genetic link in surrogacy sits uneasily alongside provisions that recognise increasing diversity in family life and that emphasise the importance of social or psychological parentage ahead of the genetic connection.¹⁴³²

Artificial fertilisation is a form of assisted reproduction that was made possible by the medical technological advances. Surrogacy as a form of assisted reproductive technology is increasingly used to overcome infertility,¹⁴³³ despite prior views by the medical profession that surrogacy should be an exceptional measure, considered as a last resort in view of all the medical, psychological and legal consequences associated with it.¹⁴³⁴ As defined in chapter 1 of this thesis, artificial fertilisation is a non-therapeutic procedure which bypasses the sterility of the male and the female without curing the cause of the sterility.¹⁴³⁵ As discussed in chapter 2, prior to the enactment of the NHA, the procedure was lawful, provided that the provisions of the now repealed Human Tissue Act and the applicable regulations at the time were adhered to.¹⁴³⁶ The Human Tissue Act controlled the removal and use of gametes and those obtained by removal or withdrawal was only allowed to be used for medical purposes or for artificial fertilisation.¹⁴³⁷ As stated already, the procedure is currently regulated by the NHA¹⁴³⁸ and the Regulations.¹⁴³⁹

¹⁴³¹ Bracken, L "Surrogacy and the Genetic Link" 2020 *Child and Family Law Quarterly* 32(3) 303.

¹⁴³² Bracken, L "Surrogacy and the Genetic Link" 2020 *Child and Family Law Quarterly* 32(3) 303.

¹⁴³³ Van Niekerk 2015 *PELJ* 398.

¹⁴³⁴ SALC Project 65 at par 7.6.3 and par 8.2.1.

¹⁴³⁵ LexisNexis *Family Law service* at par J107. SALRC Project 140 iii (summary of issue paper). Carstens and Pearmain *Foundational Principles* 181. The authors explain that assisted reproduction refer to fertility treatments which enhance the natural fertility of a person or cure their infertility and which is only one way of addressing difficulties with reproduction. The more extreme methods include AIH, AID and surrogate motherhood.

¹⁴³⁶ SALC Project 65 at par 4.2.1.

¹⁴³⁷ Sec 19 of Human Tissue Act. LexisNexis *Family Law service* at par J107.

¹⁴³⁸ 61 of 2003. Van Niekerk C "Strange (and incompatible) bedfellows: The relationship between the National Health Act and the regulations relating to artificial fertilisation of persons, and its impact on individuals engaged in assisted reproduction" 2017 *South African Journal of Bioethics and Law* 10(1) 32. The author points out that the Act and regulations has been characterised as 'flawed law'.

¹⁴³⁹ Regulations relating to artificial fertilisation of persons 2012. Van Niekerk 2017 SAJBL 33-35. Van Niekerk, referring to the Regulations Relating to Artificial Fertilisation of Persons (GN R1165 GG 40312) of 30 September 2016, argues that several flaws exist in the regulations which raises some concern for the use of assisted reproduction. One such flaw relates to reg 17 of the regulations. (The wording of reg 17 wording is similar to the wording of reg 17 of the 2012 regulations). It is assumed that reg 17 refers to instances where donor gametes have been used.

Despite initial objection to artificial fertilisation in the 1960s, it has become morally acceptable as a means of assisted reproduction in South Africa over the past five decades.¹⁴⁴⁰ An evolution in medical technology relating to artificial insemination over the past few decades has led to surrogacy becoming an option for infertile couples.¹⁴⁴¹

2. INFERTILITY AND THE BIOLOGICAL ABILITY TO REPRODUCE

2.1 Introduction

The inability of a couple to conceive a child may be because of failure to conceive or failure to carry an embryo to a viable physical state. Either of these can be ascribed to a number of physiological and psychological factors that causes infertility.¹⁴⁴² Infertility is described by the World Health Organisation as a "disease" of the male or female reproductive system, defined by the failure to fall pregnant after twelve months or more

The purpose of the notification of defects and illnesses in this regulation is unclear and raises the important question of whether liability will ensue, and if so, who will be held liable. The author mentioned a few other flaws as well as inconsistencies between the NHA and the Children's Act. She states that whilst the inconsistencies are not fatal as far as assisted reproduction is concerned, they unfortunately do suggest an incompatibility in the existing legislative framework. It has the potential to frustrate the persons who engage in this form of reproduction. Jordaan, DW "A Constitutional critique on the regulations relating to artificial fertilisation of persons" 2017 South African Journal of Bioethics and Law 10(1) 30. Jordaan, referring to the Regulations Relating to Artificial Fertilisation of Persons (GN R1165 GG 40312) of 30 September 2016, maintains that the way in which the regulations deal with the aspect of artificial fertilisation treat the different types of artificial insemination with gamete donation differently. The author further argues that gamete donation by a known donor and sperm donation are examples of over-medicalisation. As a competent person must be involved in the process of artificial fertilisation, although it is possible for a woman to inseminate at herself at home, the author regards the regulations as obstructing the right of persons who use artificial fertilisation to establish a family. He provides the example of where a lesbian couple obtaining the semen from a friend inseminate one of the women at home. Both will technically be liable to be prosecuted in terms of reg 21 because a competent person was not involved in either the fertilisation or the donation. The author also avers that medical influence and supervision over artificial fertilisation and gamete donation is legally enforced without a good and constitutionally aligned reason.

¹⁴⁴⁰ Carstens and Pearmain *Foundational Principles* 181-182. Skosana 2017 *Obiter* 263. Artificial insemination and surrogacy are the two recognised forms of artificial fertilisation in the Children's Act and the NHA.

¹⁴⁴¹ Nicholson, S and Nicholson, C "I used to have two parents and now I have three? When science (fiction) and the law meet: Unexpected complications" 2016 *Medicine and Law* 35 424.

Palm MT and Hirsch HL "Infertility and sterility: Legal implications of artificial conception" 1982 Medicine and Law 1 43. Infertility can be attributed to male causes, cervical causes, uterine causes, tubal and peritoneal causes and ovarian causes for example. It is also possible that infertility is of multiple etiology. Nosarka and Kruger 2005 SAMJ 942-943. It is explained that the most common absolute indications are patients with an absent uterus, congenital uterine abnormality, an inoperable scarred uterus and previous total abdominal hysterectomy. SALRC Project 140 14 and 106.

of regular unprotected sexual intercourse.¹⁴⁴³ Primary infertility is ascribed to a problem with the production of gametes or of the implantation of the embryo.¹⁴⁴⁴ Secondary infertility, on the other hand, occurs when at least one prior pregnancy has been achieved but is followed by the failure to fall pregnant.¹⁴⁴⁵

Ovum or sperm donation then becomes the secondary treatment of childlessness due to unsuccessful treatment of the primary condition.¹⁴⁴⁶ A couple's ability to produce is dependent upon normal interaction of a number of factors.¹⁴⁴⁷ The man must produce a large number of healthy sperm, which must traverse his reproductive system and be discharged into the healthy reproductive system of the woman.¹⁴⁴⁸ The sperm must penetrate the cervical mucus after deposit and ascend through the uterus to the fallopian tubes where fertilisation normally takes place.¹⁴⁴⁹ The woman must have an intact hypothalamic pituitary thyroid-ovarian-axis (functioning endocrine system) and she must develop and release an ovum into a patent and freely mobile fallopian tube for fertilisation.¹⁴⁵⁰ After fertilisation, the blastocyst must travel to the endometrial cavity where nidation must occur in an endometrium properly conditioned by hormone activity.¹⁴⁵¹ Unfortunately, a slight defect in any one of these basic events in either the man or the woman can lead to infertility.¹⁴⁵²

Factors which influence the prevalence of infertility are, among others, trends towards childbearing later in life; environmental factors such as infection from sexually transmitted diseases and occupational exposure; medical treatments such as those used for high blood pressure, stomach ulcer and cancer, including non-therapeutic

¹⁴⁴³ The definition is referred to to in ch 1, par 7.21.

¹⁴⁴⁴ The definition is referred to to in ch 1, par 7.21.

¹⁴⁴⁵ The definition is referred to to in ch 1, par 7.21.

¹⁴⁴⁶ Mason, McCall Smith and Laurie *Law and Medical Ethics* 68.

Palm MT and Hirsch HL "Infertility and sterility: Legal implications of artificial conception" 1982 Medicine and Law 1 43.

¹⁴⁴⁸ Palm MT and Hirsch HL "Infertility and sterility: Legal implications of artificial conception" 1982 *Medicine and Law* 1 43.

¹⁴⁴⁹ Palm MT and Hirsch HL "Infertility and sterility: Legal implications of artificial conception" 1982 *Medicine and Law* 1 43.

Palm MT and Hirsch HL "Infertility and sterility: Legal implications of artificial conception" 1982 Medicine and Law 1 43.

¹⁴⁵¹ Palm MT and Hirsch HL "Infertility and sterility: Legal implications of artificial conception" 1982 Medicine and Law 1 43.

¹⁴⁵² Palm MT and Hirsch HL "Infertility and sterility: Legal implications of artificial conception" 1982 *Medicine and Law* 1 43. Fortunately, many problems can be and are corrected surgically of by the administration of medications at 44. Artificial fertilisation and/or *in vitro* fertilisation can be used in cases that cannot be treated by the conventional means.

drugs such as narcotics, alcohol and tobacco.¹⁴⁵³ Infertility, however, may be present in one partner or both partners or it can be traced to biochemical or immunological incompatibility between partners.¹⁴⁵⁴ Most of female infertility is due to ovulation disorders; tubal blockage; endometriosis and other causes, including abnormalities of the vagina or cervix, and mucus incompatibilities with sperm.¹⁴⁵⁵ Female infertility can be treated and some of the treatments include hormone or drug therapy, surgery and medically assisted reproduction technologies.¹⁴⁵⁶ Male infertility, on the other hand, typically results from decreased numbers or an absence of sperm in the semen, abnormal motility and structural abnormalities and all of which prevent the normal fertilisation of an egg.¹⁴⁵⁷ In some cases, infertile couples can be treated by using their own genetic material and in other cases, the treatment will involve the use of donated material which can be sperm, eggs or embryos.¹⁴⁵⁸

2.2 Advances in biotechnology

Numerous advances have taken place in the sphere of human reproductive technology, as stated already.¹⁴⁵⁹ Artificial fertilisation is one of these advances. What started out as an effort to assist the infertile has transformed into a field that surpasses ordinary human expectations.¹⁴⁶⁰ Artificial fertilisation, broadly defined in chapter 1 of the thesis, involves from a clinical perspective the introduction of male sperm into a female patient by way of an intrauterine, paracervical, intravaginal or cervical cup deposition of pre-collected semen.¹⁴⁶¹ Assisted reproductive technologies have made it possible that a surrogate does not have to be biologically related to the embryo that

¹⁴⁵³ Kennedy and Grubb *Medical Law* 758.

¹⁴⁵⁴ Kennedy and Grubb *Medical Law* 758.

¹⁴⁵⁵ Kennedy and Grubb *Medical Law* 758-759.

¹⁴⁵⁶ Kennedy and Grubb *Medical Law* 759.

¹⁴⁵⁷ Kennedy and Grubb *Medical Law* 759.

¹⁴⁵⁸ Kennedy and Grubb *Medical Law* 759. The authors point out that it was the development of IVF procedures which raised concerns about the practices of medically assisted reproduction and which led to a call for regulation thereof. South African Research Council: *Guidelines on Ethics for Medical Research* 2nd Book (Reproductive Biology and Genetic Research) 2002 3

Quinn, CM "Mom, Mommy & Daddy and Daddy, Dad & Mommy: Assisted Reproductive Technologies & the Evolving Legal Recognition of Tri-Parenting" 2018 *Journal of the American Academy of Matrimonial Lawyers* 31(1) 176. Biko J and Nene Z "Ethics aspects of third-party reproduction" 2017 *Obstetrics & Gynaecology Forum* 3 12.

¹⁴⁶⁰ Van Niekerk 2017 *PELJ* 3.

¹⁴⁶¹ Palm MT and Hirsch HL "Infertility and sterility: Legal implications of artificial conception" 1982 *Medicine and Law* 1 44.

she is carrying through gestational surrogacy.¹⁴⁶² Yet, as discussed earlier in this thesis, different moral, legal and ethical controversies arise with respect to artificial fertilisation because the method of creating life is not entirely natural.¹⁴⁶³ ART's may require the use of reproductive resources, such as eggs, sperm or wombs from third parties who may not intend to play a role in the rearing of the child to be born.¹⁴⁶⁴

Heterosexual and same-sex couples, as well as single persons, who seek to have biologically related children often turn to clinics and agencies for sperm -or egg donors and gestational services.¹⁴⁶⁵ Relying on collaborative reproduction mostly involves persons unknown by the recipients, although some couples do prefer the use of a known third party.¹⁴⁶⁶ Reproduction is no longer limited to sexual intercourse as the range of technological advancements in this field and the increasing rate of infertility have made noncoital reproduction a common occurrence.¹⁴⁶⁷ As such, both artificial fertilisation and surrogacy may also be regarded as forms of noncoital reproduction.¹⁴⁶⁸

2.3 Assisted reproduction procedures

Chapter one of this thesis refers briefly to some of the procedures discussed in this chapter. Because of the complexities relating to these procedures, a clear understanding of these processes is necessary, as some of these clinical processes have related legal consequences that may impact on notions of legal parenthood in the context of surrogacy, not to mention those that may lead to medico-legal liability.

There is a difference between assisted reproductive technology (ART) and assisted reproduction procedures. ART specifically refers to fertility treatments in which eggs or

 ¹⁴⁶² Canner, S "Navigating surrogacy law in the non-united states: Why all states should adopt a uniform surrogacy statute" 2019 *Journal of Civil Rights and Economic Development* 33(2) 138.
 ¹⁴⁶³ Palm MT and Hirsch HL "Infertility and sterility: Legal implications of artificial concention" 1982

Palm MT and Hirsch HL "Infertility and sterility: Legal implications of artificial conception" 1982
 Medicine and Law 1 44.

¹⁴⁶⁴ SALRC Project 140 15. The medically assisted reproduction using third parties is also known as collaborative reproduction and it is widely recognised in many countries.

¹⁴⁶⁵ SALRC Project 140 15.

¹⁴⁶⁶ SALRC Project 140 16. The person can be unrelated or related. Where a family member is used, it is termed intrafamilial medically assisted reproduction which raises various ethical issues and it controversial.

¹⁴⁶⁷ Van Niekerk 2017 *PELJ* 3.

¹⁴⁶⁸ Van Niekerk 2017 *PELJ* 19.

embryos are handled,¹⁴⁶⁹ whereas assisted reproduction procedures refer to the different forms of artificial fertilisation and surrogacy.¹⁴⁷⁰ There are six different procedures of artificial fertilisation,¹⁴⁷¹ namely: artificial fertilisation of a wife using her husband's semen (AIH);¹⁴⁷² artificial fertilisation of a wife using a donor's semen (AID);¹⁴⁷³ uniting of the male and female gametes outside of the human body in a test tube and placing the resulting embryo in the womb of a female person (IVF and ET);¹⁴⁷⁴ gamete intra-fallopian transfer (GIFT);¹⁴⁷⁵ peritoneal oocyte and sperm transfer (POST);¹⁴⁷⁶ vaginal intro-peritoneal sperm transfer (VISPER)¹⁴⁷⁷ and surrogacy.¹⁴⁷⁸

2.3.1 Medical and legal risks

Performing IVF requires a degree of skill beyond that of the average specialist gynaecologist.¹⁴⁷⁹ The IVF¹⁴⁸⁰ and ET¹⁴⁸¹ process is not without any risks as a laparoscopy is done to remove the required ova.¹⁴⁸² This entails a risk of possible

¹⁴⁶⁹ See definition in par 7.4, ch 1.

¹⁴⁷⁰ Carstens and Pearmain *Foundational Principles* 181.

¹⁴⁷¹ Sec 1 of the National Health Act definition of artificial fertilisation (see definition in par 7.3, ch 1 above) includes artificial insemination, *in vitro* fertilisation, gamete intrafallopian tube transfer, embryo intrafallopian transfer or intracytoplasmic sperm injection.

¹⁴⁷² See par 7.18, ch 1 above for definition.

¹⁴⁷³ See par 7.19, ch 1 above for definition.

¹⁴⁷⁴ See par 7.5, ch 1 for definition. Palm MT and Hirsch HL "Infertility and sterility: Legal implications of artificial conception" 1982 *Medicine and Law* 1 48. IVF is when immature ova (oocytes) are removed from the prospective mother and placed in a culture medium and fertilised with sperm until the conceptus has reached the blastocyst stage. It is thereafter transplanted into the female's uterus where normal gestation thereafter proceeds. SALRC Project 140 14. At 183 it is also referred to as a form of medically assisted reproduction and it is stated that by 2012 an estimated 5 million babies were born because of medically assisted reproduction.

¹⁴⁷⁵ See par 7.20, ch 1 for definition. SALRC Project 140 14.

¹⁴⁷⁶ See par 7.20, ch 1 for definition.

¹⁴⁷⁷ See par 7.22, ch 1 above for definition.

¹⁴⁷⁸ See par 7.13, ch 1 above for definition. LexisNexis *Family Law service* at par J105. This procedure is when a woman, other than a man's wife, is impregnated with the man's semen by way of artificial fertilisation or by way of IVF and ET.

¹⁴⁷⁹ LexisNexis *Family Law service* at par J115. At present there is a limited number of gynaecologists worldwide who have acquired the necessary skill and have trained laboratory staff to assist them in conceiving a child *in vitro* and then successfully implanting the resulting blastocyst into the mother's uterus.

¹⁴⁸⁰ Mason, McCall Smith and Laurie Law and Medical Ethics 84. IVF is the standard treatment for childlessness because of blockage of the fallopian tubes. SALRC Project 140 14 – 15. It is stated that the success and availability of IVF raised the hopes of many infertile couples who have not been able to conceive because of infertility ascribed to blocked or absent fallopian tubes and male infertility and many more causes.

¹⁴⁸¹ Mason, McCall Smith and Laurie *Law and Medical Ethics* 84. Embryo transfer is the implantation of an embryo which has no genetic relationship either to the recipient or to her husband.

¹⁴⁸² LexisNexis *Family Law service* at par J115. Mason, McCall Smith and Laurie *Law and Medical Ethics* 85.

medical complications, including infection.¹⁴⁸³ It is possible that the recipient could be injured in the process of transplanting the blastocyst or the child may be born with defects as a result of the technical manipulations performed on the ova, or because it was conceived from a defective egg or sperm.¹⁴⁸⁴ The involvement of the gynaecologist when the procedure is completed places an additional responsibility on him or her in that he or she must conduct careful and detailed tests on the foetus to make sure that it is developing normally.¹⁴⁸⁵ Should the gynaecologist fail to do this, his or her conduct could amount to negligence and render him liable to a claim for damages should the child born be defective.¹⁴⁸⁶

2.3.2 AIH and AID

The use of AIH is medically indicated when the deposition of the husband's semen by coitus is prevented as a result of three possible factors: (i) anatomical or psychological problems; (ii) poor mobility, paucity or otherwise defective sperm cells; or (iii) too small a volume of ejaculate.¹⁴⁸⁷ The use of AID is medically indicated for psychologically fit, emotionally stable parents under circumstances of: (i) azoospermia (absolute male sterility); (ii) oligospermia (less than ten to fifteen million sperm per cubic centimetre semen with infertility of long duration); (iii) existence of some hereditary disease in the husband, which makes propagation inadvisable for eugenic reasons; and (iv) an Rh incompatibility which can be anticipated to cause an abnormal baby (when the incompatibility cannot be overcome by other techniques).¹⁴⁸⁸

Donor insemination (AID) introduces two additional concepts into the management of infertility.¹⁴⁸⁹ Firstly, the procedure expands the confines of a private matter between two persons in a close emotional relationship as it involves a third party whose

¹⁴⁸³ LexisNexis *Family Law service* at par J115.

¹⁴⁸⁴ LexisNexis *Family Law service* at par J115.

¹⁴⁸⁵ LexisNexis *Family Law service* at par J115.

¹⁴⁸⁶ LexisNexis *Family Law service* at par J115. It is also stated that an underqualified gynaecologist who attempts this technique will also expose himself to claims for damages. Nosarka and Kruger 2005 *SAMJ* 943. It is important to note that gynaecologists have no legal or moral obligation to partake in a surrogacy agreement.

¹⁴⁸⁷ Palm MT and Hirsch HL "Infertility and sterility: Legal implications of artificial conception" 1982 *Medicine and Law* 1 45.

¹⁴⁸⁸ Palm MT and Hirsch HL "Infertility and sterility: Legal implications of artificial conception" 1982 *Medicine and Law* 1 45.

¹⁴⁸⁹ Mason, McCall Smith and Laurie *Law and Medical Ethics* 76.

contribution lies at the heart of the enterprise.¹⁴⁹⁰ Secondly, the sperm donation may result in the production of an excess of gametes and, therefore, to the creation of options for their storage and later use.¹⁴⁹¹

2.2.3 Conditions for performing in vitro fertilisation (IVF)

The 2012 Regulations relating to artificial fertilisation of persons defines a competent person as a person who is registered as such in terms of the Health Professions Act¹⁴⁹² who is a medical practitioner who specialises in gynaecology with training in reproductive medicine, or a medical scientist, a medical technologist, and a clinical technologist with training in reproductive biology and related laboratory procedures.¹⁴⁹³ It is only a competent person that is allowed to remove or withdraw a gamete or cause a gamete to be removed or withdrawn from the body of the gamete donor for the purpose of artificial fertilisation.¹⁴⁹⁴ A donor of gametes may be compensated only for any reasonable expenses incurred by him or her in order to effect the donation.¹⁴⁹⁵ An electronic database shall be established by the Director-General where all the information regarding the donations of gametes and embryos will be kept.¹⁴⁹⁶

In circumstances where a maximum of six children have been conceived by using the gametes of the donor for artificial fertilisation, the competent person may not further

¹⁴⁹² 56 of 1974.

¹⁴⁹⁰ Mason, McCall Smith and Laurie *Law and Medical Ethics* 76. At 77 the authors point out there are persons who object strongly to AID on the grounds that the basis of the marriage bond is compromised by the wife's pregnancy through involving another man. They explain that adultery could not be held to have taken place in light of the fact that there was not sexual contact between the wife and the donor. Further, AID without the consent of the husband may be taken as constituting cruel and unreasonable conduct for divorce proceedings. South African Research Council: *Guidelines on Ethics for Medical Research* 2nd Book (Reproductive Biology and Genetic Research) 2002 3. The concerns in respect of a third-party gametes are mainly due to potential psychological problems, the risk of transmitting serious genetic disorders and the danger of transmitting infectious diseases, especially AIDS.

¹⁴⁹¹ Mason, McCall Smith and Laurie *Law and Medical Ethics* 76.

¹⁴⁹³ Reg 1 of Regulations relating to artificial fertilisation of persons 2012.

¹⁴⁹⁴ Reg 3(1) of Regulations relating to artificial fertilisation of persons 2012. Reg 3(2) determines that the removed gametes must be stored in a frozen state or cryopreserved.

¹⁴⁹⁵ Reg 4 of Regulations relating to artificial fertilisation of persons 2012. Biko J and Nene Z "Ethics aspects of third-party reproduction" 2017 Obstetrics & Gynaecology Forum 3 13. The author points out that the South African society of Reproductive Medicine recommends that donors of eggs be paid a stipend for the discomfort of undergoing the donation process. Importantly, the author stated that it is immoral and unethical should gametes be treated as commodities that are available to the highest bidder as this will make gamete donation inaccessible to the average citizen who really needs it.

¹⁴⁹⁶ Reg 5 of Regulations relating to artificial fertilisation of persons 2012.

remove or withdraw a gamete from that donor.¹⁴⁹⁷ In the event that the donor has conceived six children, the competent person shall inform the donor that he or she may not donate any further gametes.¹⁴⁹⁸ Gametes may not be removed or withdrawn from a donor if the competent person has not opened a gamete donor file (if one has not previously been opened for the donor):¹⁴⁹⁹ submitted the information obtained to the central data bank;¹⁵⁰⁰ and established from the central data bank that no more than six children have been conceived through artificial fertilisation of a person in using that donor's gametes (if it is a known donor).¹⁵⁰¹ Further, the donor must provide a signed statement where he or she states whether he or she has previously made a donation of gametes.¹⁵⁰² Informed consent must be obtained from the donor;¹⁵⁰³ the age of the donor must be determined¹⁵⁰⁴ and it must be established if the donor has undergone medical tests for sexually transmissible diseases and a semen analysis (male donor) on two occasions (not more than three months apart and one month prior to that donation of the gametes).¹⁵⁰⁵ The competent person must also ascertain if the female donor has undergone a gynaecological examination prior to stimulation for the withdrawal of the gametes:¹⁵⁰⁶ guestion the donor regarding his or her family history and any possible genetic conditions and mental illness¹⁵⁰⁷ and ensure that a written confirmation and a psychological evaluation of both parties are provided in the event that the donor and the recipient know each other.¹⁵⁰⁸

¹⁴⁹⁷ Reg 6(a) of Regulations relating to artificial fertilisation of persons 2012, Reg 6(c) determines that a competent person must immediately relay all the information in respect of that donor, the removal or withdrawal of a gamete and the artificial fertilisation to the central data bank.

¹⁴⁹⁸ Reg 6(b) of Regulations relating to artificial fertilisation of persons 2012.

¹⁴⁹⁹ Reg 7(a) of Regulations relating to artificial fertilisation of persons 2012. The competent person shall allocate a unique identification number in respect of the donor.

¹⁵⁰⁰ Reg 7(b) of Regulations relating to artificial fertilisation of persons 2012.

¹⁵⁰¹ Reg 7(c) of Regulations relating to artificial fertilisation of persons 2012.

¹⁵⁰² Reg 7(d) of Regulations relating to artificial fertilisation of persons 2012. The donor must indicate where and when such donation took place.

¹⁵⁰³ Reg 7(e) of Regulations relating to artificial fertilisation of persons 2012. The informed consent is in respect of a physical examination and questioning by a competent person, removal or withdrawal of a gamete for testing, analysing or other processing as may be deemed necessary by the competent person, the information as in terms of reg 8(1)(a)(ii), (iii) and (iv), (b), (c) and (f) must be made available to the recipient person and the competent person performing the artificial fertilisation, information as in terms of reg 8(2)(c) must be made available to the Director-General and the information as in terms of reg 8(2)(c) must be submitted to the central data bank.

¹⁵⁰⁴ Reg 7(f) of Regulations relating to artificial fertilisation of persons 2012.

¹⁵⁰⁵ Reg 7(g) of Regulations relating to artificial fertilisation of persons 2012.

 ¹⁵⁰⁶ Reg 7(h) of Regulations relating to artificial fertilisation of persons 2012.
 ¹⁵⁰⁷ Reg 7(i) of Regulations relating to artificial fertilisation of persons 2012. Information must be obtained regarding a possible genetic condition or carrier status thereof and mental illness in respect of any child, brother, sister, parent or grandparents of the donor.

¹⁵⁰⁸ Reg 7(j) of Regulations relating to artificial fertilisation of persons 2012.

Apart from the above prerequisites for the removal or withdrawal of gametes in terms of regulation 7, similar prerequisites are set out for the artificial fertilisation and embryo transfer in regulation 11. Before effecting the artificial fertilisation or embryo transfer, the competent person must ensure that a recipient file is opened if such file is not opened yet,¹⁵⁰⁹ informed consent must be obtained from the recipient,¹⁵¹⁰ it must be ensured that both the donor's and recipient's particulars and wishes are obeyed¹⁵¹¹ and that the recipient and donor is not a carrier of a serious genetic condition.¹⁵¹²

Regulation 12 determines that no more than three zygotes or embryos may be transferred to the recipient during an embryo transfer procedure except if a specific medical indication requires the contrary.¹⁵¹³ Regulation 14 determines that the competent person must record the request of the recipient in respect of the population and the religious group of the donor whose gametes will be used for the artificial fertilisation.¹⁵¹⁴ Regulation 17 provides for the testing on the child (born as a result of artificial fertilisation) by the authorised institution that effected the artificial fertilisation or embryo transfer if he or she displays any genetic disorder or suffers from a mental illness in order to ascertain whether the relevant disorder or mental illness results from the donor or the recipient.¹⁵¹⁵ It is important that a parent who discovers a disorder or mental illness in his or her child born as a result of artificial fertilisation.¹⁵¹⁶

¹⁵⁰⁹ Reg 11(a) of Regulations relating to artificial fertilisation of persons 2012. The recipient shall have a unique identification number allocated to him or her.

¹⁵¹⁰ Reg 11(b) of Regulations relating to artificial fertilisation of persons 2012. The consent relates to the physical examination and questioning by the competent person, removal or withdrawal of a gamete from the donor's body for purposes of testing, analysing or other processing of the gamete by the competent person, artificial fertilisation of, or the embryo transfer to herself, and information as in terms of reg 13(2)(c) must be made available to the central data bank.

¹⁵¹¹ Reg 11(c)(i) and (ii) of Regulations relating to artificial fertilisation of persons 2012.

¹⁵¹² Reg 11(c)(iii) of Regulations relating to artificial fertilisation of persons 2012. In the event that the recipient or donor should be a carrier of a serious genetic condition, they are tested to determine whether they are such genetic carriers and should both be such carriers or the donor is a carrier, a gamete of the donor may not be used for the artificial fertilisation of or the embryo transfer to the recipient.

¹⁵¹³ Reg 12 of Regulations relating to artificial fertilisation of persons 2012.

¹⁵¹⁴ Reg 14(1)(a)(iii) of Regulations relating to artificial fertilisation of persons 2012.

¹⁵¹⁵ Reg 17(1)(a) of Regulations relating to artificial fertilisation of persons 2012. Reg 17(1)(b) determines that the authorised institution that effected the donation of the gametes should be notified in writing of the disorder or mental illness and any tests that was carried out regarding the disorder or mental illness together with the results of the tests and their view on the disorder or the mental illness should the defect be traced back to the donor.

¹⁵¹⁶ Reg 17(2) of Regulations relating to artificial fertilisation of persons 2012.

Consent must be obtained before the gametes required for the artificial fertilisation, may be removed from the body of a living person.¹⁵¹⁷ In providing informed consent, such person from whom gametes are removed should have full knowledge and appreciation of the techniques, risks and consequences involved.¹⁵¹⁸

2.2.4 Prohibitions and offences

The NHA and Regulations provides for the control of and use of gametes and more specifically for artificial fertilisation of persons. Regulation 21 determines that it is an offence should any person contravene or fail to comply with any provision of the regulations.

Section 56 of the NHA determines that gametes may not be removed from a mentally ill person or a minor for any medical purpose.¹⁵¹⁹ Further, no gamete from a donor younger than eighteen years old may be used for artificial fertilisation.¹⁵²⁰ Disclosing the identity of any donor or any recipient or any matter related to the artificial fertilisation of such gametes or any reproduction that resulted from the artificial fertilisation, except in circumstances where a law provides otherwise or where a court so orders, is prohibited.¹⁵²¹

Not only are the suitably qualified medical practitioners involved with artificial fertilisation obliged to comply with the relevant provisions of the NHA and Regulations, but they must obtain their patients' informed consent before the insemination is carried out, otherwise it could, among others, be regarded as unprivileged touching.¹⁵²² Full informed consent requires that the medical practitioner must advise the couple of the medical risks and chances for success involved and the possibilities that an abnormal

¹⁵¹⁷ Sec 55(a) of the National Health Act (Previously sec 18 of the Human Tissue Act 65 of 1983).

¹⁵¹⁸ LexisNexis *Family Law service* at par J106. There is a rebuttable presumption that both persons gave their consent in this regard Palm MT and Hirsch HL "Infertility and sterility: Legal implications of artificial conception" 1982 *Medicine and Law* 1 46.

¹⁵¹⁹ Sec 56(2)(i) and (iii).

 ¹⁵²⁰ Reg 10(1)(c) of Regulations relating to artificial fertilisation of persons 2012. The gamete of a minor donor may only be used for artificial fertilisation in the case of a medical indication.
 ¹⁵²¹ Reg 10 of Regulations relating to artificial fertilisation of persons 2012.

Reg 19 of Regulations relating to artificial fertilisation of persons 2012.

¹⁵²² Palm MT and Hirsch HL "Infertility and sterility: Legal implications of artificial conception" 1982 *Medicine and Law* 1 46. The husband's consent to the AID is often the deciding factor in legal battles over custody and support should divorce follow the birth of the child.

child might be born despite all the precautions taken in selecting the donor and performing the procedure.¹⁵²³ Further, the couple should be advised of potential psychological effects which might affect them, the child or the marital and familial relationships.¹⁵²⁴ Couples who have doubts regarding the legal status of the child to be born and other legal complications which might result, are advised to consult an attorney that specialises in this field.¹⁵²⁵ Finally, should a medical practitioner artificially fertilises a woman without her consent, and the artificial fertilisation is carried out in secret, it will constitute an invasion of her bodily integrity and therefore amounts to assault.¹⁵²⁶ Such action will also constitute a violation of the woman's constitutional and common law personality rights, which could lead to a claim for damages.¹⁵²⁷ In the event that a medical practitioner is negligent in the selection of semen causing a defective child to be born, the medical practitioner could possibly also face an action for wrongful life.¹⁵²⁸

2.2.5 The donor

A gamete donor is defined as a living person from whose body a gamete or gametes are removed or withdrawn for the purpose of artificial fertilisation.¹⁵²⁹ The competent person¹⁵³⁰ must, before removal or withdrawal of a gamete, ascertain the age of the donor;¹⁵³¹ that the donor has undergone medical tests for sexually transmissible diseases; a semen analysis (male donor);¹⁵³² and a female donor has undergone a gynaecological examination prior to stimulation for the withdrawal of gametes.¹⁵³³

¹⁵²³ Palm MT and Hirsch HL "Infertility and sterility: Legal implications of artificial conception" 1982 *Medicine and Law* 1 46.

¹⁵²⁴ Palm MT and Hirsch HL "Infertility and sterility: Legal implications of artificial conception" 1982 *Medicine and Law* 1 46.

¹⁵²⁵ Palm MT and Hirsch HL "Infertility and sterility: Legal implications of artificial conception" 1982 *Medicine and Law* 1 46.

¹⁵²⁶ LexisNexis Family Law service at par J119. Dhai and McQuoid-Mason Bioethics, Human Rights and Health Law 144. The medical practitioner can be charged with sexual assault under the Sexual Offences and Related Matters Amendment Act 32 of 2007.

¹⁵²⁷ Dhai and McQuoid-Mason *Bioethics, Human Rights and Health Law* 144.

¹⁵²⁸ Dhai and McQuoid-Mason *Bioethics, Human Rights and Health Law* 144. The authors use the examples of obtaining semen from a donor who suffers from venereal disease or where the donor is too closely related to the recipient.

¹⁵²⁹ Reg 1 of Regulations relating to artificial fertilisation of persons 2012.

¹⁵³⁰ See definition in par 5.4.3 above.

¹⁵³¹ Reg 7(f) of Regulations relating to artificial fertilisation of persons 2012.

¹⁵³² Reg 7(g)(i) and (ii) of Regulations relating to artificial fertilisation of persons 2012.

¹⁵³³ Reg 7(h) of Regulations relating to artificial fertilisation of persons 2012.

Further, the competent person must enquire about the donor's family history and regarding possible genetic conditions.¹⁵³⁴

Where the donor and the recipient are known to each other, the competent person must ensure that a written confirmation is received by both the persons and a psychological evaluation is done involving both of them.¹⁵³⁵ The recording of the information is important and the necessary information, as determined by regulation 8(1)(a) to 8(g), must be recorded and the relevant documentation must be in the donor's file before a gamete is withdrawn or removed from the donor.¹⁵³⁶ Certain particulars must be made available to the recipient and the competent person who will perform the artificial fertilisation.¹⁵³⁷ The competent person is not allowed to make the gamete donor file, or any information contained therein, available to any person except someone acting under his or her supervision and except in terms of legislation or a court order.¹⁵³⁸

2.2.6 The recipient

The regulations relating to the artificial fertilisation of persons define a recipient as:

a female person in whose reproductive organs a male gamete or gametes are to be introduced by other than natural means; or in whose uterus/womb or fallopian tubes a zygote or embryo is to be placed for the purpose of human reproduction.¹⁵³⁹

Regulations 11 and 14 sets out the provisions in respect of the recipient. The recipient must provide the competent person with informed consent in respect of a physical examination; the removal or withdrawal of gametes for testing; to undergo the artificial fertilisation or embryo transfer; and that her information may be made available to the central data bank.¹⁵⁴⁰ As stated already, a maximum of three zygotes or embryos may

¹⁵³⁴ Reg 7(i) of Regulations relating to artificial fertilisation of persons 2012.

¹⁵³⁵ Reg 7(j)(i) and (ii) of Regulations relating to artificial fertilisation of persons 2012.

¹⁵³⁶ Reg 8(1) of Regulations relating to artificial fertilisation of persons 2012.

¹⁵³⁷ Reg 8(2)(b) of Regulations relating to artificial fertilisation of persons 2012. Particulars in terms of reg 8(1)(a)(ii), (iii) and (iv), (b), (c) and (f); and the identification number (reg 7(a)).

¹⁵³⁸ Reg 8(2)(d) of Regulations relating to artificial fertilisation of persons 2012.

¹⁵³⁹ Reg 1 of Regulations relating to artificial fertilisation of persons 2012.

¹⁵⁴⁰ Reg 11(b)(i) to (iv) of Regulations relating to artificial fertilisation of persons 2012.

be transferred to the recipient during an embryo transfer procedure, except in circumstances where there is a specific medical indication to the contrary.¹⁵⁴¹

3. SURROGACY VERSUS ADOPTION

Surrogacy provides a would-be parent with an opportunity to have a child that is biologically linked to him or her. Although another woman will carry the child for the commissioning mother, surrogacy remains a means through which the commissioning parent(s) can have a family that includes a child who is biologically related to them.¹⁵⁴² Children conceived by artificial fertilisation in a surrogacy arrangement are described as children who are brought into a loving and caring environment.¹⁵⁴³ Further, these children are more likely to be better adjusted than the children conceived normally and born unwanted into an undesirable environment.¹⁵⁴⁴ In surrogacy, "life is not cheapened by this process, it is created and cherished."¹⁵⁴⁵

Adoption, on the other hand, is a juristic process through which parental authority over a child is terminated and it is vested in the adoptive parents.¹⁵⁴⁶ Adoption has been suggested as an alternative to surrogacy for those persons who do not meet the genetic prerequisite of section 294.¹⁵⁴⁷ Adoption, in contrast to surrogacy, is subject to age restrictions, not to mention the fact that the supply of babies for adoption is far outstripped by the demand.¹⁵⁴⁸ As a result of the adoption process having limitations of its own, it may exclude certain persons from becoming parents, which suggests that adoption may not provide a possibility for certain infertile persons.¹⁵⁴⁹ Adoption is different from surrogacy in that surrogacy agreements are concluded prior to the child's birth, whereas adoption in most cases happens after the child's birth.¹⁵⁵⁰ Thus, in surrogacy the commissioning parents are guaranteed a new-born child compared to

¹⁵⁴¹ Reg 12 of Regulations relating to artificial fertilisation of persons 2012.

¹⁵⁴² Nicholson, S and Nicholson, C "I used to have two parents and now I have three? When science (fiction) and the law meet: Unexpected complications" 2016 *Medicine and Law* 35 424.

¹⁵⁴³ SALC Project 65 at par 2.3.1.

¹⁵⁴⁴ SALC Project 65 at par 2.3.1.

¹⁵⁴⁵ SALC Project 65 at par 2.3.1. The SALC referred to this quote from Eaton.

¹⁵⁴⁶ Report of the Ad Hoc Committee 9.

¹⁵⁴⁷ Van Niekerk 2015 *PELJ* 414.

¹⁵⁴⁸ Nicholson 2013 SAJHR 497. Louw 2013 THRHR 571.

¹⁵⁴⁹ Van Niekerk 2015 *PELJ* 415.

¹⁵⁵⁰ Van Niekerk 2015 *PELJ* 415.

adoption where a new-born child is not a guarantee.¹⁵⁵¹ Van Niekerk describes the most important difference between adoption and surrogacy as follows: with adoption, the issue is "how best to care for the adoptive child now that the birth family cannot", whereas with surrogacy it is "how best to bring a child into the world and into a family that desires a child".¹⁵⁵² The notion that one is an alternative for the other is a huge misconception given the differences between the processes.¹⁵⁵³

There are thus clear and important psychological differences between becoming a parent through adoption and having a child through surrogacy.¹⁵⁵⁴ In the case of *AB* and another *v* Minister of Social Development, Khampepe J rightly observes that:

While the end result of both processes is that a person becomes the parent of a child genetically unrelated to them, the nature of the relationship between parent and child is substantially different.¹⁵⁵⁵

It was accepted in the matter of *AB* and another *v* Minister of Social Development¹⁵⁵⁶ that an emotionally significant parent-child bond may develop between the commissioning parent and the child during the pregnancy.¹⁵⁵⁷ Thus, this difference in process may have implications for how the commissioning parent and the child relate to one another once the child is born, a difference that goes beyond the superficial similarity that both processes result in a commissioning parent(s) having a child that is not genetically related to them.¹⁵⁵⁸ Moreover, with adoption, the infertile couple are placed on a long waiting list after going through a complex selection process,¹⁵⁵⁹ whereas in the case of surrogacy, the couple may only need to wait the nine months of normal gestation in the event of a successful surrogacy agreement.¹⁵⁶⁰

¹⁵⁵¹ Van Niekerk 2015 *PELJ* 415.

¹⁵⁵² Van Niekerk 2015 *PELJ* 415.

¹⁵⁵³ Van Niekerk 2015 *PELJ* 415.

¹⁵⁵⁴ AB and another v Minister of Social Development at par 180.

¹⁵⁵⁵ AB and another v Minister of Social Development at par 180. At par 181-182 the court explained the difference between adoption and surrogacy and stated that the difference is especially important against the backdrop of infertility. Expert evidence shows that the choices made during the surrogacy process often have a strong psychological healing function that ameliorates the negative psychological effects of infertility. The second difference is that the commissioning parents are often intimately involved in the surrogate's pregnancy process.

¹⁵⁵⁶ 2017 (3) BCLR 267 (CC); 2017 (3) SA 570 (CC).

¹⁵⁵⁷ AB and another v Minister of Social Development at par 184.

¹⁵⁵⁸ AB and another v Minister of Social Development at par 184-185.

¹⁵⁵⁹ SALC Project 65 at par 2.3.2.

¹⁵⁶⁰ SALC Project 65 at par 2.3.2. It was argued that adoption provides a solution for a situation that already exist whereas surrogacy creates a situation that is prone to all kinds of problems.

There might be different reasons for different persons for wanting to use a surrogate mother. This thesis has already explained that surrogacy has become a possible option for the growing number of infertile couples and individual persons who wish to have children of their own.¹⁵⁶¹ Surrogacy provides the commissioning parent(s) with the chance to experience the pregnancy with the surrogate and to make sure that the baby receives the appropriate prenatal care.¹⁵⁶² In contrast to adoption, surrogacy intentionally brings into the world a child whose genealogy (pedigree) is blurred.¹⁵⁶³ It is clear from the provisions of section 295(a) of the Children's Act that there is no closed list for what the possible causes of the inability to give birth must be, just as long as the condition is permanent and irreversible,¹⁵⁶⁴ for example, where a woman does not have a uterus;¹⁵⁶⁵ has a structural abnormality to her uterus or has fibroids or scar tissue inside the uterus; or has no ova (eggs) or unhealthy ova.¹⁵⁶⁶ The question arises whether a commissioning mother who is fertile but who suffers from a permanent illness, would fall under this provision. Other examples are where the mother has a medical condition that would make pregnancy dangerous, for example, severe heart disease; kidney disease; diabetes; severe preeclampsia or a history of breast cancer.¹⁵⁶⁷ In the judgment of APP and another v NKP,¹⁵⁶⁸ Bozalek J concludes that a narrow interpretation of section 295(a) would be at odds with the apparent purpose of chapter 19 which were designed to provide an opportunity to persons who would not otherwise be able to have a child genetically related to them to do so through the use of surrogacy.¹⁵⁶⁹ The term 'not able to give birth' was interpreted as being unable to give birth without a significant medical risk to the health or the life of the mother.¹⁵⁷⁰ I agree with the interpretation of Bozalek J that section 295(a) must not be interpreted narrowly. It is simply unreasonable, unfair and unjustified to force a woman with a

¹⁵⁶¹ SALRC Project 140 81. Iannacci, B.R "Why New York should legalize surrogacy: A comparison of surrogacy legislation in other states with current proposed surrogacy legislation in New York" 2018 *Touro Law Review* 34(4) 1242. Surrogacy is the only option for male couples should they want to have genetically linked children.

¹⁵⁶² Iannacci, B.R "Why New York should legalize surrogacy: A comparison of surrogacy legislation in other states with current proposed surrogacy legislation in New York" 2018 *Touro Law Review* 34(4) 1242.

¹⁵⁶³ SALC Project 65 at par 2.3.2.

¹⁵⁶⁴ Discussion in par 3.5.4 above.

¹⁵⁶⁵ SALRC Project 140 81.

¹⁵⁶⁶ SALRC Project 140 82.

¹⁵⁶⁷ SALRC Project 140 82.

¹⁵⁶⁸ 2021 JDR 1650 (WCC)/(17962/2020) [2021] ZAWCHC 69 (11 March 2021). The case was discussed in detail in par 3.10.1 above.

¹⁵⁶⁹ APP and another v NKP at par 24.

¹⁵⁷⁰ APP and another v NKP at par 25.

permanent illness or medical condition to carry a baby to term and to give birth to that baby knowing the risks to both the woman and the baby. The husband and/or partner will be forced to face the risk of losing both the mother and the child. Regulations on this exact issue, namely of what constitutes a permanent inability to give birth is needed. In other words, is it a physical condition that is required, or another kind of risk with similar dire consequences?

4. ETHICAL, SOCIAL AND RELIGIOUS ISSUES

It is apt to start this section with the following quote from the Law Commission on the impact of surrogacy:

Whereas different family patterns have developed over the centuries, surrogacy was thrust unexpectedly on humanity. The formulation of adequate policy and procedures to the benefit of those concerned and society is hampered by the rapidity of developments, the complicated impact of surrogacy on established values and the uncertainty with regard to human behaviour that emanate from the new developments.¹⁵⁷¹

Disagreements regarding the morality of aspects relating to surrogacy arrangements are thus very likely,¹⁵⁷² as problems relating to reproductive health are regularly influenced by values, culture, traditions, social practices and laws.¹⁵⁷³

Dhai and McQuoid-Mason regard reproductive health not just a health issue, but also an issue of development and human rights.¹⁵⁷⁴ As a diverse nation with different population and religious groups, ethical values for different groups in the community may differ sharply—values that are acceptable to one group may not be acceptable to another group.¹⁵⁷⁵ Pyrce explains that when surrogacy turns the traditional ideas of parentage upside down, countries struggle to find effective regulations that protect their own citizens while still recognising the increasingly global nature of modern society.¹⁵⁷⁶

Biko and Nene discuss the dilemma that arises when patients have to abide by the rules of religions and faith, despite third party (donor-assisted) reproduction being

¹⁵⁷¹ SALC Project 65 at par 6.4.2.

¹⁵⁷² SALC Project 65 at par 2.2.1

¹⁵⁷³ Dhai and McQuoid-Mason *Bioethics, Human Rights and Health Law* 139.

¹⁵⁷⁴ Dhai and McQuoid-Mason *Bioethics, Human Rights and Health Law* 139.

¹⁵⁷⁵ SALC Project 65 at par 2.2.2.

¹⁵⁷⁶ Pyrce, C "Surrogacy and citizenship: A conjunctive solution to a global problem" 2016 *Indiana Journal of Global Legal Studies* 23(2) 927.

permitted in law.¹⁵⁷⁷ Some religious beliefs proscribe the conception of a child through any means other than through sexual intercourse within a lawful marriage.¹⁵⁷⁸ ART, for example, is not accepted by the Vatican as an appropriate means of creating a family and they further do not accept third party reproduction.¹⁵⁷⁹ The Catholic Church (Catholicism) teaches that children are a gift from God and the child must be conceived and carried naturally by the married woman and her husband which means that the adding of a third party to the process is immoral.¹⁵⁸⁰ They believe that modern fertility procedures do harm to the dignity of a person and the institution of marriage.¹⁵⁸¹

The Anglican and Protestant faiths appear not to object to either assisted or third-party reproduction.¹⁵⁸² Islamic law regards surrogacy as synonymous to adultery which means that the child born as a result thereof does not have a legal lineage.¹⁵⁸³ More modern Muslims, however, believe that the process of surrogacy and IVF can be

¹⁵⁷⁷ Biko J and Nene Z "Ethics aspects of third-party reproduction" 2017 *Obstetrics & Gynaecology Forum* 3 14. "Believes, religion and faith are not usually subjected to ethical principles but in secular countries like South Africa, patients have a choice to practice the faith or to reject it."

¹⁵⁷⁸ SALC Project 65 at par 2.2.3. Biko J and Nene Z "Ethics aspects of third-party reproduction" 2017 Obstetrics & Gynaecology Forum 3 14.

¹⁵⁷⁹ Biko J and Nene Z "Ethics aspects of third-party reproduction" 2017 *Obstetrics & Gynaecology Forum* 3 14. Deonandan, R "Thoughts on the ethics of gestational surrogacy: perspectives from religions, Western liberalism, and comparisons with adoption" 2020 *Journal of Assisted Reproduction and Genetics* 37 269–279.

¹⁵⁸⁰ What does religion say about the morality of surrogacy? <u>https://www.americansurrogacy.com/blog/what-does-religion-say-about-the-morality-of-</u> <u>surrogacy/#:~:text=Religious%20Views%20on%20Surrogacy&text=Instead%2C%20the%20Ch</u> <u>urch%20teaches%20that,this%20process%20is%20considered%20immoral</u>. (11 December 2017) (accessed on 5 January 2023). Sukhanova Anna (2019) "Surrogacy and religion" <u>https://surrogacybypons.com/surrogacy-and-religion/ (accessed 5 January 2023)</u>.

¹⁵⁸¹ Deonandan, R "Thoughts on the ethics of gestational surrogacy: perspectives from religions, Western liberalism, and comparisons with adoption" 2020 *Journal of Assisted Reproduction and Genetics* 37 269–279.

¹⁵⁸² Biko J and Nene Z "Ethics aspects of third-party reproduction" 2017 *Obstetrics & Gynaecology Forum* 3 14.

¹⁵⁸³ SALC Project 65 at par 2.2.8. Biko J and Nene Z "Ethics aspects of third-party reproduction" 2017 Obstetrics & Gynaecology Forum 3 14. Third party reproduction is completely outlawed in countries, such as Egypt and Saudi Arabia because the Sunni Muslims are in majority in those countries. In countries such as Iran and Lebanon third party reproduction are permissible and there, the Shi'ite Muslim are in majority. What does religion say about the morality of surrogacy? <u>https://www.americansurrogacy.com/blog/what-does-religion-say-about-the-morality-ofsurrogacy/#:~:text=Religious%20Views%20on%20Surrogacy&text=Instead%2C%20the%20Ch urch%20teaches%20that,this%20process%20is%20considered%20immoral. (11 December 2017) (accessed on 5 January 2023). Deonandan, R "Thoughts on the ethics of gestational surrogacy: perspectives from religions, Western liberalism, and comparisons with adoption" 2020 Journal of Assisted Reproduction and Genetics 37 269–279.</u>

allowed as long as the semen and the ovum come from a married couple.¹⁵⁸⁴ In general, Islam accepts ART but is opposed to surrogacy.¹⁵⁸⁵

The liberal religious groups of the Jewish faith (Judaism) accepts ART and third-party reproduction, as Jewishness is seen to be conferred by the mother through the act of gestation and giving birth to the child.¹⁵⁸⁶ The conservative Jewish groups, on the other hand, do not approve of the process of surrogacy.¹⁵⁸⁷ Views differ regarding the use of donor sperm and legitimacy remains an issue.¹⁵⁸⁸ It would seem that their belief is that the commissioning parents are excluded from recognised parentage if they have no genetic link and no physical role in the arrival of the child into the world.¹⁵⁸⁹

Buddhism accepts the use of surrogacy because it does not regard procreation as a moral duty.¹⁵⁹⁰ The Buddhist societies appear to be liberal with regard to the aspect of

¹⁵⁸⁴ What does religion say about the morality of surrogacy? <u>https://www.americansurrogacy.com/blog/what-does-religion-say-about-the-morality-of-surrogacy/#:~:text=Religious%20Views%20on%20Surrogacy&text=Instead%2C%20the%20Church%20teaches%20that,this%20process%20is%20considered%20immoral. (11 December 2017) (accessed on 5 January 2023). Sukhanova Anna (2019) "Surrogacy and religion" https://surrogacybypons.com/surrogacy-and-religion/ (accessed 5 January 2023).</u>

¹⁵⁸⁵ Deonandan, R "Thoughts on the ethics of gestational surrogacy: perspectives from religions, Western liberalism, and comparisons with adoption" 2020 *Journal of Assisted Reproduction and Genetics* 37 269–279.

¹⁵⁸⁶ Biko J and Nene Z "Ethics aspects of third-party reproduction" 2017 Obstetrics & Gynaecology Forum 3 15. What does religion say about the morality of surrogacy? <u>https://www.americansurrogacy.com/blog/what-does-religion-say-about-the-morality-of-surrogacy/#:~:text=Religious%20Views%20on%20Surrogacy&text=Instead%2C%20the%20Ch urch%20teaches%20that,this%20process%20is%20considered%20immoral. (11 December 2017) (accessed on 5 January 2023).</u>

¹⁵⁸⁷ What does religion say about the morality of surrogacy? <u>https://www.americansurrogacy.com/blog/what-does-religion-say-about-the-morality-of-surrogacy/#:~:text=Religious%20Views%20on%20Surrogacy&text=Instead%2C%20the%20Church%20teaches%20that,this%20process%20is%20considered%20immoral. (11 December 2017) (accessed on 5 January 2023). Sukhanova Anna (2019) "Surrogacy and religion" https://surrogacybypons.com/surrogacy-and-religion/ (accessed 5 January 2023).</u>

¹⁵⁸⁸ Deonandan, R "Thoughts on the ethics of gestational surrogacy: perspectives from religions, Western liberalism, and comparisons with adoption" 2020 *Journal of Assisted Reproduction and Genetics* 37 269–279.

¹⁵⁸⁹ Deonandan, R "Thoughts on the ethics of gestational surrogacy: perspectives from religions, Western liberalism, and comparisons with adoption" 2020 *Journal of Assisted Reproduction and Genetics* 37 269–279.

¹⁵⁹⁰ What does religion say about the morality of surrogacy? <u>https://www.americansurrogacy.com/blog/what-does-religion-say-about-the-morality-of-surrogacy/#:~:text=Religious%20Views%20on%20Surrogacy&text=Instead%2C%20the%20Church%20teaches%20that,this%20process%20is%20considered%20immoral. (11 December 2017) (accessed on 5 January 2023). Sukhanova Anna (2019) "Surrogacy and religion" <u>https://surrogacybypons.com/surrogacy-and-religion/</u> (accessed 5 January 2023).</u>

ART.¹⁵⁹¹ Christianity reveals a few examples of what can be referred to as "traditional surrogacy" deriving from the Bible, for example, Bilha (Jacob's sister-in-law) and Jacob and Hagar (the servant) and Abram.¹⁵⁹² Today, the surrogate will be fertilised through IVF and not through the sexual act.¹⁵⁹³ Christianity believes that marriage is between two people and children should be born from that unity.¹⁵⁹⁴ It is believed that to have children is not a right but a gift from God and to use surrogacy because of arrogance against God, constitutes a sin. However, if surrogacy is used after seeking God's will and guidance, it will be seen as an acceptable alternative for the infertile person.¹⁵⁹⁵ Christianity thus generally accepts the process of ART but is in general opposed to surrogacy.¹⁵⁹⁶

Surrogate motherhood and artificial fertilisation may also complicate the conventional parent-child arrangement. For example, a child born as a result of ART after using donated gametes and a surrogate, may have as many as five persons who could all be said to be the parents of that child: the intending rearing parents who are not genetically related to the child; the sperm donor, who is the biological father; the egg donor, who is the genetic mother; and the surrogate or gestational mother who will be carrying the child.¹⁵⁹⁷ In other instances, a three-parent IVF scenario is possible, in that

¹⁵⁹¹ Deonandan, R "Thoughts on the ethics of gestational surrogacy: perspectives from religions, Western liberalism, and comparisons with adoption" 2020 *Journal of Assisted Reproduction and Genetics* 37 269–279.

¹⁵⁹² Sukhanova Anna (2019) Surrogacy and religion: <u>https://surrogacybypons.com/surrogacy-and-religion/</u> (accessed 5 January 2023).

¹⁵⁹³ Sukhanova, A (2019) Surrogacy and religion: <u>https://surrogacybypons.com/surrogacy-and-religion/</u> (accessed 5 January 2023).

¹⁵⁹⁴ What does the bible say about being or using a surrogate mother? (2022) <u>https://www.gotquestions.org/surrogate-mother.html</u> (accessed 5 January 2023).

¹⁵⁹⁵ What does the bible say about being or using a surrogate mother? (2022) <u>https://www.gotquestions.org/surrogate-mother.html</u> (accessed 5 January 2023). Deonandan, R "Thoughts on the ethics of gestational surrogacy: perspectives from religions, Western liberalism, and comparisons with adoption" 2020 *Journal of Assisted Reproduction and Genetics* 37 269–279.

¹⁵⁹⁶ Sukhanova, A (2019) "Surrogacy and religion" <u>https://surrogacybypons.com/surrogacy-and-religion/</u> (accessed 5 January 2023). What does the bible say about being or using a surrogate mother? (2022) <u>https://www.gotquestions.org/surrogate-mother.html</u> (accessed 5 January 2023). Deonandan, R "Thoughts on the ethics of gestational surrogacy: perspectives from religions, Western liberalism, and comparisons with adoption" 2020 *Journal of Assisted Reproduction and Genetics* 37 269–279.

¹⁵⁹⁷ SALC Project 65 at par 2.1.3. Pretorius Surrogate motherhood 17. SALRC Project 140 177. Dada and McQuoid-Mason Introduction to Medico-Legal Practice 66. The authors argue that as medical technologies have advanced, the legal and social definitions of parenthood have lagged behind. Further, until the introduction of advanced reproductive technologies, the law tended to provide a child with two parents, being a mother and a father in order to protect the best interests of the child.

the woman and her husband are the intending parents, with the donor as the genetic parent.¹⁵⁹⁸ Through gestational surrogacy, a woman and her husband may create an embryo using their own gametes, but another woman (the surrogate mother) carries and gives birth to the child.¹⁵⁹⁹ Here, the husband and wife are both the intending parents as well as the genetic parents, but the surrogate is the gestational parent, thus, again three different parents to the child.¹⁶⁰⁰ Nosarka and Kruger observe that surrogacy results in a child having biological connections with two women.¹⁶⁰¹ As discussed in the introduction to this study, traditional definitions of 'mother' and 'father', whether in the legal, medical or sociological context, are no longer accurate in the context of surrogacy and artificial fertilisation.¹⁶⁰²

Deonandan captures the ethical, social and legal dilemma very aptly as follows:

As technologies become cheaper, more effective and efficient, legal, and ethical wrinkles are manifesting with increasing frequency. While traditional surrogacy has been part of the human story for some time, enriching to some extent our impressions of family, inheritance, and parentage, the advent of gestational surrogacy, has complicated this already tenuous understanding. Ethical issues abound, of course, and ancient modes of wisdom struggle to answer questions emerging from this new technology and mode of being. Uncertainties relevant to the gestational surrogacy phenomenon are manifold, and cover ground from whether paid surrogates is exploited, and whether commercial surrogacy is akin to baby selling, to whether ART is morally justifiable in a world with multitudes of orphans seeking parents and homes.

Our new era of complicated reproductive rights separates motherhood from pregnancy, sex from reproduction, and introduces third and even fourth parties into a family creation process that was mostly the domain of private couples. Lineage and family identity have been mainstays throughout human history, regardless of social class, whether as practical guidelines for the cross-generational transmission of power and property or as qualitative context for developing personal identity. Parenthood has also been a fundamental aspect of the human life identity, with paeans to motherhood in particular immortalized in art, culture, and in how many people still view their roles, powers, and identities. The new reproductive technologies have complicated these identity mainstays, and ancient ways of thinking are inconstant in their capacities to navigate this new complexity.¹⁶⁰³

Every person has the right to reproductive health, irrespective of their sexual orientation, fertility or marital status, which means that it would be unethical and

¹⁵⁹⁸ SALRC Project 140 177.

¹⁵⁹⁹ SALRC Project 140 177.

SALRC Project 140 177. Quinn, CM "Mom, Mommy & Daddy and Daddy, Dad & Mommy: Assisted Reproductive Technologies & the Evolving Legal Recognition of Tri-Parenting" 2018 *Journal of the American Academy of Matrimonial Lawyers* 31(1) 176 and 202. The involved parties by design or by choice decide that the child that they plan to create will have more than two parents. Thus, tri-parenting by design.

¹⁶⁰¹ Nosarka and Kruger 2005 SAMJ 944.

¹⁶⁰² SALC Project 65 at par 2.1.3.

¹⁶⁰³ Deonandan, R "Thoughts on the ethics of gestational surrogacy: perspectives from religions, Western liberalism, and comparisons with adoption" 2020 *Journal of Assisted Reproduction and Genetics* 37 269–279.

unlawful to deny a patient a medical procedure because of his or her infertility or marital status or sexual orientation.¹⁶⁰⁴ In light of the fact that the WHO and the American Medical Association regard infertility as a disease, I firmly believe that it cannot be seen as harmful, immoral or unethical to treat this disease with donor gametes.¹⁶⁰⁵ The advancement in medical technology is there to assist society where assistance is needed. ART's, including artificial fertilisation, are there to assist the infertile person(s) where the infertility is clearly an incurable disease.

Reproductive health is a more critical issue for women as they have complex and vulnerable reproductive systems.¹⁶⁰⁶ Dhai and McQuoid-Mason argue that a major burden of disease in women is related to their reproductive potential and function and the societal treatment of women.¹⁶⁰⁷ The authors stated that a woman's human rights are often infringed, which compromises their reproductive health.¹⁶⁰⁸ On the other hand, it is important not to view a woman as a means in the process of reproduction and further as a target in the process of fertility control.¹⁶⁰⁹

Forman argues that most of the debate regarding surrogacy has centred on the enforceability of the basic premise that a woman contractually binds herself to carry a

¹⁶⁰⁴ Section 12(2)(a) gives every person the right to bodily and psychological integrity which includes the right to make decisions concerning reproduction. Section 27(1) gives every person the right to access to health care which includes the right to reproductive health care. Biko J and Nene Z "Ethics aspects of third-party reproduction" 2017 Obstetrics & Gynaecology Forum 3 15.

¹⁶⁰⁵ Biko J and Nene Z "Ethics aspects of third-party reproduction" 2017 *Obstetrics & Gynaecology Forum* 3 15. The authors further state that the infertility rate is in excess of 10% and it is therefore essential that the disease of infertility is treated to essentially save society from the negative impact of infertility for example, alcohol abuse, reckless behaviour, depression and suicide.

¹⁶⁰⁶ Dhai and McQuoid-Mason *Bioethics, Human Rights and Health Law* 139. At 140 the authors provide a definition of reproductive health as per the United Nations as follows: "Reproductive health is a state of complete physical, mental and social well-being and not merely the absence of disease and infirmity in all matters relating to the reproductive system and its functions and processes. Reproductive health therefore implies that people are able to have a satisfying and safe sex life and that they have the capability to reproduce and the freedom to decide if, when and how often to do so. Implicit in this last condition are the right of men and women to be informed and to have access to safe, effective, affordable and acceptable methods of family planning of their choice, as well as other methods of their choice for regulation of fertility which are not against the law, and the right of access to appropriate health-care services that will enable women to go safely through pregnancy and child birth and provide couples with the best chance of having a healthy infant."

¹⁶⁰⁷ Dhai and McQuoid-Mason *Bioethics, Human Rights and Health Law* 139. The social dysfunctions impact on a woman's physical, mental and social health.

¹⁶⁰⁸ Dhai and McQuoid-Mason *Bioethics, Human Rights and Health Law* 139.

¹⁶⁰⁹ Dhai and McQuoid-Mason *Bioethics, Human Rights and Health Law* 139.

child for another person(s).¹⁶¹⁰ Surrogacy agreements contains (or may contain) provisions which regulates a range of eventualities and restrictions on the surrogate's behaviour and her decision-making authority.¹⁶¹¹ Termination of pregnancy is also addressed in the agreement and it most commonly foresees two potential scenarios.¹⁶¹² Firstly, when a foetus suffers from serious birth defects, and secondly, when multiple pregnancy arises and where foetal reduction may be recommended to improve the outcome for the remaining foetus(es).¹⁶¹³ Surrogacy agreements usually contain a related clause that restricts the surrogate's ability to terminate the pregnancy without the intended parent's consent unless the surrogate faces substantial harm from continuing the pregnancy.¹⁶¹⁴ Importantly, although the agreements often contain abortion and selective reduction provisions, practitioners routinely describe these provisions as unenforceable.¹⁶¹⁵ The choice of termination of the pregnancy legally remains the choice of the surrogate, as it is her body that is involved in the procedure.

The advancement in technology and assisted reproduction make it more difficult to determine parentage.¹⁶¹⁶ Pyrce regards the most serious issue regarding countries that use diverse methods to determine the status of a surrogacy-born child in citizenship and parentage law as that of statelessness.¹⁶¹⁷ Statelessness occurs when the commissioning parents' domestic parentage and citizenship requirements are in conflict with the laws of the surrogate's country.¹⁶¹⁸ It may also occur where the child

¹⁶¹⁰ Forman, D.L. "Abortion clauses in surrogacy contracts: Insights from a case study" 2015 *Family Law Quarterly* 49(1) 31.

¹⁶¹¹ Forman, D.L. "Abortion clauses in surrogacy contracts: Insights from a case study" 2015 *Family Law Quarterly* 49(1) 31.

¹⁶¹² Forman, D.L. "Abortion clauses in surrogacy contracts: Insights from a case study" 2015 *Family Law Quarterly* 49(1) 31 and 33.

¹⁶¹³ Forman, D.L. "Abortion clauses in surrogacy contracts: Insights from a case study" 2015 *Family Law Quarterly* 49(1) 34. It is often the case that the contract provides for the intended parents to have the right to make all termination decisions. Provisions in respect of the reduction of a foetus can be more specific, for example, to not allow the intended parents selective reduction for gender selection purposes.

¹⁶¹⁴ Forman, D.L. "Abortion clauses in surrogacy contracts: Insights from a case study" 2015 *Family Law Quarterly* 49(1) 34.

¹⁶¹⁵ Forman, D.L. "Abortion clauses in surrogacy contracts: Insights from a case study" 2015 *Family Law Quarterly* 49(1) 34. At 35 the author explained that the scholars by and large echo that provisions relating to abortion are either unenforceable altogether or at best subject to a suit for damages.

¹⁶¹⁶ Pyrce, C "Surrogacy and citizenship: A conjunctive solution to a global problem" 2016 *Indiana Journal of Global Legal Studies* 23(2) 930.

¹⁶¹⁷ Pyrce, C "Surrogacy and citizenship: A conjunctive solution to a global problem" 2016 *Indiana Journal of Global Legal Studies* 23(2) 933.

¹⁶¹⁸ Pyrce, C "Surrogacy and citizenship: A conjunctive solution to a global problem" 2016 *Indiana Journal of Global Legal Studies* 23(2) 934.

is not a citizen of the commissioning parents' home country (because its laws prohibit surrogacy) and where the child is also not a citizen in the surrogate's country (because parentage is genetically determined and the surrogate is not genetically related to the child).¹⁶¹⁹ If a child is stateless, he or she cannot gain citizenship; nor can he or she obtain a passport from either the countries which means that the child is stranded, parentless and rightless.¹⁶²⁰ A child's right to be with his or family under international and regional human rights instruments is clearly threatened when the commissioning parents' home state refuses to recognise a surrogate child's birth certificate where the commissioning parents are listed as the parent(s) instead of the surrogate.¹⁶²¹ Pyrce advocates for reforms to domestic legislation so that a middle ground may be found regarding the risk of statelessness, to avoid a situation where the impact of new technology may lead to absurd results that may force the assisted reproduction market out of the country or into the black market.¹⁶²²

5. THE ROLE OF EPIGENETICS

Epigenetics literally means "above" or "on top of" genetics.¹⁶²³ Epigenetics refers to external alterations to DNA that turn genes "on" or "off."¹⁶²⁴ Rettner explains that these alterations do not change the DNA sequence, but instead, they affect how cells "read" genes.¹⁶²⁵

¹⁶¹⁹ Pyrce, C "Surrogacy and citizenship: A conjunctive solution to a global problem" 2016 *Indiana Journal of Global Legal Studies* 23(2) 934.

¹⁶²⁰ Pyrce, C "Surrogacy and citizenship: A conjunctive solution to a global problem" 2016 *Indiana Journal of Global Legal Studies* 23(2) 934.

¹⁶²¹ Pyrce, C "Surrogacy and citizenship: A conjunctive solution to a global problem" 2016 *Indiana Journal of Global Legal Studies* 23(2) 935. It can also happen that the lack of any genetic relation to the commissioning parents can prevent the child from gaining citizenship by descent if the commissioning parents use traditional surrogacy to have a child in another country.

¹⁶²² Pyrce, C "Surrogacy and citizenship: A conjunctive solution to a global problem" 2016 *Indiana Journal of Global Legal Studies* 23(2) 952. Citizenship laws must be amended as it is necessary to specifically contemplate the surrogacy context so that children that are born abroad will be able to gain citizenship and travel home with the commissioning parents.

¹⁶²³ Rettner, Rachael: Epigenetics: Definition & Examples (June 24, 2013) accessed from https://www.livescience.com/37703-epigenetics.html (accessed on 16 January 2021)

¹⁶²⁴ Rettner, Rachael: Epigenetics: Definition & Examples (June 24, 2013) accessed from <u>https://www.livescience.com/37703-epigenetics.html</u> (accessed on 16 January 2021). Christiansen, K "Who is the mother? Negotiating identity in an Irish surrogacy case" 2015 *Medicine, Health Care and Philosophy* 18(3) 319.

¹⁶²⁵ Rettner, Rachael: Epigenetics: Definition & Examples (June 24, 2013) accessed from <u>https://www.livescience.com/37703-epigenetics.html</u> (accessed on 16 January 2021). The author gives an example of epigenetics as follows: "Epigenetic changes alter the physical structure of DNA. One example of an epigenetic change is DNA methylation — the

The Centre for Disease Control and Prevention (CDC) describes epigenetics as the study of how behaviour and environment can cause changes that affect the way genes work and, unlike genetic changes, epigenetic changes are reversible and do not change a person's DNA sequence, but they can change how a person's body reads a DNA sequence.¹⁶²⁶ The basic underlying mechanism of epigenetics involves methylation and acetylation of specific DNA base pairs or DNA-associated proteins to regulate gene activation.¹⁶²⁷ As discussed in an earlier chapter, the genetic makeup of a surrogate has a substantial impact upon the genetics of the child that she carries in her womb, irrespective of whether or not she is an ovum donor.¹⁶²⁸ Although the two biological parents of the child contribute their genetic material to the child, the genetic expression of the child's genes, in other words, how its genes are switched on, is significantly influenced by the surrogate's transcription factors.¹⁶²⁹ It is for this reason that Nicholson argues that, contrary to the view that the sperm and ovum donors are the only two biological parents of the child, where the womb for the gestation of the child is the womb of a third person, this third person becomes an additional biological parent whose input is experienced both in utero and after birth by the said child.¹⁶³⁰ The reality is that at least three people have a biological and psychological effect on the foetus.¹⁶³¹ Fischbach and Loike maintain that there are scientific evidence that shows, with reference to gestational surrogacy, how maternal-foetal cellular exchange (microchimerism) and epigenetic factors create intimate biological bonds between the surrogate and the foetus that she is carrying.¹⁶³²

addition of a methyl group, or a "chemical cap," to part of the DNA molecule, which prevents certain genes from being expressed."

¹⁶²⁶ "What is epigenetics?" accessed from <u>https://www.cdc.gov/genomics/disease/epigenetics.htm</u> (accessed on 16 January 2021). Nicholson, S and Nicholson, C "I used to have two parents and now I have three? When science (fiction) and the law meet: Unexpected complications" 2016 *Medicine and Law* 35 426. Nicholson states that epigenetic changes simply change the expression of the genes by adding small marks to the DNA.

¹⁶²⁷ Fischbach, R.L. and Loike, J.D "Maternal-fetal cell transfer in surrogacy: Ties that bind" 2014 *American Journal of Bioethics* 14(5) 35.

¹⁶²⁸ Nicholson, S and Nicholson, C "I used to have two parents and now I have three? When science (fiction) and the law meet: Unexpected complications" 2016 *Medicine and Law* 35 425.

¹⁶²⁹ Nicholson, S and Nicholson, C "I used to have two parents and now I have three? When science (fiction) and the law meet: Unexpected complications" 2016 *Medicine and Law* 35 425.

¹⁶³⁰ Nicholson, S and Nicholson, C "I used to have two parents and now I have three? When science (fiction) and the law meet: Unexpected complications" 2016 *Medicine and Law* 35 425. Rasheed, A "Confronting problematic legal fictions in gestational surrogacy" 2021 *Journal of Health Care Law and Policy* 24(2) 195.

¹⁶³¹ Rasheed, A "Confronting problematic legal fictions in gestational surrogacy" 2021 *Journal of Health Care Law and Policy* 24(2) 198.

¹⁶³² Fischbach, R.L. and Loike, J.D "Maternal-fetal cell transfer in surrogacy: Ties that bind" 2014 *American Journal of Bioethics* 14(5) 35. Rasheed, A "Confronting problematic legal fictions in

Bidirectional cellular exchange between the gestational mother and the foetus, together with the epigenetic processes, reveal that the mother and the foetus are connected beyond the defined time that the foetus stays in the womb.¹⁶³³ Surrogates are not only financially compensated for the use of their uterus, but they are undeniably establishing a biological bond with the foetus which appears to last for decades.¹⁶³⁴ Both the surrogate and the commissioning mother should understand this biological bond and they should give voluntary and valid consent especially because microchimerism is known to increase the incidence of autoimmune diseases in a child.¹⁶³⁵ The parties to a surrogacy agreement must understand that epigenetic processes resulting from the surrogate's lifestyle could have both physiological and behavioural consequences for the child.¹⁶³⁶ Possible health risks must therefore be recognised by the parties.¹⁶³⁷ Rasheed points out that the physical effects of pregnancy begin immediately, lasts indefinitely and they are by every measure profound.¹⁶³⁸ The indirect genetic influences are significant to the growing embryo because epigenetic changes may influence everything from the child's sensitivity to allergens to increased infant death and immune dysfunction.¹⁶³⁹ Prenatal programming takes place in the womb and this programming predisposes the child to certain adult diseases.¹⁶⁴⁰

Christiansen points out that what happens in the womb can activate or deactivate certain genetic traits in the child.¹⁶⁴¹ As a result of all the methylation being erased after fertilisation, the heritable epigenetics and environmental programming of the

gestational surrogacy" 2021 *Journal of Health Care Law and Policy* 24(2) 196. The author opines that even if only genetic in small amounts, the biological connection between the surrogate and the child is undeniable.

¹⁶³³ Fischbach, R.L. and Loike, J.D "Maternal-fetal cell transfer in surrogacy: Ties that bind" 2014 *American Journal of Bioethics* 14(5) 35.

¹⁶³⁴ Fischbach, R.L. and Loike, J.D "Maternal-fetal cell transfer in surrogacy: Ties that bind" 2014 *American Journal of Bioethics* 14(5) 35.

¹⁶³⁵ Fischbach, R.L. and Loike, J.D "Maternal-fetal cell transfer in surrogacy: Ties that bind" 2014 *American Journal of Bioethics* 14(5) 35.

¹⁶³⁶ Fischbach, R.L. and Loike, J.D "Maternal-fetal cell transfer in surrogacy: Ties that bind" 2014 *American Journal of Bioethics* 14(5) 35.

¹⁶³⁷ Fischbach, R.L. and Loike, J.D "Maternal-fetal cell transfer in surrogacy: Ties that bind" 2014 *American Journal of Bioethics* 14(5) 35.

¹⁶³⁸ Rasheed, A "Confronting problematic legal fictions in gestational surrogacy" 2021 *Journal of Health Care Law and Policy* 24(2) 191.

¹⁶³⁹ Rasheed, A "Confronting problematic legal fictions in gestational surrogacy" 2021 *Journal of Health Care Law and Policy* 24(2) 195.

¹⁶⁴⁰ Nicholson, S and Nicholson, C "I used to have two parents and now I have three? When science (fiction) and the law meet: Unexpected complications" 2016 *Medicine and Law* 35 435.

¹⁶⁴¹ Christiansen, K "Who is the mother? Negotiating identity in an Irish surrogacy case" 2015 *Medicine, Health Care and Philosophy* 18(3) 319.

epigenome is dependent on the womb environment.¹⁶⁴² The foetus thus has the genetic information of two genetic parents together with the imprinted gene silencing and the integration of environmental signals.¹⁶⁴³

Because of the significant role of epigenetics, one may rightly ask who should be regarded as the biological mother of a commissioned child where the ovum comes from the commissioning woman and the womb from the surrogate mother?¹⁶⁴⁴ Both women may be said to have contributed genetically to the child's development, as epigenetics allows an organism (in this scenario the foetus) to respond to its environment (in this scenario the womb) by changes in gene expression.¹⁶⁴⁵ Nicholson and Nicholson remark as follows on the role of the womb in foetal development:

Offspring often have similar methylation patterns in their DNA to those of their parents. There is mounting evidence that epigenetic programming is sensitive to nutrition, endocrine signaling, maternal stress and maternal care. The developing foetus is the point of human development most sensitive to epigenetic changes. The foetus is undergoing rapid cell proliferation, differentiation and programming of specific cells to specific tissues, increasing the need for flexibility in the epigenome.¹⁶⁴⁶ The embryo has very limited epigenetic programming to allow for the rapid gene adaptation of the foetus to the new environmental conditions and tissue specific gene expression requirements. This makes the environment in which a foetus develops vital to its proper epigenetic programming.¹⁶⁴⁷

¹⁶⁴² Nicholson, S and Nicholson, C "I used to have two parents and now I have three? When science (fiction) and the law meet: Unexpected complications" 2016 *Medicine and Law* 35 432.

¹⁶⁴³ Nicholson, S and Nicholson, C "I used to have two parents and now I have three? When science (fiction) and the law meet: Unexpected complications" 2016 *Medicine and Law* 35 431.

¹⁶⁴⁴ Nicholson, S and Nicholson, C "I used to have two parents and now I have three? When science (fiction) and the law meet: Unexpected complications" 2016 *Medicine and Law* 35 425. The authors states that it is not a simple matter of nature or nurture as the womb is significantly more than an incubator.

¹⁶⁴⁵ Nicholson, S and Nicholson, C "I used to have two parents and now I have three? When science (fiction) and the law meet: Unexpected complications" 2016 *Medicine and Law* 35 425. Rasheed, A "Confronting problematic legal fictions in gestational surrogacy" 2021 *Journal of Health Care Law and Policy* 24(2) 195. The surrogate is not biologically separate from her foetus.

¹⁶⁴⁶ The National Human Genome Research Institute (NHI) defines the term epigenome as follows: "The term epigenome is derived from the Greek word epi which literally means "above" the genome. The epigenome consists of chemical compounds that modify, or mark, the genome in a way that tells it what to do, where to do it, and when to do it. Different cells have different epigenetic marks. These epigenetic marks, which are not part of the DNA itself, can be passed on from cell to cell as cells divide, and from one generation to the next." Talking glossary of genetic terms accessed from <u>https://www.genome.gov/genetics-glossary/Epigenome</u> (accessed on 29 January 2023).

¹⁶⁴⁷ Nicholson, S and Nicholson, C "I used to have two parents and now I have three? When science (fiction) and the law meet: Unexpected complications" 2016 *Medicine and Law* 35 427.

Rasheed argues that the correct understanding of biology indicates that surrogates and other tissue donors play a genetic role that courts, and policy makers have not yet recognised.¹⁶⁴⁸

Even though the effect of epigenetics on the developing commissioned child is still not well-known, one may conclude that this development may lead to intense confusion and uncertainty in the surrogacy context. Chapter 19 of the Children's Act does not provide for circumstances where the gestational mother may claim rights over the child because she has an epigenetic connection or bond with the child. The role of epigenetic processes will play an important role in the selection of a healthy and suitable surrogate mother. The impact of epigenetics also means that maternal stress and nutrition, as well as the surrogate's epigenetic profile, must be considered when surrogates or egg donors are selected. Fischbach and Loike believe that a complete genetic analysis of both the proposed surrogate and the gamete donors should be obtained.¹⁶⁴⁹

Epigenetics also has a reach beyond the surrogate agreement. Nicholson argues that the legal and ethical implications of epigenetics cannot be underestimated because of numerous legal and ethical challenges and because a person can be influenced generations later by the circumstances and experiences of a forbearer.¹⁶⁵⁰

If the frightening reality is that the memories that are transferred by parents to their children's genes, where they are entrenched and might potentially negatively affect that child or future generations, how much more frightening does this scenario become when the child is exposed to a third set of influences through the womb of a surrogate?¹⁶⁵¹

The idea of including a third parent to the equation of surrogacy brings forth the important question of how far the commissioning couple can go with having control over the surrogate (as third parent) and her life during the gestational period in trying to protect the foetus from any harm? The surrogate, as third parent, might argue that she knows the best. How far can the commissioning couple go in doing an in-depth

¹⁶⁴⁸ Rasheed, A "Confronting problematic legal fictions in gestational surrogacy" 2021 *Journal of Health Care Law and Policy* 24(2) 191.

¹⁶⁴⁹ Fischbach, R.L. and Loike, J.D "Maternal-fetal cell transfer in surrogacy: Ties that bind" 2014 *American Journal of Bioethics* 14(5) 35.

¹⁶⁵⁰ Nicholson, S and Nicholson, C "I used to have two parents and now I have three? When science (fiction) and the law meet: Unexpected complications" 2016 *Medicine and Law* 35 432.

¹⁶⁵¹ Nicholson, S and Nicholson, C "I used to have two parents and now I have three? When science (fiction) and the law meet: Unexpected complications" 2016 *Medicine and Law* 35 432.

investigation into the life and background of the surrogate? It is evident from the above that the surrogate's way of life, her psychological state together with her health and any underlying genetic disorder all influence the foetus and thus the resultant child. Ethically, the commissioning parent(s) cannot go further than what the surrogate allows them to know. It is thus an unknown risk that the commissioning parent(s) will have to accept when using the surrogate as gestational mother of the commissioned child.

The issue of responsibility for birth defects in the child is also complicated by epigenetics which makes the environment, nutrition and medical history of the surrogate, even one whose gamete is not used, relevant.¹⁶⁵² The biological realities of gestational surrogacy, including the biological bond between the surrogate and the child, is overlooked by the law of surrogacy when policing the balance of power between private parties in surrogacy agreements.¹⁶⁵³ The difficulty lies in giving the 'third parent' any rights in respect of the commissioned child in light of the biological bond that exists between the resultant child and the surrogate, especially in circumstances where it was a gestational surrogacy. Where can one draw the line as the commissioning parent(s) still holds the genetic link to the child and it is the commissioned child? As is clear from the above, the surrogate is the bearer of the most responsibility in respect of the commissioned child during the gestation period but it is a choice that she made.

Further, if one takes the view that surrogacy is a simple matter of contract, would commissioning parents be able to rely on product liability if the child proves to have some predisposition to psychological or physical conditions inherited from the surrogate?¹⁶⁵⁴

In conclusion, although requesting a complete genetic analysis from the surrogate mother and/or gamete donors may be advisable, succeeding with this is unlikely in the

¹⁶⁵² Nicholson, S and Nicholson, C "I used to have two parents and now I have three? When science (fiction) and the law meet: Unexpected complications" 2016 *Medicine and Law* 35 434.

¹⁶⁵³ Rasheed, A "Confronting problematic legal fictions in gestational surrogacy" 2021 *Journal of Health Care Law and Policy* 24(2) 196.

¹⁶⁵⁴ Nicholson, S and Nicholson, C "I used to have two parents and now I have three? When science (fiction) and the law meet: Unexpected complications" 2016 *Medicine and Law* 35 434.

South African context. Commissioning parent(s) are already required to spend a lot of money on the drafting of the agreement and the application for confirmation by the court, not to mention the costs that they will incur for the pregnancy and birth of the child. Should the commissioning couple select not to request a complete genetic analysis and it appears after the birth of the child that the surrogate did have existing health issues (which she was unaware of) which may have caused harm to the child, the commissioning couple will arguably not be able to hold the surrogate responsible for the harm to child because they elected to accept the risk of not knowing when they concluded the surrogate agreement.

6. CONCLUSION

This chapter explored the salient issues relating to assisted reproduction and related ethical, social and religious issues with reference to the NHA and regulations. The difference between adoption and surrogacy has also been canvassed. A pertinent issue that may significantly impact on the health of the commissioned child, is the role of epigenetics, which directly relates to the surrogate mother's genetic profile. It is imperative that all parties to a surrogate agreement take note of the impact of epigenetics, as this may require a revision of existing agreements with regard to the type and scope of medical tests that may be required. Epigenetics also changes the dynamic between the parties, as her biological link to the commissioned child is stronger than previously thought.

The next chapter will turn to a comparative discussion of the regulation of surrogacy in the United Kingdom, Canada and India with the objective of tracing best practices in these jurisdictions that may assist in addressing some of the identified *lacunae* arising from the surrogate motherhood legal framework in South Africa.

CHAPTER 6

COMPARATIVE ANALYSIS OF SURROGACY REGULATION IN THE UNITED KINGDOM, CANADA AND INDIA

1. INTRODUCTION

Many jurisdictions have introduced legislation to regulate surrogate motherhood. It is well-established that law reform often relies on comparative law elsewhere to address shortcomings in domestic legislation.

In this chapter, the legal systems regulating surrogate motherhood in the United Kingdom, Canada and India will be compared with the purpose of identifying best practices in these systems that may clarify or address some of the unresolved or ambiguous issues relating to assisted reproduction and surrogate motherhood in South Africa. The selection of the UK, Canada and India is motivated by the fact that the UK and Canada, like South Africa, ban any form of commercial surrogacy but allow altruistic surrogacy. India is one of the countries that initially opened its doors to commercial surrogacy, subject to guidelines that regulate assisted reproduction. Like South Africa, the UK, Canada and India are all committed to the protection of rights and freedoms.

2. THE UNITED KINGDOM

2.1 Regulation of surrogate motherhood

Surrogacy has been allowed as a form of reproduction in the UK since 1985. The UK is one of the first jurisdictions in the world to introduce surrogacy legislation.¹⁶⁵⁵ The first statute that provided for the use of surrogacy was the Surrogacy Arrangements Act of 1985.¹⁶⁵⁶ The Human Fertilisation and Embryology Act 1990 ("HFEA 1990")

¹⁶⁵⁵ Fenton-Glynn, C "The regulation and recognition of surrogacy under English Law: an overview of the case-law" 2015 *Child and Family Law Quarterly* 27(1) 83.

¹⁶⁵⁶ Fenton-Glynn, C "The regulation and recognition of surrogacy under English Law: an overview of the case-law" 2015 *Child and Family Law Quarterly* 27(1) 84.

amended certain provisions of the Surrogacy Arrangements Act. The aim of the HFEA 1990 was to make provision for fertilisation and human embryos and any subsequent development of such embryos; to prohibit certain practices regarding embryos and gametes; to establish a Human Fertilisation and Embryology Authority; to provide for persons who under certain circumstances are to be treated in law as the parents of a child; and to amend the Surrogacy Arrangements Act 1985.¹⁶⁵⁷ Section 5 of the HFEA created the Human Fertilisation and Embryology Authority to regulate and license activities involving human embryos.¹⁶⁵⁸ Other pertinent issues covered by the Act are, for example, consent, legal status and the welfare of the resulting child and the necessity of potential parents having proper understanding of the course upon which they were embarking.¹⁶⁵⁹ The Act further introduced a statutory requirement that the welfare of the child (born as a result of the fertility treatment) was to be taken into account.¹⁶⁶⁰

The Human Fertilisation and Embryology Act 2008 ("HFEA 2008") amended certain provisions of the HFEA 1990 and specifically made amendments to the sections relating to surrogacy.¹⁶⁶¹ The aim of the 2008 amendments was to provide for the persons who in certain circumstances are to be treated in law as the parents of a child and for connected purposes.¹⁶⁶² It is important to note that the requirement for the consideration of the child's welfare remained part of the revised Act.¹⁶⁶³ The reading

 ¹⁶⁵⁷ Introduction of HFEA 1990 accessed from <u>https://www.legislation.gov.uk/ukpga/1990/37/introduction</u> (accessed on 21 January 2023). *JP v LP and others* [2014] EWHC 595 (Fam) at par 17. Kennedy and Grubb *Medical Law* 768. This regulatory framework took effect on 1 August 1991. This framework was only applicable in so far as a surrogate birth was achieved through the use, in part or whole, of donated genetic material or using IVF techniques. Otherwise, the practice of surrogacy was regulated, in part, by the Surrogacy Arrangements Act 1985.

¹⁶⁵⁸ Sec 5 of Human Fertilisation and Embryology Act 1990 (HFEA 1990). *JP v LP and others* at par 17. Mason, McCall Smith and Laurie *Law and Medical Ethics* 69.

¹⁶⁵⁹ JP v LP and others at par 17.

¹⁶⁶⁰ Sec 13(5) determined that a woman shall not be provided with treatment services, other than basic partner treatment services, unless account has been taken of the welfare of any child who may be born as a result of the treatment (including the need of that child for a father), and of any other child who may be affected by the birth. Code of practice 8th ed accessed from <u>https://portal.hfea.gov.uk/media/1593/hfea-code-of-practice-8th-edition-r8.pdf</u> (accessed on 11 January 2023).

JP v LP and others at par 19. Fenton-Glynn, C "The regulation and recognition of surrogacy under English Law: an overview of the case-law" 2015 *Child and Family Law Quarterly* 27(1) 84.
 Introduction of HFE Act 2008 accessed from

https://www.legislation.gov.uk/ukpga/2008/22/introduction (accessed on 21 January 2023).

¹⁶⁶³ Sec 13(5) of the HFEA 1990 and as amended by sec 14(2) of HFE Act 2008. *JP v LP and others* at par 19.

of the HFEA 2008 is complicated by the fact that it did not repeal the HFEA 1990 in its entirety. The Act made amendments which should be read into the HFEA 1990 and the Surrogacy Arrangement Act. The most relevant provisions regarding surrogacy will be discussed, also considering the Surrogacy Arrangement Act, the HFEA 1990 and the HFEA 2008.

It is a condition of the granting of a treatment license to a fertility clinic that a woman shall not be treated unless she, and any partner who are being treated together, have been given a suitable opportunity to receive proper counselling about the implications of proceeding with the proposed treatment steps, and have been provided with such relevant information as is appropriate.¹⁶⁶⁴ Similar to South Africa, at least one of the two commissioning parents must contribute a gamete for the purposes of conception.¹⁶⁶⁵

The UK Surrogacy Arrangements Act of 1985 discourages the use of surrogacy by making surrogacy agreements unenforceable and by criminalising certain acts relating to surrogacy agreements.¹⁶⁶⁶ Although surrogate agreements are defined in the 1985 Act, it does not include agreements concluded after the child is conceived and which are intended to result in the child being handed over by the surrogate mother.¹⁶⁶⁷ The intention of the parties to the surrogate agreement that the commissioned child be handed over to the commissioning parents (after the said child's birth) is part and parcel of the purpose of any surrogacy agreement. Lyon sets out the provisions which are likely to be included in a surrogacy agreement: arrangements relating to

¹⁶⁶⁴ Sec 13(6) of HFEA 1990 as amended by Sec 14(3) of HFEA 2008. The 8th Code of Practice, Part 3 specifically refers to counselling being provided at every stage of the treatment process by a qualified counsellor.

¹⁶⁶⁵ Sec 54(1)(b) of HFEA 2008 in cases where there are two applicants and sec 54A(1)(b) in cases where there is only one applicant.

¹⁶⁶⁶ Chapter 49 of the Surrogacy Arrangements Act 1985, was enacted on 16 July 1985 to regulate certain activities in connection with arrangements made with a view to women carrying children as surrogate mothers. Sec 1A. "No surrogacy arrangement is enforceable by or against any of the persons making it." Louw 2013 *THRHR* 572. Strauss *Doctor, Patient and the Law* 188. Mason, McCall Smith and Laurie *Law and Medical Ethics* 99. Fenton-Glynn, C "The regulation and recognition of surrogacy under English Law: an overview of the case-law" 2015 *Child and Family Law Quarterly* 27(1) 84. Elder, AH "Wombs to rent: Examining the jurisdiction of international surrogacy" 2014 16(2) *Oregon Review of International Law* 365. Brown, A "Trans parenthood and the meaning of 'mother', 'father' and 'parent' – R (McConnell and YY) v Registrar General for England and Wales [2020] EWCA Civ 559" 2021 *Medical Law Review* 29(1) 160.

¹⁶⁶⁷ Kennedy and Grubb *Medical Law* 846. The writers point out that such an arrangement would remain governed by the common law.

conception, pregnancy, birth and post-birth; expenses and a will by the intending parents.¹⁶⁶⁸

In terms of the common law, the woman who gestates and gives birth to a child is the legal mother of child.¹⁶⁶⁹ Parenthood was defined by genetic make-up which meant that where sperm was donated, the donor was considered the legal father of the child. If the child was born within a marriage, the husband would be deemed the father by virtue of a rebuttable presumption.¹⁶⁷⁰ Further, regarding the egg or embryo donation, the common law's unfamiliarity with these practices (which separate the gestational from the genetic) would probably have meant that it would have preferred the claims of the gestational mother.¹⁶⁷¹ The common law rule in respect of sperm donation was reversed by section 27 of the Family Law Reform Act 1987.¹⁶⁷² In terms of this section, the husband of a woman who was artificially fertilised was regarded as the father of any child born unless it was proved that he did not consent to the said procedure.¹⁶⁷³

The current position in respect of the commissioned child's status is set out in the HFEA 2008. Section 33 of HFEA 2008 defines "mother" as:

The woman who is carrying or has carried a child as a result of the placing in her of an embryo or of sperm and eggs, and no other woman, is to be treated as the mother of the child. 1674

"Father" is described with reference to the scenario below:

If at the time of the placing in her of the embryo or of the sperm and eggs or of her artificial insemination, W was a party to a marriage [with a man][or a civil partnership with a man], and the creation of the embryo carried by her was not brought about with the sperm of the other party to the marriage [or civil partnership], then, subject to section 38(2) to (4), the other party to the marriage [or civil partnership] is to be treated as the father of the child unless it is shown that he did not consent to the placing in her of the embryo or the sperm or to her artificial insemination (as the case may be).¹⁶⁷⁵

¹⁶⁶⁸ Lyon, F "The surrogacy journey" 2020 *New Law Journal* 170(7900) 12.

¹⁶⁶⁹ Brown, A "Trans parenthood and the meaning of 'mother', 'father' and 'parent' – R (McConnell and YY) v Registrar General for England and Wales [2020] EWCA Civ 559" 2021 *Medical Law Review* 29(1) 160.

¹⁶⁷⁰ Kennedy and Grubb *Medical Law* 815.

¹⁶⁷¹ Kennedy and Grubb *Medical Law* 815. The writers point out that the inconsistency between the two positions could have persuaded a court to opt for the genetic determination of parenthood regarding both the mother and the father.

¹⁶⁷² Family Law Reform Act 1987 Chapter 42. This Act was enacted on 15 May 1987 to reform the law relating to the consequences of birth outside marriage; to make further provision with respect to the rights and duties of parents and the determination of parentage; and for connected purposes. Kennedy and Grubb *Medical Law* 816.

¹⁶⁷³ Sec 27 of the Family Law Reform Act 1987. Kennedy and Grubb *Medical Law* 816.

¹⁶⁷⁴ Sec 33(1) of HFEA 2008.

¹⁶⁷⁵ Sec 35(1) of HFEA 2008.

Because the surrogate (and if married, her husband or partner) is/are seen as the parent(s) of the child that is born of surrogacy, it is necessary for the commissioning parents to apply to court for a parental order so that they can be regarded as the legal parents of the commissioned child. A woman will not be considered as the parent of a child in circumstances where she only donated an ovum, without carrying the child.¹⁶⁷⁶

Provisions in terms of the HFEA 2008 that are relevant to surrogacy are sections 54 and 59. Section 54 provides for the application for parental orders and section 59 amends certain provisions in the Surrogacy Arrangements Act. Parties to a surrogacy agreement must ensure that they meet the requirements of a parental order before going the surrogacy route, as there might otherwise be complications at the end (discussed elsewhere in this chapter with reference to case law).¹⁶⁷⁷

Where two applicants apply to court for a parental order, the court may make an order if the following conditions are met: the child was carried by a woman other than one of the applicants and as a result of her artificial fertilisation or the placing in her of an embryo or sperm and eggs;¹⁶⁷⁸ at least one of the applicants' gametes was used for the conception of the embryo;¹⁶⁷⁹ the applicants must be married (husband and wife)¹⁶⁸⁰ or they may be civil partners of each other¹⁶⁸¹ or they may live as partners in an enduring family relationship.¹⁶⁸² Further, the application must be brought within six months from the date of the child's birth.¹⁶⁸³ The child must be living with the applicants at the time that the application is brought and the order is made.¹⁶⁸⁴ One or both of the applicants must be domiciled in the UK, the Channel Islands or the Isle of Man at the

¹⁶⁷⁶ Sec 47 of HFEA 2008. Except where she is married (or in a civil partnership) to the woman being artificially fertilised with her ovum or where she agrees to parenthood of the child as in terms of sec 43 and 44 of the Act.

¹⁶⁷⁷ Lyon, F "The surrogacy journey" 2020 *New Law Journal* 170(7900) 13.

¹⁶⁷⁸ Sec 54(1)(a) of HFEA 2008. Sec 54(10) determines that this provision applies even if the woman was not in the UK at the time of placing in her of the embryo or the sperm and the eggs or her artificial fertilisation.

¹⁶⁷⁹ Sec 54(1)(b) of HFEA 2008.

¹⁶⁸⁰ Sec 54(2)(a) of HFEA 2008.

¹⁶⁸¹ Sec 54(2)(b) of HFEA 2008.

¹⁶⁸² Sec 54(2)(c) of HFEA 2008. The applicants must not be within a prohibited degree of relationship with each other.

¹⁶⁸³ Sec 54(3) of HFEA 2008.

¹⁶⁸⁴ Sec 54(4)(a) of HFEA 2008.

time of the application.¹⁶⁸⁵ At the time that the order is granted, both the applicants must be older than eighteen years.¹⁶⁸⁶

Section 54(6) determines that the court must be satisfied that the surrogate and any other person being a parent of the child (not one of the applicants) have freely and with full understanding of what is involved, agreed unconditionally to the granting of the order.¹⁶⁸⁷ In circumstances where the surrogate cannot be found or she is incapable of giving her agreement, her agreement is not required.¹⁶⁸⁸ Importantly, the surrogate's agreement will be ineffective if given by her less than six weeks after the child's birth.¹⁶⁸⁹ Before a parental order can be made, the court must be satisfied there was no money paid (except for reasonable expenses) or any other benefit has been given or received by either of the applicants for or in consideration of making of the order;¹⁶⁹⁰ any agreement required from the surrogate (and the other parent)¹⁶⁹¹ for the handing over of the child;¹⁶⁹² or the making of arrangements with a view to making of the order,¹⁶⁹³ except where the court approves such payment.¹⁶⁹⁴ Moreover, section 54(8A) determines that no order in respect of the child must previously have been made under sections 54 or 54A, except where the order has been quashed or an appeal against the order has been allowed.¹⁶⁹⁵

A parental order allows the child to be considered in law as the child of the commissioning parents (the applicants) as it extinguishes the parental status of the surrogate (and the other parent).¹⁶⁹⁶

¹⁶⁹⁰ Sec 54(8)(a) of HFEA 2008.

- ¹⁶⁹² Sec 54(8)(c) of HFEA 2008.
- ¹⁶⁹³ Sec 54(8)(d) of HFEA 2008.
- ¹⁶⁹⁴ Sec 54(8) of HFEA 2008.
- ¹⁶⁹⁵ Sec 54(8Å) of HFEA 2008.

¹⁶⁸⁵ Sec 54(4)(b) of HFEA 2008.

¹⁶⁸⁶ Sec 54(5) of HFEA 2008.

¹⁶⁸⁷ Sec 54(6) of HFEA 2008.

¹⁶⁸⁸ Sec 54(7) of HFEA 2008.

¹⁶⁸⁹ Sec 54(7) of HFEA 2008. Louw 2013 *THRHR* 572. Fenton-Glynn, C "The regulation and recognition of surrogacy under English Law: an overview of the case-law" 2015 *Child and Family Law Quarterly* 27(1) 89. The author argues that the six-week rule is there to protect women from making life-changing decisions immediately after birth. This would exploit the woman's weakness at a time when she is most vulnerable and it would be unfair to put any weight on her consent at such time.

¹⁶⁹¹ Sec 54(8)(b) of HFEA 2008.

¹⁶⁹⁶ Louw 2013 *THRHR 572.* Elder, AH "Wombs to rent: Examining the jurisdiction of international surrogacy" 2014 16(2) *Oregon Review of International Law* 366.

In circumstances where there is a single applicant who applies for a parental order, section 54A governs the process. The court may make a parental order on application of one person if the following requirements are met:¹⁶⁹⁷ the child was carried by a woman other than the applicant and because of her artificial fertilisation or the placing in her of an embryo or sperm and eggs,¹⁶⁹⁸ and the applicant's gametes were used for the conception of the embryo¹⁶⁹⁹. The application must be brought within six months from the date of the child's birth.¹⁷⁰⁰ The child must be living with the applicant at the time that the application is brought and the order is made.¹⁷⁰¹ The applicant must be domiciled in the UK, the Channel Islands or the Isle of Man at the time of the application.¹⁷⁰² The applicant must be eighteen years old at the time that the order is granted.¹⁷⁰³

The court must further be satisfied that the surrogate and any other person being a parent of the child (but is not the applicant) have freely and with full understanding of what is involved, agreed unconditionally to the making of the order.¹⁷⁰⁴ Where the surrogate cannot be found or she is incapable of giving her agreement, it is not required to get her agreement before the order can be made.¹⁷⁰⁵ Here too, the surrogate's agreement will be ineffective if given by her less than six weeks after the child's birth.¹⁷⁰⁶ Before the court can grant a parental order, it must be satisfied there was no money paid (except for reasonable expenses) or any other benefit has been given or received by the applicant for or in consideration of making of the order, ¹⁷⁰⁷ for any agreement required from the surrogate (and the other parent)¹⁷⁰⁸ for the handing

¹⁶⁹⁷ Sec 54A(1) of HFEA 2008.

¹⁶⁹⁸ Sec 54A(1)(a) of HFEA 2008. Sec 54A(10) determines that this provision applies even if the woman was not in the UK at the time of placing in her of the embryo or the sperm and the eggs or her artificial fertilisation.

¹⁶⁹⁹ Sec 54A(1)(b) of HFEA 2008.

¹⁷⁰⁰ Sec 54A(2) of HFEA 2008.

¹⁷⁰¹ Sec 54A(3)(a) of HFEA 2008. ¹⁷⁰² Sec 54A(3)(b) of HEEA 2008.

¹⁷⁰² Sec 54A(3)(b) of HFEA 2008.

¹⁷⁰³ Sec 54A(4) of HFEA 2008. ¹⁷⁰⁴ Sec 54A(5) of HEEA 2008

¹⁷⁰⁴ Sec 54A(5) of HFEA 2008.

¹⁷⁰⁵ Sec 54A(6) of HFEA 2008.

¹⁷⁰⁶ Sec 54A(6) of HFEA 2008.

¹⁷⁰⁷ Sec 54A(7)(a) of HFEA 2008. ¹⁷⁰⁸ Sec 54A(7)(b) of HEEA 2008

¹⁷⁰⁸ Sec 54A(7)(b) of HFEA 2008.

over of the child¹⁷⁰⁹ or for making of arrangements with a view to making of the order,¹⁷¹⁰ except where the court approves such payment.¹⁷¹¹

Commercial surrogacy is expressly prohibited in the UK. Section 2(1) of the Surrogacy Arrangements Act contains the relevant provisions which were amended by the HFEA 2008. Where a person contravenes the provisions set out in section 2(1), he or she will be guilty of an offence and liable on summary conviction to a fine not exceeding level 5 on the standard scale or to imprisonment for a period up to three months, or both.¹⁷¹²

Regarding the right to know one's origins, the position is as follows: persons born through gamete donation are allowed to apply for non-identifying information at the age of sixteen and identifying information at the age of eighteen years.¹⁷¹³ Such an application must be made to the Human Fertilisation and Embryology Authority for the said information.¹⁷¹⁴ Advertising surrogacy services are prohibited in terms of section 3 of the Surrogacy Arrangements Act. It is not allowed to advertise an indication that a person is or may be willing to enter into a surrogacy agreement or to negotiate or facilitate the making of a surrogacy agreement.¹⁷¹⁵ It is further not allowed to advertise an indication that a person is looking for a woman willing to act as surrogate or for persons wanting a woman to act as surrogate for them.¹⁷¹⁶

¹⁷⁰⁹ Sec 54A(7)(c) of HFEA 2008.

¹⁷¹⁰ Sec 54A(7)(d) of HFEA 2008.

¹⁷¹¹ Sec 54A(7) of HFEA 2008.

¹⁷¹² Sec 4 of Surrogacy Arrangements Act 1985.

¹⁷¹³ Sec 24 of the HFEA 2008 https://www.legislation.gov.uk/ukpga/2008/22/section/24 (accessed 13 2023) and sec 31ZA(1) and sec 31ZA(4) of **HFEA** Januarv 1990 https://www.legislation.gov.uk/ukpga/1990/37/section/31ZA (accessed on 13 January 2023). Sec 24 of HFEA 2008 amended sec 31 of the HFEA 1990. Human Fertilisation and Embryology Authority (Disclosure of Donor Information) Regulations 2004 (come into force on 1st of July 2004) https://www.legislation.gov.uk/uksi/2004/1511/regulation/2/made (accessed 13 January 2023). Reg 2(2) of Human Fertilisation and Embryology Authority (Disclosure of Donor Information) Regulations 2004 determines what kind of information may be made available. Wade, K "Reconceptualising the interest in knowing one's origins: a case for mandatory disclosure" 2020 Medical Law Review 28(4) 738.

¹⁷¹⁴ Sec 24 of the HFEA 2008 and sec 31ZA of HFEA 1990. Wade, K "Reconceptualising the interest in knowing one's origins: a case for mandatory disclosure" 2020 *Medical Law Review* 28(4) 738.

¹⁷¹⁵ Sec 3(1)(a) of the Surrogacy Arrangements Act 1985 as amended by the HFEA 1990 and the HFEA 2008.

¹⁷¹⁶ Sec 3(1)(b) of the Surrogacy Arrangements Act 1985 as amended by the HFEA 1990 and the HFEA 2008.

The biggest differences between the provisions of the South African Children's Act and the provisions of the different Acts described above, are that in South Africa, the commissioning parent(s) must first obtain a court order, confirming the surrogacy agreement, before the surrogate may be artificially fertilised. Also, only where the surrogate is genetically linked to the commissioned child will she be entitled to a cooling off period of sixty days after the child's birth.

It is submitted that chapter 19 of the Children's Act provides more legal security for the different parties to a surrogacy agreement. The important advantage is that a surrogacy agreement is contractually enforceable, and it is scrutinised by a court before artificial fertilisation of the proposed surrogate may take place.

The following judgment explains the importance of a parental order and complying with sections 54 and 54A of HFEA 2008.

2.2 AB v DE¹⁷¹⁷

This case concerns an application for a parental order in respect of a child born as a result of IVF treatment (performed in Moscow) on a married Russian surrogate.¹⁷¹⁸ The first applicant's sperm was used together with the eggs from an anonymous Russian donor.¹⁷¹⁹ The applicants had 10 failed cycles of IVF treatment in the UK using donor eggs and a further failed IVF treatment in Russia.¹⁷²⁰ They also experienced two failed attempts at conception by an unmarried surrogate in Russia.¹⁷²¹ The applicants were thereafter put in touch with a married surrogate and she became pregnant after IVF treatment that was followed by her giving birth to the child ("C") in Moscow in 2012.¹⁷²² C was registered as a British citizen upon application and the family returned to the UK after receipt of his passport.¹⁷²³

¹⁷²³ *AB v DE* at par 5.

¹⁷¹⁷ [2013] EWHC 2413 (Fam).

¹⁷¹⁸ *AB v DE* at par 1.

¹⁷¹⁹ *AB v DE* at par 2.

¹⁷²⁰ *AB v DE* at par 3.

¹⁷²¹ *AB v DE* at par 4.

¹⁷²² *AB v DE* at par 4.

The court observes that because surrogacy agreements are legal in Russia, the applicants were thus treated as the legal parents of C under Russian law as they were legitimately registered as such on C's Russian birth certificate with the consent of the first respondent.¹⁷²⁴ Considering a parental order application, the court had to be satisfied that the criteria under section 54 of HFEA 2008 were fulfilled and that such an order would meet the lifelong welfare needs of C.¹⁷²⁵ The first applicant's gametes were used in the conception of C and he was carried by the first respondent.¹⁷²⁶ The applicants are married.¹⁷²⁷ This application was made within six (6) months of C's birth.¹⁷²⁸ The child C has been in the continuous care of the applicants and he had his home with them at the time of the application and of the granting of the order.¹⁷²⁹ The applicants are domiciled in the UK and they are older than eighteen (18) years.¹⁷³⁰ Because the surrogate was married at the time of conception, the court had to be satisfied that both the respondents had consented to the parental order.¹⁷³¹ Section 35 of the HFEA 2008 determines that because the respondents were married at the time of C's conception, the second respondent is C's legal father and he is therefore 'the other person who is a parent of the child', meaning that his consent was also required for the parental order.¹⁷³²

The court next turned to the aspect of payment for the surrogacy arrangement. Section 54(8) of the Act, the final requirement of section 54, deals with the issue of payments.¹⁷³³ An amount of €50,000.00 was paid by the applicants to the agency in terms of the surrogacy agreement.¹⁷³⁴ The surrogate received 196,269.00 RR for her actual expenses which included costs relating to travel, medication, clothes, child care,

AB v DE at par 6. It is important to note that neither of the respondents are treated as a legal parent of C in Russia.

¹⁷²⁵ AB v DE at par 7.

¹⁷²⁶ *AB v DE* at par 8. In terms of sec 54(1).

¹⁷²⁷ *AB v DE* at par 9. In terms of sec 54(2).

¹⁷²⁸ AB v DE at par 10. In terms of sec 54(3).

¹⁷²⁹ AB v DE at par 11. In terms of sec 54(4)(a).

¹⁷³⁰ AB v DE at par 11 and 12. In terms of sec 54(4)(b) and sec 54(5).

¹⁷³¹ AB v DE at par 13. In terms of sec 54(6)(a) and 7.

¹⁷³² AB v DE at par 13. In terms of sec 54(6)(b).

AB v DE at par 14. Sec 54(8) determines that the court must be satisfied that no money or other benefit (other than for expenses reasonably incurred) has been given or received by either of the applicants for or in consideration of (a) the making of the order; (b) any agreement required by subsection (6) above; (c) the handing over of the child to the applicants, or (d) the making of any arrangements with a view to the making of the order, unless authorised by the court.

¹⁷³⁴ *AB v DE* at par 15.

including a further 400,000.00 RR as compensation.¹⁷³⁵ The court had to consider the following factors when exercising its discretion to authorise the said payments: (a) was the said sum disproportionate to reasonable expenses: (b) were the applicants acting in good faith and without 'moral taint' while dealing with the surrogate; and finally, (c) were the applicants part of any attempt to defraud the authorities.¹⁷³⁶ It was clear to the court that about half of the global fee paid went to the agency.¹⁷³⁷ The court was satisfied that the amount paid to the first respondent was not unusually high in the context of what other surrogates received elsewhere in Russia.¹⁷³⁸ The court further stated that the payments made were not so disproportionate as to amount to an affront to public policy or to have overborne the will of the first respondent.¹⁷³⁹ The court had to exercise its discretion regarding authorising the payments made by the applicants to the agency, which included payments other than for expenses reasonably incurred.¹⁷⁴⁰ The court was satisfied that the evidence pointed to a warm and caring relationship between the applicants and the first respondent and that there was no suggestion of bad faith or moral taint on the side of the applicants.¹⁷⁴¹ The applicants requested a parental order to fully regularise C's status and their legal parenthood status, as at the time of application for the parental order, they had no strong legal connection with C as the child's parents, except for a residence order made by the court which gave the applicants parental responsibility.¹⁷⁴²

With regard to the child C, the court stated that its paramount consideration in relation to this application was C's lifelong welfare as set out in section 1(4) of the Adoption and Children Act 2002.¹⁷⁴³ The court confirmed that C's legal parents remain the respondents (as per section 33(1), (3) and section 35 of HFEA 2008) and that they

¹⁷³⁵ *AB v DE* at par 15.

¹⁷³⁶ *AB v DE* at par 16.

¹⁷³⁷ *AB v DE* at par 17. On the papers before the court it was explained that payments in surrogacy are not regulated by Russian law and there is no prohibition of commercial surrogacy in Russian law.

¹⁷³⁸ *AB v DE* at par 18.

¹⁷³⁹ *AB v DE* at par 21. The information regarding the first respondent's actions did support the conclusion that she entered into the surrogacy agreement of her own free will.

¹⁷⁴⁰ *AB v DE* at par 29.

¹⁷⁴¹ *AB v DE* at par 23.

AB v DE at par 25 and par 32 and 34. In light of the fact that the applicants are not C's legal parents, C lacks a lifelong connection with the applicants for matters such as inheritance, financial support and his wider identity. Without a parental order, he would be left in something of a legal vacuum without full legal membership of any family anywhere in the world.

¹⁷⁴³ *AB v DE* at par 30.

also shared parental responsibility for him (as per section 2(1) of the Children Act 1989).¹⁷⁴⁴ In terms of section 33(1), the woman who carries the child is seen as the child's mother and the second applicant is thereby excluded from being regarded as the parent of C.¹⁷⁴⁵ The first applicant is excluded from being considered C's father (even though he is C's biological father) by virtue of section 38(1).¹⁷⁴⁶ The court further observed that the respondents are not C's legal parents under Russian law and neither did they have a biological connection with C.¹⁷⁴⁷ The court concluded by stating that the welfare needs of C would clearly not be met by the respondents remaining his legal parents in the UK when they were not recognised as such in Russia. In addition, they had no intention of having any future parental role in C's life.¹⁷⁴⁸

C's future is in the long-term care of the applicants, they are his de facto legal parents and his welfare demands their relationship is given lifelong security which can only be achieved by making a parental order.¹⁷⁴⁹

2.3 CC v DD¹⁷⁵⁰

This matter relates to an application for a parental order in respect of Q, a one-year old boy born through an international surrogacy agreement.¹⁷⁵¹ The applicants were British and French, respectively, and Q was born in the US while the applicants lived in France.¹⁷⁵² The application was made in the UK.¹⁷⁵³ The first applicant was raised in the UK and lived there until 2006 when she met the second applicant and moved to France.¹⁷⁵⁴ The applicants were married in 2011 and the first applicant continued to return to the UK on a regular basis and maintained significant connections in the

¹⁷⁴⁴ *AB v DE* at par 31.

¹⁷⁴⁵ *AB v DE* at par 31.

AB v DE at par 31. "Section 38(1) provides 'Where a person is to be treated as the father of the child by virtue of section 35 of 36, no other person is to be treated as the father of the child'.

AB v DE at par 31. The court referred to the Russian export report which states that the applicants are considered as C's legal parents and the respondents have no parental rights in respect of him.
 AB v DE at par 36

¹⁷⁴⁸ *AB v DE* at par 36.

¹⁷⁴⁹ *AB v DE* at par 36.

¹⁷⁵⁰ [2014] EWHC 1307 (Fam).

¹⁷⁵¹ *CC v DD* at par 1 and 3.

¹⁷⁵² *CC v DD* at par 6.

¹⁷⁵³ CC v DD at par 3. The surrogate agreement involved legal procedures between two US States.

¹⁷⁵⁴ *CC v DD* at par 4.

UK.¹⁷⁵⁵ The surrogacy agreement was entered into under the law of Iowa (where the respondents lived at the time of application) and Minnesota (where the surrogacy agency is based).¹⁷⁵⁶ The second respondent (as surrogate) donated her egg to the applicants which resulted in the conception and birth of Q.¹⁷⁵⁷ The necessary legal steps were taken in Minnesota to secure the applicants' status as Q's legal parents.¹⁷⁵⁸ This include an order made in line with the normal post-birth procedure in Minnesota, confirming the first applicant as Q's legal father with the consent of the surrogate mother, extinguishing her legal motherhood.¹⁷⁵⁹ A stepparent adoption order was made under the law of Minnesota whereby the second applicant's position as Q's mother was secured.¹⁷⁶⁰ The applicants were recorded as Q's legal parents on his birth certificate which was issued in Iowa.¹⁷⁶¹ By virtue of the Full Faith and Credit Clause of the US Constitution, the orders made by the court in Minnesota are recognised in other US States, including Iowa.¹⁷⁶²

The court stated that it had to be satisfied about two matters in considering this application, namely, (a) whether the requirements of section 54 of HFEA 2008 are satisfied, and (b) whether Q's lifelong welfare needs will be secured by this court making a parental order (section 1 of the Adoption and Children Act 2002).¹⁷⁶³ The court had to consider the relevance, or not, of the adoption order that was granted in Minnesota.¹⁷⁶⁴

Although a parental order has the effect of conferring parental responsibility on the child's new legal parents, the primary function of the application is to transfer legal parenthood from the surrogate and her husband to the applicants.¹⁷⁶⁵ Thus, the relevant legal provision governing this application is primarily concerned with parentage and status and not parental responsibility. The court was satisfied that the

¹⁷⁶⁰ CC v DD at par 8.

¹⁷⁶³ *CC v DD* at par 9.

¹⁷⁵⁵ *CC v DD* at par 5.

¹⁷⁵⁶ *CC v DD* at par 6. Q resided in Minnesota temporarily after his birth.

¹⁷⁵⁷ CC v DD at par 7.

¹⁷⁵⁸ CC v DD at par 8.

¹⁷⁵⁹ *CC v DD* at par 8.

¹⁷⁶¹ CC v DD at par 8.

¹⁷⁶² CC v DD at par 8.

¹⁷⁶⁴ CC v DD at par 11.

¹⁷⁶⁵ CC v DD at par 21.

surrogate and her husband had consented freely, unconditionally and with full understanding to the making of a parental order.¹⁷⁶⁶ Regarding the question of payments, a total of approximately \$35, 780.00 was paid in respect of the surrogacy agreement.¹⁷⁶⁷ The amount of \$19,200.00 was paid towards compensation rather than expenses and this would be the focus of the court requiring authorisation under section 54(8).¹⁷⁶⁸ The court was satisfied that the sums paid were not so disproportionate to the expenses that were incurred and that the applicants acted in good faith and the payments were therefore authorised.¹⁷⁶⁹ Documents submitted to the court confirmed that surrogacy was not illegal in either Minnesota or Iowa.¹⁷⁷⁰

Two novel points raised by the existence of the adoption order in this case are: (a) whether the applicants had committed any breach of UK domestic adoption law pursuant to section 83 of the Adoption and Children Act 2002, despite the fact that in the US the adoption was not considered as an international adoption (subject to the Hague Convention), and (b) whether the applicants were already recognised as Q's legal parents under the UK law by virtue of the adoption order and, if so, what impact that would have on the making of a parental order.¹⁷⁷¹ The court found that there had been no breach of section 83 of the Adoption and Children Act 2002 because neither of the applicants was or has been habitually resident in the British Isles, with the result that this section did not apply.¹⁷⁷² Because the US is a designated country for the purpose of an overseas adoption, the adoption order should be recognised in the UK.¹⁷⁷³

Because the second applicant was not the subject of the adoption order, his status as Q's legal father was not recognised in the UK by virtue of section 35 of HFEA 2008,

¹⁷⁶⁶ CC v DD at par 29. Sec 54(6) and (7).

¹⁷⁶⁷ *CC v DD* at par 30.

¹⁷⁶⁸ *CC v DD* at par 30. Commercial organisations (the agency) are lawfully allowed to operate in the US.

¹⁷⁶⁹ *CC v DD* at par 31.

¹⁷⁷⁰ *CC v DD* at par 34.

¹⁷⁷¹ *CC v DD* at par 35. The court pointed out in par 36 that sec 83 of the Adoption and Children Act 2002 creates an offence for persons who are habitually resident in the British Islands to adopt a child abroad unless they have complied with the provisions of that section and all the relevant regulations made there under.

¹⁷⁷² *CC v DD* at par 36.

¹⁷⁷³ CC v DD at par 38.

as the surrogate was married.¹⁷⁷⁴ A parental order would, however, confer legal parenthood on both the applicants.¹⁷⁷⁵ In granting the parental order, the court concluded by stating that it is clear that Q's lifelong welfare could only be met by securing his relationship with both the applicants in a lifelong way that would give them equal status and would endure for the rest of their lives.¹⁷⁷⁶

2.4 JP v LP and others¹⁷⁷⁷

The applicant was the mother of a little boy, CP, born because of an informal surrogacy agreement.¹⁷⁷⁸ The surrogate (SP) was a friend of the applicant. The surrogate was artificially fertilised at home with the sperm of the commissioning father, whereafter she became pregnant.¹⁷⁷⁹ The court mentioned that under section 1(A) of the Surrogacy Arrangements Act 1985, surrogacy agreements are not enforceable by law.¹⁷⁸⁰ At the time the surrogate was about to give birth to CP after the hospital became aware that the birth would follow as a result of a surrogacy agreement, the hospital asked the parties to conclude a surrogacy agreement.¹⁷⁸¹ The agreement had to be provided to the hospital.¹⁷⁸²

After the birth, CP was handed over to the applicant and the father.¹⁷⁸³ The surrogate registered CP's birth (the birth certificate showed her as the registered mother and the commissioning father as CP's father).¹⁷⁸⁴ Although the applicant and the father had planned to issue an application for a parental order in terms of section 54 of HFEA 2008, the applicant failed to lodge the application within the prescribed six months

¹⁷⁷⁹ JP v LP and others at par 4.

¹⁷⁷⁴ *CC v DD* at par 40(1).

¹⁷⁷⁵ CC v DD at par 40(1).

¹⁷⁷⁶ CC v DD at par 43.

¹⁷⁷⁷ [2014] EWHC 595 (Fam).

 $^{^{1778}}$ JP v LP and others at par 1 – 3. The applicant is however not the birth nor the genetic mother of CP. The applicant had a hysterectomy and the applicant and the respondent father (LP) wanted to conceive through partial surrogacy.

¹⁷⁸⁰ JP v LP and others at par 5.

JP v LP and others at par 6. At par 7. The agreement was prepared by a firm of Birmingham solicitors. The solicitors were actually committing a criminal offence as, whilst such agreements can lawfully be drawn up free of charge, the solicitors in preparing and charging for the preparation of the said agreement were negotiating surrogacy arrangements on a commercial basis which is in contravention of sec 2 of the Surrogacy Arrangements Act 1985.

¹⁷⁸² JP v LP and others at par 6.

¹⁷⁸³ JP v LP and others at par 8.

¹⁷⁸⁴ *JP v LP and others* at par 8. At par 9-10. After the relationship between the applicant and the father broke down, a shared residence order was made in favour of the applicant and the father.

after CP's birth.¹⁷⁸⁵ The applicant made an application for sole residence of CP, despite that no parental order was yet made.¹⁷⁸⁶

The court had to address the issues which arose in circumstances where there was no parental order regularising the legal status of each of the parties involved.¹⁷⁸⁷ The court also had to consider how this impacted on the exercise of parental responsibility by each of the parties (the applicant (child's mother), the father and the surrogate).¹⁷⁸⁸

The court referred to full surrogacy involving *in vitro* fertilisation and stated that this must take place in a licenced fertility clinic, as it must fall within the remit of the HFEA regulations.¹⁷⁸⁹ Partial surrogacy, on the other hand, may be carried out in an informal manner between the parties themselves and without involving a licenced fertility clinic.¹⁷⁹⁰ Thus, such partial surrogacy cases are not controlled or regulated.¹⁷⁹¹ The HFEA 2008 spells out the legal effect of such an informal arrangement, despite that the arrangement might have taken place outside the structure of the Act.¹⁷⁹²

In terms of section 33(1) of the HFEA, the surrogate mother is regarded as the child's (CP) legal mother, unless the child is adopted after birth or parenthood is transferred through a parental order.¹⁷⁹³ The father is seen as the genetic and social father of CP.¹⁷⁹⁴ The commissioning mother does not have parental responsibility without any

¹⁷⁸⁵ JP v LP and others at par 10 and 11.

¹⁷⁸⁶ JP v LP and others at par 15.

¹⁷⁸⁷ JP v LP and others at par 16.

¹⁷⁸⁸ JP v LP and others at par 16.

 $^{^{1789}}$ JP v LP and others at par 21. The commissioning couple's sperm and egg are fertilised in vitro and implanted into the surrogate mother. Further see ch 1, par 1.2.5 above.

¹⁷⁹⁰ JP v LP and others at par 22. The parties in this matter opted for partial surrogacy in that the surrogate mother's egg was inseminated with the commissioning father's sperm and it was done at home.

JP v LP and others at par 22. In these circumstances, there is no third party consideration of the welfare of the child to be born and none of the three people (commissioning parents and the surrogate mother) involved received the necessary information about critical issues for instance, who will be the legal parents of the child or information or counselling to ensure they understand the challenges presented to parents about issues, such as identity, consequent upon a surrogate birth.

¹⁷⁹² JP v LP and others at par 23.

 $^{^{1793}}$ JP v LP and others at par 23 (i). The surrogate mother has and retains parental responsibility without a subsequent adoption or a parental order.

¹⁷⁹⁴ *JP v LP and others* at par 23 (ii). Because the surrogate mother was not married and because she was not treated in a UK licenced clinic, she was not in the category of relationship which would satisfy the 'fathership' conditions in terms of sec 37 of the Act. The relationship could otherwise have the effect of making her husband or partner the legal father in place of the genetic father.

legal intervention and she has no status other than the emotional and social status of being CP's psychological mother.¹⁷⁹⁵ The court pointed out that prior to November 1994, the only means by which the commissioning parents of a commissioned child could obtain parental status of the child, was through a full adoption procedure.¹⁷⁹⁶

Because section 54 of HFEA 2008 now regulates the application process for parental orders, an application for a parental order in this case was made on 16 July 2010.¹⁷⁹⁷ The only requirements in terms of section 54 that were satisfied by the applicants were sections 54(1)(a),¹⁷⁹⁸ 54(2)(a),¹⁷⁹⁹ 54(5),¹⁸⁰⁰ 54(6),¹⁸⁰¹ and 54(8).¹⁸⁰² The requirements that were not satisfied related to sections 54(3)¹⁸⁰³ and 54(4)(a).¹⁸⁰⁴ The parties did recognise that the policy and the purpose of parental orders were to provide for the speedy consensual regularisation of the legal parental status of a child's carers following a birth through a surrogacy agreement.¹⁸⁰⁵ The court stated that the policy did not fit comfortably with extensions of time in this case, which inevitably resulted in the continued involvement over a protracted period of the surrogate in the lives of the commissioning couple and their child.¹⁸⁰⁶ The court concluded that a parental order was not an option for the applicants.¹⁸⁰⁷ The court further explained that an adoption

¹⁷⁹⁵ *JP v LP and others* at par 23 (iii). The effect is that she does not have parental responsibility which means that she cannot give consent to medical treatment, register CP for school or take a myriad of decisions in relation to CP which parents usually do without a thought as to whether or not they have the authority to do so.

JP v LP and others at par 24. The court explained in par 25 that The Parental Orders (Human Fertilisation and Embryology) Regulations 1994 was brought in effect by sec 30 of the HFEA 1990. The regulations created parental orders which provided for a consensual 'fast-track' way for the transfer of legal parenthood to a couple in the case where the child was conceived using the gametes of at least one of the couple. Sec 30 was repealed and the new provisions for parental orders can be found in sec 54 of HFEA 2008. The Act expanded the categories of individuals in respect of whom parental orders could be made so as to include married couples, civil partners, unmarried opposite- sex couples and same sex couples not in a civil partnership.

¹⁷⁹⁷ *JP v LP and others* at par 25 and 27.

¹⁷⁹⁸ CP was carried by a woman (artificially fertilised with the sperm of the father) and she was not one of the applicants.

¹⁷⁹⁹ The applicants are husband and wife.

¹⁸⁰⁰ The applicants are older than 18 years.

¹⁸⁰¹ The surrogate mother consented unconditionally to the making of the parental order.

¹⁸⁰² JP v LP and others at par 27. No money or benefit had been given or received.

¹⁸⁰³ The application was not made within the six-month period. The court pointed out that sec 54(3) states the parties must apply for the order during the prescribed time and there is no provision within the Act to provide for a discretionary extension to the statutory six-month time limit and no one sought to argue that the court could seek to circumnavigate the mandatory provisions of the statute.

¹⁸⁰⁴ *JP v LP and others* at par 28. The question regarding if CP's home was with the applicants at the time of the application was not argued although there is a shared residence order.

¹⁸⁰⁵ JP v LP and others at par 30.

¹⁸⁰⁶ JP v LP and others at par 30.

¹⁸⁰⁷ JP v LP and others at par 31.

order did not provide a solution to the mother's status of irrevocable parental responsibility.¹⁸⁰⁸ A special guardianship order was also not an option because the applicant would then have been able to exercise her newly granted parental responsibility to the exclusion of the surrogate and the first respondent (as the father).¹⁸⁰⁹ Although a shared residence order was granted and parental responsibility was conferred upon the applicant, the residence order did confer legal motherhood upon her.¹⁸¹⁰ The shared residence order merely regulated where CP would live and it gave the applicant parental responsibility, yet the surrogate retained legal motherhood and parental responsibility of CP in terms of section 33 of HFEA 2008.¹⁸¹¹

The parties subsequently reached a shared care arrangement in respect of CP on the following terms: that CP would be made and remain a ward of court until further order; a shared residence order be made between the mother and father (applicant and first respondent); that all issues of parental responsibility be delegated to the mother and the father jointly; and the surrogate be prohibited from exercising any parental responsibility for CP without the leave of the court.¹⁸¹²

This case is very interesting in that it shows how the issue of legal parenthood is handled completely different in the UK compared to South Africa. The UK framework resembles the legal position in South Africa prior to the promulgation of chapter 19 of the Children's Act. Without a successful parental order, the commissioning parents will not be considered the legal parents of their commissioned child.

2.5 *Re A*¹⁸¹³

¹⁸⁰⁸ *JP v LP and others* at par 32. The court explained that if the mother were to adopt CP alone, the father's parental responsibility would be extinguished pursuant to sections 46 and 67 of the Adoption and Children Act 2002. The applicant and first respondent cannot adopt together because they got divorced and they are not living together as partners in an enduring family relationship.

¹⁸⁰⁹ JP v LP and others at par 33.

¹⁸¹⁰ JP v LP and others at par 34. In the event that she ceases to have a residence order, she would lose her parental responsibility over CP as well.

¹⁸¹¹ JP v LP and others at par 34.

¹⁸¹² *JP v LP and others* at par 36. The court explained in par 37 that given the wholly exceptional circumstances of this case, wardship is the most appropriate way in which to manage the overall use of parental responsibility as between the father, the legal mother and the psychological mother of CP.

¹⁸¹³ [2015] EWHC 1756 (Fam).

This is an application for a parental order in relation to A, a little girl, born as a result of a surrogacy agreement concluded in South Africa.¹⁸¹⁴ There were three particular features to this application: (a) it was the court's first experience of dealing with a surrogate agreement from South Africa; (b) the court agreed to proceed with the application despite that A was not present in UK by allowing her to join via video link; and (c) the Parental Order Reporter was assisted by a social worker in South Africa to assist with the welfare enquiries that had to be made.¹⁸¹⁵

Two matters that the court had to be satisfied of, were compliance with the requirements under section 54 of HFEA 2008; and whether A's lifelong welfare needs as per section 1 of the Adoption and Children Act 2002 would be met by the court granting the parental order.¹⁸¹⁶ The court stated that the surrogate had cooperated entirely with the South African process overseen by the High Court in advance of A's conception, which extinguished the surrogate's status as A's legal mother.¹⁸¹⁷ The court was satisfied that all the criteria in respect of section 54 were met.¹⁸¹⁸

The court remarked that the lifelong needs of A included ensuring that her legal relationship with those who care for her was put on the securest footing possible.¹⁸¹⁹ As stated above, unless an order is made under UK law, the respondent surrogate would remain A's legal mother, although this would not be the legal position in South Africa.¹⁸²⁰ In granting the parental order, the court expressed the view that it was therefore clearly in A's lifelong interests that there was consistency regarding her legal status in relation to those who care for her across the relevant jurisdictions where she was likely to live.¹⁸²¹

¹⁸¹⁹ *Re A* at par 21.

¹⁸¹⁴ Re A at par 1 and 2. The surrogate mother is a widowed lady residing in South Africa. A was born in South Africa after she was conceived through IVF treatment with an embryo created from the sperm of the father and an egg from a donor.

¹⁸¹⁵ *Re A* at par 2. The court pointed out in par 3 that this was the second application made by the applicants in that they also made application for a parental order in 2011 in respect of a little boy who was born as a result of a surrogacy agreement concluded in South Africa.

¹⁸¹⁶ *Re A* at par 4.

¹⁸¹⁷ *Re A* at par 16.

¹⁸¹⁸ *Re A* at par 19. The court did describe the procedure that has to be followed in South Africa in par 12 to 15.

¹⁸²⁰ *Re A* at par 21.

¹⁸²¹ *Re A* at par 21.

Although the only permissible payments in terms of a surrogacy agreement are for reasonable expenses, the judgments above seem more flexible than courts applying chapter 19 of the Children's Act in South Africa.

The regulation of surrogacy in Canada will be examined next, pointing out differences and similarities where relevant.

3. CANADA

3.1 Regulation of surrogate motherhood

As is the case in South Africa and the UK, altruistic surrogacy is the only form of assisted reproduction allowed in Canada. The relevant legislation and guidelines that govern surrogacy and artificial fertilisation are the Assisted Human Reproduction Act;¹⁸²² Consent for the Use of Human Reproductive Material and In Vitro Embryos Regulations;¹⁸²³ Administration and Enforcement (Assisted Human Reproduction Act) Regulations;¹⁸²⁴ Reimbursement Related to Assisted Human Reproduction Regulations;¹⁸²⁵ and the Safety of Sperm and Ova Regulations.¹⁸²⁶ The following sections of Assisted Human Reproduction Act are relevant to surrogacy: section 6 (the payment for surrogacy services and related matters); section 12 (the reimbursement of expenditures)¹⁸²⁷ and section 60 (offences in terms of the Act and punishment). The most relevant regulations to surrogacy are Reimbursement Related to Assisted Human Reproduction Act and Punishment).

¹⁸²² S.C. 2004, c. 2. This Act was assented to on 29 March 2004. Accessed from <u>https://laws-lois.justice.gc.ca/PDF/A-13.4.pdf</u> (accessed 20 January 2023).

¹⁸²³ SOR/2007-137. Accessed from <u>https://laws-lois.justice.gc.ca/PDF/SOR-2007-137.pdf</u> (accessed on 20 January 2023).

¹⁸²⁴ SOR/2019-194. Accessed from <u>https://laws-lois.justice.gc.ca/PDF/SOR-2019-194.pdf</u> (accessed on 20 January 2023). These Regulations came into force on date of registration, 10 June 2019.

 ¹⁸²⁵ SOR/2019-193. Accessed from <u>https://laws-lois.justice.gc.ca/PDF/SOR-2019-193.pdf</u> (accessed on 20 January 2023). These Regulations came in force with sec 12 of the Assisted Human Reproduction Act. See reg 13 of Reimbursement Related to Assisted Human Reproduction Regulations. Carsley, S "Regulating reimbursements for surrogate mothers" 2021 *Alberta Law Review* 58(4) 812. Carsley pointed out that these regulations came into force sixteen years after Assisted Human Reproduction Act received royal assent on 29 March 2004.
 ¹⁸²⁶ SOR/2010, 102

¹⁸²⁶ SOR/2019-192.

¹⁸²⁷ Sec 12 came into force on 9 June 2020. Sec 78 of Assisted Human Reproduction Act.

The surrogate is assumed to be the legal parent of the child born from a surrogacy agreement.¹⁸²⁸ Interesting to note, and similar to the position in the UK, is that the surrogate is only allowed to consent to the transfer of her parental rights in respect of the child to the commissioning (intended) parents *after* the birth of the child.¹⁸²⁹ Provincial legislation in Canada determines the parentage or filiation of a child born through surrogacy.¹⁸³⁰ For example, in Quebec, parental rights are relinquished through 'special consent' adoption, whereas in Ontario and British Columbia, the commissioning parents may be registered as the parents of the child if the surrogate consented thereto after the child's birth and if the parties concluded a written agreement prior to conception, setting out the intentions regarding the child's parentage.¹⁸³¹ Ontario further determines that the surrogate must receive independent legal advice and she is only allowed to consent seven days after the birth of the child.¹⁸³² To make matters more difficult, some provincial statutes determine that surrogacy agreements are not legally binding and that a surrogate cannot be forced to relinquish her parental status solely based on a surrogacy agreement.¹⁸³³

In contrast to chapter 19 of the Children's Act in South Africa, the Canadian Assisted Human Reproduction Act does not deal with the aspect of surrogacy with the same amount of detail. The Canadian Act determines that a woman is not allowed to be paid to act as a surrogate and no person shall offer to pay a surrogate or advertise that she will be paid.¹⁸³⁴ No person is allowed to act as go-between in accepting payment for arranging for the services of a surrogate or to offer to make such an arrangement or to advertise that such services can be arranged.¹⁸³⁵ It is further not allowed for a person to pay a go-between for arranging surrogacy services nor to offer to pay or

¹⁸²⁸ Carsley, S "Surrogacy in Canada: Lawyers' experiences and practices" 2022 *Canadian Journal of Women and the Law* 34(1) 45.

¹⁸²⁹ Carsley, S "Surrogacy in Canada: Lawyers' experiences and practices" 2022 *Canadian Journal* of Women and the Law 34(1) 45.

¹⁸³⁰ Carsley, S "Surrogacy in Canada: Lawyers' experiences and practices" 2022 *Canadian Journal of Women and the Law* 34(1) 45.

¹⁸³¹ Carsley, S "Surrogacy in Canada: Lawyers' experiences and practices" 2022 *Canadian Journal* of Women and the Law 34(1) 45 (fn 13).

¹⁸³² Carsley, S "Surrogacy in Canada: Lawyers' experiences and practices" 2022 *Canadian Journal* of Women and the Law 34(1) 45 (fn 13).

¹⁸³³ Carsley, S "Surrogacy in Canada: Lawyers' experiences and practices" 2022 *Canadian Journal* of Women and the Law 34(1) 45.

 ¹⁸³⁴ Sec 6(1) of Assisted Human Reproduction Act.
 ¹⁸³⁵ Sec 6(2) of Assisted Human Reproduction Act.

¹⁸³⁵ Sec 6(2) of Assisted Human Reproduction Act.

advertise to pay for the said arrangement.¹⁸³⁶ Importantly, a proposed surrogate may not be younger than 21 years.¹⁸³⁷

Section 12 sets out the provisions relating to the repayment for expenditures relating to the donation of ova or sperm,¹⁸³⁸ the maintenance or transport of an *in vitro* embryo,¹⁸³⁹ and costs relating to a woman acting as surrogate.¹⁸⁴⁰ Reimbursement may only be paid in accordance with the Reimbursement Related to Assisted Human Reproduction Regulations.¹⁸⁴¹ No reimbursement may be made by any person for the expenses unless he or she received a receipt for the said expenses.¹⁸⁴² No person is allowed to reimburse a surrogate for a loss of income that she suffered during her pregnancy, except where a qualified medical practitioner provides a written confirmation that the continuation of work may pose a risk to the surrogate's health or to that of the embryo or the foetus.¹⁸⁴³

Any person breaching the provisions set out in section 6 will be guilty of an offence which is punishable. The person who is convicted on indictment is liable to a fine not exceeding \$500,000 or imprisonment for a period not exceeding ten years or both.¹⁸⁴⁴ On summary conviction, the person is liable to a fine not exceeding \$250,000 or to imprisonment for a period not exceeding ten years, or to both.¹⁸⁴⁵

Regulations 4 to 11 of Reimbursement Related to Assisted Human Reproduction Regulations¹⁸⁴⁶ provide for surrogacy expenses and the reimbursement thereof. The expenses incurred by a surrogate relating to her surrogacy that may be lawfully reimbursed are set out in regulation 4. The following expenses are provided for:

¹⁸³⁶ Sec 6(3) of Assisted Human Reproduction Act.

¹⁸³⁷ Sec 6(4) of Assisted Human Reproduction Act.

¹⁸³⁸ Sec 12(1)(a) of Assisted Human Reproduction Act.

¹⁸³⁹ Sec 12(1)(b) of Assisted Human Reproduction Act.

¹⁸⁴⁰ Sec 12(1)(c) of Assisted Human Reproduction Act.

¹⁸⁴¹ SOR/2019-193.

¹⁸⁴² Sec 12(2) of Assisted Human Reproduction Act.

¹⁸⁴³ Sec 12(3)(a) of Assisted Human Reproduction Act. The repayment may only be made in accordance with the said Regulations.

¹⁸⁴⁴ Sec 60(a) of Assisted Human Reproduction Act.

¹⁸⁴⁵ Sec 60(b) of Assisted Human Reproduction Act.

¹⁸⁴⁶ SOR/2019-193.

travelling;¹⁸⁴⁷ care of dependants or pets;¹⁸⁴⁸ counselling;¹⁸⁴⁹ legal services and disbursements;¹⁸⁵⁰ obtaining any drug or device (as defined in section 2 of the Food and Drugs Act);¹⁸⁵¹ obtaining products or services that a person (authorised under the laws of a province to assess, monitor and provide health care to a pregnant woman) provided or recommended in writing,¹⁸⁵² and for getting the recommendation.¹⁸⁵³ Also covered are: services of a midwife or doula;¹⁸⁵⁴ groceries (except any non-food items);¹⁸⁵⁵ maternity clothes;¹⁸⁵⁶ telecommunications;¹⁸⁵⁷ prenatal exercise classes¹⁸⁵⁸ linked with the delivery;¹⁸⁵⁹ health, disability, travel or life insurance cover;¹⁸⁶⁰ and to obtain or confirm medical or other records.¹⁸⁶¹

Before a person is allowed to reimburse any of the expenses as set out in regulation 4, he or she must have obtained the following documents:¹⁸⁶² a declaration by the person requesting repayment (dated and signed by the person)¹⁸⁶³ setting out his or her name and address,¹⁸⁶⁴ the nature of each expense,¹⁸⁶⁵ the amount incurred for each expense and the amount requested for reimbursement if the amount is less than the incurred amount;¹⁸⁶⁶ the date of each expense;¹⁸⁶⁷ where transportation was

¹⁸⁴⁷ Reg 4(a) Reimbursement Related to Assisted Human Reproduction Regulations.

¹⁸⁴⁸ Reg 4(b) Reimbursement Related to Assisted Human Reproduction Regulations.

 ¹⁸⁴⁹ Reg 4(c) Reimbursement Related to Assisted Human Reproduction Regulations.
 ¹⁸⁵⁰ Reg 4(d) Reimbursement Related to Assisted Human Reproduction Regulations.

 ¹⁸⁵⁰ Reg 4(d) Reimbursement Related to Assisted Human Reproduction Regulations.
 ¹⁸⁵¹ Reg 4(d) Reimbursement Related to Assisted Human Reproduction Regulations.

 ¹⁸⁵¹ Reg 4(e) Reimbursement Related to Assisted Human Reproduction Regulations.
 ¹⁸⁵² Reg 4(f) Reimbursement Related to Assisted Human Reproduction Regulations.

 ¹⁸⁵² Reg 4(f) Reimbursement Related to Assisted Human Reproduction Regulations.
 ¹⁸⁵³ Reg 4(g) Reimbursement Related to Assisted Human Reproduction Regulations.

¹⁸⁵³ Reg 4(g) Reimbursement Related to Assisted Human Reproduction Regulations.

¹⁸⁵⁴ Reg 4(h) Reimbursement Related to Assisted Human Reproduction Regulations. A doula can be described as a person who is a trained professional who supports a woman before, during and after she had a baby. Doulas however do not deliver health care services, but they do offer physical and emotional support to a pregnant woman who is preparing to welcome a new baby into her and her family's life. A doula assists and advocates for the mother or birthing parent. See https://my.clevelandclinic.org/health/articles/23075-doula (accessed on 7 June 2023).

¹⁸⁵⁵ Reg 4(i) Reimbursement Related to Assisted Human Reproduction Regulations.

¹⁸⁵⁶ Reg 4(j) Reimbursement Related to Assisted Human Reproduction Regulations.

 ¹⁸⁵⁷ Reg 4(k) Reimbursement Related to Assisted Human Reproduction Regulations.
 ¹⁸⁵⁸ Reg 4(l) Reimbursement Related to Assisted Human Reproduction Regulations.

 ¹⁸⁵⁸ Reg 4(I) Reimbursement Related to Assisted Human Reproduction Regulations.
 ¹⁸⁵⁹ Reg 4(m) Reimbursement Related to Assisted Human Reproduction Regulations.

 ¹⁸⁵⁹ Reg 4(m) Reimbursement Related to Assisted Human Reproduction Regulations.
 ¹⁸⁶⁰ Reg 4(n) Reimbursement Related to Assisted Human Reproduction Regulations.

¹⁸⁶¹ Reg 4(o) Reimbursement Related to Assisted Human Reproduction Regulations.

¹⁸⁶² Reg 6 Reimbursement Related to Assisted Human Reproduction Regulations.

¹⁸⁶³ Reg 6(a) Reimbursement Related to Assisted Human Reproduction Regulations.

¹⁸⁶⁴ Reg 6(a)(i) Reimbursement Related to Assisted Human Reproduction Regulations.

¹⁸⁶⁵ Reg 6(a)(ii) Reimbursement Related to Assisted Human Reproduction Regulations.

¹⁸⁶⁶ Reg 6(a)(iii) Reimbursement Related to Assisted Human Reproduction Regulations.

¹⁸⁶⁷ Reg 6(a)(iv) Reimbursement Related to Assisted Human Reproduction Regulations.

used;¹⁸⁶⁸ the addresses of the points of departure and the destination,¹⁸⁶⁹ together with the total distance in kilometres between the two points;¹⁸⁷⁰ a statement that states each of the expenses was incurred in relation to surrogacy;¹⁸⁷¹ a statement indicating that the amount of each expense has not been paid to him or her by any other source, in full or in part;¹⁸⁷² and a statement that confirms all of the information set out in the declaration is correct and complete to the best of his or her knowledge.¹⁸⁷³ Further, a receipt for each expense for which reimbursement is requested that identifies the date of each expense is also required.¹⁸⁷⁴

Regulation 7 determines that a person who reimburses an expense as set out in the declaration must indicate on the document the amount of each expense that was repaid, and he or she must sign it to confirm the information. The preliminary requirements for the reimbursement of a surrogate for her loss of income is set out in regulation 8. The following documents are necessary before a person may reimburse a surrogate for a loss of income suffered by her during the pregnancy:¹⁸⁷⁵ the surrogate must provide the person with a declaration dated and signed by her and setting out her name and address;¹⁸⁷⁶ the start and end dates of the period that she could not work during her pregnancy (for a reason certified by a qualified medical practitioner in terms of section 12(3) of the Act);¹⁸⁷⁷ the amount that she is requesting for repayment;¹⁸⁷⁸ a statement wherein the surrogate indicate that she has not received compensation from any other source, in full or in part, for her loss of income;¹⁸⁷⁹ and a statement confirming that all the information set out in the declaration is correct and complete to the best of her knowledge.¹⁸⁸⁰ Further, supporting evidence must be

¹⁸⁶⁸ Reg 6(a)(v) Reimbursement Related to Assisted Human Reproduction Regulations. If no receipt was provided by the transport service.

¹⁸⁶⁹ Reg 6(a)(v)(A) Reimbursement Related to Assisted Human Reproduction Regulations.

¹⁸⁷⁰ Reg 6(a)(v)(B) Reimbursement Related to Assisted Human Reproduction Regulations.

¹⁸⁷¹ Reg 6(a)(vi) Reimbursement Related to Assisted Human Reproduction Regulations. This regulation is also regarding expenses incurred in the course of donating ova or sperm and in the maintenance or transport of an *in vitro* embryo.

¹⁸⁷² Reg 6(a)(vii) Reimbursement Related to Assisted Human Reproduction Regulations.

¹⁸⁷³ Reg 6(a)(viii) Reimbursement Related to Assisted Human Reproduction Regulations.

¹⁸⁷⁴ Reg 6(d) Reimbursement Related to Assisted Human Reproduction Regulations. A receipt is not necessary in the case of transport expenses.

¹⁸⁷⁵ Reg 8 Reimbursement Related to Assisted Human Reproduction Regulations. This repayment can be made for loss of income under sec 12(3) of Assisted Human Reproduction Act.

¹⁸⁷⁶ Reg 8(a)(i) Reimbursement Related to Assisted Human Reproduction Regulations.

¹⁸⁷⁷ Reg 8(a)(ii) Reimbursement Related to Assisted Human Reproduction Regulations.

¹⁸⁷⁸ Reg 8(a)(iii) Reimbursement Related to Assisted Human Reproduction Regulations.

¹⁸⁷⁹ Reg 8(a)(iv) Reimbursement Related to Assisted Human Reproduction Regulations.

¹⁸⁸⁰ Reg 8(a)(v) Reimbursement Related to Assisted Human Reproduction Regulations.

provided of the income that the surrogate would have earned had she not been absent from her work (for the period set out in regulation 8(a)(ii))¹⁸⁸¹ and a copy of the certification that was provided by a qualified medical practitioner in terms of section 12(3)(a) of Assisted Human Reproduction Act.¹⁸⁸²

The person who reimburses the surrogate for her loss of income as set out on the declaration must indicate on the document the amount of the loss of income that was repaid, the date that it was paid and he or she must sign it to confirm the information.¹⁸⁸³

South Africa will most definitely benefit from detailed regulations guiding the courts on the exact type of expenses that fall under the different categories of section 301 of the Children's Act. Requiring the surrogate to provide the commissioning parent(s) with proof of actual expenses protects the commissioning parent(s) from the surrogate trying to claim expenses unrelated to the surrogacy agreement. Furthermore, it is also submitted regulations be proposed, like the Canadian example above, where it becomes a requirement that the surrogate provides the commissioning parent(s) with an affidavit setting out the exact expenses together with the receipt for every expense. Further, the surrogate in South Africa should also be required to provide an affidavit with a medical certificate and proof of loss of income should she claim loss of income from the commissioning parent(s). Surrogacy is already an expensive exercise for the commissioning parents and these proposed regulations will protect them from unnecessary expenses should the surrogate inflate expenses or a loss of income.

Carsley argues that there are certain kinds of expenses for which surrogates have been reimbursed and they are likely to continue receiving these kinds of payments in spite of the Regulations, because Health Canada does not require proof of what the

¹⁸⁸¹ Reg 8(b) Reimbursement Related to Assisted Human Reproduction Regulations.

¹⁸⁸² Reg 8(c) Reimbursement Related to Assisted Human Reproduction Regulations.

¹⁸⁸³ Reg 9 Reimbursement Related to Assisted Human Reproduction Regulations. Reg 11(1) determines that the person who repaid expenses as referred to in the regulations must keep a record of all the documents obtained in respect of the repayments for a period of six years after the date of the repayment of each expense. Reg 11(2) determines that the person who repaid the surrogate for her loss of income must, for each repayment made, maintain a record of all the documents obtained for the purposes of repayment for a period of six years after the date of repayment.

surrogate's monthly expenses are prior to her pregnancy.¹⁸⁸⁴ Examples of expenses that were claimed are: lawn mowing, fitness classes, babysitting and snow clearing.¹⁸⁸⁵ These expenses are for services incurred independent of a surrogacy agreement.¹⁸⁸⁶ The author further argues that surrogates are also likely to continue receiving bonuses for being an experienced surrogate, giving birth to twins and undergoing a caesarean section, because Health Canada has done little to enforce the Assisted Human Reproductive Act.¹⁸⁸⁷ A Guidance Document was released alongside the Regulations to explain that the provision regarding expenses is intentionally broad so that it can cover different circumstances.¹⁸⁸⁸ It would, however, appear that this Guidance Document is causing confusion, for example, the Regulations provide for loss of income during the surrogate's pregnancy, yet the Guidance Document indicates that surrogates may be reimbursed for loss of income during the pre-pregnancy and the post-partum period.¹⁸⁸⁹ The Guidance Document does not have the force of law and judges are hence not bound by it (although it may be persuasive).¹⁸⁹⁰ In instances where a discrepancy or conflict arises between a statute and/or regulations and the Guidance Document, the first-mentioned will prevail.1891

A recent legal development worth mention is the introduction of "Bill 2" in Quebec in October 2022 that aims to create a framework regulating surrogacy by requiring the conclusion of a notarised surrogacy agreement before the onset of pregnancy. It would also require obtaining (after the birth of the child) the consent of the person who gave

¹⁸⁸⁴ Carsley, S "Regulating reimbursements for surrogate mothers" 2021 *Alberta Law Review* 58(4) 834.

¹⁸⁸⁵ Carsley, S "Regulating reimbursements for surrogate mothers" 2021 *Alberta Law Review* 58(4) 834.

¹⁸⁸⁶ Carsley, S "Regulating reimbursements for surrogate mothers" 2021 *Alberta Law Review* 58(4) 834.

 ¹⁸⁸⁷ Carsley, S "Regulating reimbursements for surrogate mothers" 2021 Alberta Law Review 58(4) 834.
 ¹⁸⁸⁸ Carsley, S "Begulating reimbursements for surrogate mothers" 2021 Alberta Law Review 58(4)

 ¹⁸⁸⁸ Carsley, S "Regulating reimbursements for surrogate mothers" 2021 Alberta Law Review 58(4) 833.
 ¹⁸⁸⁹ Carsley, S "Bagulating reimburgements for surrogate mothers" 2021 Alberta Law Review 58(4)

 ¹⁸⁸⁹ Carsley, S "Regulating reimbursements for surrogate mothers" 2021 Alberta Law Review 58(4) 835.
 ¹⁸⁹⁰ Carsley, S "Begulating reimbursements for surrogate mothers" 2021 Alberta Law Review 58(4)

 ¹⁸⁹⁰ Carsley, S "Regulating reimbursements for surrogate mothers" 2021 Alberta Law Review 58(4) 835.
 ¹⁸⁹¹ Carsley, S "Begulating reimbursements for surrogate mothers" 2021 Alberta Law Review 58(4)

¹⁸⁹¹ Carsley, S "Regulating reimbursements for surrogate mothers" 2021 *Alberta Law Review* 58(4) 835.

birth that she relinquishes any legal rights to the child.¹⁸⁹² A surrogate mother will also need to be at least 21 years old and would be compensated for loss of work income and for several expenses (yet not paid for her services).¹⁸⁹³ She would be able to change her mind and terminate the contract at any time without risking a lawsuit.¹⁸⁹⁴

Comparing the legal position in South Africa with that of Canada reveals interesting, yet not surprising similarities and differences. Firstly, for surrogacy to be lawful in both jurisdictions, it may only be altruistic surrogacy. Both South Africa and Canada criminalises commercial surrogacy. The legal status of the parties in a surrogacy arrangement resembles those in the UK, where federal law in Canada gives parental rights to the surrogate mother upon the birth of the commissioned child and not the commissioning parents. This is different from the South African position in terms of which legal parenthood is bestowed on the commissioning parents via legislation (chapter 19 of the Children's Act). Canada does not restrict surrogacy services to married, heterosexual couples because discrimination on grounds of sexual orientation or marital status is prohibited. Intended parents usually find the process of acquiring legal parental rights over children born of surrogacy, accommodating. It is curious that a lot of attention is placed on certain logistical matters, such as which payments may be made to surrogates, as well as the supporting documentation that is required. This approach is not observed in South Africa or the United Kingdom.

3.2 B.A.N. v. J.H.¹⁸⁹⁵

The petitioners in this case were husband and wife and they were seeking a declaratory order of parentage of children born as a result of a surrogacy

¹⁸⁹² West, R "Proposed changes to Quebec Surrogacy laws under Bill 2" <u>https://canadafertility.law/blog/proposed-changes-to-quebec-surrogacy-laws-under-bill-2/</u> (accessed 28 January 2023).

¹⁸⁹³ West, R "Proposed changes to Quebec Surrogacy laws under Bill 2" <u>https://canadafertility.law/blog/proposed-changes-to-quebec-surrogacy-laws-under-bill-2/</u> (accessed 28 January 2023).

¹⁸⁹⁴ National Assembly of Quebec, Second Session, 42nd Legislature. Bill 2 (2022, chapter 22). An Act respecting family law reform with regard to filiation and amending the Civil Code in relation personality riahts and civil status. Assented to to on 8 June 2022. http://www2.publicationsduquebec.gouv.gc.ca/dynamicSearch/telecharge.php?type=5&file=202 2C22A.PDF (accessed 25 January 2023).

¹⁸⁹⁵ 2008 BCSC 808 accessed on <u>https://www.canlii.org/en/bc/bcsc/doc/2008/2008bcsc808/2008bcsc808.html</u> (accessed on 30 January 2023).

agreement.¹⁸⁹⁶ The surrogate was only the gestational carrier as the husband's sperm was used together with donor ova.¹⁸⁹⁷ The donor and the surrogate both stated that consent was given freely and voluntarily for the intended parents to be the sole parents and guardians of the children and further that they lay no claim and wish that the intended parents be declared as the sole parents and guardians of the children.¹⁸⁹⁸ The surrogacy agreement that was concluded between the parties confirms the intended parents of the children.¹⁸⁹⁹ Further, the surrogate and her husband would have no parental relationship with the children.¹⁹⁰⁰ The court pointed out that:

The definition of "birth" in s. 1 of the *Vital Statistics Act*, R.S.B.C. 1996, c. 479, presumes the birth mother to be the mother of a child. The surrogate mother could of course register the birth, showing herself as the birth mother and the intended and genetic father as the birth father. Then, given that all five persons involved are consenting, for the intended mother to go through a second-parent adoption under s. 29(2) of the *Adoption Act*, R.S.B.C. 1996, c. 5. This would serve to sever any legal relationships that could then exist between the children and the egg donor or the surrogate mother by operation of s. 37(2)(b) of the *Adoption Act*.¹⁹⁰¹

The petitioners however did not want to follow the adoption route. The Vital Statistics Agency indicated in a letter that it approves the registration of the intended parents as the birth parents had indicated that at least one of them was a genetic parent of the children.¹⁹⁰² The court found that the province did not have legislation that dealt specifically with surrogacy and that, in light of the fact that the Vital Statistics Agency had approved the approach that was taken by the petitioners, and the other parties had agreed to the relief sought, the court declared that the intended parents are the parents of the children and also the parents to the exclusion of the egg donor and the surrogate.¹⁹⁰³

3.3 MAM v TAM¹⁹⁰⁴

¹⁸⁹⁶ B.A.N. v. J.H. at par 1. All the parties agreed to the relief sought by the petitioners.

¹⁸⁹⁷ B.A.N. v. J.H. at par 3. ¹⁸⁹⁸ B.A.N. v. J.H. at par 3.

¹⁸⁹⁸ *B.A.N. v. J.H.* at par 3. ¹⁸⁹⁹ *B.A.N. v. J.H.* at par 4.

¹⁸⁹⁹ B.A.N. v. J.H. at par 4.
¹⁹⁰⁰ B.A.N. v. J.H. at par 4.

¹⁹⁰¹ *B.A.N. v. J.H.* at par 8.

¹⁹⁰² *B.A.N. v. J.H.* at par 19.

¹⁹⁰³ *B.A.N. v. J.H.* at par 22-24.

 ¹⁹⁰⁴ 2015 NBQB 145 accessed on https://www.canlii.org/en/nb/nbqb/doc/2015/2015nbqb145/2015nbqb145.html?searchUrlHash= <u>AAAAAQAfYXNzaXN0ZWQgaHVtYW4gcmVwcm9kdWN0aW9uIGFjdAAAAAAB&resultIndex=1</u> <u>2</u> (accessed 30 January 2023).

This was an application wherein the applicants were seeking relief in the form of declaratory orders.¹⁹⁰⁵ The second applicant was to be recognised as the father of the children and that he is their natural father; the first applicant was to be recognised as the mother of the children and that she is their natural mother; the second respondent is not the natural father of the children and the first respondent is not the natural mother of the said children.¹⁹⁰⁶ The application was brought in terms of the Family Services Act, S.N.B., 2003, Chap.F-22, the Assisted Human Reproduction Act, S.C. 2004, C.2 and the Assisted Human Reproduction (section 8 consent) Regulations, SOR/2007-137.¹⁹⁰⁷ The applicants were married but they were unable to conceive naturally and the first applicant's uterus was unable to carry a pregnancy.¹⁹⁰⁸ The first respondent agreed to act as gestational surrogate for the applicants (donor ova and the sperm of the second applicant was used for conception) whereafter twin boys were born.¹⁹⁰⁹ The applicants took the children with them to Ontario, where they were residing.¹⁹¹⁰ The children's birth certificates however indicated the first respondent, the surrogate, and second applicant as the father.¹⁹¹¹ The parties concluded a surrogacy agreement wherein it was stated that the respondents agreed to relinquish any parental rights or obligations after the children were born.¹⁹¹² Further, the applicants agreed to pay the first respondent for expenses incurred as a result of the surrogacy and the first respondent would act altruistically.¹⁹¹³ The applicants concluded a reimbursement of expenses agreement with the donor.¹⁹¹⁴ The respondents were in agreement with the relief that the applicants were seeking.¹⁹¹⁵

The court pointed out that the Family Service Act, which was assented to on 16 July 1980, did not contemplate that children would be created with the sperm of the father and the ova of an anonymous donor.¹⁹¹⁶ The court found that the applicants have been the children's caregivers and it is evident, from the agreements concluded between

- ¹⁹¹⁰ MAM v TAM at par 8.
- ¹⁹¹¹ *MAM v TAM* at par 9.
- ¹⁹¹² MAM v TAM at par 13.
- ¹⁹¹³ *MAM v TAM* at par 13.
- ¹⁹¹⁴ *MAM v TAM* at par 11.
- ¹⁹¹⁵ *MAM v TAM* at par 15.
- ¹⁹¹⁶ *MAM v TAM* at par 27.

¹⁹⁰⁵ *MAM v TAM* at par 1.

 ¹⁹⁰⁶ MAM v TAM at par 1.
 ¹⁹⁰⁷ MAM v TAM at par 2.

MAM v TAM at par 2.
 MAM v TAM at par 4 and 6.

 $^{^{1909}}$ MAM v TAM at par 6-7.

the parties, that there was a clear intention that the applicants be the parents of the children.¹⁹¹⁷ Further, the court found that the applicants were both the social and the intended parents of the children and that there is a parent and child relationship between them and the children.¹⁹¹⁸ The court concluded that, taking into account also the best interests of the children, the applicants are declared to be the children's mother and father.¹⁹¹⁹ The court did not want to declare the first applicant the natural mother of the children but it held that the respondents are not to the parents of the children.¹⁹²⁰

The regulation of surrogacy in India will be discussed next.

4. INDIA

4.1 Regulation of surrogate motherhood

The reproductive right of persons is regarded as a basic right in India.¹⁹²¹ With commercial surrogacy not prohibited in India, Indian fertility clinics have often been accused of becoming "baby factories."¹⁹²² Some scholars regard the Indian model of commercial surrogacy as great success that is founded upon the notion of a liberal market model.¹⁹²³ However, in 2015 the Indian government prohibited commercial surrogacy.¹⁹²⁴

https://cdnbbsr.s3waas.gov.in/s3ca0daec69b5adc880fb464895726dbdf/uploads/2022/08/2022 081094-1.pdf (accessed 13 January 2023).

¹⁹¹⁷ *MAM v TAM* at par 30-31.

¹⁹¹⁸ *MAM v TAM* at par 32.

¹⁹¹⁹ *MAM v TAM* at par 33.

¹⁹²⁰ *MAM v TAM* at par 34 and 38.

¹⁹²¹ Law Commission of India (2009) *Need for legislation to regulate assisted reproductive technology clinics as well as rights and obligations of parties to a surrogacy* (Report no.228) 12 accessed from

¹⁹²² Elder, AH "Wombs to rent: Examining the jurisdiction of international surrogacy" 2014 16(2) Oregon Review of International Law 368. Ex parte application WH and others 2011 (4) All SA 630 (GP) at par 45. Nicholson 2013 SAJHR 508. Biko, J & Nene, Z 'Ethics aspects of third-party reproduction' Obstetrics & Gynaecology Forum 2017, Issue 3, p 14. The commercial surrogacy is leading to a boom in the fertility tourism trade. Law Commission of India (2009) Need for legislation to regulate assisted reproductive technology clinics as well as rights and obligations of parties to a surrogacy (Report no.228) 11 accessed from https://cdnbbsr.s3waas.gov.in/s3ca0daec69b5adc880fb464895726dbdf/uploads/2022/08/2022

¹⁹²³ 081094-1.pdf (accessed 13 January 2023). Nicholson 2013 SAJHR 508.

¹⁹²⁴ Gola, S "One Step forward or one step back? Autonomy, agency and surrogates in the Indian Surrogacy (Regulation) Bill 2019" 2021 *International Journal of Law in Context* 17(1) 59.

Previously, India did not have laws regulating surrogacy apart from guidelines that governed assisted reproductive technology procedures.¹⁹²⁵ The Assisted Reproductive Technology (Regulation) Bill, 2020, was drafted with the objective of regulating assisted reproduction and surrogacy.¹⁹²⁶ With the enactment of the Surrogacy (Regulation) Act¹⁹²⁷ and the Assisted Reproductive Technology (Regulation) Act¹⁹²⁸ surrogacy agreements and assisted reproduction were finally regulated more comprehensively.¹⁹²⁹

Chapter 2 of the Surrogacy (Regulation) Act regulates the registration of surrogacy clinics. Only a registered surrogacy clinic is allowed to conduct activities in respect of surrogacy and surrogacy procedures may only be performed in a place registered in terms of the Act.¹⁹³⁰ Commercial surrogacy is now completely prohibited, both domestically and internationally.¹⁹³¹ It is prohibited to promote, publish, canvass, propagate or advertise or cause to be promoted, published, canvassed, propagated or advertised, with the intention to induce or is likely to induce a woman to act as surrogate;¹⁹³² to seek or intend to seek a woman to act as a surrogate¹⁹³³; or to say or

¹⁹²⁵ Elder, AH "Wombs to rent: Examining the jurisdiction of international surrogacy" 2014 16(2) *Oregon Review of International Law* 369.

¹⁹²⁶ The Assisted Reproductive Technology (Regulation) Bill, 2020 to be accessed from <u>https://www.prsindia.org/billtrack/assisted-reproductive-technology-regulation-bill-2020</u> (date accessed 9 February 2021). The Bill was introduced in Lok Sabha on 14 September 2020 and its aim was to provide for the regulation of assisted reproductive technology services in the country.

¹⁹²⁷ 47 of 2021. Accessed from <u>https://www.indiacode.nic.in/handle/123456789/17046?sam_handle=123456789/1362</u> (accessed on 20 January 2023).

¹⁹²⁸ 42 of 2021. The Act was published on 20 December 2021. Accessed from <u>https://www.indiacode.nic.in/handle/123456789/17031?sam_handle=123456789/1362</u> (accessed on 20 January 2023).

¹⁹²⁹ Elder, AH "Wombs to rent: Examining the jurisdiction of international surrogacy" 2014 16(2) *Oregon Review of International Law* 368.

¹⁹³⁰ Sec 3(i), 3(iv) and sec 11 of The Surrogacy (Regulation) Act.

¹⁹³¹ Sec 3(ii) of The Surrogacy (Regulation) Act. "[n]o surrogacy clinic, paediatrician, gynaecologist, embryologist, registered medical practitioner or any person shall conduct, offer, undertake, promote or associate with or avail of commercial surrogacy in any form." Sec 3(v) "[n]o surrogacy clinic, registered medical practitioner, gynaecologist, paediatrician, embryologist or any other person shall promote, publish, canvass, propagate or advertise or cause to be promoted, published, canvassed, propagated or advertised which"- Sec 3(v)(b) "[i]s aimed at promoting a surrogacy clinic for commercial surrogacy or promoting commercial surrogacy in general". Sec 3(v)(e) advertise commercial surrogacy in any form. New laws in India regulates assisted reproduction and surrogacy <u>https://reproductiverights.org/assisted-reproduction-and-surrogacy-in-india/</u> (accessed 26 January 2023).

¹⁹³² Sec 3(v)(a) of The Surrogacy (Regulation) Act.

¹⁹³³ Sec 3(v)(c) of The Surrogacy (Regulation) Act.

suggest that a woman is willing to act as surrogate.¹⁹³⁴ It is important to note that no surrogacy clinic, registered medical practitioner, gynaecologist, paediatrician, embryologist, intending couple or any other person shall perform or cause an abortion to be performed during the gestation period of the surrogacy period unless written consent of the surrogate and approval by the appropriate authority concerned was obtained.¹⁹³⁵

Chapter 3 of the Act regulates surrogacy and surrogacy procedures. Surrogacy is only allowed for purposes as set out in section 4(ii) of the Act, and after satisfying all the conditions as set out in section 4(iii).¹⁹³⁶ No surrogacy or surrogacy procedure is permitted except when gestational surrogacy is required as a result of a medical indication of the intending couple.¹⁹³⁷ A couple of Indian origin or an intending woman who plans to use surrogacy must first apply to the National Assisted Reproductive Technology and Surrogacy Board for a certificate of recommendation.¹⁹³⁸ Only altruistic surrogacy is permitted.¹⁹³⁹ Surrogacy or surrogacy procedures will not be allowed for commercial purposes or for the commercialisation of surrogacy or surrogacy procedures.¹⁹⁴⁰ Surrogacy or surrogacy procedures are furthermore not permitted for producing children for sale, prostitution or for any other form of exploitation.¹⁹⁴¹ However, surrogacy or surrogacy procedures may be permitted for any other condition or disease specified by the regulations.¹⁹⁴²

Furthermore, surrogacy and surrogacy procedures are only allowed in circumstances where the Director or person in-charge of the surrogacy clinic and the person that is qualified to do so are satisfied that the required conditions have been fulfilled.¹⁹⁴³ The intending couple must have a "certificate of essentiality" issued by the authority upon confirmation that the following conditions are met, which are:¹⁹⁴⁴ that a certificate of a

¹⁹³⁴ Sec 3(v)(d) of The Surrogacy (Regulation) Act.

¹⁹³⁵ Sec 3(iv) of The Surrogacy (Regulation) Act.

¹⁹³⁶ Sec 4(i) of The Surrogacy (Regulation) Act.

¹⁹³⁷ Sec 4(ii)(a) of The Surrogacy (Regulation) Act.

¹⁹³⁸ Sec 4(ii)(a) of The Surrogacy (Regulation) Act.

¹⁹³⁹ Sec 4(ii)(b) of The Surrogacy (Regulation) Act.

¹⁹⁴⁰ Sec 4(ii)(c) of The Surrogacy (Regulation) Act.

¹⁹⁴¹ Sec 4(ii)(d) of The Surrogacy (Regulation) Act.

¹⁹⁴² Sec 4(ii)(e) of The Surrogacy (Regulation) Act.

¹⁹⁴³ Sec 4(iii) of The Surrogacy (Regulation) Act.

¹⁹⁴⁴ Sec 4(iii)(a) of The Surrogacy (Regulation) Act.

medical indication in favour of one or both of them or the intending woman necessitating gestational surrogacy as provided by a District Medical Board is issued;¹⁹⁴⁵ an order of a court regarding the parentage and custody of the commissioned child is made;¹⁹⁴⁶ and that insurance cover in favour of the proposed surrogate for a period of thirty-six months covering postpartum delivery complications is provided.¹⁹⁴⁷

The surrogate must have an eligibility certificate issued by the appropriate authority, which will be issued if the following conditions are met:¹⁹⁴⁸ the surrogate must be between the age of 25 to 35 years; she must be an "ever-married woman" having a child of her own on the day of implantation;¹⁹⁴⁹ she shall act as surrogate and will be permitted to undergo surrogacy procedures in terms of the provisions of the Act, provided that the necessary application is submitted to the authority;¹⁹⁵⁰ she is not allowed to use her own gametes;¹⁹⁵¹ she may only act as surrogate once in her lifetime;¹⁹⁵² and she must produce a certificate issued by a registered medical practitioner testifying to her medical and psychological fitness for surrogacy and the surrogacy procedures.¹⁹⁵³

Further to the above, the intending couple's eligibility certificate is supplied separately by the appropriate authority upon fulfilment of the following conditions:¹⁹⁵⁴ they must be married; the woman must be between the ages of 23 to 50 years and the male between the ages 26 to 55 years on the day of certification;¹⁹⁵⁵ and they may not have any surviving child (biologically or through adoption or through earlier surrogacy). The latter provision shall not affect a couple who have a child who is mentally or physically

¹⁹⁴⁵ Sec 4(iii)(a)(I) of The Surrogacy (Regulation) Act.

¹⁹⁴⁶ Sec 4(iii)(a)(II) of The Surrogacy (Regulation) Act. The intending couple or intending woman and the proposed surrogate shall bring an application in this regard. The order shall be the birth affidavit after the commissioned child is born.

 ¹⁹⁴⁷ Sec 4(iii)(a)(III) of The Surrogacy (Regulation) Act. The insurance must be from an insurance company or an agent that is recognised by the Insurance Regulatory and Development Authority.
 ¹⁹⁴⁸ Sec 4(iii)(b) of The Surrogacy (Regulation) Act.

¹⁹⁴⁸ Sec 4(iii)(b) of The Surrogacy (Regulation) Act.

¹⁹⁴⁹ Sec 4(iii)(b)(I) of The Surrogacy (Regulation) Act. The woman can then help in surrogacy be being a donor.

¹⁹⁵⁰ Sec 4(iii)(b)(II) of The Surrogacy (Regulation) Act.

¹⁹⁵¹ Sec 4(iii)(b)(III) of The Surrogacy (Regulation) Act.

¹⁹⁵² Sec 4(iii)(b)(IV) of The Surrogacy (Regulation) Act.

¹⁹⁵³ Sec 4(iii)(b)(V) of The Surrogacy (Regulation) Act.

¹⁹⁵⁴ Sec 4(iii)(c) of The Surrogacy (Regulation) Act.

¹⁹⁵⁵ Sec 4(iii)(c)(I) of The Surrogacy (Regulation) Act.

challenged or suffers from a life-threatening disorder or fatal illness with no permanent cure, the latter to be confirmed and approved by the appropriate authority, supported and confirmed by a medical certificate from a District Medical Board.¹⁹⁵⁶

The provisions in the South African Children's Act are different in that there is no limit on the amount of times that the surrogate can act as surrogate mother. Moreover, the surrogate in South Africa is allowed to donate her gametes for the surrogacy process and commissioning parents are allowed to have other children when they use the surrogacy process for reproduction. There is merit in the argument that a surrogate should not be allowed to donate her gametes for use by the commissioning parents, as the commissioned child will not be genetically linked to her. This may obviate the need for the cooling-off period which the surrogate has in terms of section 298 of the Children's Act. It will also provide more security to the commissioning parents as the potential risk that the surrogate may refuse to hand the commissioned child over after birth will be minimised.

It is prohibited for any person to seek or encourage to conduct any surrogacy or surrogacy procedures on her except for the purposes as set out in section 4(ii) of the Act.¹⁹⁵⁷ The latter provision applies to the surrogate mother. Furthermore, no person is permitted to seek or conduct surrogacy procedures unless¹⁹⁵⁸ he or she has explained all the known side effects, as well as the after effects of the procedures, to the surrogate before she is to undergo the procedures.¹⁹⁶⁰ The surrogate, however, shall have the option to withdraw her consent for the surrogacy before the implantation of the embryo in her womb.¹⁹⁶¹ The intending couple or intending woman is prohibited from abandoning the commissioned child for any reason whatsoever, within or outside India.¹⁹⁶² The commissioned child shall be deemed to be the biological child of the

¹⁹⁵⁶ Sec 4(iii)(c)(II) of The Surrogacy (Regulation) Act.

¹⁹⁵⁷ Sec 5 of The Surrogacy (Regulation) Act.

¹⁹⁵⁸ Sec 6(1) of The Surrogacy (Regulation) Act.

¹⁹⁵⁹ Sec 6(1)(i) of The Surrogacy (Regulation) Act.

¹⁹⁶⁰ Sec 6(1)(ii) of The Surrogacy (Regulation) Act. The consent must be in the prescribed form and in the language that the surrogate understands.

¹⁹⁶¹ Sec 6(2) of The Surrogacy (Regulation) Act.

¹⁹⁶² Sec 7 of The Surrogacy (Regulation) Act. The reason includes, but is not limited to, any genetic defect, birth defect, any other medical condition, the defects developing subsequently, sex of the child or conception of more than one baby and the like.

intending couple or intending woman.¹⁹⁶³ The child shall thus be entitled to all the rights and privileges available to a natural child under any law from time being in force.¹⁹⁶⁴ It is also prohibited for any person, organisation, surrogacy clinic, laboratory or clinical establishment of any kind, to force the surrogate to undergo an abortion at any stage of surrogacy except in prescribed conditions.¹⁹⁶⁵

Like the position in South Africa (and after an order of confirmation of the surrogacy agreement is granted), the commissioned child in India will be deemed to be the child of the commissioning parent(s), or the "intending parents". It is submitted that this explicit recognition of the child's legal status provides more security for the different parties vis-à-vis the position in the UK where a parental order application must first be brought after the birth of the child to transfer parental rights to the commissioning parents. The Indian "eligibility certificate requirement" that applies to each of the parties to the surrogacy agreement may be helpful for a court in South Africa when an application for the confirmation of a surrogacy agreement serves before a High Court.

Chapter 7 of the Surrogacy (Regulation) Act sets out the offences and penalties in terms of the Act. Section 38 prohibits commercial surrogacy: No person, organisation, surrogacy clinic, laboratory or clinical establishment of any kind shall – (a) undertake commercial surrogacy, provide commercial surrogacy or its related component procedures or services in any form or run a racket or an organised group to empanel or select surrogate mothers or use individual brokers or intermediaries to arrange for surrogate mothers and for surrogacy procedures, at such clinics, laboratories or at any other place.¹⁹⁶⁶

Commercial surrogacy may not be advertised in any manner by any means whatsoever, scientific or otherwise.¹⁹⁶⁷ It is further prohibited to abandon, disown or exploit or cause to be abandoned, disowned or exploited in any form, the child(ren) that are born through surrogacy.¹⁹⁶⁸ The surrogate and the commissioned child may

¹⁹⁶³ Sec 8 of The Surrogacy (Regulation) Act.

¹⁹⁶⁴ Sec 8 of The Surrogacy (Regulation) Act.

¹⁹⁶⁵ Sec 10 of The Surrogacy (Regulation) Act.

¹⁹⁶⁶ Sec 38(1)(a) of The Surrogacy (Regulation) Act.

¹⁹⁶⁷ Sec 38(1)(b) of The Surrogacy (Regulation) Act.

¹⁹⁶⁸ Sec 38(1)(c) of The Surrogacy (Regulation) Act.

not be exploited in any manner.¹⁹⁶⁹ The person contravening any provision of the Act, other than section 38, shall be punishable with imprisonment for a period up to five years and with a fine up to ten lakh rupees.¹⁹⁷⁰ In instances where a person wants to follow the surrogacy route, but does not want to adhere to altruistic surrogacy or where the person is conducting surrogacy procedures for commercial purposes, that person shall be punishable with imprisonment for a period of up to five years and with a fine up to five lakh rupees.¹⁹⁷¹ The Act also provides for the maintenance and preservation of records by the surrogacy clinic for a period of twenty-five years.¹⁹⁷²

The purpose of the Assisted Reproductive Technology (Regulation) Act is, among others, the regulation and supervision of ART clinics and the ART banks. The clinics have the duty to make the commissioning couple aware of the rights of a child that is born through the use of ART.¹⁹⁷³ Information about the commissioning couple, woman and donor shall be kept confidential and the information regarding the treatment shall not be disclosed to any person, except to the database of the National Registry; if there is a medical emergency and at the request of the commissioning couple or by an order of court.¹⁹⁷⁴ No treatment or procedure shall be performed by a clinic without the written informed consent of all the parties seeking ART.¹⁹⁷⁵ Any one of the commissioning couple may withdraw his or her consent any time before the embryos or gametes are transferred to the woman's uterus.¹⁹⁷⁶ The commissioning or intending couple shall be deemed the biological parent(s) of a child born through ART, which means that the child shall be entitled to all the rights and privileges that is available to a natural child.¹⁹⁷⁷ A gamete donor shall, however, relinquish all parental rights over the child(ren) which may be born from his or her gametes.¹⁹⁷⁸

¹⁹⁶⁹ Sec 38(1)(d) of The Surrogacy (Regulation) Act.

 ¹⁹⁷⁰ Sec 39(1) of The Surrogacy (Regulation) Act. Ten lakh rupees (1 000 000 Indian rupees) are equivalent to 230,363.00 South African Rand (ZAR) and 12,115.26 United State Dollars (USD).
 ¹⁹⁷¹ Sec 40 of The Surrogacy (Regulation) Act. A subsequent offence will be punichable with

¹⁹⁷¹ Sec 40 of The Surrogacy (Regulation) Act. A subsequent offence will be punishable with imprisonment of up to ten years and a fine with up to ten lakh rupees.

¹⁹⁷² Sec 46 determines that the clinic shall maintain all records, charts, forms, reports, consent letters, agreements and all other documents under the Act.

¹⁹⁷³ Sec 21 of The Assisted Reproductive Technology (Regulation) Act.

¹⁹⁷⁴ Sec 21(e) of The Assisted Reproductive Technology (Regulation) Act.

¹⁹⁷⁵ Sec 22(a) of The Assisted Reproductive Technology (Regulation) Act.

¹⁹⁷⁶ Sec 22(4) of The Assisted Reproductive Technology (Regulation) Act.

¹⁹⁷⁷ Sec 31 of The Assisted Reproductive Technology (Regulation) Act. This provision is similar to sec 8 of The Surrogacy (Regulation) Act.

¹⁹⁷⁸ Sec 31(2) of The Assisted Reproductive Technology (Regulation) Act.

Important legal questions of parentage and citizenship had to be answered by the courts in the below judgments.¹⁹⁷⁹ It is evident that India's new regulatory framework will answer these legal questions.

4.2 Baby Manji Yamada v. Union of India¹⁹⁸⁰

In this matter the paternal grandmother filed a petition for custody in respect of a baby that was born through surrogacy.¹⁹⁸¹ The petition was filed under article 32 of the Constitution of India, 1950.¹⁹⁸² A surrogacy agreement was concluded between the biological parents (from Japan) and a surrogate (in India).¹⁹⁸³ The birth certificate indicated the biological father as the parent.¹⁹⁸⁴ The biological parents separated and the biological father had to return to Japan as a result of his visa expiring.¹⁹⁸⁵ This matter is important as the lack of a legal framework for surrogacy was pointed out by the third respondent, an NGO.¹⁹⁸⁶ This was at a time that commercial surrogacy was still allowed in India and was a booming industry. The NGO argued that there was no law in India that governed surrogacy and there were many irregularities being committed as a result of the regulatory gaps.¹⁹⁸⁷ The NGO further argued that the Union of India should enforce stringent laws that regulate surrogacy in India.¹⁹⁸⁸ The question that arose during the hearing of the matter was who can be deemed to be the mother of the child—was it the gamete donor or the surrogate?¹⁹⁸⁹ The identity document that was issued only indicated the father's name as the father and no one was indicated to be the mother on the document.¹⁹⁹⁰ The court, as a result of the

 ¹⁹⁷⁹ Gola, S "One Step forward or one step back? Autonomy, agency and surrogates in the Indian Surrogacy (Regulation) Bill 2019" 2021 *International Journal of Law in Context* 17(1) 58.
 ¹⁹⁸⁰ 2008 (42) SCC 548

¹⁹⁸⁰ 2008 (13) SCC 518.

¹⁹⁸¹ Baby Manji Yamada v. Union of India at par 2-3.

¹⁹⁸² Baby Manji Yamada v. Union of India at par 2.

¹⁹⁸³ Baby Manji Yamada v. Union of India at par 2.

¹⁹⁸⁴ Baby Manji Yamada v. Union of India at par 2.

¹⁹⁸⁵ Baby Manji Yamada v. Union of India at par 2.

¹⁹⁸⁶ Baby Manji Yamada v. Union of India at par 3. The NGO filed a Division Bench Habeas Corpus Writ Petition before the High Court of Rajasthan, Jaipur Bench. Singh, J "Commercial Surrogacy: The baby Manji Yamada case" <u>https://www.legallore.info/post/commercial-surrogacy-the-baby-manji-yamada-case</u> (accessed 1 June 2023).

¹⁹⁸⁷ Baby Manii Yamada v. Union of India at par 3.

¹⁹⁸⁸ Baby Manji Yamada v. Union of India at par 3.

¹⁹⁸⁹ Jan Balaz v. Anand Municipality and others at par 14.

¹⁹⁹⁰ Jan Balaz v. Anand Municipality and others at par 14.

NGO's petition, issued directions on the custody or planning of Baby Manji.¹⁹⁹¹ It was as a result of this petition that Baby Manji's paternal grandmother lodged a writ challenge in the Supreme Court wherein she was contesting the High Court's order.¹⁹⁹² The court observed that the Commission for the Protection of Child Rights Act of 2005 was formed to protect children's rights and further to expedite the prosecution of crimes against children.¹⁹⁹³ The Commission thus had to make a decision in this case should a decision be deemed necessary.¹⁹⁹⁴ The court further held that there was no complaint made regarding Baby Manji and the order requiring her presence before the court was held to be invalid.¹⁹⁹⁵ The court furthermore ordered that a passport be granted by the Regional Passport Authority to enable Baby Manji's travel out of India.¹⁹⁹⁶

4.3 Jan Balaz vs. Anand Municipality and others¹⁹⁹⁷

The question in this matter was whether a child, born from a surrogacy arrangement where the surrogate was an Indian national and the biological father a foreign national, would be eligible for Indian citizenship by birth.¹⁹⁹⁸ The father was registered as the father on the birth certificates of the two children born from the surrogacy agreement and the surrogate was registered as the mother of the children.¹⁹⁹⁹ In terms of section 3 of the Citizenship Act of 1955, children born from a surrogate agreement could not

¹⁹⁹¹ Singh, J "Commercial Surrogacy: The baby Manji Yamada case" <u>https://www.legallore.info/post/commercial-surrogacy-the-baby-manji-yamada-case</u> (accessed on 1 June 2023).

¹⁹⁹² Singh, J [']Commercial Surrogacy: The baby Manji Yamada case" <u>https://www.legallore.info/post/commercial-surrogacy-the-baby-manji-yamada-case</u> (accessed on 1 June 2023).

¹⁹⁹³ Singh, J [']Commercial Surrogacy: The baby Manji Yamada case" <u>https://www.legallore.info/post/commercial-surrogacy-the-baby-manji-yamada-case</u> (accessed 1 June 2023).

¹⁹⁹⁴ Singh, J "Commercial Surrogacy: The baby Manji Yamada case" <u>https://www.legallore.info/post/commercial-surrogacy-the-baby-manji-yamada-case</u> (accessed 1 June 2023).

¹⁹⁹⁵ Ratchaya, S.K "Facts of the case: Baby Manji Yamada v. Union of India" <u>https://liberttem.in/blog/surrogacy-in-india-baby-manji-yamada-case-and-surrogacy-regulation-bill</u> (accessed on 1 June 2023).

¹⁹⁹⁶ Ratchaya, S.K "Facts of the case: Baby Manji Yamada v. Union of India" <u>https://liberttem.in/blog/surrogacy-in-india-baby-manji-yamada-case-and-surrogacy-regulation-bill</u> (accessed on 1 June 2023).

¹⁹⁹⁷ Gujarat H.C. 2009.

¹⁹⁹⁸ Jan Balaz v. Anand Municipality and others at par 1.

¹⁹⁹⁹ Jan Balaz v. Anand Municipality and others at par 3.

be deemed as Indian citizens.²⁰⁰⁰ Here, the biological parents were not Indian citizens which meant that the children were also not Indian citizens.²⁰⁰¹ It was pointed out that the central government of India was yet to legalise surrogacy and therefore, children born from a surrogacy agreement could not be deemed Indian citizens although they were born in India.²⁰⁰² The court pointed out that it is primarily concerned with the rights of the biological parents, the surrogate and the gamete donor.²⁰⁰³ The legitimacy of the children was an issue that the court had to deal with.²⁰⁰⁴ The public pressure for comprehensive legislation to define, for instance, the rights of a child born through surrogacy, was gaining momentum.²⁰⁰⁵ The court found that in the absence of legislation to the contrary, the court was more inclined to recognise the gestational surrogate, who gave birth to the children, as their natural mother.²⁰⁰⁶ Importantly, a gamete donor will not become a natural mother of the resultant child.²⁰⁰⁷ Further, the court stated that the only conclusion that was possible is that the gestational mother, who had blood relations with the child, was more deserving to be called as the natural mother of the child as she had carried the embryo for ten months and she had nurtured the babies through the umbilical cord.²⁰⁰⁸ The children were entitled to Indian citizenship although they were illegitimate (the biological father was not married to the surrogate).²⁰⁰⁹ The important consideration was that they were born from a Indian national.2010

5. CONCLUSION

A comparison of the regulation of surrogacy in UK to the position in South Africa reveals that South Africa's regulation of surrogacy through the provisions of the Children's Act constitutes a better structured regulatory vehicle. The UK regulates surrogacy through three different statutes: the Surrogacy Arrangement Act, the HFEA

Jan Balaz v. Anand Municipality and others at par 6.

Jan Balaz v. Anand Municipality and others at par 6.

Jan Balaz v. Anand Municipality and others at par 6.

Jan Balaz v. Anand Municipality and others at par 9.

Jan Balaz v. Anand Municipality and others at par 9.

Jan Balaz v. Anand Municipality and others at par 14.

Jan Balaz v. Anand Municipality and others at par 16.

Jan Balaz v. Anand Municipality and others at par 16.

Jan Balaz v. Anand Municipality and others at par 16.

Jan Balaz v. Anand Municipality and others at par 17.

Jan Balaz v. Anand Municipality and others at par 17.

1990 and the HFEA 2008. The amendments that the HFEA 2008 brought to the regulation of surrogacy must be read into the two older Acts which complicates the clear understanding of the provisions meant to be relevant to surrogacy. Surrogacy agreements are unenforceable in the UK and the definition of a surrogate agreement does not include an agreement which intends to result in the surrogate handing the commissioned child over to the commissioning parent(s). This situation, compared to what the reason for using a surrogate is, obscures the regulation of surrogacy. Further, the commissioning parent(s) must first obtain a parental order before he/she/they can be regarded as the legal parents of the commissioned child and this order can only be obtained six weeks after the commissioning parent(s) in that a court order must first be obtained before the surrogate mother may be artificially fertilised. The commissioning parent(s) will have clarity regarding the legal status of the commissioned child before the child is born.

A legal comparison between the legal frameworks governing surrogacy in South Africa and Canada is complicated by the fact that surrogacy in Canada is regulated under the federal Assisted Human Reproduction Act, yet certain provincial laws also play a role regarding who is legally recognised as the parents of a child. This regulatory picture is further complemented by four different sets of Regulations that apply to all provinces in Canada. Chapter 19 of the Children's Act in South Africa, however, is one comprehensive chapter in an Act that attempts to regulate surrogacy in more detail. It is interesting that the locus of the surrogacy regulation in South Africa falls within the ambit of the Children's Act and not the NHA. Because surrogacy straddles issues relating to both family law and health law, one would expect the legislation to fall into either private-law legislation or health legislation. The clinical aspects regarding artificial fertilisation in the context of surrogacy are governed by the National Health Act and Regulations in South Africa.

Gleaning from the Canadian example, the list of expenses that is set out in regulation 4 of Reimbursement Related to Assisted Human Reproduction Regulations may be helpful to courts in South Africa. This chapter proposes that when regulations to chapter 19 of the Children's Act are promulgated, a list of allowed expenses could be included that resembles the Canadian example.

In contrast to the position in India, South Africa allows partial surrogacy. This chapter proposes that the South African legislator should consider limiting surrogacy in Chapter 19 to allow full surrogacy only. By prohibiting the surrogate from donating her gametes, the commissioning parent(s) may be better protected and the risk that a surrogate may refuse to hand over the commissioned child after birth, may be circumvented. Furthermore, the Indian provisions that aim to secure the position of the commissioned child after birth by prohibiting any effort by the commissioning parents to abandon the commissioned child, should be commended. Finally, India's recent ban on commercial surrogacy should also be welcomed.

This chapter has initially sought to find answers to some of the lacunae in chapter 19 of the Children's Act. Apart from the few instructive comparative examples mentioned above, one's impression of the comparative legal overview is that chapter 19 of the Children's Act constitutes a solid and comprehensive framework compared to its foreign counterparts. It is submitted that the remaining gaps that this thesis has identified, may hopefully be closed in the near future.

CHAPTER 7

CONCLUSION AND RECOMMENDATIONS

1. INTRODUCTION

The focus of this thesis was to consider the constitutionality of assisted reproduction and surrogate motherhood and related medico-legal issues. Medico-legal issues in the context of surrogate motherhood are primarily concerned with the artificial fertilisation of the surrogate mother after confirmation of the surrogate agreement by the court.

Chapter 19 of the Children's Act provides some guidance on the conditions required for the legality of the surrogacy agreement, including the legal consequences of both lawful agreements (that comply with the requirements of chapter 19) and unlawful agreements (where the requirements are not met). Legal consequences include the impact of lawful and unlawful surrogacy arrangements on the status of the commissioning couple, the surrogate mother and the commissioned child.

Surrogacy and artificial fertilisation are examples of noncoital reproduction. The right to reproduction includes the right to choose whether or not to have a child and is part and parcel of personal autonomy. The Constitution of South Africa protects the right to bodily and psychological integrity and to make decisions regarding reproduction. Even though the global population has grown steadily over the past few decades, infertility has become increasingly common due to social and environmental changes. Infertility has a severe psychological and physical effect on infertile persons, not to mention the societal pressure they may face to conceive a child of their own. As this thesis has shown, adoption does not provide a satisfactory alternative because of the fundamental differences between the two arrangements.

2. A NEED FOR LEGAL REFORM

It is trite that technology develops faster than the law, leading to legal gaps in regulatory frameworks. The South African regulatory framework regarding surrogate motherhood and artificial fertilisation is a clear example of this, demonstrated in this thesis by an

overview of the legal-historical development of South African law regarding surrogate motherhood and artificial fertilisation and by examining some of the legislative amendments made over decades to close some of these legal ambiguities. Canvassing the legal-historical development is necessary for an understanding of the evolution of some of the key legal principles and rules that found their way into chapter 19 of the Children's Act.

Chapter 19 of the Children's Act provides the current legal framework governing surrogate motherhood. Chapter 2 of the thesis traces the development of laws preceding the promulgation of chapter 19 of the Children's Act, highlighting some of the thorny legal dilemmas that arose from, among others, the now repealed Children's Status Act and Child Care Act concerning the legal status of the commissioned child. The Children's Act has introduced a complete change to the common law position that a child born to a married woman is considered her legal child and the accompanying legal presumption that the husband of such child is the legal father.

3. REGULATORY CLARITY WITH SOME CONSTITUTIONAL CAVEATS

Chapter 3 of the thesis examines the current legal regulatory framework relating to the Children's Act together with relevant provisions in the NHA and Regulations relating to artificial fertilisation. The Children's Act seeks to provide regulatory clarity regarding the rights of all the parties to the surrogacy agreement. Clarity is required because artificial fertilisation can cause a commissioned child to have as many as six parents in some instances. Chapter 19 attempts to provide certainty to all parties involved in a surrogacy agreement regarding their responsibilities and rights towards one another and in respect of the commissioned child. This certainty aims to protect the rights and interests of the commissioned child should there be a dispute after his or her birth.

One of the rights that children generally may lay claim to concerns the right of a child to know his or her biological or genetic origins, which derive from the South African Constitution, the Promotion of Access to Information Act and the Children's Act. For example, children older than sixteen years may request access to hospital files and if younger, a person with parental rights may make such request on the child's behalf. The type of information that may be accessed by a child determines when it may be requested. Whilst the Children's Act distinguishes between medical information and "any other information concerning the child's genetic parents" (Section 41), a child must be eighteen years old before he or she may be lawfully provided information other than medical information regarding his or her genetic parents. A child is not allowed to receive information regarding the identity of the gamete donors or of the surrogate.

Chapter 3 of the thesis refers extensively to relevant High Court judgments that adjudicated several provisions of the Children's Act relating to surrogate motherhood. In the matter of Ex Parte MS and another; In Re: Confirmation of surrogate mother agreement, the importance of first having the surrogacy agreement confirmed before the surrogate is artificially fertilised, came to the fore. The contractual parties have to understand the legal consequences in the event that the court refuses to confirm the surrogacy agreement post-fertilisation. The best interests of the commissioned child, who is the reason for the surrogate agreement in the first instance, must always remain the priority. Legally, confirmation of the surrogate agreement should precede the fertilisation of the surrogate mother, although chapter 3 has discussed instances where the court confirmed an agreement after fertilisation of the agreement, yet before the commissioned child's birth. Where a surrogate is artificially fertilised prior to the surrogacy agreement being confirmed by the court, the risk arises that the court may refuse to confirm the agreement on the ground that the agreement is invalid. The legal consequences of such invalid agreement will be that the commissioned child will be deemed to be the child of the surrogate for all purposes and not the child of the commissioning parent(s). In this instance, the position will revert to the common law position prior to the promulgation of the Children's Act where the commissioning parent(s) would need to adopt the commissioned child. It is therefore imperative that the parties comply with the requirements as set out in chapter 19 of the Act.

The requirements of a valid surrogacy agreement are set out in sections 292 – 296 of the Children's Act. The parties to such an agreement must comply with the said requirements as non-compliance will unfortunately result in the agreement to be invalid and unenforceable between the parties. It is to be noted that the domicile requirement in terms of section 292(1)(d) of the Act does not prevent a pregnant surrogate from leaving South Africa. This is a risk which the commissioning parents must be made

aware of and accept as there will be no form of protection for them should the surrogate leave the country after conclusion of the agreement and the fertilisation already carried out.

One specific (and controversial requirement) in the Act is section 294, which requires that a genetic link exists between the commissioning parent or parents and the commissioned child. The use of only donor gametes in the surrogacy process is not permitted. This study argues that although there may be valid reasons for requiring a genetic link in a surrogacy agreement, it cannot be seen as reasonable and fair to exclude a person, who is both "conception- and pregnancy infertile", from having access to the surrogacy process. The rationale that such parents should rely on adoption to have a child of their own, is untenable, for various reasons relating to the inherent differences to the adoption and surrogacy processes. This thesis submits that this requirement should be removed from chapter 19 of the Children's Act and that a court should exercise its discretion and that each case be decided on a case-by-case approach.

Section 295(d) is an important provision in that it provides for certain requirements regarding the commissioned child's contact, care, upbringing and general welfare, to be included in a surrogacy agreement, including that the best interests of the commissioned child must be considered at all times. It is the court's responsibility to have regard to both the personal, social and financial circumstances and the family situations of the parties to the surrogacy agreement, guided by the best interests of the commissioned child in relation to each of the instances.

With the Constitution at the apex of all laws in South Africa, the thesis also explores the relevant rights of parties to the surrogacy arrangement. Chapter 4 of the thesis identifies a first potentially unconstitutional provision in Section 294 of the Children's Act, which unjustifiably, unfairly and unreasonably draws a distinction between conception and pregnancy infertile persons by denying a person who is biologically unable to offer a genetic link to the commissioned child, access to surrogacy. This prohibition denies such person his or her right guaranteed under section 12(2)(a) of the Constitution. Other fundamental rights relevant to parties to a surrogacy agreement, discussed in chapter 4 are section 10 of the Constitution that protects a

person's right to human dignity, which is potentially violated by section 294 of the Children's Act that treats persons differently in an unjustifiable manner because of inherent personal traits attached to specific infertile persons. Section 14 determines that every person has the right to privacy. The right to equality, protected in section 9 of the Constitution, is potentially infringed by Section 294 of the Children's Act that requires commissioning parent(s) to have a genetic link with the commissioned child and those infertile persons that are unable to provide their own gametes.

The right of access to health care services, which includes the right to reproductive health care, is entrenched in section 27(1) of the Constitution. The right of a person to reproduce non-coitally and their right to reproductive freedom, the latter recognised in section 12 of the Constitution, is linked to section 27(1) of the Constitution in that "health care services" should include services to ameliorate infertility, which is a medical condition. Because fertility treatment (for example, artificial fertilisation) is not a health priority in the health care system, the realisation of reproductive health care and reproductive freedom of persons still has a long way to go.

Chapter 4 of the thesis has placed emphasis on the best interests of the child principle, which is also entrenched in Section 28 of the Constitution (providing an omnibus protecting the rights of children). Courts that have been approached for the confirmation of surrogate agreements have routinely expressed the importance of being provided with sufficient information to enable it to decide in this regard, as it needs to assess the impact of all of the facts on the commissioned child. Based on this assessment, the court must envisage the future of the commissioned child by ensuring that adequate provision is made in the surrogacy agreement for the child's psychological and physical well-being and his or her protection in the long run.

The constitutionality of certain provisions in chapter 19 of the Children's Act came under the spotlight in recent judgments. As alluded to above, section 294 was challenged in the case of *AB* and another *v* Minister of Social Development on the ground that it unjustifiably infringes upon the constitutional rights to equality, dignity, reproductive autonomy, privacy and the right of access to health care services. The minority found that section 294 is unconstitutional and invalid because of its effect on persons who are both conception- and pregnancy infertile. The majority, on the other

hand, concluded that no limitation on any of the applicant's rights could be demonstrated and that section 294 is thus not unconstitutional.

In the case of *EJ* and others v Haupt NO, which dealt with the situation of children born via artificial insemination to a lesbian couple in a permanent life partnership, the court ruled that a partner who was not the biological parent was to be regarded as a natural parent and guardian and that the children were legitimate in law. Both spouses automatically acquire rights and responsibilities in respect of the child that was born from artificial fertilisation.

The focus of the case, *KB* and another *v* Minister of Social Development, was again section 294, but this time on the ground that the provision does not provide for a genetic link between siblings, but only requires a genetic link with one or both of the parents. The court held that section 294 has nothing to do with the right of a minor child to have a sibling with the same genetic link and reiterated that the purpose of the section is to provide clarity regarding the origin of a child. The court pointed out that clarity regarding one's origin is important to the self-identity and self-respect of the child.

Section 40 of the Children's Act again came under fire in *VVJ and another v Minister* of Social Development and Another. The court found that the Children's Act remains conservatively lagging in terms of artificial fertilisation and the acknowledgement of partners as parents. Thus, section 40 was found to be unconstitutional in that the provision unfairly discriminates based on marital status in terms of its treatment of children born in or out of wedlock. The court declared section 40 inconsistent with the Constitution to the extent that the said section does not make provision for a "permanent life partner" after the word "spouse" and "husband" wherever such words appear in the section. Not only the Children's Act was in the court's purview regarding constitutionality issues with surrogacy. In the recent case of *Surrogacy Advisory Group v Minister of Health*, certain provisions of the 2012 Regulations Relating to the Artificial Fertilisation of Persons (regulations 7(j)(ii), 13 and 19) and the 2012 Regulations Relating to the Use of Human Biological Material (regulation 6) were declared unconstitutional, on a range of human rights infringements.

The conclusion drawn from the constitutional challenges to the Children's Act is that

despite the Act's laudable intentions, judicial scrutiny has steadily been closing some of the gaps. The Act is still far from perfect but should be considered a "work in progress". One aspect still requiring attention is the Act's failure to accommodate the different and diverse forms of families that exist in present times.

A closer understanding of the nuanced differences between the types of assisted reproductive technologies is necessary, as different legal, ethical and social issues attach to each of these. Chapter 5 of the thesis explores some of these diverse views pertaining to understandings of what a "family" should constitute, including different religious views on ART and surrogate motherhood and the morality of using donor gametes in establishing a family. Surrogacy and artificial fertilisation challenge traditional parent-child relationships because a commissioned child may technically have multiple parents. Gestational surrogacy also results in the commissioned child having a biological bond with two women. The thesis advances the argument that because the WHO regards infertility as a disease, the use of advanced medical technology that aims to treat infertility with the use of donor gametes should be regarded as a form of assistance. Chapter 5 also explores the exciting consequences of epigenetics in the context of surrogacy. Science has convincingly demonstrated the impact of the genetic make-up of the surrogate mother on the health and development of the commissioned child, irrespective of whether she is an ovum donor or not. Contrary to the traditional view that the gamete donors are the only biological parents of the commissioned child where a surrogate is only used for gestation, epigenetic research shows that the surrogate should rightly be considered an additional biological parent of the child that she carries. With this scientific discovery still very novel, this study argues that it should be considered an important factor by parties concluding a surrogate agreement, as the epigenetics of the surrogate will play a determining factor on the developing child during pregnancy and after birth. Not only should the effect of epigenetics be considered in a surrogate agreement, but the Children's Act and proposed regulations to the Children's Act should envisage any claims that may follow from this biological determinant. This study proposed that because of the unforeseen health risks that may arise because of the role of epigenetics, a complete genetic analysis of the proposed surrogate should be undertaken before selection of a suitable surrogate. This would unfortunately be a costly exercise and not a possibility for all commissioning parent(s).

4. LEGAL COMPARATIVE LESSONS

In the final instance, the thesis adopts a comparative legal approach by analysing the legal frameworks regarding surrogate motherhood in three foreign, yet comparable jurisdictions, namely the United Kingdom, Canada and India. The purpose of this comparison was to explore best practices in these legal frameworks that may assist in closing some of the legal gaps that exist in the Children's Act.

This legal comparison reveals interesting findings. Turning to the United Kingdom's framework, the thesis concludes that chapter 19 of the Children's Act, which comprehensively regulates surrogacy in South Africa, provides more legal security to the parties to a surrogacy agreement. Under UK law, a surrogate and her husband or partner (if any) will be deemed the legal parents of the commissioned child that is born because of a surrogacy agreement. The commissioning parent(s) would need to apply for a parental order in terms of the HFEA 2008, after the birth of the child, for the extinguishing of the parental status of the surrogate and her husband or partner.

In Canada, similar to the position in the UK, the surrogate is assumed to be the legal parent of the commissioned child. The surrogate is only allowed to agree to the transfer of her parental rights in respect of the commissioned child after the birth of the child. Any reimbursement for expenses in terms of the surrogate agreement may only be paid in accordance with the specific requirements set out in the Reimbursement Related to Assisted Human Reproduction Regulations.

India only allows a surrogate to act as a surrogate mother once in her life and she is not allowed to donate her own gametes, thus only allowing gestational surrogacy and full surrogacy, but not partial surrogacy. As is the case in South Africa, the commissioned child will be deemed to be the child of the commissioning parent(s) after the surrogacy confirmation order.

The jurisdictions of UK and Canada retain the common law position regarding the legal status of the commissioned child, requiring an additional legal step to change the child's status *after* the child's birth via a parental order (UK) or transfer of parental rights (Canada). In South Africa and India, a confirmation of a surrogate agreement precedes

the artificial fertilisation of the surrogate, as well as the birth of the commissioned child. It is a requirement in India that the confirmation order (regarding the parentage and custody of the commissioned child) forms part of the parties' certificate of essentiality.

5. **RECOMMENDATIONS**

This study proposes the following recommendations to address the identified gaps and inconsistencies in the current legal framework governing surrogate motherhood and artificial fertilisation:

- (1) First, the Children's Act lacks regulations to provide more granularity to the Act. Specific sections of the Children's Act require more detail to improve compliance by parties to a surrogacy agreement. Although the judgments discussed in the thesis provide some guidance on gaps and inconsistencies, these have done so on a case-by-case basis. For example, regulations to the Act could include more detail on the type of costs that may lawfully be provided to a surrogate mother in terms of section 301 of the Children's Act, based on the Canadian model. For example, the proposed regulations could make it a requirement that the surrogate provides the commissioning parent(s) with an affidavit that sets out the exact expenses together with a receipt for each expense. If the surrogate claims a loss of income, she must provide an affidavit with proof of loss of income together with a medical certificate indicating her time off from work.
- (2) Second, regulations are necessary regarding any supporting documentation that the parties to a surrogacy agreement must submit to the court for confirmation of such agreement.
- (3) Third, regulations are needed to provide some form of protection to the commissioning parent(s) if the surrogate decides to leave South Africa whilst pregnant with the commissioned child. It is recommended that the surrogate at least be required to discuss her intention of leaving South Africa with the commissioning parent(s) prior to her leaving.

- (4) Fourth, section 293 of the Children's Act requires the consent of both the commissioning mother and the surrogate's husband, wife or partner. Regulations should also define what a "permanent life partnership" entails, to enable the parties to a surrogacy agreement to provide the consent as required in terms of this section.
- (5) Fifth, in respect of the genetic link requirement set out in section 294 of the Act, it is recommended that either the genetic link requirement regarding the commissioning parent(s) be removed in the Children's Act, or, alternatively, that the genetic link requirement be considered for commissioning parent(s) who desire to have more children via surrogacy that are genetically linked to the existing commissioned child(ren). Proposed regulations could provide guidance on how this requirement in the Act be applied in surrogacy confirmation applications on a case-by-case basis. This is of particular importance for commissioning parent(s) who are pregnancy and conception infertile.
- (6) Sixth, because section 295(a) requires that the commissioning parent(s) should have a medical condition that makes her unable to give birth to a child, which condition must be permanent and irreversible, regulations should provide detail on this to assist the court on what the possible causes of this inability to give birth could be. The regulations should be clear whether this inability must be because of a physical condition or any other kind of risk that has the same consequences. A closed list of conditions is not advisable, as this may complicate a confirmation application. It is recommended that this section also be dealt with on case-by-case approach as no two applications will be same. It is also important to provide guidance as to the required evidence that the commissioning parent(s) must provide that will suffice as proof of the inability to give birth.
- (7) Seventh, this study recommends that the South African legal framework only allow gestational and full surrogacy. Allowing gestational surrogacy will provide more security for the commissioning parent(s) in that the surrogate will not be genetically linked to the commissioned child which means that the cooling-off period will fall away and she will not be able to refuse to hand over the commissioned child after his or her birth.

(8) Eighth, requiring an eligibility certificate from each party to the surrogacy agreement may be helpful to the High Court when dealing with a surrogacy agreement confirmation application. (The eligibility certificate may arguably not be necessary in South Africa currently, considering the required medical reports that have to be included in the confirmation of a surrogate motherhood application. However, such certificate may strengthen the application process in South Africa. The relevant national department and custodian of the Children's Act, the Department of Social Development, may consider appointing a specific body to issue such a certificate.)

The confirmation of surrogate agreements by the courts since the enactment of chapter 19 of the Children's Act has been instructive in clarifying the information that should accompany a confirmation application to the court. This include the following:

- (i) detailed information regarding who the commissioning parents are;
- (ii) their financial position and support systems;
- (iii) their living conditions;
- (iv) how they will take care of the commissioned child;
- (v) an expert assessment report from a social worker;
- (vi) an expert report regarding the suitability of the surrogate;
- (vii) a police clearance certificate of the commissioning parent(s);
- (viii) regarding the permitted payments, a detailed list of surrogacy expenses must be provided setting out expenses with sufficient specificity; and
- (ix) if an agency was involved in the surrogacy process, the parties must explain the relationship between the agency and the commissioning parent(s) and the manner in which the agency operates;
- (x) furthermore, detailed information must be presented to show that the commissioning parent(s) will be providing the commissioned child with a stable home environment, as well as how the commissioning parent(s) will function as a family unit and whether they a living together. Should they not live together, they need to explain why this will not impact the commissioning parent(s) and the commissioned child to live as a family unit and why it will be in the best interests of the child; and finally,

(xi) medical reports regarding the proposed surrogate's mental and physical health must be provided.

With reference to international surrogacy, it is worth noting that there is no public or private law instrument that regulates international surrogacy and parties to a surrogacy agreement must be wary of the possibility of the commissioned child ending up being stateless and parentless after his or her birth because of the different legal consequences of each jurisdiction, particularly where the commissioned child is born following a surrogacy agreement in a country where surrogacy agreements are unenforceable or surrogacy is banned.

This thesis has also demonstrated that South Africa is not that far behind in regulating surrogacy and artificial fertilisation compared to other jurisdictions, especially considering that South Africa is a lower- and middle- income country, compared to the UK and Canada as high-income countries. Despite the few regulatory and constitutional gaps that remain, there is no doubt that as soon as these lacunae are closed, South Africa will have an exemplary model for the regulation of surrogate motherhood and artificial fertilisation.

BIBLIOGRAPHY

1. BOOKS AND CONTRIBUTIONS IN BOOKS

Bosman-Sadie and Corrie *Practical approach to the new Children's Act* Bosman-Sadie, H and Corrie, L *Practical approach to the new Children's Act* (Van Schaik 2012)

Carstens and Pearmain Foundational Principles Carstens, P and Pearmain, D Foundational Principles of South African Medical Law (LexisNexis 2007)

Dada and McQuoid-Mason Introduction to Medico-Legal Practice Dada, MA and McQuoid-Mason, DJ Introduction to Medico-Legal Practice (Butterworths 2001)

Dhai and McQouid-Mason *Bioethics, Human Rights and Health Law* Dhai, A and McQuoid-Mason D *Bioethics, Human Rights and Health Law, Principles and Practice* (2 ed Juta & Co 2020)

Kennedy and Grubb *Medical Law* Kennedy, I and Grubb, A *Medical Law* (2 ed Butterworths 1994)

Mason, McCall Smith and Laurie *Law and Medical Ethics* Mason JK, McCall Smith RA and Laurie GT *Law and Medical Ethics* 6th ed (Butterworths 2002)

Pretorius Surrogate motherhood Pretorius, D Surrogate motherhood: a worldwide view of the issues (Charles C Thomas, 1994)

Slabbert and Roodt "South Africa"

Slabbert M and Roodt C 'South Africa' in Trimmings K and Beaumont P International Surrogacy Arrangements Legal Regulation at the International Level (Hart Publishing 2013) 325-345

Sloth-Nielsen "Surrogacy in South Africa"

Sloth-Nielsen, J "Surrogacy in South Africa" in Scherpe J, Fenton-Glynn C and Kaan T *Eastern and Western perspectives on surrogacy* (Intersentia 2019) 185-202

South African Research Council: *Guidelines on Ethics for Medical Researc*h 2nd Book (Reproductive Biology and Genetic Research) 2002 2.

Strauss Doctor, Patient and the Law Strauss SA Doctor, Patient and the Law (3 ed JL Van Schaik 1991)

Strauss and Maré *Mediese Reg* Strauss, SA and Maré, MC *Mediese Reg* (Study Guide, LCR404-U, University of South Africa)

2. JOURNAL ARTICLES

Biko J and Nene Z "Ethics aspects of third-party reproduction" 2017 *Obstetrics* & *Gynaecology Forum* 3 12-16.

Baase 2019 PELJ

Baase M "The ratification of inadequate surrogate motherhood agreements and the best interest of the child" 2019 *Potchefstroom Electronic Law Journal* 22 1-26

Boniface 2017 *TSAR* Boniface, A.E "The genetic link requirement for surrogacy: A family cannot be defined by genetic lineage" 2017 *Journal of South African Law* 2017(1) 190-206

Bonthuys and Broeders 2013 SALJ

Bonthuys, E and Broeders, N "Guidelines for the approval of surrogate motherhood agreements: Ex parte WH" 2013 *South African Law Journal* 130(3) 485-495

Bracken, L "Surrogacy and the Genetic Link" 2020 *Child and Family Law Quarterly* 32(3) 303-319

Brown, A "Trans parenthood and the meaning of 'mother', 'father' and 'parent' – R (McConnell and YY) v Registrar General for England and Wales [2020] EWCA Civ 559" 2021 *Medical Law Review* 29(1) 157-171

Canner, S "Navigating surrogacy law in the non-united states: Why all states should adopt a uniform surrogacy statute" 2019 *Journal of Civil Rights and Economic Development* 33(2)113-140

Carsley, S "Surrogacy in Canada: Lawyers' experiences and practices" 2022 *Canadian Journal of Women and the Law* 34(1) 41-81

Carsley, S "Regulating reimbursements for surrogate mothers" 2021 *Alberta Law Review* 58(4) 811-836

Cascão, R "The challenges of international commercial surrogacy: from paternalism towards realism?" 2016 *Medicine and Law* 35(2) 151-164

Clarke 2000 Stell LR

Clarke, B "A "golden thread"? Some aspects of the application of the standard of the best interest of the child in South African family law" 2000 *Stellenbosch Law Review* 2000(11) 3-19

Christiansen, K "Who is the mother? Negotiating identity in an Irish surrogacy case" 2015 *Medicine, Health Care and Philosophy* 18(3) 317-327

Deonandan, R "Thoughts on the ethics of gestational surrogacy: perspectives from religions, Western liberalism, and comparisons with adoption" 2020 *Journal of Assist Reproduction and Genetics* 37 269–279

Elder, AH "Wombs to rent: Examining the jurisdiction of international surrogacy" 2014 16(2) *Oregon Review of International Law* 347-374

Fenton-Glynn, C "The regulation and recognition of surrogacy under English Law: an overview of the case-law" 2015 *Child and Family Law Quarterly* 27(1) 83-96

Fischbach, R.L. and Loike, J.D "Maternal-fetal cell transfer in surrogacy: Ties that bind" 2014 *American Journal of Bioethics* 14(5) 35-36

Florescu, S and Sloth-Nielsen, J "Visions on surrogacy – From North to South: The approach of the Netherlands and South Africa to the issue of surrogacy and the child's right to know his origin" 2017 *International Survey of Family Law* 2017 239-258

Forman, D.L. "Abortion clauses in surrogacy contracts: Insights from a case study" 2015 *Family Law Quaterly* 49(1) 29-53

Gola, S "One Step forward or one step back? Autonomy, agency and surrogates in the Indian Surrogacy (Regulation) Bill 2019" 2021 *International Journal of Law in Context* 17(1) 58-74

Heaton 2015 THRHR

Heaton, J "The Pitfalls of international surrogacy: A South African family law perspective" 2015 *Journal for Contemporary Roman-Dutch Law* 78(1) 24-46

Hisano, EY "Gestational surrogacy maternity disputes: Refocusing on the child" 2011 *Lewis & Clark Law Review* 15(2) 517-552

Iannacci, B.R "Why New York should legalize surrogacy: A comparison of surrogacy legislation in other states with current proposed surrogacy legislation in New York" 2018 *Touro Law Review* 34(4) 1239-1270

Jordaan 2016 SAMJ

Jordaan, DW "Surrogate Motherhood in illness that does not cause infertility" 2016 South African Medical Journal 106(7) 684-685

Jordaan 2017 SAJBL

Jordaan, DW "A Constitutional critique on the regulations relating to artificial fertilisation of persons" 2017 *South African Journal of Bioethics and Law* 10(1) 29-31

Labuschaigne, M., Mahomed, S., and Dhai, A. "Evolving capacity of children and their best interests in the context of health research in South Africa: An ethico-legal position" 2022 *Developing World Bioethics* 1-9

Louw 2014 De Jure

Louw, A "Ex Parte MS 2014 JDR 1012: Surrogate motherhood agreements, condonation of non-compliance with confirmation requirements and the best interests of the child" 2014 *De Jure* 47(1) 110-118

Louw 2013 THRHR

Louw, A "Surrogacy in South Africa: Should we reconsider the current approach?" 2013 *Journal for Contemporary Roman-Dutch Law* 76 564-588

Lyon, F "The surrogacy journey" 2020 New Law Journal 170(7900) 12-14

Lupton, M.L "Artificial wombs: medical miracle, legal nightmare" 1997 *Medicine and Law* 16 621-633

Meyerson, D "Surrogacy, Geneticism and Equality: The case of AB Minister of Social Development" 2019 *Constitutional Court Review* 9 317-342

Mills 2010 *Stell LR* Mills, L "Certainty about surrogacy" 2010 *Stellenbosch Law Review* 21(3) 429-437

Nicholson 2013 SAJHR Nicholson, C "When moral outrage determines a legal response: Surrogacy as labour" 2013 South African Journal on Human Rights 29(3) 496-514

Nicholson and Bauling 2013 De Jure

Nicholson, C and Bauling, A "Surrogate motherhood agreements and their confirmation: A new challenge for practitioners?" 2013 *De Jure* 46(2) 510-538

Nicholson, S and Nicholson, C "I used to have two parents and now I have three? When science (fiction) and the law meet: Unexpected complications" 2016 *Medicine and Law* 35 423-440

Nosarka and Kruger 2005 *SAMJ* Nosarka S, Kruger TF "Surrogate Motherhood" 2005 *South African Medical Journal* 95(12) 942-946

Nöthling Slabbert 2012 SAJBL

Nöthling Slabbert, M "Legal issues relating to the use of surrogate mothers in the practice of assisted conception" 2012 *South African Journal of Bioethics and Law* 5(1) 27-32

Oluwaseyi and Oladimeji 2021 *Obiter* Oluwaseyi, O.O. and Oladimeji, O "Surrogacy agreements and the rights of children in Nigeria and South Africa" 2021 *Obiter* 42(1) 20-38

Palm MT and Hirsch HL "Infertility and sterility: Legal implications of artificial conception" 1982 *Medicine and Law* 1 43-56

Pillay and Zaal 2013 SALJ

Pillay, R and Zaal, FN "Surrogate motherhood confirmation hearings: The advent of a fundamentally flawed process" 2013 *South African Law Journal* 130(3) 475-485

Pyrce, C "Surrogacy and Citizenship: A Conjunctive solution to a global problem" 2016 Indiana Journal of Global Legal Studies 23(2) 925-952

Quinn, CM "Mom, Mommy & Daddy and Daddy, Dad & Mommy: Assisted Reproductive Technologies & the Evolving Legal Recognition of Tri-Parenting" 2018 *Journal of the American Academy of Matrimonial Lawyers* 31(1) 175-207 Rasheed, A "Confronting problematic legal fictions in gestational surrogacy" 2021 *Journal of Health Care Law and Policy* 24(2) 179-202

Rautenbach 2017 TSAR

Rautenbach, I.M "Overview of constitutional court judgments on the bill of rights – 2016" 2017 *Journal of South African Law* 2017(2) 352-369

Rosenberg 2020 *TSAR* Rosenberg, W "Does the right to know one's origins exist and can it be limited?" 2020 *Journal of South African Law* 2020(4) 724-746

Skosana 2017 *Obiter* Skosana, T "A donor-conceived child's right to know its genetic origin: A South African perspective" 2017 *Obiter* 38(2) 261-274

Sloth-Nielsen 2019 *De Jure* Sloth-Nielsen, J "Children's rights Jurisprudence in South Africa – a 20 year retrospective" 2019 *De Jure Law Journal* 52 501-520

Sloth-Nielson, J and van Heerden, B "The constitutional family developments in South African child and family law 2003-2013" 2014 *International Journal of Law, Policy and the Family* 28(1) 100-120

Slabbert, M.N "Are the human embryo and the foetus *extra uterum* sufficiently protected in terms of South African law?" 2001 *Journal of South African Law* 2001(3) 495-512

Strauss, SA "Triplets to a Surrogate Grandmother in South Africa: Legal Issues" 1989 International Legal Practitioner 14(3) 70-71

Tager 1986 SALJ Tager,L "Surrogate Motherhood, Legal Dilemma" 1986 South African Law Journal 103(3) 381-404

Thaldar 2018 De Rebus

Thaldar D "Surrogate Motherhood and scaremongering: Is South Africa entering an age of post-truth jurisprudence?" 2018 *De Rebus* 27-29

Thaldar 2018 SAJBL

Thaldar, DW "The need to develop objective criteria for suitability as a surrogate mother: Reflections on *Ex Parte KAF*" 2018 *South African Journal of Bioethics and Law* 11(1) 35-36

Thaldar 2019 SAJBL

Thaldar, DW "Criteria for assessing the suitability of intended surrogate mothers in South Africa: Reflections on Ex Parte KAF II" 2019 *South African Journal of Bioethics and Law* 12(2) 61-65

Van Niekerk 2015 PELJ

Van Niekerk, C "Section 294 of the Children's Act: Do roots really matter?" 2015 *Potchefstroom Electronic Law Journal* 18(2) 398-428

Van Niekerk 2017 SAJBL

Van Niekerk, C "Strange (and incompatible) bedfellows: The relationship between the National Health Act and the regulations relating to artificial fertilisation of persons, and its impact on individuals engaged in assisted reproduction" 2017 *South African Journal of Bioethics and Law* 10(1) 32-35

Van Niekerk 2017 PELJ

Van Niekerk, C "Assisted reproductive technologies and the right to reproduce under South African law" 2017 *Potchefstroom Electronic Law Journal* 20 1-31

Wade, K "Reconceptualising the interest in knowing one's origins: a case for mandatory disclosure" 2020 *Medical Law Review* 28(4) 731-752

3. OLD AUTHORITIES AND EARLIER SOURCES

Grotius Inleidinge 2 27 28

Hahlo, H.R "Some Legal Aspects of Human Artificial Insemination" 1957 South Africa Law Journal 167

Masters, N.C "Artificial insemination" 1953 South Africa Law Journal 375

Spiro, E "Artificial insemination and the law" Acta Juridica (1972) 213

4. LAW REPORTS

South Africa

AB and another v Minister of Social Development 2017 (3) SA 570 (CC)

Ex parte WH and others 2011 (4) All SA 630 (GNP)

L v J 1985 (4) All SA 20 (C)

Ex parte MS and others; In re: Confirmation of surrogate mother agreement 2014 (2) All SA 312 (GNP)

CM v NG 2012 (3) All SA 104 (WCC)

J and Another v Director General, Department of Home Affairs and others 2003 (5) BCLR 463 (CC); 2003 (5) SA 621 (CC)

In re confirmation of three surrogate agreements 2011 (6) SA 22 (GSJ)

Ex parte SA and others 2014 (JDR) 2616 (GP)

Ex parte HPP and others; Ex parte DME and others 2017 (4) SA 528 (GP); [2017] 2 All SA 171 (GP)

Ex parte CJD and others 2018 (3) SA 197 (GP)

V v R 1979(3) SA 1006 (T)

Ex parte KAF and others 2019 (2) SA 510 (GJ)

Ex parte KAF and others (14341/17) [2017] ZAGPJHC 227 (10 August 2017)

KB and another v Minister of Social Development (966/2022) [2023] ZAMPMBHC 12

(20 February 2023)

VVJ and another v Minister of Social Development and another

[Unreported case, North Gauteng High Court, Pretoria, case no 27706/2021]

APP and another v NKP 2021 JDR 1650 (WCC)/ (17962/2020) [2021] ZAWCHC 69 (11 March 2021)

EJ and others v Haupt N.O 2022 (1) SA 514 (GP)

Surrogacy Advisory Group v Minister of Health (50683/2020) [2022] ZAGPPHC 558 (19 July 2022)

National Coalition for Gay and Lesbian Equality and Another v Minister of Justice and others 1999(1) SA 6 (CC)

NM and others v Smith and others (Freedom of expression institute as Amicus Curiae) 2007 (5) SA 250 (CC)

Bernstein and others v Bester and others NNO 1996 (2) SA 751 (CC)

Dawood and Another V Minister of Home Affairs and others: Shalabi Affairs and Another Minister of Home and others; V Thomas and Another v Minister of Home Affairs and others 2000 (3) SA 936 (CC)

Minister of Home Affairs and another v Fourie and another (Doctors for Life International and others Amici Curiae); Lesbian and Gay Equality project and others v Minister of Home Affairs and others 2006 (1) SA 524 (CC)

Minister of Health and others v Treatment Action Campaign and others

Bannatyne v Bannatyne (Commission for gender equality, as *Amicus Curiae*) 2003 (2) SA 363 (CC)

Harksen v Lane N.O 1998(1) SA 300 (CC)

Green v Fitzgerald, Fitzgerald v Green 1914 AD 88

United Kingdom

AB v DE [2013] EWHC 2413 (Fam) *Re A* [2015] EWHC 1756 (Fam) *CC v DD* [2014] EWHC 1307 (Fam) *JP v LP and others* [2014] EWHC 595 (Fam)

Canada

B.A.N. v. J.H. 2008 BCSC 808. https://www.canlii.org/en/bc/bcsc/doc/2008/2008bcsc808/2008bcsc808.html MAM v TAM 2015 NBQB 145.

https://www.canlii.org/en/nb/nbqb/doc/2015/2015nbqb145/2015nbqb145.html?searc hUrlHash=AAAAAQAfYXNzaXN0ZWQgaHVtYW4gcmVwcm9kdWN0aW9uIGFjdAA AAAAB&resultIndex=12

India

Baby Manji Yamada v. Union of India & Anr., (2008) 13 SCC 518. https://indiankanoon.org/doc/854968/

Jan Balaz vs. Anand Municipality and others Gujarat H.C. 2009. (Jan Balaz Vs. Anand Municipality & Ors., Letters Patent Appeal No.2151 of 2009.) https://www.casemine.com/judgement/in/56b48efa607dba348fff6a16#:~:text=Anand

<u>%20Municipality%20%26%206%20(S),-</u>

Judgement&text=K.%20S.%20RADHAKRISHNAN%2C%20C.J.%20%3A%2D%20T he,2.

5. LEGISLATION

5.1 STATUTES AND REGULATIONS: SOUTH AFRICA

Births, Marriages and Death Registration Act 81 of 1963

Human Tissue Act 65 of 1983

Regulations regarding the artificial insemination of persons, and related matters (Regulations regarding the artificial insemination of persons and related matters, published in GG No R 1182 of 20 June 1986 (GG 10283)

Child Care Act 74 of 1983

Children's Status Act 82 of 1987

Children's Act 38 of 2005

Regulations relating to the artificial fertilisation of persons published under Government Notice R175 in Government Gazette 35099 of 2 March 2012

The Constitution of the Republic of South Africa 103 of 1996

Regulations relating to assisted conception of persons published under Government Gazette No 44321 on 25 March 2021 (proposed)

National Health Act 61 of 2003 (NHA)

Choice on Termination of Pregnancy Act 92 of 1996

5.2 LEGISLATION: FOREIGN

5.2.1 United Kingdom

Surrogacy Arrangements Act of 1985 https://www.legislation.gov.uk/primary+secondary?title=surrogacy

The Human Fertilisation and Embryology Act 1990 ("HFEA 1990") <u>https://www.legislation.gov.uk/ukpga/1990/37/introduction</u> accessed on 21 January 2.

The Human Fertilisation and Embryology Act 2008 ("HFEA 2008") <u>https://www.legislation.gov.uk/ukpga/2008/22/introduction</u> accessed on 21 January 2023.

Code of practice 8th ed https://portal.hfea.gov.uk/media/1593/hfea-code-of-practice-8th-edition-r8.pdf accessed on 11 January 2023.

Human Fertilisation and Embryology Authority (Disclosure of Donor Information) Regulations 2004 https://www.legislation.gov.uk/uksi/2004/1511/regulation/2/made accessed 13 January 2023.

5.2.2 Canada

Reimbursement Related to Assisted Human Reproduction Regulations SOR/2019-193.

https://laws-lois.justice.gc.ca/PDF/SOR-2019-193.pdf accessed on 20 January 2023.

Assisted Human Reproduction Act S.C. 2004, c. 2. <u>https://laws-lois.justice.gc.ca/PDF/A-13.4.pdf</u> accessed 20 January 2023.

5.2.3 India

The Surrogacy (Regulation) Act, 47/2021.

https://www.indiacode.nic.in/handle/123456789/17046?sam_handle=123456789/136 2 accessed on 20 January 2023.

The Assisted Reproductive Technology (Regulation) Act,42/2021. <u>https://www.indiacode.nic.in/handle/123456789/17031?sam_handle=123456789/136</u> <u>2</u> accessed on 20 January 2023.

The Assisted Reproductive Technology (Regulation) Bill, 2020 <u>https://www.prsindia.org/billtrack/assisted-reproductive-technology-regulation-bill-</u> 2020 date accessed 9 February 2021.

6. INTERNATIONAL DOCUMENTS/INSTRUMENTS

United Nations Convention on the Rights of a Child (UNCRC)

Fast facts UNCRC:

https://www.parliament.gov.za/storage/app/media/Pages/2019/november/19-11-2019_30_Year_Commemoration_of_the_United_Nations_Convention_on_the_Right s of the Child/docs/FAST FACTS UNCRC draft 2 19 November 2019final.pdf accessed on 10 August 2021.

The convention

https://www.ohchr.org/en/professionalinterest/pages/crc.aspx accessed on 10 August 2021.

7. REGIONAL DOCUMENTS/INSTRUMENTS

African Charter on the Rights and Welfare of the Child (ACWRC)

8. REFERENCE WORKS

LexisNexis Law of South Africa (subscription-based)

LexisNexis Bill of Rights Compendium (looseleaf edition)

LexisNexis Family Law service VI (subscription-based; looseleaf edition)

Davel and Skelton *Commentary* Davel, CJ and Skelton, AM Commentary on the Children's Act (Jutastat e-publications, revision service 9, 2018 chapter 19) 5-36

9. MISCELLANEOUS AND OTHER DOCUMENTS

Report of the Ad Hoc Committee

Report of the Ad Hoc Committee on Report of South African Law Commission on Surrogate Motherhood (11 February 1999)

SALC Project 65

South African Law Commission Report, Project 65, Surrogate Motherhood (11 November 1992)

SALC Project 110

South African Law Commission, Project 110, Review of the Child Care Act (December 2002),

https://www.justice.gov.za/salrc/reports/r_pr110_01_2002dec.pdf.

SALRC Project 140

South African Law Reform Commission Issue paper 32, Project 140, The Right to know one's own biological origins (20 May 2017)

10. WEBSITES AND INTERNET SOURCES

https://fertilitypedia.org/edu/reproductive-cells/oocyte#/Fertilitypedia-description accessed 4 January 2023.

Semen vs Sperm <u>http://www.andrologycenter.in/2018/10/24/semen-vs-sperm/</u> accessed on 4 January 2023.

Infertility

https://www.who.int/news-room/fact-sheets/detail/infertility accessed on 3 January 2023.

https://www.oxfordlearnersdictionaries.com/definition/english/parent_1?q=parent accessed on 4 January 2023.

World Economic Forum: The law can't keep up with new tech. Here's how to close the gap (21 June 2018). <u>https://www.weforum.org/agenda/2018/06/law-too-slow-for-new-tech-how-keep-up/</u> accessed 4 January 2023.

Wheeler MA, McGrath MJ, Haslam N (2019) Twentieth century morality: The rise and fall of moral concepts from 1900 to 2007. PLOS ONE 14(2): e0212267. https://doi.org/10.1371/journal.pone.0212267 (accessed 4 January 2023).

Rettner, Rachael: Epigenetics: Definition & Examples (June 24, 2013) <u>https://www.livescience.com/37703-epigenetics.html</u> accessed on 16 January 2021.

What is epigenetics?

https://www.cdc.gov/genomics/disease/epigenetics.htm. accessed on 16 January 2021.

Talking glossary of genetic terms

https://www.genome.gov/genetics-glossary/Epigenome accessed on 16 January 2021.

https://dictionary.cambridge.org/dictionary/english/dignity accessed on 4 January 2023.

https://www.vocabulary.com/dictionary/dignity accessed on 31 January 2023.

WHO/International Classification of Functioning, Disability and Health (ICF) (2018) <u>https://www.who.int/classifications/international-classification-of-functioning-</u> <u>disability-and-health</u> accessed on 19 July 2022.

What does religion say about the morality of surrogacy?

https://www.americansurrogacy.com/blog/what-does-religion-say-about-the-moralityof-

surrogacy/#:~:text=Religious%20Views%20on%20Surrogacy&text=Instead%2C%20t he%20Church%20teaches%20that,this%20process%20is%20considered%20immor al. (11 December 2017) accessed on 5 January 2023.

Sukhanova Anna (2019) "Surrogacy and religion" https://surrogacybypons.com/surrogacy-and-religion/ accessed 5 January 2023.

West, R "Proposed changes to Quebec Surrogacy laws under Bill 2" <u>https://canadafertility.law/blog/proposed-changes-to-quebec-surrogacy-laws-under-bill-2/</u> accessed on 28 January 2023.

National Assembly of Quebec, Second Session, 42nd Legislature. Bill 2 (2022, chapter 22). An Act respecting family law reform with regard to filiation and amending the Civil Code in relation to personality rights and civil status. Assented to on 8 June 2022.

http://www2.publicationsduquebec.gouv.qc.ca/dynamicSearch/telecharge.php?type= <u>5&file=2022C22A.PDF</u> accessed on 25 January 2023.

New laws in India regulates assisted reproduction and surrogacy <u>https://reproductiverights.org/assisted-reproduction-and-surrogacy-in-india/</u> accessed 26 January 2023.

Law Commission of India (2009) Need for legislation to regulate assisted reproductive technology clinics as well as rights and obligations of parties to a surrogacy (Report no.228) 12

https://cdnbbsr.s3waas.gov.in/s3ca0daec69b5adc880fb464895726dbdf/uploads/202 2/08/2022081094-1.pdf accessed on 13 January 2023.

What is assisted reproductive technology?

https://www.cdc.gov/art/whatis.html accessed on 1 July 2022.

Wolf, D "Gamete and embryo cryopreservation" accessed from the Global Library of Women Medicine

https://www.glowm.com/section_view/heading/gamete-and-embryocryopreservation/item/365#:~:text=Cryopreservation%E2%80%94the%20ability%20t o%20freeze,nitrogen%20at%20%2D196%C2%B0C accessed on 14 February 2021.

Doula

https://my.clevelandclinic.org/health/articles/23075-doula accessed on 7 June 2023.

Singh, J "Commercial Surrogacy: The baby Manji Yamada case" <u>https://www.legallore.info/post/commercial-surrogacy-the-baby-manji-yamada-case</u> accessed on 1 June 2023.

Ratchaya, S.K "Facts of the case: Baby Manji Yamada v. Union of India" <u>https://liberttem.in/blog/surrogacy-in-india-baby-manji-yamada-case-and-surrogacy-regulation-bill</u> (accessed on 1 June 2023)

ATTACHMENT 1: ETHICAL CLEARANCE



UNISA 2022 ETHICS REVIEW COMMITTEE

Date: 2022:11:14 Dear Elaine Auret

ERC Reference No.: ST131 Name: E Auret

Decision: Ethics Approval from 2022:11:14 to 2025:11:14

Researcher: Adv Elaine Auret

Supervisor: Prof M Labuschaigne

Co-supervisor: Dr N Mabeka

The constitutionality of, and medico-legal issues regarding assisted reproduction and surrogate motherhood

Qualification: LLD

Thank you for the application for research ethics clearance by the Unisa 2022 Ethics Review Committee for the above mentioned research. Ethics approval is granted for 3 years.

The **negligible risk application** was **reviewed** by the CLAW Ethics Review Committee on 14 November 2022 in compliance with the Unisa Policy on Research Ethics and the Standard Operating Procedure on Research Ethics Risk Assessment.

The proposed research may now commence with the provisions that:

- The researcher will ensure that the research project adheres to the relevant guidelines set out in the Unisa Covid-19 position statement on research ethics attached.
- The researcher(s) will ensure that the research project adheres to the values and principles expressed in the UNISA Policy on Research Ethics.



University of South Africa Prelier Sanat, Muchanak Ricta, Cay of Thware PD Box 192 UNISA 0001 South Africa Telephone: +37 12 429 3111 Factoriale: +37 12 429 4150 www.ofitbout.28

- Any adverse circumstance arising in the undertaking of the research project that is relevant to the ethicality of the study should be communicated in writing to the CLAW Committee.
- The researcher(s) will conduct the study according to the methods and procedures set out in the approved application.
- 5. Any changes that can affect the study-related risks for the research participants, particularly in terms of assurances made with regards to the protection of participants' privacy and the confidentiality of the data, should be reported to the Committee in writing, accompanied by a progress report.
- 6. The researcher will ensure that the research project adheres to any applicable national legislation, professional codes of conduct, institutional guidelines and scientific standards relevant to the specific field of study. Adherence to the following South African legislation is important, if applicable: Protection of Personal Information Act, no 4 of 2013; Children's act no 38 of 2005 and the National Health Act, no 61 of 2003.
- Only de-identified research data may be used for secondary research purposes in future on condition that the research objectives are similar to those of the original research. Secondary use of identifiable human research data requires additional ethics clearance.
- No field work activities may continue after the expiry date 2025:11:14. Submission
 of a completed research ethics progress report will constitute an application for
 renewal of Ethics Research Committee approval.

Note:

The reference number ST131-2022 should be clearly indicated on all forms of communication with the intended research participants, as well as with the Committee.

Yours sincerely,



Prof L Fitz Chair of CLAW ERC E-mail: fitzig@unisa.ac.za Tel: (012) 433-9504

the last

Prof QJ Kole Acting Executive Dean: CLAW E-mail: koleoj@unisa.ec.za Tel: (012) 429-6305



University of South Africa Prelier Street, Muddensel: Ridge, City of Tihwane PO Box 192 UNISA DDOS South Africa Telephone: +27 12 429 3111 Facsimile +27 12 429 4150 www.unisa.ac.ta

ATTACHMENT 2: TURNITIN REPORT



